

Oxidation of low density lipoprotein by ferritin at lysosomal pH and its possible role in atherosclerosis

Oluwatosin Ololade Ojo

Thesis submitted for the Degree of Doctor of Philosophy School of Biological Sciences

February 2019

Declaration

All of the data presented in this thesis is my own work. None of the content has been submitted for a diploma, degree or other qualifications at any other University. The use of all materials from other sources have been acknowledged.

Oluwatosin Ololade Ojo

Acknowledgements

My special gratitude goes to my supervisor, Professor David Leake for his advice, encouragement and support throughout this PhD research work. I hope I continue to enjoy his mentorship. I would also like to thank my supervisory committee members, Dr Sam Boateng and Dr Phil Dash for their useful comments and suggestions.

My appreciation goes to Tertiary Education Trust Fund (TETFund) for the funding received. I would also like to thank Professor J.S.A Osho for the financial support received to augment my tuition fees. I am also grateful to the management of Federal University Ndufu Alike-Ikwo (FUNAI) now known as Alex Ekweme Federal University Ndufu Alike-Ikwo for the opportunity to access the grant.

I would like to thank Dr John McKendrick for the opportunity to use ICP-MS in his laboratory, Andrew Dodson, Graham Luke, and Carlos Poveda, for the inductions provided. Many thanks to Chris J Humphrey for setting up the AAS machine, Nicholas Michael for his inputs in trouble shooting HPLC systems, Dr Xiao Yi for the gift of the novel lysosomal targetable probe and Dr Sakthivel Vaiyapuri for the gift of some of the reagents used for SDS-PAGE analysis.

Our colleagues who donated their blood made the majority of the work possible. Without them, there would be no LDL isolated. Particular thanks go to Ms. Rada Mihaylova and Dr. Kim Jackson for skilfully taking blood from our volunteers. I thank my colleagues who have either gone through this journey or currently pursuing their degree Hadeel Alboaklah, Khalid Alyodawi, Rob Mitchel, Laura Thei, James Tomkins, Feroz Ahmad, Eleni Kaisi, Jono Sheard and Wouter Eilers for their kind support.

Special thanks to my friends Dr. Omolola Oyenihi, Yewande Suberu, Aderonke Ajayi-Smith, Olusola Genesis, Jamal Alabi and Ian Common. The encouragement and exchange of experience were comforting.

Finally, I would like to thank my family for their immense support, Mr and Mrs Adefuye, Mrs Ivackovic, my cousins Mrs Jacklyn Strafford and Mr Peter Adefuye. My parents for their constant support and prayers. My siblings, Mr Olusegun & Abiodun Ojo, Mrs Tumininu Komolafe and Dr Temitope Ojo, you people stood behind me like a rock throughout this journey. Above all, I thank almighty God for keeping me alive to complete this journey.

Publications and Posters

Full paper

• Ojo, O. O. & Leake, D. S. 2018. Low density lipoprotein oxidation by ferritin at lysosomal pH. *Chem Phys Lipids*, 217, 51-57.

Published abstracts

- Ojo, O.O. & Leake, D.S. 2018. P27 Oxidation of LDL by ferritin in lysosomes increases oxidative stress in macrophages. *Cardiovascular Research*, 114 S9-S9.
- Ojo, O. O. & Leake, D. S. 2018. Oxidation of low density lipoprotein by ferritin at lysosomal pH. *Atherosclerosis Supplements*, 32, 99.
- Ojo, O. O, Ahmad, F, Leake, D. S. 2016. 194 Ferritin oxidises low density lipoprotein at lysosomal pH. *Heart*, 102: A131.

Posters

- Ferritin oxidises low density lipoprotein at lysosomal pH. *Joint BCS/BSCR/BAS conference*. Manchester. 2016.
- Oxidation of low density lipoprotein by ferritin at lysosomal pH. *XVIIIth International Symposium on Atherosclerosis*. Toronto. 2018.

Full papers to be prepared and submitted

- Vitamins C and E do not effectively inhibit lysosomal LDL oxidation by ferritin at lysosomal pH.
- Effects of LDL degradation on its oxidation by ferritin at lysosomal pH
- LDL oxidation by ferritin in lysosomes of macrophages: effects on macrophage function in relation to atherosclerosis.

Abstract

Atherosclerosis leading to coronary heart disease and thrombotic stroke is the leading cause of death worldwide. Oxidised low density protein is considered important by many in this disease. Our laboratory has shown that LDL can be oxidised by iron in the lysosomes of macrophages. Others have demonstrated the presence of oxidised products similar to those formed *in vitro* by catalytically active iron in advanced atherosclerotic lesions. Ferritin is the body's main iron storing protein. The role of ferritin in lysosomal LDL oxidation and explored the possible consequences for atherosclerosis were investigated. Ferritin oxidised LDL effectively at lysosomal pH (pH 4.5), much faster than at pH 7.4, as shown by increased formation of oxidised lipids (HPLC and tri-iodide assay) and conjugated dienes (automated spectrophotometry). Ferritin spontaneously released iron at lysosomal pH and iron chelators supressed the oxidation of LDL by ferritin.

The degradation of apolipoproteinB-100 and cholesteryl esters of LDL speeded up the oxidation of LDL by ferritin. The lysosomotropic antioxidant cysteamine was shown to be a more appropriate antioxidant to effectively inhibit LDL oxidation by ferritin compared to ascorbate, α -tocopherol and 4-hydroxy-2,2,6,6-tetramethylpiperidine-N-oxyl (tempol) and N, N'- diphenyl 1,4-phenylenediamine (DPPD).

The role of ferritin in lysosomal LDL oxidation within macrophages and macrophage function were accessed. Incubation with ferritin and LDL led to increased intracellular lipid, ceroid and lipid peroxidation in lysosomes. Ferritin and LDL treatment increased glycolysis and THP-1 macrophage-like cells became metabolically activated, as shown by a Seahorse analyser. Ferritin-oxidised LDL induced reactive oxygen species formation and apoptosis in macrophages, suggesting LDL oxidation by ferritin in lysosomes might be atherogenic.

Table of Contents

Declarationi
Acknowledgementsii
Publications and Postersiv
Abstractv
List of Figuresxvi
List of Abbreviationsxx
Chapter 1- General introduction 1
1.1 Importance of Subject Area2
1.2 Overview of Atherosclerosis
1.2.1 Histology of human artery
1.2.2 Pathogenesis of Atherosclerosis
1.2.2.1 Features of atherosclerotic lesion
1.2.2.2 Aetiology
1.2.3.2 Lesion initiation, inflammatory response and foam cell formation
1.2.3.4 Atherosclerotic lesion macrophages10
1.2.3.5 Formation of fibrous plaques11
1.2.3.6 Plaque progression and vulnerability
1.2.4 Cell death in atherosclerosis
1.2.5 Epidemiology and risk factors
1.2.5.1 Smoking
1.2.5.2 Dyslipidaemia and Hyperlipidaemia

	1.2.5.	3 Diabetes	17
	1.2.5.	4 Hypertension	18
	1.2.5.	5 Age	18
	1.2.5.	6 Gender	19
	1.2.5.	7 Obesity	19
	1.2.5.	8 Sedentary lifestyle and lack of exercise	20
	1.2.5.	9 Other factors	20
1.3	Lipo	oproteins and Atherosclerosis	22
1	.3.1	Introduction to lipoproteins	22
1	.3.2 Lij	pid Metabolism	23
	1.3.2.	1 Endogenous lipid pathway	23
	1.3.2.	2 Exogenous lipid pathway	24
	1.3.2.	3 Reverse cholesterol transport	24
1	.3.3	Low density lipoprotein (LDL)	25
1	.3.4	LDL metabolism	26
1	.3.5	Role of LDL in atherosclerosis	27
1.4	Oxidat	ive Modification of LDL and Atherosclerosis	28
1	.4.1	LDL modification hypothesis	28
1	.4.2	Proatherogenic effects of oxidised LDL	30
1	.4.3	Oxidation of LDL	32
1	.4.4 Re	active oxygen species and atherosclerosis	38

1.	.4.5 Po	tential Mechanisms of LDL modification	38
	1.4.5.1	Myeloperoxidase	39
	1.4.5.2	Lipoxygenase	39
	1.4.4.3	Reactive nitrogen species	40
	1.4.5.4	Glycation of LDL	41
	1.4.5.5	Transition metals	41
1.5	Lysoso	mal oxidation of LDL	43
1.	.5.1 Gener	ral structure and function of lysosomes	43
1.	.5.2 Lysos	somes as a source of redox active iron	46
1.	.5.3 Lysos	somal storage disorder and atherosclerosis	47
1.	.5.4 Evide	ence for LDL oxidation by iron in lysosomes	47
1.	.5.5 LI	DL aggregation and lysosomal LDL oxidation	49
1.6	Iron and a	therosclerosis	51
1.	.6.1 Iron i	n atherosclerotic plaques	53
1.	.6.2 Irc	on metabolism	54
1.	.6.3 Irc	on binding proteins and LDL oxidation	55
1.7	Ferritin	1	56
1.	.7.1 Iron i	ncorporation into ferritin	58
1.	.7.2 Contr	ol of cellular iron availability by ferritin	58
1.8	The use	e of Antioxidants in atherosclerosis	59
1.9 ′	Thesis hyp	pothesis and aims	62

Chap	ter 2	- General materials and methods
2.0	Ma	terials and Methods
2.1	Che	emicals
2.1	.1	Solutions
2.1	.2 La	boratory equipment
2.2	Pre	paration of LDL
2.2	2.1 Et	hics and blood collection71
2.2	2.2	Enrichment of LDL with α-tocopherol
2.2	2.3	Lowry protein assay
2.2	2.4	Preparation of sphingomyelinase aggregated LDL74
2.2	2.4	Preparation of lipoprotein-deficient Serum (LPDS)
2.3	Spe	ectrophotometric measurement of LDL oxidation by ferritin at lysosomal pH 76
2.3	8.1	Comparison of oxidation of LDL by ferritin at pH 4.5 and pH 7.4
2.3	8.2	Measurement of LDL aggregation in the presence of ferritin
2.3	8.3	Atomic absorption spectrophotometric quantification of iron in ferritin and
apo	oferri	tin 77
2.3	8.4	Comparison of iron released from ferritin at pH 4.5 and 7.4 using iron chelator.77
2.3	8.5	Comparison of iron released from ferritin by the ultrafiltration method77
2.3	8.6	The effect of DTPA and EDTA on LDL oxidation by ferritin
2.3	8.7	Effect of antioxidants on LDL oxidation by ferritin
2.3	8.8	Effect of cysteamine on iron released by ferritin
2.4	Me	asurement of LDL oxidation by reverse-phase HPLC analysis of lipid species 79

2.4	4.1	LDL oxidation with ferritin for HPLC analysis	79
2.4	4.2	Extraction of lipids from the oxidised sample for HPLC analysis	79
2.4	4.3	HPLC Measurement	80
	2.4.3.	1 Conditions for analysis of targeted lipid species	80
2.5	Mea	asurement of LDL oxidation by determination of total lipid hydroperoxide co	ontent
by th	ne tri-i	odide assay	82
2.6	Pro	teolytic degradation of ferritin	82
2.	6.1	Enzyme digestions	82
	2.6.1.	1 Incubation of ferritin with cathepsin D	82
	2.6.1.	2 Incubation of ferritin with cathepsin D and B	83
	2.6.1.	3 Incubation of LDL with cathepsin D	83
	2.6.1.	3 Co-Incubation of LDL with cholesteryl esterase	83
2.	6.2	Electrophoresis	83
2.	6.3	Effect of proteases and cholesterol esterase on LDL oxidation by ferritin	83
2.7	Cel	l culture	84
2.	7.1	Cryopreservation of THP-1 monocytes.	85
2.7	7.2	Conversion of THP-1 cells to THP-1 derived macrophages	85
2.8	Isol	ation of human monocyte-derived macrophages	85
2.9	Det	ermination of the effect of hepcidin on intracellular iron in macrophages	87
2.9	9.1 Tr	eatment of THP-1 derived macrophage cells with hepcidin	87
2.	9.2 De	etermination of intracellular protein concentrations	88
	2.9.2.	1 Preparation of cell lysate	88

2.9.2.2.Determination of protein content with Bio-Rad DC TM protein assay kit
2.9.3 Inductively coupled plasma mass spectrometry (ICP-MS) for measuring total iron
in macrophages
2.10 Measurement of lysosomal lipid peroxidation in THP-1 cells and human monocyte-
derived macrophages
2.11 Measurement of intracellular lipid and ceroid levels after treatment with ferritin90
2.11.1 Preparation of sterilised coverslips90
2.11.2 Measurement of intracellular lipids and intralysosomal ceroid
2.11.3 Oil Red O Staining for intracellular lipids
2.11.4 Oil Red O Staining for Ceroid
2.12 Determination of intracellular reactive oxygen species formation induced by ferritin-
oxidised LDL
2.13 Determination of the effect of oxidation of LDL by ferritin on cellular
respiration/metabolism using a Seahorse analyser
2.14 Determination of apoptosis in macrophages treated with ferritin-oxidised LDL95
2.15 Statistical analysis
Chapter 3 - Mechanisms of oxidation of LDL by ferritin at lysosomal pH
3.0 Background and rational
3.1 Objectives
3.2 Methods 103
3.3 Results
3.3.1 Spectrophotometric measurement of LDL oxidation by ferritin at lysosomal pH 104

3.3.2 Measurement of LDL aggregation during oxidation by ferritin
3.3.3 Comparison of oxidation of LDL by Ferritin at pH 4.5 and pH 7.4 109
3.3.4 Determination of the quantity of iron contained in ferritin
3.3.5 Measurement of iron released from ferritin at pH 4.5 versus pH 7.4 using the iron
chelator bathophenanthroline (BP)113
3.3.6 Ultrafiltration method for determination of iron released
3.3.6 Effect of iron chelators on LDL oxidation by ferritin at pH 4.5 117
3.3.7 Oxidation of LDL in the presence of apoferritin at lysosomal pH 119
3.3.8 Formation of oxidised lipids by LDL oxidised by ferritin at lysosomal pH 121
3.4 Discussion
Chapter 4: Degradation of ferritin and LDL in lysosomes: implications for lysosomal LDI
oxidation by ferritin
4.0 Background and rational
4.1 Objectives
4.2 Methods:
4.3 Results
4.3.1 Proteolytic degradation of ferritin by cathepsins in vitro (pH 4.5)
4.3.2 Effects of treatment of ferritin with proteases on LDL oxidation by ferritin at lysosoma
pH137
4.3.3 Proteolytic degradation of LDL by cathepsin D
4.3.4 Effect of treatment of LDL with proteases on LDL oxidation by ferritin at lysosoma
pH143

4.3.5 Effect of co-incubation with cholesteryl esterase on LDL oxidation by ferritin at
lysosomal pH 145
4.3.6 Effects of co-incubation with cholesteryl esterase and proteases on LDL oxidation by
ferritin at lysosomal pH148
4.4 Discussion
Chapter 5: The role of antioxidants in preventing LDL oxidation by ferritin
5.0 Background and rational156
5.1 Objectives
5.2 Methods 162
5.3 Results
5.3.1 Effect of α-tocopherol in LDL oxidation mediated by ferritin
5.3.2 Effects of vitamin C (ascorbate and dehydroascorbate) on LDL oxidation by ferritin at
lysosomal pH165
5.3.3 Effects of existing oxidised lipids and pH on antioxidant effects of ascorbate 168
5.3.4 Effects of tempol on LDL oxidation by ferritin at lysosomal pH
5.3.5 Effects of DPPD on LDL oxidation by ferritin at lysosomal pH
5.3.6 Effect of cysteamine on LDL oxidation mediated by ferritin at lysosomal pH 176
5.3.6.1 Effect of lower concentrations of cysteamine on LDL oxidation by ferritin at
lysosomal pH176
5.3.6.2 Effect of higher concentrations of cysteamine on LDL oxidation by ferritin at
lysosomal pH178
5.3.7 Effect of cysteamine on ferrous complex formation by ferritin

5.4 Discussion
Chapter 6: The effects of lysosomal oxidation of LDL by ferritin in macrophages
6.0 Background and rational
6.1 Objectives
6.2 Methods
6.3 Results
6.3.1 Effect of hepcidin on intracellular iron in macrophages
6.3.2 Effects of ferritin on lipid peroxidation within lysosomes of macrophages
6.3.3 Effects of ferritin on intracellular lipids and ceroid
6.3.4 Effects of ferritin-oxidised LDL on intracellular ROS in macrophages
6.3.5 Effect of ferritin oxidation of LDL on cellular respiration/metabolism in THP1-cell
6.3.6 Effect of ferritin-oxidised LDL on macrophages cell death
6.4 Discussion
0.4 Discussion
Chapter 7: General Discussion
Chapter 7: General Discussion
0.4 Discussion 22 Chapter 7: General Discussion 22 7.0 General Discussion 23 7.1 Critical evaluation of the present study 24
0.4 Discussion 22 Chapter 7: General Discussion 22 7.0 General Discussion 23 7.1 Critical evaluation of the present study 24 7.1.1 Summary of main findings 24
0.4 Discussion 22 Chapter 7: General Discussion 22 7.0 General Discussion 23 7.1 Critical evaluation of the present study 24 7.1.1 Summary of main findings 24 7.1.2 Limitations of the study 24
Chapter 7: General Discussion 22 7.0 General Discussion 23 7.1 Critical evaluation of the present study 24 7.1.1 Summary of main findings 24 7.1.2 Limitations of the study 24 7.2 Possible future work 24
Chapter 7: General Discussion 22 7.0 General Discussion 23 7.1 Critical evaluation of the present study 24 7.1.1 Summary of main findings 24 7.1.2 Limitations of the study 24 7.2 Possible future work 24 References 24

Appendix 1: List of chemicals and suppliers	
Appendix 2: Quantity of iron contained in ferritin	300
Appendix 3: Quantity of iron contained in Apoferritin	302

List of Figures

Chapter 1

Figure 1.1: Schematic representation of the progression of atherosclerotic lesion	23
Figure 1.2: Structure of LDL	26
Figure 1.3: Schematic representation of peroxidation reaction in polyunsaturated	fatty
acid	35
Figure 1.4: Structure of ferritin	56

Chapter 2

Figure 2.1 Measurement of aggregation in SMase-LDL compared to native LDL75
Figure 2.2: Example chromatogram showing 7-ketocholesterol81
Figure 2.3: Example chromatogram showing cholesteryl linoleate hydroperoxide81

Chapter 3

Figure 3.1: The oxidation of LDL by different concentrations of ferritin
Figure 3.2: Effect of LDL oxidation by ferritin on LDL aggregation108
Figure 3.3: Comparison of oxidation of LDL by ferritin at pH 4.5 and pH 7.4110
Figure 3.4: Standard plot of iron determination in ferritin assay by atomic absorption
spectroscopy112
Figure 3.5.1: Ferrous complex formation at pH 7.4 and pH 4.5114
Figure 3.5.2: Iron released from ferritin at pH 4.5 measured by ultrafiltration and AAS116
Figure 3.6: Effect of EDTA and DTPA on oxidation of LDL by ferritin at lysosomal pH118
Figure 3.7: Oxidation of LDL by apoferritin at lysosomal pH120
Figure 3.8: Oxidised lipids formed from oxidation of LDL by ferritin at pH 4.5122

Figure 3.9: Total hydroperoxides formed by LDL oxidised by ferritin at pH 4.5.....123

Chapter 4

Chapter 5

Figure 5.1: The structures of α -tocopherol, ascorbate and dehydroascorbate157
Figure 5.2 Structure of tempol159
Figure 5.3: Structures of DPPD and cysteamine160
Figure 5.4. The effect of α -tocopherol on LDL oxidation by ferritin at pH 7.4 and pH 4.5164
Figure 5.5: The effect of ascorbate on LDL oxidation by ferritin at lysosomal pH166
Figure 5.6: The effect of dehydroascorbate on LDL oxidation by ferritin at lysosomal pH.167
Figure 5.7. The effect of existing oxidised lipids on the effect of ascorbate on LDL oxidation
by ferritin at lysosomal pH170

Figure 5.8. The effect pH on ascorbate protecting LDL from copper-mediated LDL

oxidation1	71
Fig. 5.9: The effect of tempol on LDL oxidation by ferritin at lysosomal pH1	73
Figure 5.10: The effect of DPPD on LDL oxidation by ferritin1	75
Fig 5.11: The effect of cysteamine on LDL oxidation by ferritin1	77
Fig 5.12: The effect of higher concentrations of cysteamine on LDL oxidation by ferritin	ı at
lysosomal pH1	79
Fig. 5.12: The effect of cysteamine on ferrous complex formation by ferritin at pH 4.51	.81

Chapter 6

Figure 6.1: Untreated THP1 monocytes and macrophages
Figure 6.2: Intracellular protein content of THP1 macrophages treated with hepcidin200
Figure 6.3: Intracellular iron content of THP-1 macrophages treated with hepcidin201
Figure 6.4: Two-way flow cytometry analysis of lipid peroxidation in THP-1 macrophages.204
Figure 6.5: Measurement of lipid peroxiation in THP-1 macrophages in the presence of native
LDL and ferritin
Figure 6.6: Measurement of lipid peroxiation in THP-1 macrophages and HMDM in the
presence of SMase-LDL and ferritin
Figure 6.7: Detection of intracellular lipids in THP-1 macrophages209
Figure 6.8: Measurement of percentage intracellular lipids in THP-1 macrophages210
Figure 6.9: Detection of ceroid in lysosomnes of THP-1 macrophages211
Figure 6.10: Measurement of percentage ceroid in THP-1 macrophages212
Figure 6.11: Effect of ferritin-oxidised LDL on THP-1 cells intracellular ROS214

Figure 6.12: Effect of cysteamine on ROS formation induced by ferritin-oxidised LDL215
Figure 6.13: Time course of effect of ferritin and LDL on metabolism of macrophages217
Figure 6.14: Effects of LDL and ferritin on metabolism of macrophages
Figure 6.15: Flow cytometry analysis of ThP1 macrophages cell death
Figure 6.16: Effects of ferritin-oxidised LDL on apoptosis in THP-1 macrophages

List of Abbreviations

AAS	Atomic absorption spectrophotometer
Acetyl LDL	Acetylated LDL
ADH	Autosomal dominant hypercholesterolaemia
BHT	Butylated hydroxytoluene
BP	Bathophenthroline
CAD	Coronary artery disease
CHD	Coronary heart disease
CLOOH	Cholesteryl linoleate hydroperoxide
CVD	Cardiovascular Disease(s)
DHE	Dihydroethidium
DMT 1	Divalent metal transporter 1
DTPA	Diethylenetriaminepentaacetic acid
DPPD	N, N'- diphenyl – p-phenylenediamine
EC	Endothelial cells
ECM	Extracellular matrix
FCS	Fetal calf serum
FH	Familial hypercholesterolaemia
HD	Huntington Disease
HDL	High density lipoprotein
HDS	High density Solution
HIF-1a	Hypoxia-inducible factor
HMDM	Human monocyte derived macrophages
HNE	Hydroxynonenal
IDL	Intermediate density lipoprotein

iNOS	Inducible nitric oxide synthase
PL	Phospholipid
PMA	Phorbol 12-myristate-12-acetate
IFN γ	Interferon γ
IHD	Ischaemic heart disease
LDL	Low density lipoprotein
LIP	Labile iron pool
LOX1	Lectin-type oxidised LDL receptor 1
MCSF	Macrophage colony-stimulating factor
MCP-1	Monocyte chemoattractant protein-1
MDA	Malondialdehyde
MMP	Matrix metalloproteinase
MI	Myocardial infarction
MPO	Myeloperoxidase
oxLDL	oxidised low density lipoprotein
TG	Triglycerides
TGF-β	Tumour growth factor β
TNF-α	Tumour necrosis factor a
PCSK9	Proprotein convertase subtilisin-kexin 9
PDGF	Platelet derived growth factor
PPAR	Peroxisome proliferator-activated receptor
PUFA	Polyunsaturated fatty acid(s)
SDS	Sodium dodecyl sulphate
SMase-LDL	Sphingomyelinase-aggregated LDL
SMC	Smooth muscle cells

- SRA Scavenger receptor A
- SRB1 Scavenger receptor class B member 1
- SREBP Sterol regulatory binding protein
- TEMPOL 4-hydroxy-2,2,6,6-tetramethylpiperidine-N-oxyl
- WHHL Watanabe-Heritable Hyperlipidaemic
- VCAM-1 Vascular cell adhesion molecule-1
- VLDL Very low density lipoprotein

Chapter 1- General introduction

1.1 Importance of Subject Area

Cardiovascular diseases (CVD) which arise from disorders of the heart, arteries and other blood vessels remain one of the leading causes of death worldwide, despite scientific advances. In the year 2012, the global burden of CVD had increased to 17.9 million from a previous value of 14.4 million in 1990 (WHO, 2015). Atherosclerosis, characterised by the pathological transformation of the arteries is the underlying cause of the majority of the deaths arising from CVD (Stone, 2012). It is a complex chronic disease in which there is an accumulation of fibrofatty deposits made up of lipids and fibrous element mainly in the inner lining (intima) of arteries (Lusis, 2000, Wang and Bennett, 2012). Atherosclerosis is a multifactorial disease; studies conducted over the past five decades have revealed different genetic and environmental factors associated with the disease (Goldbourt and Neufeld, 1988, Assmann *et al.*, 1999, Gerhard and Duell, 1999). Until recently, CVD has been overshadowed by infectious diseases such as HIV/AIDS. This is evident by its omission in the United Nations Millennium Development Goals in 2000.

CVD in the UK has declined since the early 1970s but despite the reduction, it remains relatively high when compared to other western European countries. CVD still remains the biggest killer in the UK with about 180,000 deaths in 2010 (Townsend *et al.*, 2012). It has been predicted that by the year 2030, non-communicable diseases will be responsible for three-quarters of death occurring worldwide (Fuster and Kelly, 2010). In low and middle-income countries, CVD alone will be responsible for more deaths compared to other diseases (Beaglehole and Bonita, 2008). Due to the high mortality rate and incidence of heart diseases, the EU economy is estimated to spend almost 196 billion Euros a year on CVD (Nichols *et al.*, 2012). Over the last decades, there has been a significant rise in cardiovascular diseases mainly

in developing countries. These changes have been attributed to lifestyle changes and nutritional transition (Herrington *et al.*, 2016). The clinical consequence of atherosclerosis, heart attack and thrombotic strokes continues to be a huge burden. In spite of the progress made in decreasing mortality rate in the UK, the National Health Service in England incurred a cost of about 6.8 billion pounds from spending on CVD between the year 2012 and 2013 (Bhatnagar *et al.*, 2015). CVD continues to be a burden on the country, in terms of cost and health.

Despite the lifestyle modification initiatives and development of effective drugs that can effectively reduce some of the risk associated with the disease such as high plasma cholesterol and blood pressure (Olsen *et al.*, 2016, Cesena *et al.*, 2017), atherosclerosis and its clinical consequence, still remains a global burden (Herrington *et al.*, 2016). The global burden of atherosclerotic cardiovascular disease in people living with HIV have been recently been suggested to be double that of the healthy population, as they are twice at risk of developing cardiovascular disease (Shah Anoop *et al.*, 2018). Hence the scenario of atherosclerotic cardiovascular disease is of global concern. The key issues to be addressed include systemic inflammation, clotting factors, hyperlipidaemia and the oxidation of low density lipoprotein (LDL) (Sharma *et al.*, 2013). LDL is believed to be a major contributor to the pathogenesis of atherosclerosis (Lewington *et al.*, 2007).

1.2 Overview of Atherosclerosis

1.2.1 Histology of human artery

The large artery is made up of three distinct layers surrounding the arterial lumen. Each of these layers is made up of different cells and extracellular matrix (ECM). The innermost layer adjacent to the lumen is called the intima and is bound by a single layer of endothelial cells (ECs) on the side of the lumen and bordered on the peripheral side by the internal

elastic lamina, a sheet of elastic fibres. The intima is made up of proteoglycans and collagen as the main connective tissue matrix (Lusis, 2000). ECs are the main cellular component of this layer, although smooth muscle cells (SMCs) are sometimes found in this layer. The EC regulate a number of events including vascular tone (Sandoo *et al.*, 2010) and leukocyte movement (Muller, 2014), among others. After the internal elastic lamina is the second layer, the media (the middle layer) which consist of layers of SMCs and elastic lamellae. The number of layers present is dependent on the size of the artery. The ECM, which binds the SMCs to each other, contains mainly collagen and elastic fibres with a lesser amount of proteoglycans. The outermost layer, the adventitia contains a loose matrix of SMCs, fibroblast, and collagen (Lusis, 2000, Stocker and Keaney, 2004).

1.2.2 Pathogenesis of Atherosclerosis

1.2.2.1 Features of atherosclerotic lesion

Atherosclerosis is characterised by hardening of the arteries which arises from the accumulation of fibrofatty deposits (made up of lipids and fibrous element) in the inner lining and the middle lining of the arteries (Lusis, 2000). These abnormalities in the arteries (atherosclerotic lesions) are classified based on their histological features and composition. The lesion is classified into six types and designated with roman figures according to their stage and sequential progression.

The first observable change is the type I lesion which contains enough lipids to trigger the formation of scattered foam cells (lipid-laden macrophages). These changes are seen more in areas that are susceptible to the lesion and capable of adaptive thickenings, (the thickening are common in everyone at birth but are only formed in response to mechanical force) (Stary *et al.*, 1995). The accumulation of the scattered foam cells leads to progression to type II lesions,

fatty streaks (which are accumulated lipid-filled macrophages) which are formed at this stage. Continuous formation of lipid-laden cells (foam cells) and macrophages death leads to the formation of Type III lesions which contain extracellular droplets of lipids in addition to the foam cells. These early stage lesions can be observed within the first decade of life (Stary, 1983, Stary *et al.*, 1995). The extracellular lipids in type III form the basis for the larger lipid core that characterises the type IV lesion. The core lipid is separated from the lumen by the thin tissue layer, which advances and becomes thickened to form a fibrous connective tissue (the lesion cap). They mature into the more advanced lesion (Type VI lesion) characterised by calcification of the fibrous region and appearance of ulcerations that are often visible (Stary *et al.*, 1995).

1.2.2.2 Aetiology

The aetiology of atherosclerosis is often complex. In the last fifteen decades, there have been concerted efforts by researchers to explain the series of complex events that are associated with the onset of the disease. To explain these events several hypotheses have emerged over the years. Early studies proposed passive deposition of cellular components. The early hypothesis of atherosclerosis **"incrustation hypothesis"** by Rokitansky suggest that an atherosclerotic event arises as a result of deposition of fibrin in the arterial intima (Rokitansky, 1849). Virchow in 1856 proposed the **"lipid infiltration hypothesis"** which suggests the deposition of lipids in the arterial wall is responsible for the disease. The deposited lipid is taken up by macrophages and SMCs resulting in the formation of foam cells (Virchow, 1989).

Ross and Glomset in 1973, proposed the **"response to injury hypothesis"**. They suggested that the early event in atherosclerosis is due to endothelial damage which has a negative impact on normal endothelial function (Ross and Glomset, 1973). The injury increases the adhesion

of leukocytes and platelets to the endothelium. The recruited leukocytes bring about an inflammatory response by releasing a number of cytokines and growth factors. The inflammatory area is often associated with smooth muscle cells proliferation and migration from the media giving rise to an intermediate lesion. The macrophages and T lymphocytes recruited are the principal mediators of the early inflammatory response (Jonasson *et al.*, 1986, Keaney Jr, 2000). The hallmark of atherosclerosis involves the uptake of LDL by macrophages and to some extent SMC to form foam cells (lipid-laden macrophages). The accumulation of foam cells leads to the continuation of the inflammatory response and formation of a localised lesion. Cellular apoptosis or necrosis occurs as a result of the continued inflammatory process. The associated release of growth factor, cytokines and proteolytic enzymes bring about the enlargement of the lesion. Blood flow can be impaired at some point when the artery is unable to compensate for the encroachment of the lumen.

The response-to-injury hypothesis previously believed the key initiating event in atherosclerosis is as a result of the endothelium desquamation (Stocker and Keaney, 2004). Later on, there was evidence that when atherosclerotic lesions develop, they are covered by an endothelial cell layer that is intact. This led to the refinement of this hypothesis. In 1999, Ross proposed that endothelial dysfunction can initiate atherosclerosis through enhanced permeability of the endothelium to LDL (Ross, 1999). This hypothesis was challenged by the fact that more LDL entered the normal artery than the amount that accumulated there (Carew *et al.*, 1984). The rate of entry was similar but areas with a large accumulation of lipoproteins are prone to developing lesions due to their ability to retain apoB-containing lipoproteins which are atherogenic in nature (Schwenke and Carew, 1989, Schwenke and Zilversmit, 1989). These observations lead to the development of an alternative hypothesis, the **"response to retention hypothesis"** which suggests that the inciting event for atherosclerosis is the retention of

lipoprotein with the recognition that endothelial permeability may have a role in the initiation of the disease if a portion of the infiltrated material is retained. Other functional modification of the endothelial layer which has been documented often appear later. For example, in rabbits, the expression of the vascular cell adhesion molecule-1 (VCAM-1) by endothelial cells on lesions appeared after four days of severe hypercholesterolaemia and foam cell formation (Li *et al.*, 1993). On the other hand, aggregation and retention of lipoproteins were observed within hours of the onset of hypercholesterolaemia. This lead to the conclusion the endothelial changes during atherosclerosis such as expression of cellular adhesion molecules (CAM) is likely to be a consequence of the lipoprotein retained initially within the arterial wall (Williams and Tabas, 1998).

The link between blood flow and atherogenesis in the response-to- injury hypothesis prompted the studies on the effect of shear stress on cultured endothelial cells. Many alterations have been reported *in vitro* (Williams and Tabas, 1998), such as increased endothelial cell division, production of growth factors and adhesion molecules. Findings from *in vitro* studies support the contribution of shear stress to alterations observed in the endothelium in atherogenesis. *In vivo* studies, however, demonstrated that endothelial alterations induced by shear stress are not enough by themselves for the initiation of atherosclerosis. Possibly, stress-induced endothelial changes can contribute to the pathogenic process of atherosclerosis. The most relevant changes identified at pre-lesional sites are altered proteoglycan structure and retention of lipoprotein (Keaney Jr, 2000). The retention of lipoprotein appears to be closely associated with the constituents of the extracellular matrix. The apolipoprotein B-100 (apoB-100) component of LDL is retained within the arterial wall with tight linkage to proteoglycans that promote lipoprotein retention (Williams and Tabas, 1998). Lipoprotein oxidation has also been proposed to be central to atherogenesis. The discoveries that modified LDL was toxic to endothelial cells (Hessler *et al.*, 1983) and that ox-LDL is taken up faster by macrophages (Steinbrecher *et al.*, 1984) lead to **"the oxidative modification hypothesis"**, which proposed that LDL enters the intima, becomes oxidised and can cause endothelial dysfunction, inflammation and foam cell formation (Steinberg *et al.*, 1989). The hypothesis is discussed in detail in (section 1.4.1). All the hypotheses have attempted to explain the complex nature of the cause of atherosclerosis focusing on a critical initiating event. A common feature is the deposition of lipids, particularly LDL. The oxidative hypothesis, in addition, showed the importance of oxidative events in the genesis of the disease.

1.2.3.2 Lesion initiation, inflammatory response and foam cell formation.

Studies with the cultured cells have suggested the ways in which the disease process begins and progresses. The major cause of atherosclerosis is believed to be the accumulation of lipoprotein in the subendothelial intima (Tabas *et al.*, 2007). The accumulation of LDL, oxLDL, and other forms of modified LDL is believed to be the primary event of initiation. The build-up of oxLDL contributes to the infiltration of macrophages and formation of foam cells in the intima. Lipoprotein with an additional apolipoprotein (a) namely lipoprotein (a) (Lp(a)) has also been confirmed to be atherogenic (Morrisett, 2000). Studies with cultured cells have also suggested the initiating role of endothelial cells in mediating inflammation. oxLDL is toxic to the cells and hence induce endothelial damage.

The toxicity of oxLDL elicits a compensatory response from the immune system. The endothelial cells (ECs) subsequently become pro-inflammatory cells, thereby increasing the endothelial adherence of leukocytes and platelets through cell adhesion molecules (CAM),

such as selectins, and the production of growth factors, such as cytokines. The signalling proteins (cytokines and chemokines) trigger the movement of lymphocytes and monocytes to the site of activation on the surface of ECs. Chemokines play a major role in movement into the subendothelial region after binding to CAM on the surface. Vascular cell adhesion molecule 1 (VCAM-1) is specific for binding monocytes and T lymphocytes found in the newly developing lesion. Chemokines such as interferon (IFN γ), macrophage colony-stimulating factor (M-CSF), and monocyte chemoattractant protein-1 (MCP-1) are commonly found in the atherosclerotic lesion (Libby, 2002, McLaren *et al.*, 2011, Ilhan and Kalkanli, 2015). The role of CAM and growth factors in atherosclerosis has been shown by studies mice lacking selectins (Dong, 1998, Collins, 2000) and MCP-1 (Gu, 1998, Gosling *et al.*, 1999, Ohman *et al.*, 2010). There was a reduction in atherosclerotic lesions of apoE deficient mice lacking both intercellular adhesion molecule (ICAM)-1 and P-selectin (Collins, 2000).

Macrophages do not take up native LDL enough to generate foam cells but oxLDL (Goldstein *et al.*, 1979a). Foam cells are formed from the uptake of massive amount of oxLDL. The mechanisms of oxidation of LDL and its relation to atherosclerosis is discussed in detail in section 1.4. In the event of atherosclerosis, lymphocytes and monocytes are recruited to the arterial wall. The importance of inflammation in atherosclerosis has been confirmed clinically and experimentally. Many lines of evidence support the important role of the inflammatory cells especially monocyte derived macrophage (Lusis, 2000, Wang *et al.*, 2012, Ilhan and Kalkanli, 2015) due to the fact that cholesterol-enriched macrophages are a major indication of the early atherosclerotic lesion. The link between atherosclerosis and inflammation was recently reaffirmed by a large human trial (CANTOS trial) by Ridker et al which demonstrated that anti-inflammatory therapy, canakinumab significantly reduced incidents of cardiovascular disease without any influence on the level of lipids compared to placebo (Ridker *et al.*, 2017).

However, the role of oxidised LDL as a mediator of inflammation in macrophages hence promoting atherosclerotic events (Schwarz *et al.*, 2017, Lara-Guzmán *et al.*, 2018) has remained of interest to researchers.

1.2.3.4 Atherosclerotic lesion macrophages

As described above, under atherosclerotic conditions, monocytes are recruited into the arterial wall. They differentiate into macrophages in a process regulated by specific cytokines with M-CSF as the most common growth factor involved (Waldo *et al.*, 2008). Macrophages are phagocytic cells found in nearly all tissues, which function in the removal of cellular debris, clearance of necrotic cells and apoptotic cells. They originate from myeloid progenitor cells that produce the neutrophils (another phagocytic cell) and dendritic cells involved in antigen presentation (Mosser and Edwards, 2008). Macrophages play a key role in all the stages of lesion development and progression. They are the major type of cells present in early lesions, contribute to the progression of the lesion through the rapid uptake of oxidised or modified LDL and the remnants of lipoproteins leading to the formation of foam cells (Lusis, 2000, Mosser and Edwards, 2008).

Modified LDL is taken up through receptors that recognize a wide range of ligand (scavenger receptors). The scavenger receptors are regulated by cytokines such as tumour necrosis factor (TNF- α) and interferon (IFN- Υ) and region for oxidised fatty acids. The peroxisome-proliferator activated receptor (PPAR), a transcription factor containing a binding discovery of scavenger receptors confirmed the cells as the starting point for the formation of fatty streak, an indication of the onset of the disease and the centre for development of plaque (Libby, 2002, McLaren *et al.*, 2011). The emergence of atherosclerotic lesion is largely dependent on monocyte-derived macrophages, with macrophages as the most abundant type of cells present

in atherosclerotic plaque, they possess the ability to contribute to the progression of atherosclerosis (Ilhan and Kalkanli, 2015). Due to the great influence they have on inflammation, cholesterol accumulation and transformation into lipid-laden foam cells, formation of necrotic core and degradation of the ECM.

1.2.3.5 Formation of fibrous plaques.

Vascular smooth muscle cells (VSMCs) contribute to the development of the fatty streaks into a more advanced form (fibrous plaques), this is usually covered with a fibrous cap. The core of the fibrous cap is made up of mainly cholesterol and its esters (Lusis, 2000), modified LDL and cells that are prone to apoptosis and necrosis due to their inability to obtain enough nutrient for their survival (McLaren *et al.*, 2011). The SMCs contribute to the growth of the plaque through the formation of the fibrous cap and the production of extracellular matrix (ECM) (Wang *et al.*, 2012). SMCs facilitate the migration of lymphocytes and monocytes by producing adhesion molecule (VCAM-1). Growth factors and cytokines such as plateletderived growth factor (PDGF), MCP-1, transforming growth factor- β (TGF- β) produced by T cells and macrophages are important in the activation of leukocytes, the production of ECM, promotion of endothelial dysfunction and enhancing the proliferation of SMCs (Libby, 2002)

Recent studies have demonstrated the important role of CD40 (protein present on antigen presenting cells) expressed by T cells and macrophages in promoting advanced lesions. The CD40 interacts with its ligand to promote the production of proteases, which can degrade the matrix, adhesion molecules and inflammatory cytokine (Lusis, 2000, McLaren *et al.*, 2011, Wang *et al.*, 2012). Although the interaction was first seen in T and B cells, it has also been demonstrated in ECs and SMCs (Lusis, 2000). The formation of the fibrous cap is highly

dependent on the migration and proliferation of SMCs, which is in turn regulated by factors that are produced by T cells, endothelial cells and resident macrophages.

1.2.3.6 Plaque progression and vulnerability

It has been known for about four decades that plaque rupture is not only a determinant of clinical events but also a factor for the progression of the plaque. The association of angina and MI with plaque rupture in arteries with reduced blood flow was demonstrated by Davies and Thomas; the studies showed that the clinical manifestation was not only due to the narrowing of the lumen but also the change in morphology of the plaque. Pathological studies revealed that clinical events are triggered mainly by the vulnerability of the plague and its composition rather than the event of stenosis (narrowing of the blood vessel) (Davies and Thomas, 1985, Libby, 2002).

Advanced atherosclerotic plaques are classified as stable or vulnerable. As the plaque continues to build up, sudden rupture or break off can occur and the plaque becomes unstable. The plaques that are likely to become unstable and prone to present symptoms are termed vulnerable plaques. The progression of atherosclerosis later present symptoms which occur when the coronary blood flow is decreased to the point that the metabolic needs of the organs cannot be met. Restriction of blood flow to the heart or the brain may be due to blood thrombus formation (thrombosis) that can obstruct blood follow. The disturbance of blood flow initially presents as angina (pain from the heart) described as pressure, pain or squeezing arising from the heart. This can be stable for several years but the sudden rupture of the plaque brings about acute coronary clinical events of unstable angina, heart attack or myocardial infarction(MI) meaning (death of the heart muscle) (Arroyo and Lee, 1999). An acute event, such as thrombotic stroke, can occur when there is reduced blood flow to the brain.

Stable plaque consists of a thick fibrous cap, a small lipid core with few inflammatory cells, as opposed to the vulnerable plaques that are characterised with a thin fibrous cap, considerably large lipid core and numerous macrophages and T cells (Arroyo and Lee, 1999, Wang and Bennett, 2012). The stability of the plaque is dependent on the thickness of the fibrous cap, whose structure is maintained by matrix synthesis and degradation. Low synthesis and increased degradation of the matrix lead to a weakening of the fibrous cap, both of these processes are likely to be influenced by inflammatory cells (Arroyo and Lee, 1999). For example, macrophages are the main cells in lesions that produce matrix metalloproteinase (MMP). The MMPs are zinc dependent enzymes that degrade the ECM. T cells produce interferon γ (IFN- γ) which inhibits the synthesis of matrix by SMCs. ECM components are mainly produced by SMCs, hence the migration and proliferation of SMCs influence the stability of the plaque (Lusis, 2000, McLaren *et al.*, 2011).

The intima of advanced atherosclerotic lesions becomes mineralized as their complexity increases. Mineralization of the lipid core is regulated by enzymes involved in regulation of calcification and bone tissue formation (ossification) (Wang et al., 2012). Calcification and neovascularisation (formation of new blood cells), can also influence the stability of atherosclerotic lesion (Lusis, 2000). The clinical consequence such as heart attack and stroke that occur in atherosclerosis are not usually due to the blockage of the lumen by the build-up of the advanced plaque but a manifestation of an event arising from plaque rupture or endothelial erosion and the initiation of thrombosis. The progression of atherosclerotic lesions from onset to well-advanced form is depicted in Fig. 1.1. Maintaining the stability of advanced plaques, hence preventing rupture seems a plausible way of preventing the occurrence of acute clinical events (Lusis, 2000, Halvorsen *et al.*, 2008). The many factors involved in the destabilisation of the plaque makes this challenging.


Figure 1.1: Schematic representation of the progression of atherosclerotic lesion.

The normal human coronary artery is characterised by three layers. The development of fatty streak arises from the deposition of lipids (Apo B-LP) mainly oxLDL which incites the recruitment of inflammatory cells. The accumulation of the lipids and the inflammatory cells (mainly lipid-laden macrophages) leads to a formation of a lipid core. As the lesion progresses, SMCs and T-cells infiltrate the intima leading to the formation of established lesions. The SMCs migration and proliferation leads to the formation of a fibrous cap. The vulnerable plaques (with thinning fibrous cap) results in a formation of necrotic core from the accumulated apoptotic cells. The thinning of the fibrous cap leads to the formation of unstable plaques susceptible to rupture, hence the development of a thrombus, which eventually brings about the acute coronary clinical event such as unstable angina, heart attack. Adapted from (Moore and Tabas, 2011).

1.2.4 Cell death in atherosclerosis

Apoptotic cell death has been well demonstrated in atherosclerotic plaque. The consequence of apoptosis is however dependent on the cells involved, where the cells are located within the plaque and stage of progression on the atherosclerotic lesions. Apoptosis of macrophages might possibly be beneficial in early lesions, but not in advanced lesions if the apoptotic bodies cannot be phagocytosed and cause inflammation (Tabas, 2005). Smooth muscle cells and macrophages have been shown to go through apoptosis in plaques (Boyle, 1999, Akishima *et al.*, 2005).

Over the years several studies have emphasised the importance of apoptosis in contributing to plaque stability and the progression of atherosclerosis (Kockx and Herman, 2000, Tabas, 2005, Martinet et al., 2011). Factors such as hypoxia, oxidative stress, cholesterol overload and IFN- γ have been implicated in instigating apoptosis in atherosclerosis (Mallat and Tedgui, 2000, Van Vre *et al.*, 2012). The increased turnover of EC in atherosclerotic plaque is linked to enhanced apoptosis. Loss of SMC can contribute to the weakening of fibrous cap and the disappearance of SMCs in plaque has been attributed to apoptosis. Apoptotic macrophages are mainly present in cellular regions which are rich in macrophages which synthesise DNA (Kockx and Herman, 2000). Macrophages represent a greater number of cell deaths occurring in atherosclerotic lesions compared to leukocytes, SMCs and ECs, making up over forty percent of cell deaths (Kolodgie *et al.*, 2000).

1.2.5 Epidemiology and risk factors

Atherosclerosis affects people with certain risk factors more than others. Over the years, many risk factors have been identified as a predictive factor for the development of atherosclerotic cardiovascular disease (Homma, 2004). Epidemiological studies over five decades have focused on the associated risk factors, largely grouped as genetic and environmental factors. Genetic predisposition to the disease can commonly be associated with the family history of CVD. The importance of genetic factors has been emphasised over the years. Asides the impact of genetic variations involved in lipid metabolism, other genes that

predispose to other risks factors of atherosclerosis are equally considered of importance (Indolfi, 2002). The heritability of coronary disease and the importance of genetic factors have been suggested by studies in twins (Marenberg *et al.*, 1994, Zdravkovic *et al.*, 2002).

Some of the factors that are strongly correlated to atherosclerosis are non-modifiable. Nonmodifiable factors such as age and genetic factors e.g. familial hypercholesterolaemia cannot be changed. Modifiable factors such as diabetes, smoking, hyperlipidaemia and hypertension can change depending on certain lifestyle changes. It is compelling to note that most risk factors for atherosclerosis are risk factors for heart failure. The involvement of some of the risk factors as predictive factors for the development of CVD is described below:

1.2.5.1 Smoking

The risk of developing CVD from smoking is dependent on the level of smoking in terms of the number of cigarettes smoked per day and the age at which the person began smoking. The link between smoking and heart diseases dates as far back as studies which clearly linked smoking to the incidence of heart attack (Doll and Hill, 1956, Hammond and Horn, 1958). Evidence from in vivo and *in vitro* studies demonstrated the ability of cigarette smoke to induce LDL oxidation (Yokode *et al.*, 1996, Yamaguchi *et al.*, 2001). Cessation of smoking leads to a reduction of risk of a heart attack in smokers, in two years the risk status of ex-smokers was near that of nonsmokers(Gaziano, 1996). Mons et al showed increased association of cardiovascular mortality in current smokers compared to former smokers and never smokers with smoking cessation having a beneficial effect in reducing the risk of cardiovascular mortality (Mons *et al.*, 2015). Smoking is often classified as an important risk factor for CVD

1.2.5.2 Dyslipidaemia and Hyperlipidaemia

The term hyperlipidaemia or hyperlipoproteinaemia is described as the presence of abnormal elevated concentration of lipids or lipoprotein in blood and is the most common form of dyslipidaemia (abnormal level of lipids in the blood). The abundance of lipoprotein in plasma is of importance in atherosclerosis, as high level of atherogenic lipoprotein serves as a precursor in most form of the disease. Individuals with an elevated level of LDL due to genetic inheritance or lifestyle have been confirmed to be a high risk of myocardial infarction(Goldstein and Brown, 1986).

The relationship between LDL-cholesterol and atherosclerosis is well established, lowering blood cholesterol leads to decrease incident of cardiovascular deaths in men with hypercholesterolemia (Shepherd *et al.*, 1995). Human studies on the lipid lowering drugs statins demonstrate a reduction in the incidence of CVD due to decreased LDL-cholesterol (Baigent *et al.*, 2010, Mihaylova *et al.*, 2012). A meta-analysis of large longitudinal cohort study showed the link between lipids and the risk of ischaemic heart disease (IHD) with LDL cholesterol having a direct association to IHD whereas HDL cholesterol was inversely related (Lewington *et al.*, 2007, Di Angelantonio *et al.*, 2009). LDL is clearly causal, but it is not clear if HDL is causally protective.

1.2.5.3 Diabetes

Diabetic patients are three to five times more at risk of coronary atherosclerosis in spite of controlling other confounding factors (Mazzone *et al.*, 2008). The oxidation and glycation of LDL in diabetic patients can lead to the formation of modified LDL that is taken up by macrophages and generates foam cells which are associated with the onset of atherosclerosis (Knott *et al.*, 2003). Other risks such as abnormal lipids and hypertension are commonly found in diabetic patients (Bierman, 1992). There is an increased lipid peroxides level and increased serum cholesterol in diabetics patients (Suryawanshi *et al.*, 2006), which might predispose them to more cardiovascular disease. Both type 1 and type 2 diabetes promotes atherosclerosis with the level of blood glucose as the causal factor (Chait and Bornfeldt,

2009). Hence, diabetes is a major contributor to atherosclerosis with most deaths of diabetic patients arising from atherosclerosis.

1.2.5.4 Hypertension

Hypertension is defined as diastolic and systolic blood pressure above 90mmHg and 140mmHg, respectively. It is a risk factor for CHD, stroke and other CVD (WHO, 1996). Reduction of blood pressure (systolic blood pressure (SBP) below 140mmHg and diastolic blood pressure (DBP) below 90mmHg) has been seen to reduce the onset of CVD (Turnbul, 2003). In 2002 a longitudinal cohort study demonstrated a direct association between DPB and SBP and both IHD and stroke (Lewington *et al.*, 2002). Hypertension might lead to thickening of the arterial wall or lesion fissuring. The benefit of treating hypertension was seen in both mild and severe cases at all ages within five years of treatment, the incidence of CHD and stroke was reduced to 14% and 40% respectively. The hypertension optimal treatment study demonstrated that aggressive antihypertensive therapy reduced the occurrence of a cardiovascular event in high-risk patients (Hansson *et al.*, 1998).

1.2.5.5 Age

The risk of CVD has been proven to increase with age, atherosclerosis is often regarded as a very good example of age-related disease. Atherosclerosis progresses with age, becomes more severe and universally observed more in older humans (Stout, 1987). The link is best observed when comparing the average risk of developing CVD at a certain age compared to increased age perhaps monitored over a period over ten years. Comparing the average risk of men at the age of 30-34 years in the US to men at the age of 60-64 years showed a seven fold increase (Wilson *et al.*, 1998). Examination of the aorta of elderly with magnetic resonance imaging (MRI) demonstrated that age might promote atherosclerosis (Chen *et al.*, 2013). Age is an important non-modifiable factor, as atherosclerosis is often regarded as part of the process of aging. Other factors such as diabetes and hypertension also increase with age.

1.2.5.6 Gender

Data from observational studies showed that males are at higher risk when compared to females of the same age. Both human and animal studies have documented the gender difference in the predisposition to CVD. The Framingham study by Kannel *et al.* showed that the rate at which males below the age of 60years developed CVD doubled that of women (Kannel *et al.*, 1976). Hayashi *et al* showed that the formation of atherosclerotic lesions in male rabbits was greater than that of female rabbits fed the same high cholesterol diet (Hayashi *et al.*, 1995). The incidence is seen to accelerate in postmenopausal women, supported by the work of Kannel *et al.*, which showed that the event of CVD was almost twice that of premenopausal women (Kannel *et al.*, 1976). It is controversial if sex hormones play a role in the development of CHD in women, as hormone replacement therapy did not protect against CVD in postmenopausal women (Manson *et al.*, 2003), although there is ample of evidence that oestrogen prevents atherosclerosis in animal models (Williams *et al.*, 1995). It is also evident that the deficiency of testosterone predisposes to atherosclerotic events (Oskui *et al.*, 2013). Over the last decade there is greater emphasis laid on the crucial role, gender might have on the occurrence of CVD (Spence and Pilote, 2015).

1.2.5.7 Obesity

Obesity is a term used to describe excess body weight. Humans are classified as obese based on their body mass index (BMI), A range between 18.5-24.9 kg/m² is considered healthy or normal while values above 30 kg/m² are considered obese (Kopelman, 2000). Obesity is regarded as the starting point for some other risk factors such as dyslipidaemia, hypertension, insulin resistance and glucose intolerance (Wilson *et al.*, 1999). A Study by McGill *et al* demonstrated that obesity is linked to increased atherosclerosis in young adult men (McGill *et* *al.*, 2002). The link between obesity and CVD is of growing concern due to the increase in the prevalence of obesity in developed society. Other studies have shown that depletion of visceral fat reduces risk factors associated with obesity such as impaired glucose metabolism, hypertension and dyslipidaemia hence preventing CVD (Shimano, 2009). Other factors such as endothelial dysfunction, inflammation, inflammasome activation, and adipokine imbalance have been proposed to link obesity to atherosclerosis (Lovren *et al.*, 2015).

1.2.5.8 Sedentary lifestyle and lack of exercise

Lack of physical exercise is a predisposing factor to CHD, aerobic exercise is associated with a lower risk of CVD. It has been revealed that individuals who expend more energy weekly are less likely to develop CHD (Thompson *et al.*, 2003). Exercising regularly promotes antiatherosclerotic properties such as increased HDL cholesterol concentration, less body fat, lower level of blood pressure and glucose, decreased LDL cholesterol and higher sensitivity to insulin (Assmann *et al.*, 1999). Large population studies showed that physical inactivity doubles the risk of CVD compared to physically active individuals (Powell *et al.*, 1987, Wannamethee *et al.*, 1998). Increased level of physical fitness is inversely correlated with CHD and starting an exercise regimen has been suggested to reduce the risk of CVD. (Paffenbarger *et al.*, 1993, Wannamethee *et al.*, 1998, Schroeder *et al.*, 2007).The possible mechanistic events that might enhance reduced susceptibility to CVD was suggested to include but not limited to the release of less inflammatory mediators from adipose tissue, decreased thrombotic events, stabilised vulnerable plaques and enhancement of the endothelial function (Bowles and Laughlin, 2011).

1.2.5.9 Other factors

Small dense LDL, which is one of the phenotypes of LDL known as pattern B with a high proportion of small dense LDL particles, has been demonstrated to be linked with increased CAD (Austin *et al.*, 1988, Krauss, 2010, Arai *et al.*, 2013). *In vitro* studies have shown that dense LDL had increased susceptibility to oxidation hence promotes increased atherogenic

effects (de Graaf *et al.*, 1991). There is evidence on the possible link between accumulation Lp(a), the variant form of LDL (which contains additional apo (a) attached to apoB-100) and the level of fibrinogen with increased events of CHD (Thompson *et al.*, 1995). The similarity in the sequence of its protein and that of plasminogen and relation to fibrinolysis suggest its important role in thrombosis (Gerald and Daphne, 2012). The evidence that Lp(a) is a cause of CAD was strengthened by Clarke *et al.*, who discovered an association between isoforms of Lp(a) and the development of CAD (Clarke *et al.*, 2009). Atherogenic properties of Lp(a) may be due to its similarity to LDL and the transport of cholesterol to areas with vessel injury and its high propensity to aggregate (Hajjar and Nachman, 1996).

Homocysteine produced as an intermediate product in the metabolism of sulphur containing proteins. Elevation of plasma homocysteine caused by homocystinuria is associated with atherosclerosis and premature thrombosis. Gerhard and Duell showed that an elevated homocysteine level is an independent risk factor for atherosclerosis. However, the contribution of homocysteine to the disease is still under investigation (Gerhard and Duell, 1999). Hyperhomocysteinaemia has been linked to endothelial dysfunction in humans (Tawakol *et al.*, 1997, Woo *et al.*, 1997). The clinical trial of homocysteine lowering showed no effect (Armitage *et al.*, 2010). Other studies have found high levels of homocysteine mediating vascular damage such as promoting mitogenesis of vascular smooth muscle cells (Tsai *et al.*, 1996), causing damage to endothelial cells (Starkebaum and Harlan, 1986), increased platelet aggregation (Durand *et al.*, 1997).

C-reactive protein (CRP) and fibrinogen have also been identified as a novel risk factor for atherosclerosis (Ridker *et al.*, 2001). Inflammation which regulates the acute phase proteins

such as serum amyloid A, fibrinogen, and CRP, have a vital role in atherosclerosis (Gabay and Kushner, 1999, Libby *et al.*, 2002). Some population and clinical studies have demonstrated the association between these inflammatory mediators and increased cardiovascular events (Ridker *et al.*, 1998, Koenig *et al.*, 1999). The relevance of the functional effects of CRP as a possible causal factor in atherosclerosis was discussed by Paffen and deMatt (Paffen and deMaat, 2006)

These factors can work independently or together to increase susceptibility to the risk of atherosclerosis. The interactions between the associated risk factors make an investigation into causal factors difficult, hence most common forms of CHD occur as a result of genetic susceptibility, aging, and environmental factors. Of note is the important link between LDL and most forms of the disease.

1.3 Lipoproteins and Atherosclerosis

1.3.1 Introduction to lipoproteins

Lipoproteins are formed from the combination of lipids and protein. Lipids are insoluble in aqueous solution. The combination with protein is necessary for the transportation of lipid through the blood stream to tissues. In general, lipoproteins are spherical particles in which the outer component is made up of polar proteins, phospholipids and non-esterified cholesterol (free cholesterol). This surrounds nonpolar molecules triacylglycerols (TG), free cholesterol and cholesteryl esters. The protein parts are often called apolipoproteins, which aside from their structural roles facilitate the cellular uptake of lipoproteins by serving as ligands for the lipoprotein receptors.

Lipoproteins differ in their composition, size, electrophoretic mobility and density. They are classified into chylomicrons, very low density lipoproteins (VLDL), low density lipoprotein (LDL) and high density lipoproteins (HDL). VLDL is synthesized in the liver and transport TG produced endogenously in the liver to peripheral tissues for energy needs, they contain apoB, apoC-II and apoE (apoC-II and apoE are acquired after secretion). LDL is mostly derived from delipidated VLDL when it forms VLDL remnants that can be degraded further to form LDL, but a small portion is directly released from the liver. It transfers cholesterol from the liver to peripheral tissues. It contains only ApoB-100 proteins. HDL is the main transport form of cholesterol from peripheral tissues to the liver. They contain apoA1 with apoA2, apoC and apoE (Goldstein and Brown, 1977, Gerald and Daphne, 2012).

The function of HDL and LDL in cholesterol transport gives them clinical significance in atherosclerosis, the major component of arterial lesion is cholesterol-enriched cells. Elevated level of serum HDL is negatively correlated with incidence of CVD. The HDL is said to be protective and anti-atherogenic in nature, although this has not been proven. However, LDL is positively correlated with the incidence of CVD and pro atherogenic in nature (Gerald and Daphne, 2012).

1.3.2 Lipid Metabolism

1.3.2.1 Endogenous lipid pathway

Lipoproteins can be synthesised from triglycerides and cholesterol in the hepatocytes. In the liver triglycerides and cholesterol are attached to apo B to form VLDL. It is formed through lipidation of ApoB-100, these VLDL rich in triglyceride enters the plasma to deliver triacylglycerol to muscle and adipose tissue through lipoprotein lipase activity (Feingold and Grunfeld, 2000). VLDL is degraded by lipoprotein lipase found on endothelial cells to form VLDL remnants or intermediate density lipoprotein (IDL) leading to the detachment

of Apo C while ApoE remains attached to the IDL particle (Murdoch and Breckenridge, 1996). The IDL in circulation can be taken up by the liver through the interaction of ApoE/ApoB-100 with the IDL receptor (Mahley and Ji, 1999) or can be further degraded to form triglycerides and fatty acids by hepatic lipase (Zambon *et al.*, 2003) leaving the IDL remnant known as LDL behind. The metabolism of LDL is discussed in section 1.3.4.

1.3.2.2 Exogenous lipid pathway

The exogenous lipid pathway begins in the intestine where lipase hydrolyses dietary triglycerides into monoacylglycerol and free fatty acids emulsified with cholesterol, fat soluble vitamins, plant sterol and bile acids to form micelles which are then transported into the intestinal cells. The uptake of plant sterol and cholesterol is mediated by Niemann-Pick C1-like protein (Feingold and Grunfeld, 2000). The enterocytes convert the fatty acids and monoacylglycerol back to triglycerides. ApoB-48 and triglycerides are incorporated into chylomicrons through microsomal transfer protein (MTP) (Hussain, 2014). Chylomicrons bypass the liver circulation entering the blood stream via the lymph where they acquire Apo C's and Apo E from HDL(Ramasamy, 2013). ApoC2 activates lipoprotein lipase present in the endothelium of capillaries, which removes triglycerides from the core to produce a chylomicron remnant. The chylomicron which passes through the capillary endothelium in the liver and then taken up through the binding of ApoE to the LDL receptor family.

1.3.2.3 Reverse cholesterol transport

HDL mediates the transportation of cholesterol from extrahepatic tissues back to the liver for it to be excreted in bile, this process is known as reverse cholesterol transport. The formation of nascent HDL occurs in the hepatocytes and enterocytes through the addition of free cholesterol and phospholipids to ApoA1 by the ATP-binding cassette transporter A1 (ABCA1) (Kiss *et al.*, 2003, Hussain, 2014). The phospholipid transfer protein transfers phospholipid to

HDL from chylomicrons when they become hydrolysed to chylomicron remnants (Huuskonen *et al.*, 2001). The phospholipid-rich HDL becomes matured by ABCA1 adding free cholesterol to them (Ge *et al.*, 2013). Most of the cholesterol is then converted to cholesteryl ester by lecithin-cholesterol acyltransferase (LCAT) bound to HDL and the esterification causes them to relocate to the core of the HDL particle. The cholesteryl ester transfer protein (CETP) exchanges the esters with triglycerides derived from other classes of lipoprotein (Chajek and Fielding, 1978). Alternatively, the cholesteryl esters and cholesterol can be released in the liver by binding scavenger receptor class B1 (SR-B1) (Leiva *et al.*, 2011).

1.3.3 Low density lipoprotein (LDL)

LDL is a small particle, which may vary in diameter in diameter between 20-25 nm and molecular weight of about 2 to 3.5 million Daltons. The density ranges between 1.019 g/ml – 1.063 g/ml, the variation in size is due to the extent of metabolism by hepatic lipase (Goldstein, 1977). Generally, the LDL particle (Fig 1.2) is made up of 75% lipids, mainly cholesteryl esters and 25% protein. It has an outer monolayer made up of phospholipid, free cholesterol and ApoB-100 protein (Steinberg, 2002). It contains one molecule of Apo B and on average about 185 molecules of TG, 700 molecules of phospholipids (PL), mainly phosphatidylcholine, 1600 molecules of cholesteryl esters and 600 molecules of non-esterified cholesterol (free cholesterol) (Steinberg, 1997b). It also contains a certain amount of antioxidants, such as α - and γ -tocopherol, β -carotene and lycopene (Jessup *et al.*, 2004, Siess, 2006). LDL has long been identified as a causal factor in atherosclerosis. Elevated LDL level in plasma was first recognised to be strongly associated with atherosclerosis by Gofman and colleagues in 1950 (Gofman *et al.*, 1950).



Figure 1.2 Structure of LDL

Source: (Rausch and Kortleever, 2011)

1.3.4 LDL metabolism

The uptake of LDL is dependent on LDL receptor-mediated endocytosis. The LDL receptor was discovered by Brown *et al.* in 1973 from the study on cultured human cells (Brown *et al.*, 1973). The most abundant LDL receptors are present in the hepatic cells, but they are present in most other cells as well. The receptors are located in clathrin-coated pits in specialized regions. The binding of LDL to its receptor is through the apoB-100 region of the molecule. The process by which LDL binds its receptor and LDL cholesterol uptake in cells is a highly regulated event. The apo-B-100 protein binds to the receptor, the LDL receptor complex is internalized via endocytosis. The endocytotic vesicle formed fuses with a lysosome. The content of LDL along with its apoB-100 content are hydrolyzed by enzymes present in

lysosomes (hydrolases), to form free cholesterol and amino acids. The LDL receptor is recycled back to the surface of the cell where it can bind more LDL particles (Brown *et al.*, 1973, Goldstein and Brown, 1977, Goldstein and Brown, 1979).

1.3.5 Role of LDL in atherosclerosis

Lipoproteins, particularly LDL are a common feature of the hypothesis on atherogenesis. Individuals with high concentrations of LDL due to genetic inheritance or lifestyle are more susceptible to heart attack. Genetic, epidemiological and clinical studies have provided some evidence in support of the association between LDL and atherosclerosis. The most convincing link of LDL to was seen in familial hypercholesterolaemia (FH). This is a genetic disease and if homozygous and untreated, the patient's death from coronary atherosclerosis occurs in the second decade of life due to the accumulation of LDL arising from the absence of the LDL receptor (Goldstein and Brown, 1973).

Autosomal dominant hypercholesterolaemia (ADH) is also a genetic disease associated with increased levels of LDL cholesterol and CVD. The mutation of three genes (ApoB, LDL receptor and proprotein convertase subtilisin-kexin type 9 (PCSK9)) is associated with ADH (Marduel *et al.*, 2010). PCSK9 is important in the regulation of LDL-cholesterol, it interacts with LDL receptors in the liver and targets the LDL receptors for degradation in lysosomes and regulates LDL concentration in plasma. Inactivation of PCSK9 extends the life span of the LDL receptors and leads to decreased LDL cholesterol plasma concentration and reduced CHD (Farnier, 2014, Tavori *et al.*, 2015, Shapiro and Fazio, 2017). The effect of PCSK9 on LDL receptor led to the development of active drugs to lower plasma LDL cholesterol thereby reducing CVD.

Several epidemiological studies have supported the proatherogenic role of the Apo B-100 containing lipoprotein (Kannel et al., 1984) and the fact that clinical studies have shown that lipid-lowering drugs, such as statin therapy, decrease the risk of CVD is an important indication of the association between LDL and CVD (Serruys et al., 2002, Baigent et al., 2010). It was anticipated that LDL will show an atherogenic effect *in vitro* because of its role as a major cholesterol transporter to tissues and the fact that early lesion of atherosclerosis is characterised by cholesterol-laden macrophages. This was opposed by the failure of LDL to increase cholesterol accumulation in macrophages despite the high concentration of LDL incubated with the cells (Goldstein et al., 1979b, Fogelman et al., 1980). In vitro studies demonstrated that macrophage uptake of LDL is too low to cause the sufficient accumulation found *in vivo* and these led to the suggestion that LDL may be modified *in vivo* to a form that is rapidly taken up through a different receptor (Goldstein and Brown, 1979). The down regulation of the LDL receptor by the increased intracellular amount of cholesterol via the sterol regulatory element binding protein 2 (SREBP2) might be the reason to the inability to increase the cholesterol level of macrophages by way of LDL receptor (Steinberg *et al.*, 1989, Steinberg, 1997a). The LDL is likely to undergo changes that can affect its usual pathways to cells and alter their interaction with surface receptors on the cells (Chait, 1987).

1.4 Oxidative Modification of LDL and Atherosclerosis

1.4.1 LDL modification hypothesis

The hypothesis originated in 1979 when Goldstein and Brown discovered that patients with homologous FH who lack functional LDL receptors were still able to accumulate foam cells like hypercholesterolaemic patients that possess normal LDL receptor (Goldstein *et al.*, 1979b). They hypothesized that LDL might be altered prior to the uptake. To test the hypothesis, they

tried modifying LDL to identify the forms of LDL that will lead to accumulation of cholesterol in macrophages. Acetylated LDL (acetyl-LDL) provided higher uptake of LDL and the uptake was mediated by receptors identified and characterised as scavenger receptor A (SRA) in macrophages (Brown and Goldstein, 1983). Henriksen *et al.*, discovered that incubation of endothelial cells with LDL for 24 hours led to modification of LDL (Henriksen *et al.*, 1981) to a form that was recognized by peritoneal macrophages via the acetylated receptor similar to that which was previously described by Goldstein and colleagues (Goldstein *et al.*, 1979b). Later studies showed that LDL was oxidatively modified by endothelial cells and other cell types (Steinbrecher *et al.*, 1984, Witztum, 1993) and these findings led to the proposal of the LDL modification hypothesis.

The LDL oxidation hypothesis proposed that oxidative modification is involved in the onset of atherosclerosis and also contributes to the progression of the disease. The presence of oxLDL in circulation is a known marker of atherosclerosis, hence a risk factor for the development of CVD (Stocker and Keaney, 2004). There is compelling evidence that atherosclerotic lesions of both animal models and humans contain oxLDL. Small amounts of oxidised LDL can be observed immunologically in plasma, using specific monoclonal antibodies such as FOH1a/DLH3 that can bind to the epitopes of oxidised lipids bound to apo B-100 (Itabe *et al.*, 1994). The amount of detectable oxLDL was significantly elevated in diabetes, renal and coronary heart disease (Itabe, 2003, Tsimikas, 2006, Maiolino *et al.*, 2013). The use of antibodies staining against oxLDL has demonstrated the presence of oxLDL in atherosclerotic lesions and not in the arterial wall in hypercholesterolaemic rabbits (Boyd *et al.*, 1989), humans (Ylä-Herttuala *et al.*, 1989).

Evidence of oxLDL as the main culprit in atherogenesis, rather than native LDL has been well demonstrated in several studies which showed that modified forms of LDL such as oxidised and acetylated form are insatiably taken up by macrophages through the scavenger receptors. The principal receptors responsible for the binding and uptake of oxLDL are the scavenger receptor class A and CD36 (Collot-Teixeira et al., 2007). Unlike the LDL receptor, the scavenger receptors are not downregulated by the presence of oxLDL hence the substantial accumulation of cholesterol in macrophages leading to the formation of foam cells (Witztum and Steinberg, 1991, Yoshida et al., 1998). There are very strong line of evidence, that the in vivo modification of LDL increases its ability to induce inflammatory response and promote atherosclerosis, this is supported by the accumulation of oxidised LDL (oxLDL) in animal models with atherosclerosis (Steinberg et al., 1989, Harats et al., 2000) and the presence of oxidised lipids in both chemical and immunological analysis of atherosclerotic plaques in humans (Kritharides et al., 1998). LDL oxidation leads to the particles of LDL having immunogenic epitopes promoting the formation of antibodies against them, these antibodies have been found in many advanced atherosclerotic lesions of patients (Erkkila et al., 2000, Inoue et al., 2001). The main indication of this hypothesis is peroxidation of LDL promotes atherosclerosis.

1.4.2 Proatherogenic effects of oxidised LDL

Many lines of evidence support the role of oxLDL in foam cell formation and lesion advancement. There are strong indications, that the *in vivo* modification of LDL increases its ability to induce inflammatory response and promote atherosclerosis, this is supported by the accumulation of oxidised LDL (oxLDL) in animal models with atherosclerosis (Ylä-Herttuala *et al.*, 1989) and the presence of oxLDL oxidised lipids in both chemical and immunological analysis of atherosclerotic plaques in humans (Kritharides *et al.*, 1998). They induce

inflammatory responses by activating genes for the synthesis of cellular adhesion molecules and chemotactic proteins (Boullier *et al.*, 2001). Extensively oxidised LDL can also inhibit the outward movement of macrophages from the arteries (Quinn *et al.*, 1987), hence they remain in the arteries and further accumulate or undergo apoptosis in them (Tabas, 2005).

OxLDLs are cytotoxic to many cells, Chisolm *et al.* revealed that oxidised LDL was cytotoxic to the cells of the arterial wall, and thereby proposed that oxidised LDL may be significant for atherogenesis (Hessler et al., 1979). The ECs are susceptible to LDL toxicity, the accumulation of oxLDL can thus lead to endothelial dysfunction (Pech et al., 1992) and promote a further inflammatory response. The ability of LDL to cause cell death or injure cells is a general result of oxidation irrespective of the mode of oxidation. The potency of OxLDL to kill cells is often attributed to the degree of oxidation, although moderately oxidised LDL can also kill cells (Morel et al., 1983, Hessler et al., 1983, Siow et al., 1999b). Two decades later Chisolm and colleagues identified the oxidised cholesterol content (mainly cholesterol hydroperoxides) as a major factor responsible for the injurious nature of oxidised LDL (Colles et al., 2001b). Katouah *et al* recently showed that oxidised LDL triggered cell death in human macrophages via upregulation of intracellular production of reactive oxygen species (ROS) and inhibition of the major catabolic enzymes glyceraldehyde 3-phosphate dehydrogenase (GAPDH) and aconitase (Katouah et al., 2015). Berliner and others reported that minimally oxidised LDL increased the binding of monocytes to endothelial cells through the upregulation of production of the differentiating factor macrophage-colony stimulating factor (M-CSF) and the chemokine monocyte chemotactic factor (MCP-1) (Cushing et al., 1990, Berliner et al., 1990). OxLDL upregulates the production of metalloproteinases, major factors in thrombosis due to their contribution to destabilising fibrous cap in advanced lesions (Rajavashisth et al., 1999). The association of oxLDL with peripheral, coronary, preclinical atherosclerosis and vulnerable

plaques has been demonstrated in humans (Ylä-Herttuala *et al.*, 1989). All this evidence put together signifies the potential importance of oxLDL as a biomarker for atherosclerosis and its contribution to atherogenesis.

Despite the presence of small amounts of oxidatively modified LDL in plasma and its isolation from the atherosclerotic lesion of animals and man and its proatherogenic nature, the mechanism by which LDL is oxidatively modified *in vivo* is still largely debated. Most studies on the process are done *in vitro*. Understanding how LDL is oxidised is critical in inhibiting the process and its contribution to the onset and progression of the disease.

1.4.3 Oxidation of LDL

Oxidised LDL has been broadly classified into minimally oxidised LDL (MM-LDL) and fully or extensively modified LDL (oxLDL). The MM-LDL is less modified and can still be recognized by the LDL receptor. However, it is capable of increasing M-CSF and MCP-1 release by endothelial cells (Liao *et al.*, 1991). The extensively oxidised OxLDL has lost its recognition by the LDL receptor but can be recognized by scavenger receptors such as SRAI/II and CD36 on vascular cells (Levitan *et al.*, 2010). MM-LDL and OxLDL differ from each other in terms of what they contain and their effect in the body, both types differ from native LDL. Modification of LDL includes aggregation, lipolysis, proteolysis and oxidation. LDL is readily denatured, aggregated and easily prone to oxidation. LDL can be extensively aggregated by vortexing for few seconds (Khoo *et al.*, 1988)

The oxidation of LDL is often complex as both the protein and the lipid component can be modified. It was originally described as LDL containing modified protein product. Evidence from previous research suggested the formation of hydroperoxides of proteins prior to the formation of lipid hydroperoxides and the modified protein might play a role in the oxidation of lipids (Gebicki *et al.*, 2000, Gieseg *et al.*, 2000). However, the evidence implicating oxLDL in atherosclerosis has been attributed mainly to the lipid component. LDL with oxidised protein without changes to its lipid component is yet to be identified whereas oxidation of LDL can occur with little or no change to the apoB-100 (Parthasarathy *et al.*, 2010). Hence oxLDL has been defined as LDL containing the products of lipid oxidation such as the peroxides and their catabolic products (Levitan *et al.*, 2010). A recent review by Davies, however, suggests that depending on the concentration and reactivity protein is the main target for biological oxidants although this has been linked to the pathology of some human diseases the link to causality of the diseases is yet to be confirmed (Davies, 2016).

A number of lipid peroxidation products arise from the peroxidation of lipids (Esterbauer et al., 1992). Lipid peroxidation is the process whereby initiators of oxidation or radicals abstract an electron from lipids. Radicals are highly reactive chemical species that can exist independently and possess one or more lone pair electron (Aruoma, 1999, Sparrow *et al.*, 1992). The lipid peroxidation reaction occurs in three major overlapping steps: initiation, propagation, and termination steps. In the first step, a lipid radical is produced. At the propagation stage, the unstable lipid radical produced readily reacts with molecular oxygen to create a peroxyl-lipid radical, another unstable species that reacts with another lipid to yield a lipid radical and lipid-peroxide. The reaction proceeds as the next lipid radical reacts in a similar manner. The reaction of the radical with a non-radical continues to yield another radical (chain reaction mechanism). The termination reaction occurs when radicals react with each other to form a non-radical (Schafer *et al.*, 2000).

L-H + oxidant⁻ \longrightarrow L⁻ + oxidant-H (initiation)

$L^{\cdot} + O_2 \longrightarrow LOO^{\cdot}$

$LOO' + LH \longrightarrow LOOH + L'$ (propagation)

L' + L' ---- Non radical products (NRPs) (termination)

$L^{\cdot} + LOO^{\cdot} \longrightarrow NRPs$

Equation adopted from Schafer *et al*, (Schafer *et al*., 2000) L-H = lipids, L[·] = lipid radical, LOOH=hydroperoxide, LOO[·] = peroxyl radical



Fig 1.3: Schematic representation of peroxidation reaction in polyunsaturated fatty acid

Diagram created with ChemSketch software. Modified from (Young and McEneny, 2001)

The major component susceptible to oxidation is the polyunsaturated fatty acids (PUFA) of the lipids present in LDL (the mechanism involved is illustrated in figure 1.3). PUFAs are more susceptible to oxidation because they possess multiple bonds. The autoxidation of PUFA leads to the rearrangements of double bonds and the formation of conjugated dienes (a product containing two conjugated carbon double bonds) and their products. The carbon centred conjugated radicals have very short lifetimes. The rearrangement of bonds and reaction with molecular oxygen give rise to hydroperoxides the first stable products that are likely to be conjugated dienes. For example, the oxidation of linoleic acid leads to formation of well-known conjugated hydroperoxides, hydroxyoctadecanoic acids (HPODE) such as 9-HPODE and 13-HPODE which can further be transformed by isomerisation or chain cleavage to for further oxidation products or reacts with a lipid radical to promote the propagation step (Schneider, 2009). The chain reactions lead to formation of different products of oxidation.

Products of lipid modification include oxidised free fatty acids (FA), sphingolipid, and phospholipid products, oxysterols, cholesteryl ester oxidised on the fatty acyl moiety and short chain aldehydes. Further oxidation gives rise to hydroxenonenal (HNE) one of the most abundant aldehydes found in oxLDL (Levitan *et al.*, 2010). The amino acids present in lipoproteins can react with aldehydes to form adducts. HNE can readily react with cysteine and histidine residues (Uchida and Stadtman, 1992). Malondialdehyde (MDA) is also an important aldehyde product of lipid peroxidation and are able to react with lysine residues of apoB. MDA-modiodified LDL correlated with LDL-cholesterol in plasma of patients with CHD (Viigimaa *et al.*, 2010).

The oxidation of the core of rich in cholesteryl esters leads to formation of more lipid hydroperoxides which are further degraded into aldehydes (Leitinger, 2003). Cholesteryl ester hydroperoxides components have been suggested to enhance pro-inflammatory response in macrophages and identified as major constituents of LDL that are moderately oxidised (Harkewicz et al., 2008). Phospholipids within LDL are targets of lipid peroxidation from attack by oxidant or free radicals. The oxidation of phospholipids and the derived oxidation products depends on the type of oxidant involved in the initiation process, the length of fatty acyl group attached and the link that is present between the glycerol backbone and the fatty acyl group (Spickett et al., 2011, Reis and Spickett, 2012). Phosphatidylcholine derivative 1palmitoyl-2-(5-oxo-pentanoyl)-3-glycero-PC (POVPC) and other derivatives similar in structure were found in atherosclerotic plaques and minimally oxidised LDL (Watson, 1997, Hoff et al., 2003). The role of oxidised phospholipids products in atherosclerosis was reviewed by Lee and colleagues (Lee et al., 2012), they suggested that the effects of oxidised phospholipid could be attributed to different signalling pathways and gene regulation among other things. Oxidised phospholipid products obtained from phospholipid oxidation mediated by HOCL was shown to have anti proliferative effects, cytotoxic and increased levels of reactive oxygen species in endothelial cells (Robaszkiewicz et al., 2014). The oxidised phospholipids products can be detected by electrospray mass spectrometry which can detect oxidation products using the fragmentation patterns and can detect the phospholipid head groups, the chlorine and hydroperoxide contents of the oxidised phospholipids (Reis and Spickett, 2012). Ceroid is also a product of LDL oxidation that has been found in atherosclerotic lesion as a product of LDL oxidation. It contains complexes of protein and polymerized insoluble lipid (Mitchinson, 1982).

1.4.4 Reactive oxygen species and atherosclerosis

The possible role of reactive oxygen species (ROS) in atherogenesis has been examined over the last three or four decades and this has often been linked to the LDL oxidation theory. There is also evidence which suggests that certain risk factors of atherosclerosis, such as aging, smoking diabetes mellitus and arterial hypertension, leads to increased production of ROS from endothelial cells, adventitial cells and smooth muscle cells (Ohara et al., 1993, Gozin et al., 1998, Vogiatzi et al., 2009). The main origin of ROS and oxidants in atherosclerotic vessels are macrophages and smooth muscle cells. Several studies have demonstrated that oxLDL stimulates the production of ROS in macrophages, VSMCs and endothelial cells (Hsieh et al., 2001, van Aalst et al., 2004). It has also been suggested that ROS derived from macrophages are possible regulators of matrix metalloproteinase which can then lead to plaque instability (Rajagopalan et al., 1996). The role of oxidative stress has been evident in humans with observed significantly lower levels of malondialdehyde in healthy patients as compared to patients with myocardial infarction and unstable angina (Dubois-rande et al., 1994). Superoxide anion was suggested to participate in LDL oxidation in human monocytes and enhanced formation of cytotoxic modified LDL (Cathcart et al., 1989). High levels of extracellular and intracellular reactive nitrogen and oxygen species might play a vital role in development of atherosclerosis through vascular homeostasis.

1.4.5 Potential Mechanisms of LDL modification

Several mechanisms have been proposed for the oxidation of LDL. Most of the oxidative enzymes have been implicated in the oxidation by cells and also the non-enzymatic oxidation by transition metals and other catalysts. *In vitro* oxidations of LDL have been studied extensively using peroxidative enzymes and redox active transition metals (Yoshida and Kisugi, 2010).

1.4.5.1 Myeloperoxidase

Myeloperoxidase (**MPO**), an enzyme linked to oxidative stress and inflammation can generate hypochlorous acid (HOCl) and other chlorinated biological molecules from hydrogen peroxide and chloride. 3-Chlorotyrosine a product of MPO-catalysed reaction was shown to be significantly increased in LDL that was extracted from atherosclerotic lesions (Himmelfarb *et al.*, 2001). The previous study by Carr *et al* suggests that MPO may contribute to modification of LDL due to its ability to bind to LDL in the presence of robust antioxidants such as albumin, urate, and ascorbate (Carr *et al.*, 2000a). The modification of LDL by MPO mainly transforms the apolipoprotein moiety through tyrosines (nitrotyrosine, chlorotyrosine, and dityrosine). The study by Spickett and colleagues, however, suggested that the modification of the lipid component is largely determined by pre-existing hydroperoxides and pH (Spickett *et al.*, 2000, Zouaoui Boudjeltia *et al.*, 2004). The fact that macrophages and neutrophils can express MPO, generates H₂O₂ suggest that MPO mediated system might be involved in physiological oxidation of LDL (Schindhelm *et al.*, 2009). MPO-oxidised LDL can trigger an inflammatory response in endothelial cells and macrophages by inducing TNF- α and IL-8 hence promoting atherosclerosis (Delporte *et al.*, 2013).

1.4.5.2 Lipoxygenase

Lipooxygenase (**LOX**), is one of the oxidative enzymes in cells that directly oxidises PUFA. *In vitro*, LOX has been shown to oxidize LDL, through direct oxidation and indirect oxidation (non-enzymatically) by the production of reaction products that can further promote the peroxidation of the lipids (Heydeck *et al.*, 2001). The mRNA of the protein and the protein itself has been found to be present in the human atherosclerotic tissues. A slight increase in the expression of the protein in rabbit resulted in an increased level of antigenic determinants for oxidation products (Yla-Herttuala *et al.*, 1990, Yla-Herttuala *et al.*, 1995). LOX with the antigenic determinant of oxLDL has also been identified in human atherosclerotic lesion. All these suggest that the enzyme might have a role in the physiological oxidation of LDL and the onset of atherosclerosis.

The relevance of LOX was supported when Cyrus *et al.*, showed that apoE deficient mice lacking the genes for 12/15 LPO enzyme showed decreased lipid peroxidation product in plasma and the urinary concentration of isoprostanes had a positive correlation with the extent of the disease and enzyme activity (Cyrus, 1999). Lipoxygenase might be able to contribute to LDL oxidation by reacting with its lipid component directly or stimulating the production of ROS which can oxidise LDL (Delporte *et al.*, 2013). In contrast, overexpression of LOX in both Watanabe-heritable hyperlipidaemia (WHHL) and cholesterol-fed rabbits were protected against atherosclerosis (Shen *et al.*, 1996). Also, the knockout of LOX in apoE deficient mice did not reduce atherosclerosis (Merched *et al.*, 2008). This contradictory evidence suggests that LOX can protect or promote atherosclerosis. The dual role of LOX makes its role in the oxidation of LDL *in vivo* controversial.

1.4.4.3 Reactive nitrogen species

Reactive nitrogen species (RNS) have also been implicated in the oxidative modification process. Nitric oxide (NO) produced by ECs of the artery is a potent antioxidant possessing a number of anti-atherogenic properties such as reducing the vascular pressure and preventing the adherence of platelets and leukocyte to the endothelium (Heinecke, 1998). However, in the presence of superoxide anion (O_2^{-}) , NO can be pro-atherogenic. It generates peroxynitrite (ONOO⁻).

ONOO⁻ + H⁺ ⇐ ONOOH

ONOOH \longrightarrow NO₂ + OH⁻

Peroxynitrite is a strong oxidant that can nitrate the protein and lipid component of LDL to a form that is not recognised by the LDL receptor but can be recognized by the scavenger receptor (Darley *et al.*, 1992, Graham *et al.*, 1993). 3-Nitrotyrosine a product of the protein nitrating effect of ONOO⁻ has been shown to be present in atherosclerosis (Graham *et al.*, 1993, Beckman *et al.*, 1994). Hence, peroxynitrite may have a role in the *in vivo* oxidation of LDL.

1.4.5.4 Glycation of LDL

The non-enzymatic addition of carbohydrate molecules to LDL can produce a pro-atherogenic form of LDL. Lysine, a major amino acid component of the apoB-100 component of LDL, can undergo glycation leading to the formation of glycated lysine-LDL (Younis *et al.*, 2008). There is a higher proportion of glucose bound lysine in apoB-100 in diabetics (Tames *et al.*, 1992). Chronic hyperglycaemia has been implicated in increased lipoprotein oxidation. Glucose contribution to peroxidation of lipids in LDL is mediated through a superoxide radical pathway (Kawamura *et al.*, 1994). Moreover, LDL isolated from diabetics patients has been proven to be more susceptible to oxidation (Colas *et al.*, 2010). Glycated LDL has been proven to be more susceptible to oxidation than native LDL (Sobal *et al.*, 2000). A recent study by Younis *et al* suggests that THP-1 derived macrophages incubated with glycated LDL had more cholesterol accumulation compared to control cells incubated with non-glycated LDL (Younis *et al.*, 2009). This evidence put together suggest that glycation of LDL speeds up its oxidation and promotes its atherogenicity and therefore may be important in atherosclerosis.

1.4.5.5 Transition metals

LDL has been found to undergo oxidative modification when incubated with vascular cells (smooth muscle cells (Heinecke *et al.*, 1984), endothelial cells(Steinbrecher *et al.*, 1984) and macrophages (Leake and Rankin, 1990), but this often than not occurs in the presence of

transition metals in the medium. Copper or iron is required in cell-mediated oxidation and contributes to oxidation even when present in micromolar concentration. Iron has been demonstrated to catalyse the oxidation of LDL by smooth muscle cells (Heinecke et al., 1986), macrophages (Henriksen et al., 1983) and endothelial cells (Morel et al., 1983). The ability of metal chelating agents in the medium to stop the oxidation of LDL suggests the presence of metal ion might be necessary for the process of LDL oxidation within cells (Steinbrecher al et., 1984). Free metal ions can oxidize LDL even in the absence of cells when present in high concentrations. Oxidation of LDL initiated by copper and iron are the most studied models in vitro. Copper-mediated LDL oxidation is, however, the most widely used method for oxidation of LDL in vitro. The oxidation of LDL by copper often with the incubation of 2 to 25 μ M concentrations of copper in the culture medium or phosphate Saline buffer or MOPS buffer with LDL concentrations between 50 g protein/ml and 2 mg protein/ml. The Kinetics of oxidation of LDL by metal ions is often measured by the continuous change in diene adsorption at 234 nm (Esterbauer et al., 1989b). Our Laboratory has previously described a rapid method to generate copper-oxidised LDL rich in hydroperoxides or oxysterols (Gerry *et al.*, 2008). The products of oxidation by copper at pH 7.4 is well characterised (Esterbauer et al., 1990b, Brown et al., 1996).

Presence of free iron in the body is hazardous due to its ability to transport electrons and interconvert from iron (II) to iron (III). Iron has a deleterious effect in lipid peroxidation as it has the capacity to both initiate and increases LPO. It can mediate this effect in several ways. Iron can initiate lipid peroxidation by the formation of hydroxyl radicals or directly decomposing lipid hydroperoxides.

 $Fe^{2+} + LOOH \rightarrow Fe^{3+} + OH^- + LO^-$

$Fe^{3+} + LOOH \rightarrow Fe^{2+} + LOO^{\cdot} + H^+$

Ferrous Iron can react with hydrogen peroxide (H_2O_2) to form a hydroxyl ion and hydroxyl radical.

$Fe^{2+} + H_2O_2 \rightarrow Fe^{3+} + OH^- + OH^-$

The hydroxyl radical is a powerful oxidant, highly reactive and reacts directly where it is formed. Iron can also react directly with molecular oxygen (O_2) to form superoxide anion.

$$Fe^{2+} + O_2 \rightarrow Fe^{3+} + O_2^{--}$$

The superoxide itself, is a less reactive and powerful oxidant which has the capacity of generating more of ferrous iron (Fe^{2+}) by acting as a reducing agent for ferric iron (Fe^{3+}), hence, more of the OH radical is generated from the Fe^{2+} produced (Schafer *et al.*, 2000). The role of iron in initiating lipid peroxidation in liposomes has been explored (Tang *et al.*, 2000, Mozuraityte *et al.*, 2008). The characterisation of iron-mediated oxidation and the role of iron in mediating lipid peroxidation within lysosome is now generating attention within our laboratory.

1.5 Lysosomal oxidation of LDL

1.5.1 General structure and function of lysosomes

Lysosomes are single membrane bounded spherical organelles discovered by Christian de Duve in 1955 (De Duve *et al.*, 1955). The lysosomes are important constituents found in nearly all eukaryotic cells, where they are present in the cytosol as dense bodies. They in shape and size depending on the type of cells, they appear spherical in shape or sometimes tubular. Their size in most cells exceeds 1 μ m, although in cells such as macrophages their diameter can be

more than several microns. The size increase in macrophages is likely to be due to the accumulation of materials that are undigested. The lysosomes are separated from the cytosolic environment by a phospholipid- bilayer 7-8 nm in diameter (Saftig and Klumperman, 2009, Appelqvist *et al.*, 2013).

The lysosomes are made up of many hydrolytic enzymes that can digest damaged or unwanted materials. The enzymes present in lysosomes possess the ability to degrade polysaccharides, lipids, proteins, RNA, and DNA. Genetic mutation of the genes encoding the lysosomal enzymes which are responsible for a different genetic disease in humans which are classified as lysosomal storage disorders (Pu et al., 2016). The inside of lysosomes is acidic in nature (about pH 4.5) (Mindell, 2012). ATP- dependent proton pumps on the membrane of the organelle, the vacuolar H+- ATPase, maintain the pH of this compartment. The H+- ATPase pump creates and preserves the acidity of the lumen of lysosomes by utilising free energy derived from the hydrolysis of ATP to pump protons into the lumen and the "counterion flux" made up secondary ion movements is used to prevent the creation of membrane potential that could arise from proton accumulation. The maintenance of its characteristic acidic pH is important for the activities of lysosomal enzymes most of which are active at pH of about 4.5-5.0 but no the cytosolic neutral pH of about 7.2 (Ohkuma et al., 1982, Mindell, 2012, Ishida et al., 2013). Lysosome was initially identified as mere waste bags in cells. Advances in science have over time allowed them to be classified as important organelles that are associated with a large number of cellular events. Lysosomes play a major role in the digestion of extracellular materials taken up by endocytosis, lysosomes are also involved in the degradation of cellular contents through autophagy. In addition to endocytosis and autophagy, lysosomes are also utilised by a phagocytic cell such as macrophages to degrade materials that are phagocytosed including cellular debris, long lived cells, and pathogens such as bacteria. They are believed to

be vital regulators of cellular homeostasis, they are involved in cholesterol homeostasis as proteins required for cholesterol efflux from endolysosomes. Niemann-Pick disease type C $_1$ (NPC1) and NPC $_2$ are located in lysosomes. NPC $_1$ is found in lysosomes and endosomes while NPC $_2$ is found in the lumen of lysosomes (Appelqvist *et al.*, 2013). Lysosomes also play avital role in mediating cell death, although this has been suggested to be induced under certain pathological conditions (Kreuzaler *et al.*, 2011). The role of lysosomes in the pathogenesis of disorders such as lysosomal storage disorders, neurodegenerative disorders (Bellettato and Scarpa, 2010) and cardiovascular diseases (Lutgens *et al.*, 2007) is becoming more evident. These findings might create new therapeutic approaches to these diseases.



Figure 1.4: Lysosome structure

Pictorial representation of the lysosome depicting various lysosomal enzymes, biomolecules such as lipids and proteins, warn out organelles such as mitochondrial and other materials targetted for degradation.

1.5.2 Lysosomes as a source of redox active iron

Some studies have demonstrated the presence of redox active iron in lysosomes (Yuan *et al.*, 1996, Yu *et al.*, 2003, Zhang and Lemasters, 2013). Lysosomes and endosomes are major sites for accumulation of chelatable iron (Zhang and Lemasters, 2013). Petrat *et al.* detected higher amount of redox active iron was present in lysosomes compared other organelles. The study showed that lysosomes contained as much as 16 μ M iron compared to 9.2 μ M and 7.3 μ M concentrations of iron found in mitochondria and cytosol, respectively (Petrat *et al.*, 2001).

The presence of redox-active iron ferrous/ferric stored in nonheme form of resident macrophages in different tissue was demonstrated by Meguro et al (2005). They suggested that most ferrous mediated reactions were localised in the phagolysosomes(pH 5.3) of iron-loaded and normal rats which implies that the acidic pH may facilitate the release of ferrous iron from hemosiderin and the insolvent of lysosomes (Meguro et al., 2005). The lysosome is the site for intracellular recycling of damaged, worn-out or old proteins and organelles through autophagy (Kurz et al., 2007). Autophagocytosed iron-rich proteins deliver iron to lysosomes (Yu et al., 2003). Degradation of ferritin in lysosomes could also account for some of this iron (Radisky and Kaplan, 1998, Yu et al., 2003, Kidane et al., 2006). The lysosomal compartment is important in macrophages as they are scavenger cells. The use of ESR combined with increased lysosomal pH to prevent proteolysis in lysosomes or lysosomotropic iron chelators showed that most intracellular labile iron is derived from lysosomes (Persson et al., 2003, Yu et al., 2003). The presence of this redox-active iron makes the lysosomes susceptible to Fenton type reaction, promotes the associated peroxidative process of materials as they are degraded and when exposed to oxidative stress the resultant iron centred radicals or hydroxyl radicals could destabilise the lysosomal membrane (Öllinger and Brunk, 1995, Persson et al., 2001b, Persson et al., 2001a).

1.5.3 Lysosomal storage disorder and atherosclerosis

The lysosomes of macrophages are crucial to the onset and progression of atherosclerosis due to the role they play in the degradation of lipids and the regulation of signalling pathways. A defective lysosomal function has been identified as a characteristic of atherosclerotic plaque development (Sergin *et al.*, 2015). Lysosomal storage disorders are caused by defects in lysosomal membrane protein and lysosomal enzymes which cause impaired functions in lysosomes and accumulation of undigested materials in the lysosomes (Schultz *et al.*, 2011). The majority of the excess cholesterol contained in atherosclerotic foam cell macrophages are found in lysosomes that are swollen as a result of accumulating the cholesterol (Jerome, 2010). The study by Bobryshev et al confirms that changes in lysosomal function have a role in the onset of atherosclerosis with consequential changes in CD68 antigen in lysosomes in fatty streak present in the lesion (Bobryshev *et al.*, 2013). Accumulation of cholesterol in lysosomes has recently been linked to coronary atherosclerosis in mice (Xu *et al.*, 2016).

1.5.4 Evidence for LDL oxidation by iron in lysosomes

LDL was previously generally believed to be oxidised within the interstitial fluid of the arterial wall and then bound to the scavenger receptors on macrophages, but this belief is challenged by the robust presence of antioxidants, such as ascorbate, urate, and albumin, in the extracellular fluid (Levitan *et al.*, 2010). The fact that serum or interstitial fluid at low level have been shown to be protective against LDL oxidation mediated by copper or iron *in vitro* (Dabbagh and Frei, 1995, Patterson and Leake, 1998, Namazi, 2009, Rodriguez *et al.*, 2009). Wen and Leake were the first to demonstrate the oxidation of LDL within the lysosomes of macrophages. They showed that LDL aggregated by vortexing was rapidly taken up by both mouse macrophage–like cell lines (J774) and human monocyte-derived macrophages (HMDM), then oxidised within the cells. Droplets of lipid were absent in control macrophages while large amount of lipid droplets were found in the cells incubated with aggregated LDL.

Co-incubation with fluorescent dextran revealed the dextran-labelled lysosomes as the main site for the lipids with the presence of cholesteryl linoleate and arachidonate, which were absent in control cells. About a 40% rise in free cholesterol was observed compared to control cells. Ceroid was formed in the lysosomes of both HMDM and J774 cells. Confirmatory experiments with acetylated LDL internalized into lysosomes of cells and incubated without access to LDL in the extracellular space still produced increased levels of oxysterols, which implies that they were mainly oxidised intracellularly (Wen and Leake, 2007).

Wen and Leake also demonstrated the importance of iron in the lysosomal oxidation, as LDL oxidation was inhibited in cells by desferrioxamine (an iron chelator). In a cell free system, ferrous sulphate extensively oxidised LDL at pH 4.5 (similar to the pH of lysosomes) but not at the pH of plasma (pH 7.4), although pH exerted the opposite effect with copper sulphate (Wen and Leake, 2007). In 2012, the group further demonstrated the implication of extensive oxidation of LDL by iron at lysosomal pH in atherosclerosis. Satchel and Leake again showed the effects of iron in the two oxidation states (Fe²⁺ and Fe³⁺), iron chelating agents, antioxidants on the oxidation of LDL and the chemical changes mediated by iron at lysosomal pH. To check if the oxidation was dependent on the presence of iron, iron chelating agent was added at different phases during oxidation and it was suggested that iron was involved in the three phases of oxidation (Satchell and Leake, 2012).

The oxidation of LDL was described by Esterbauer in 1989. LDL is said to undergo three phases of oxidation the lag phase (during which there is little or no increase in formation of dienes), the propagation phase (which involves a rapid increase in the formation of dienes) and the decomposition phase when lipid hydroperoxides breaks down to form other oxidation

products (Esterbauer *et al.*, 1989a). In 2012, Satchel and Leake demonstrated four phases involved in oxidation mediated by iron at lysosomal pH (lag, rapid oxidation, slow oxidation, aggregation and sedimentation). This is because at acidic pH oxidised LDL undergoes extensive aggregation and the aggregates then sediment towards the bottom of the cuvettes. The characterisation of the product of oxidation by iron at lysosomal pH revealed increased amounts of cholesteryl ester hydroperoxides and 7-ketocholesterol confirming that LDL oxidised at this pH might be atherogenic in nature (Satchell and Leake, 2012).

1.5.5 LDL aggregation and lysosomal LDL oxidation

Maor *et al.* demonstrated that inhibition of both LDL aggregation and oxidation is important in reducing aortic lesion in apoE deficient mice (Maor *et al.*, 1997). There is evidence supporting the involvement of aggregated LDL in the pathogenesis of atherosclerosis and their existence of LDL aggregates in arterial (Hoff and Morton, 1985, Hoff and O'Neil, 1991, Steinbrecher and Lougheed, 1992) atherosclerotic lesions (Lu and Gursky, 2013). The core of LDL containing non-polar lipids and hydrophobic in nature surrounded by a monolayer containing apoB, free cholesterol (unesterified). Hence, understanding the involvement of the different constituents of the LDL particle in aggregation is important. Changes in the structure of the apoB component and the lipid component of LDL leads to its aggregation and enhanced lipid droplet formation. These lead to increased uptake by macrophages and promotes an inflammatory response with subsequent formation of the atherosclerotic lesion. LDL aggregation is considered to begin with, the rearrangement of the surface components (Lu and Gursky, 2013).

Modification of LDL by the enzymatic or non-enzymatic process can lead to aggregation and fusion of LDL (Oorni *et al.*, 2000). Treatment of LDL with hydrolytic enzymes and substances
that are pro-oxidative can promote LDL aggregation (Oorni et al., 2000), lipolysis, proteolysis, glycation, oxidation, prolonged storage and other biochemical modification such as acetylation can mediate LDL aggregation. Proteolytic modification of the surface ApoB can cause conformational changes in the functional groups of lipid and protein component which can influence the interaction between different LDL particles and cause them to aggregate (Lu and Gursky, 2013). The oxidation of LDL induces different conformational changes of the lipid and protein component of LDL and these changes often leads to the aggregation of LDL (Maor et al., 1997, Xu and Lin, 2001). The lipid peroxides products such as aldehydes can form adducts with protein. The reactions between aldehydes and proteins have been demonstrated to induce LDL aggregation (Hoff and O'Neil, 1991). The degradation of the hydrophobic part of ApoB and its redistribution to the surface can lead to instability in the structure of LDL which can result in aggregation of LDL(Oorni et al., 2000). Lipoprotein lipase and/or sphingomyelinase (SMase) can mediate aggregation of LDL (Tabas, 1999). Phospholipase C (PLC) hydrolysis of phospholipids causes the rearrangement of non-polar diacylglycerol from LDL surface to the core and the release of phosphocholine and these can rearrangement of the hydrophobic content can promote LDL aggregation (Lu and Gursky, 2013). Aggregation of LDL is mostly found in electronegative LDL, phospholipase A₂ (PLA₂) and PLC has been linked to the observation of electronegative LDL in human plasma (Bancells et al., 2010, Sanchez-Quesada et al., 2012).

Pentikainem and colleagues showed that copper ion induced LDL oxidation, vortexing and treatment with SMase produced aggregated LDL (Pentikainen *et al.*, 1996). Sphingomyelinase enzyme can promote LDL aggregation *in vivo*, up to 50 times increase in ceramide, the product SMase hydrolysis of sphingomyelin was observe in LDL derived from atherosclerotic lesion compared to LDL isolated from plasm of healthy individuals (Schissel *et al.*, 1996). Ceramide

induced LDL aggregation might be as a result of its non-polar nature which can cause the disruption of existing balance between the polar surface and the hydrophobic core (Lu and Gursky, 2013).

Recent work in our laboratory demonstrated the lysosomal oxidation of sphingomyelinaseaggregated LDL (SMase-LDL) (Ahmad and Leake, 2018). Sphingomyelinase-aggregated LDL is a better candidate of modified LDL likely to be present in disease state compared vortexed or acetylated LDL. Findings from our laboratory have demonstrated that SMase-LDL can be oxidised to a great extent in in the lysosomes, as the presence of a large amount of ceroid was observed in the lysosomes after seven days of incubation with SMase-LDL. The recent HPLC analysis of *in vitro* SMase-LDL oxidised by iron at also revealed its ability to generate atherogenic products such as oxysterol and lipid hydroperoxides. Modification of LDL by sphingomyelinase also contributed to increased lipid peroxidation in lysosomes (Wen *et al.*, 2015, Ahmad, 2016, Ahmad and Leake, 2019)

1.6 Iron and atherosclerosis

Iron was first linked to heart diseases by Sullivan in 1981. The iron hypothesis proposed that sustained deficiency of iron protected against ischaemic heart disease while iron overload, on the other hand, promoted CVD. The theory was based on the gender difference in the risk of CVD, higher risk observed in males and the fact that post-menopausal women showed reduced protection. Based on the Framingham study (Kannel *et al.*, 1976), Sullivan argued that the lower risk observed in pre-menopausal women was not the effect of oestrogen but rather due to regular loss of iron due to menstrual bleeding (Sullivan, 1981, Sullivan, 1989). The

hypothesis was supported by the failure of oestrogen replacement therapy to protect against coronary events in post-menopausal women (Hulley *et al.*, 1998).

After over 30years of its proposal, the iron hypothesis remains a subject of debate. While some epidemiological studies have found an association between iron and CVD, others have found no association (Moore *et al.*, 1995, Eichner *et al.*, 1998). Depletion of iron has been shown to reduce the risk of MI, and other events associated with CVD, as this was demonstrated by Facchini and Saylor with the use of phlebotomy bimonthly or monthly to achieve the depleted iron state (Facchini and Saylor, 2002). There may be no consistent relationship between plasma iron and human atherosclerosis because in an inflammatory state, iron sequestered in macrophages and the plasma concentration falls (Ganz, 2005). The most common measure of iron status is ferritin, the main iron storing protein which is directly proportional to tissue iron stores (You and Wang, 2005). Kiechl *et al.* assessed the relationship between body iron stores and carotid atherosclerosis and identified the plasma concentration of the iron storing protein ferritin as the major indication of CAD. They found that increased levels of ferritin correlated with increased CAD and hypercholesterolemia (Kiechl *et al.*, 1994). Salonen *et al.* reported the correlation between the level of ferritin in serum and risk of MI, men with higher ferritin levels had a two-fold increased risk compare to other men (Salonen *et al.*, 1995).

The link between iron and atherosclerosis was demonstrated in animal studies. Lee et al confirmed the association between the deposition of iron and the progression of atherosclerotic lesion in ApoE deficient mice (Lee *et al.*, 1999). Other studies have reported no correlation between iron and CVD. Sempos and colleagues argued that there was no correlation between the marker of iron and the onset of CAD (Sempos *et al.*, 1994), although this study did not

measure the level of serum ferritin. More *et al* found no association between the thickening of the intima and serum ferritin (Moore *et al.*, 1995). The role of iron in atherosclerosis continues to be a controversial topic, hence there is a need for more researchers to focus on this relationship.

1.6.1 Iron in atherosclerotic plaques

The atherosclerotic lesion environment has been demonstrated to contain copper and iron which can mediate lipid peroxidation and free radical reactions (Smith *et al.*, 1992, Lamb *et al.*, 1995). A study by Pang et al demonstrated deposition of iron is important in the formation of advanced atherosclerotic lesion (Pang *et al.*, 1996). Lee *et al.* discovered the presence of both iron and ceroid in foam cells present in atherosclerotic lesion (Lee *et al.*, 1998), their results demonstrated a close association between deposition of iron in human aortic wall and progression of atherosclerosis. The deposits of ceroid and iron were more prominent in advanced atherosclerotic lesion suggesting iron can mediate oxidative reactions in atherosclerosis (Lee *et al.*, 1998).

Another study by Lee *et al* in 1999 used an animal model, apoE deficient mice, to investigate the role of iron deposition in atherosclerosis. Iron deposition was observed in intermediate lesions and the deposition increased as the lesions became advanced lesions. They also demonstrated that iron restriction in the diet reduced the susceptibility of LDL to *in vitro* oxidation and decreased oxidative stress (Lee *et al.*, 1999). Yuan *et al.* previously showed the occurrence of iron in early atherosclerotic lesions using Perl's method and the more sensitive modified sulphide silver method. The sulphide silver method showed that the majority of the foam cells contained iron, while few were found to contain iron with Perl's method. More iron was seen for macrophages that had engulfed erythrocytes with both methods (Yuan *et al.*, 1996). Stadler and his colleagues' quantified copper and iron in carotid lesions compared with normal controls using electron paramagnetic resonance (EPR) and inductively coupled mass spectroscopy (ICPMS). The results showed elevated iron and copper in carotid lesions compared to the controls (Stadler *et al.*, 2004). This further confirms the presence of transition metals, particularly iron and copper, in atherosclerotic plaques.

1.6.2 Iron metabolism

Iron is an important element as it forms a crucial part of many redox enzymes utilised in metabolic processes. It serves as a component of haemoglobin and myoglobin (oxygen transport proteins). Iron is an important cofactor to many enzymes required for DNA repair and synthesis (Puig *et al.*, 2017). The majority of iron is contained in the haemoglobin of erythrocytes and developing erythroid cells. Notable amounts are contained in macrophages and myoglobin while excess iron in the body is stored in the liver. The body loses iron through general bleeding, menstruation, sloughing of epithelial cells of the intestine, and desquamation of the skin. (Nemeth and Ganz, 2006, Schmidt, 2015).

Since humans do not have an effective iron excreting mechanism, body iron must be closely regulated. Iron is absorbed in the small intestine by enterocytes and then transported into cells in need of supply (Munoz *et al.*, 2009). Following the release to the bloodstream, iron binds to transferrin (a plasma glycoprotein that has the capacity to bind two atoms of iron). The iron-transferrin complex binds to the transferrin receptor and it is then endocytosed. In the acidic compartment of the late endosomes, iron is released from the complex and transported into the cytoplasmic pool of labile iron by divalent metal transporter 1 (DMT1). Iron is then transported to ferritin (an iron binding protein) for storage or to sites where macromolecules containing iron are synthesized mainly mitochondria (Kurz *et al.*, 2011).

Transferrin-iron in plasma and extracellular fluid are kept at a stable concentration by systemic iron regulation. The systemic iron regulation maintains the flow of iron into plasma by controlling major iron sources such as iron released from hepatocytes. Macrophage and hepatocytes iron are stored in intracellularly in ferritin and can be mobilised in situations where there is high demand for iron. Excess iron and deficiency of iron can lead to damages to cells and dysfunctional organs. Hence regulation of iron is essential. Hepcidin is a major regulator of iron homeostasis in the whole body; it is predominantly synthesised in the liver before been released into circulation (Collins *et al.*, 2008). It can also be synthesised in other tissue such as kidney (Kulaksiz et al., 2005), adipose tissue (Bekri et al., 2006) and the heart (Qian et al., 2007, Merle et al., 2007). Although hepcidin is mainly secreted by hepatocytes (De Domenico et al., 2008), other cells such as myeloid cells (Peyssonnaux et al., 2006), alveolar and splenic macrophages (Liu et al., 2005), monocytes (Theurl et al., 2008) and adipocytes (Bekri et al., 2006) also synthesise hepcidin but at very low rate compared to hepatocytes. Hepcidin has the ability to regulate the systemic metabolism of iron. The synthesis of hepcidin is directly linked to inflammation and increased level of iron but inversely related to erythropoiesis. It regulates plasma iron by binding to ferroportin, the iron exporter which is highly expressed on macrophages and duodenal enterocytes. Hepcidin has been demonstrated to inhibit iron efflux into plasma from hepatocytes, enterocytes and macrophages by binding ferroportin and leads to conformational changes that cause both molecules to be endocytosed and targeted for degradation. Hepcidin is therefore considered to regulate iron absorption and recycling (Nemeth et al., 2004b, Knutson et al., 2005, Collins et al., 2008).

1.6.3 Iron binding proteins and LDL oxidation

The iron binding proteins may have a role to play in the availability of iron under certain conditions. Studies have suggested the release of iron from ferritin (Minotti and Aust, 1987), and transferrin (Lamb and Leake, 1994c) and the iron released from transferrin oxidised LDL

in vitro. Moreover, iron containing proteins such as myoglobin (Rodriguez-Malaver *et al.*, 1997) and haemoglobin (Paganga *et al.*, 1992) and copper containing protein such as caeruloplasmin (Lamb and Leake, 1994a) have also oxidised LDL *in vitro* (Yoshida and Kisugi, 2010). Ferryl myoglobin oxidised LDL as shown by increased electrophoretic mobility of LDL (Rice-Evans *et al.*, 1993). Since the proposal of the iron hypothesis by Sullivan, the relationship between the iron storing proteins and atherosclerosis continues to be a topic largely debated in the literature. Hence, the role of ferritin as the major iron store should be further evaluated.

1.7 Ferritin

Ferritin (Fig.1.5) is the main protein for the intracellular storage of iron found in almost all organisms. It is a protein of molecular weight of about 500 kDa with an inside diameter of 7-8 nm and 12-13 nm on the outside.



Figure 1.5: Three-dimensional representation of ferritin (Casiday and Frey, online)

Ferritin is made up of 24 polypeptide chains with light (L) subunits and heavy (H) subunits with a molecular weight of 19 kDa and 21 kDa, respectively (Theil, 1987, Zarjou *et al.*, 2010). It is considered the main iron storage protein because of its large cavity for iron storage. The H subunit plays a major role in the binding and release of intracellular iron, iron transport and sequestration because of its ease in the binding and release of iron as compared to L subunit. It possesses a ferroxidase activity which converts Fe (II) to Fe (III) for storage while the Light subunit is involved in mineralization storage for a long term and the nucleation using iron atoms (Rucker *et al.*, 1996, Arosio *et al.*, 2017).

Ferritin is found mainly in organs that store iron, such as the spleen and liver. It is also found in organs that have low iron levels such as pancreas and heart (Kohgo *et al.*, 1980, Bomford *et al.*, 1981). Ferritin is mainly located in the cytosol but has also been seen in animal mitochondria, plant plastids, nucleus, insect endoplasmic reticulum and circulating plasma (Arosio *et al.*, 2017). Ferritin synthesis is regulated by iron (decreased synthesis in the presence of low iron and vice versa) and oxidative stress. Cytokines regulate ferritin at the transcriptional and translational level particular TNF α and interleukin 1 α have been demonstrated to stimulate the synthesis of the heavy chain of ferritin in human muscle cells, mouse adipocytes and other types of cells (Wei *et al.*, 1990, Tsuji *et al.*, 1991). This suggests that the level of ferritin may be upregulated in inflammation.

Since the proposal of the iron hypothesis, many studies have measured the relationship between ferritin as an iron store and atherosclerosis. The evidence remains controversial, however, recent molecular studies in humans and rabbits revealed elevated mRNA for both L and H chains in atherosclerotic lesion compared to normal arteries(Pang *et al.*, 1996). Direct evidence

obtained with proteomics by measuring the level of ferritin in CAD, found higher level of ferritin in diseased arteries (You *et al.*, 2003). The role of ferritin in lysosomal oxidation is still in its infancy and can be further explored for its implication in the onset and progression of atherosclerosis.

1.7.1 Iron incorporation into ferritin

The ability of iron to induce oxidative damage is prevented by storing it in a form that reduces the concentrations of redox-active iron and at the same time ensure its availability for biochemical reactions when needed (Bouton *et al.*, 1996, Rudeck *et al.*, 2000). Ferritin binds iron to inhibit its catalytical activity and protect cells from iron-mediated damage. However, damage to ferritin can lead to ferritin losing its physiological role hence (Grune *et al.*, 1997, Rudeck *et al.*, 2000), malfunction and releasing iron-to mediate toxic reactions.

Incubation with iron chelators and reducing agent can remove iron from ferritin while the addition of iron in the presence of oxygen can reform the core. This event may be similar to what happens *in vivo*. Fe (II) is delivered through the hydrophilic channel to the centre where it forms a diferric-peroxo complex after an encounter with O_2 or H_2O_2 . This complex rapidly forms two ferric ions and H_2O_2 , which is used locally for iron oxidation and hence no free radical is formed. The oxidised iron then relocates into the cavity where hydrolysis and nucleation occur (Theil *et al.*, 2013, Honarmand Ebrahimi *et al.*, 2015).

1.7.2 Control of cellular iron availability by ferritin

Evidence suggests that cellular activities are affected by cellular iron availability controlled by the ferroxidase activity of ferritin. Accumulation of iron through ferritin activity has been shown to be important in atherosclerosis. Four types of cells are involved in plaque formation which includes macrophages, lymphocytes, smooth muscle cells and endothelial cells. The main cause of plaque formation is the alteration of endothelial cell function. Endothelial dysfunction has been linked with the uptake of iron by endothelial cells (Rooyakkers *et al.*, 2002). Accumulation of iron leads to alteration of ECs structurally and activates the recruitment of monocytes (Kamanna *et al.*, 2012). Iron overload in macrophages causes polarisation of macrophages to pro inflammatory M1 macrophages which expresses higher amounts of ferritin and reduced iron turn-over (Cairo *et al.*, 2011).

1.8 The use of Antioxidants in atherosclerosis

Generally, antioxidants delay or inhibit LDL oxidation in vitro and in cultured cells. They exert varying effects depending on the means of LDL oxidation and the properties of the antioxidant. The risk of atherosclerosis might be alleviated by inhibiting the process of oxidation of LDL independently of lowering blood cholesterol level. Nutritional and pharmacologic antioxidants have been suggested to play a role in the prevention of atherosclerosis. Studies on animals as well as cultured cells showed that probucol, a drug with antioxidant and cholesterol lowering properties, prevented the oxidative modification of LDL (Parthasarathy et al., 1986). The results from probucol experiments remain controversial: Bird and colleagues showed that probucol did not decrease atherosclerosis but rather increased atherosclerosis even in the presence of vitamin E in LDL receptor deficient mice. Another study by Choy et al. demonstrated a positive effect of probucol in decreasing apoE-deficient mice, which was attributed to the anti-inflammatory properties of probucol (Choy et al., 2005). However, results from the human clinical trial suggest that probucol did not significantly reduce atherosclerosis (Walldius et al., 1994). The use of probucol in western countries was withdrawn, as its efficacy in atherosclerotic cardiovascular diseases is challenged by the failure of the clinical trial in humans and the arrival of more potent lipid lowering drugs such as statins. However, Yamashita and others

have designed a prospective study yet to be concluded to examine the efficacy and safety of probucol in preventing secondary cardiovascular diseases (Yamashita *et al.*, 2016). Recent evidence from our laboratory has shown that probucol offered no protection for initial oxidation of LDL by iron ions at lysosomal pH (Ahmad and Leake, 2018), which might be the reason for the failure in human trials.

Antioxidants such as β -Carotene, ascorbate, and α -tocopherol are important antioxidants, as their levels can be considerably increased without causing side effects in the system. β -Carotene is a hydrocarbon carotenoid derived from plants with the capacity to trap free radicals (Burton and Ingold, 1984, Krinsky, 1989). β -Carotene has been demonstrated to inhibit LDL oxidation mediated by human monocyte macrophages and copper ions (Jialal *et al.*, 1991). The role of β -carotene in inhibiting LDL oxidation is controversial. A study by Gaziano and colleagues demonstrated that *in vitro* and *in vivo* supplementation with β carotene was not protective against LDL oxidation (Gaziano *et al.*, 1995).

Vitamin C has been established as a cofactor for enzymes and antioxidants, but the mechanisms by which this vitamin can improve chronic disease state is less understood. Vitamin C has been demonstrated to reduce the adhesion of monocytes to the endothelium (Weber *et al.*, 1996) and also improve the production of nitric oxide by endothelium (Kashiba-Iwatsuki *et al.*, 1996), these effects may in turn decrease atherosclerosis. Previous work by Horsley *et al* showed that the oxidised product of vitamin C, dehydroascorbate switches from antioxidant to pro-oxidant during oxidation of LDL by copper (Horsley *et al.*, 2007). Clinical trials result showed no protective effect of vitamin C against CVD (Cook *et al.*, 2007, Sesso *et al.*, 2008). The trials by cook and Sesso *et al.* also included vitamin E but showed no effect.

 α -Tocopherol is considered the main antioxidant contained in LDL particles with each LDL particle containing about five to nine molecules of vitamin (Esterbauer et al., 1990a). The α -tocopherol are believed to protect LDL from oxidants (Meydani, 2001) as it is considered as the most important and most active antioxidant extracted in LDL (Esterbauer et al., 1990a), which can break the chain reactions of free radical. It can exert this effect by scavenging alkoxyl radicals and peroxyl radicals formed from LDL lipids (Sies et al., 1992, Liebler, 1993) and prevent them from promoting the chain reactions of lipid peroxidation. Supplementation with vitamin E has been shown to decrease atherosclerotic lesions in apolipoprotein E-deficient mice (Peluzio et al., 2001) and prevented diet induced atherosclerosis in rabbits (Schwenke et al., 2002). In vitro studies have demonstrated a complex role for a-tocopherol in the oxidation of LDL. Some studies have shown that increased α -tocopherol content in LDL made them less susceptible to oxidation by copper ions (Esterbauer et al., 1991b) and macrophages (Jessup et al., 1990). Other studies have shown that Vitamin E could be a pro-oxidant and mediate LDL peroxidation, which can be caused by α -tocopherol radical (α -Toc) reacting with a lipid group (LH) in LDL (Bowry and Stocker, 1993, Kontush et al., 1996). Evidence from some small clinical trials showed that vitamin E could protect against CVD (Stephens et al., 1996, Boaz et al., 2000). The beneficial role of vitamin E in preventing CVD is often challenged by the lack of protection seen in large clinical trials against CVD, such as the HOPE (Yusuf et al., 2000, Investigators, 2005) and GISSI (Jialal et al., 1999, Collins et al., 2002). Additional negative evidence was equally obtained from a randomised trial with a low dose of aspirin in combination with vitamin E (de Gaetano, 2001).

Prevention and management of cardiovascular diseases have focussed largely on lifestyle changes and reducing risk factors. Drugs that can effectively lower high blood pressure and cholesterol concentration have emerged. Despite the better understanding of the pathogenesis of atherosclerosis, many challenges remain in the treatment of the disease. The recent lysosomal oxidation hypothesis raises further questions on the mechanisms of the lysosomal oxidation *in vivo*, the key players in the oxidation and the possibility of a novel therapy for atherosclerosis from antioxidants that can accumulate in the lysosomes.

1.9 Thesis hypothesis and aims

LDL oxidation by ferritin in the lysosomes of macrophages is important in the pathogenesis of atherosclerosis. The overall hypothesis of this study was that ferritin as the major iron-storing protein can mediate LDL oxidation at lysosomal pH and within cells to contribute to the progression of atherosclerosis and some antioxidants can ameliorate this effect. The concept of lysosomal LDL oxidation by catalytically active iron was recently proposed and there is a need to explore the relevant sources of iron for this oxidation *in vivo*. The iron binding proteins may serve as a source of iron under certain conditions. Some studies have demonstrated the release of iron from ferritin and transferrin. Iron released from transferrin has contributed to oxidising LDL *in vitro*. Moreover, iron-containing proteins such as myoglobin and haemoglobin and copper-containing protein such as caeruloplasmin have also oxidised LDL *in vitro*. Since the proposal of the iron hypothesis of atherosclerosis by Sullivan, the relationship between the iron-storing proteins and atherosclerosis continues to be a topic actively debated in the literature. Hence the role of ferritin as the major iron store can be further evaluated. This thesis addressed the following aims:

Aim 1: The first aim of this thesis was to investigate the role of ferritin on LDL oxidation at lysosomal pH by exploring the mechanisms by which ferritin can catalyse LDL oxidation in

the lysosomes. monitoring its effect on the formation of lipid peroxidation products (conjugated dienes, cholesteryl ester hydroperoxides and 7-ketocholesterol) at lysosomal pH *in vitro*. The ability of ferritin to release catalytically active iron during the oxidation of LDL was monitored. The levels of these oxidised lipids formed in the presence ferritin might enable us to predict the potential atherogenicity of LDL oxidised by ferritin.

Aim 2: The degradation of many intracellular and extracellular proteins occurs in the lysosomes. Autophagocytosed ferritin would be degraded in lysosomes. There is evidence suggesting the release of iron from ferritin after different treatment such as proteases. The effects of degradation of ferritin or LDL on the oxidation of LDL by ferritin under the lysosomal pH condition *in vitro* were assessed. This was important to explore in the lysosomal oxidation theory, as both LDL and ferritin are targeted for degradation in lysosomes. It is possible that degradation of ferritin, apoB and lipid component (mainly cholesterol esters) component and of LDL by lysosomal enzymes will affect the rate of oxidation of LDL by ferritin at lysosomal pH

Aim 3: It was relevant to explore the role of antioxidants in inhibiting lysosomal LDL oxidation mediated by ferritin as the oxidative modification theory of atherogenesis is constantly been challenged by failure of some previously used antioxidants to protect against CVD in large clinical trials. Antioxidants which can accumulate in lysosomes might be a more potent candidate in inhibiting LDL oxidation. The ability of antioxidants in to protect LDL against oxidation by ferritin *in vitro* was assessed using the water-soluble antioxidants cysteamine (a lysosomotropic antioxidant) and vitamin C, the lipophilic antioxidants α -tocopherol and N, N'-diphenyl-p-phenylenediamine (DPPD) and the amphipathic compound, Tempol (4-hydroxy-2,2,6,6-tetramethylpiperidine 1-oxyl).

Aim 4: Some evidence supports the role of oxLDL in foam cell formation in atherosclerotic lesion advancement. There were strong indications that the *in vivo* modification of LDL increases its ability to induce an inflammatory response and promote atherosclerosis. In addition, there is ample evidence in support of the proatherogenic effects of oxLDL and its ability to induce cellular damage. Lysosomes possess the ability to process ferritin by autophagy. Understanding the role of ferritin in lysosomal oxidation is still in its infancy and can be further explored for its implication in the onset and progression of atherosclerosis. It was important to understand the effects ferritin-oxidised LDL will have at the intracellular level. Our final aim was to investigate the effects of oxidation of LDL by ferritin in lysosomes on macrophage function. The link between upregulation of iron in macrophages by hepcidin and lysosomal LDL oxidation was explored. We tested the ability of ferritin to increase lipid peroxidation in lysosomes. The relevance of ferritin-oxidised LDL on the intracellular generation of reactive oxygen species and induction of apoptosis was explored. We assessed the effect of lysosomal LDL oxidation by ferritin on cellular respiration in cultured human macrophages.

Chapter 2 - General materials and methods

2.0 Materials and Methods

2.1 Chemicals

All laboratory reagents obtained were of analytical grade. Ferritin and all antioxidants used were prepared in the appropriate buffer for the particular experiment just before use. General laboratory reagents and the companies that supplied them are listed in appendix 1, the lysosomal lipid peroxidation probe (Foam-LPO) was a gift from Dr Yi Xiao, Dalian University of Technology, China.

2.1.1 Solutions

Cryopreservation medium

The medium was prepared with Dimethyl sulphoxide (DMSO), foetal calf serum and RPMI-1640 ($\frac{10}{40}$, $\frac{v}{v}$).

Lipid hydroperoxide assay (colour reagent)

Colour reagent was prepared with sodium azide NaN₃ (150 μ M), potassium phosphate monobasic KH₂PO₄ (163 mM), dipotassium hydrogen orthophosphate K₂HPO₄ (37 mM), potassium iodide KI (120 mM), ammonium molybdate (NH₄)₂MoO₄ (10 μ M), Trition X-100 (2g/L) and Benzakonium chloride C₆H₅CH₂N(CH₃)₂RCl (0.1g/L). The pH was then adjusted to pH 6.0.

<u>DAPI</u>

Ultrapure water (2 ml) was added to 10 mg/ml stock to give 5 mg/ml. 1 μ g/ml solution was prepared in fluorescence mounting medium (DAKO S3023).

Buffer for dialysing LDL

10 Litres of dialysis buffer (140mM NaCl, 8.1 mM Na₂HPO₄, 1.9 mM NaH₂PO₄ and 100 μ M EDTA) were prepared by addition of 81.83 g NaCl, 14.42g Na₂HPO₄.2H₂O, 2.964 NaH₂PO₄.2H₂O and 372.2 mg EDTA. It was made up to 10 litres and the pH adjusted to 7.4.

Dihydroethidium (DHE)

DHE (10 mg) was dissolved in 1585 μ L Dimethyl sulphoxide (DMSO) to make 20 mM stock solution.

Flow cytometer buffer

5mM EDTA with 0.6% BSA in PBS.

High-density solution (HDS).

1L of HDS (2.97 M KBr, 2.62 M NaCl and 297 μ M EDTA) was prepared by the addition of 354 g KBR, NaCl 153g, EDTA.Na₂.2H₂O 110.5mg and then made up with distilled water to 1 litre and pH was adjusted to pH 7.4.

1.063 g/ml NaCl/KBr/EDTA solution

 $V_{HDS} = V_{LDS} (D_{REQ} - D_{CURR}) / D_{HDS} - D_{REQ})$

Vol of HDS = $0.057 \times 2,000 \text{ ml of LDS}$

The density of HDS - 1.063

The volume of HDS obtained was added to 2 litre LDS and the pH was adjusted to pH 7.4 and the density was checked with calibrated 100ml flask at 4 °C to ensure it was 1.063 g/ml. Adjustment where made where necessary.

HPLC oxidation mobile phase

HPLC grade acetonitrile, isopropanol and water at 44/54/2 % by volume. To make 1 litre 440 ml acetonitrile, 540 ml isopropanol and 20 ml water were mixed.

Stabilising buffer for incubating LDL with sphingomyelinase

100 ml of stabilising buffer was prepared with HEPES 8 0ml of 150 mM/L NaCl, 10 ml of 50 mM HEPES and 10 ml of 100 mM MgCl₂ and then pH to 7.4.

Low density solution (LDS)

10 litres of LDS with density 1.006 g/ml (150mM NaCl and 297 μ M EDTA) was prepared by adding 87.68 g NaCl, 110.5 mg EDTA and then made up to 10 litres. The pH was adjusted to pH 7.4.

1.019 g/ml NaCl/KBr/EDTA solution

 $V_{HDS} = V_{LDS} (D_{REQ} - D_{CURR}) / D_{HDS} - D_{REQ}$

Vol of HDS = $0.013 \times 2,000 \text{ ml of LDS}$ The density of HDS - 1.019

The volume of HDS obtained was added to 2 litre LDS and the pH was adjusted to pH 7.4 and the density was checked with a calibrated 100 ml flask at 4 °C to ensure it was 1.019 g/ml.

Lowry A

Lowry A solution (0.2 M Na₂CO₃, 0.1 M NaOH, 5.7 mM sodium tartrate and 35 mM SDS) was prepared with 5 g Na₂CO₃, 1 g NaOH, 0.328 g sodium tartrate and 2.523 g SDS made up to final volume (250 ml) with water.

Lowry B

To make 250 ml, 0.16 M CuSO₄, 6.384 g was dissolved in ultra-pure water and made up to 250 ml.

MOPS buffer pH7.4

NaCl, 8.8 g (150 mM) and 2.09 g 3-N-[Morpholinopropane] sulphonic acid (MOPS) (10 mM) with pre-washed Chelex-100 (1 g) was made up to 1 litre and adjusted to pH 7.4, mixed overnight with stirring. The Chelex was removed by filtration and the pH was readjusted. Chelex-100 is used to remove transition metals that might be present.

Oil Red O stain

500mg of Oil Red O powder was dissolved in 100 ml of 99% Isopropanol solution and placed in a warm water bath to dissolve. To prepare working solution from this stock solution, 3 parts of Oil red O stock was mixed with 2 parts ultra-pure water (depending on volume needed for each experiment). This was then incubated at room temperature for 10 minutes and filtered using a $0.45 \,\mu m$ Minisart filter. The working solution was used within two hours of preparation.

Paraformaldehyde cell fixation solution (4%)

4g of paraformaldehyde was dissolved in 60ml PBS and kept in a warm water bath at 37 °C in a fume hood overnight. The pH was then adjusted to pH 7.4 and the volume was made up to 100 ml.

PBS (Phosphate buffer saline) 1X

NaCl, 8.0 g (137 mM), 0.20 g KCl (2.7 mM), 0.23 g NaH₂PO₄ (1.9 mM), 0.12 g Na₂HPO₄ (0.8 mM) was dissolve in 1 litre and the pH was adjusted to pH 7.4

Reducing sample treatment buffer 6X RSTB

SDS (12%), 30% B mercaptoethanol, 30 % stacking gel buffer, 30% glycerol) 50 ml was prepared with 6 g SDS, 15 ml B mercaptoethanol, 15 ml stacking gel buffer, 15 ml glycerol and 5 ml ultra-pure water. A trace of bromophenol was added.

Resolving gel buffer

1.5M Tris-HCl (500 ml) pH 8.8 was prepared with 181.5 g of Tris and made up with ultra-pure water and pH was adjusted with only HCl to pH 8.8.

RIPA lysis buffer

Lysis buffer was prepared with 150 mM NaCl, 1 mM EDTA, 50 mM Tris pH 8.0, 1.0 % Triton X-100, 0.5 % Sodium deoxycholate and SDS (0.1 %).

Sodium chloride/sodium acetate buffer pH 4.5

NaCl, 8.8 g (150 mM) and 0.82g (10 mM) Sodium acetate with pre-washed Chelex-100 (1g) was made up to 1L and adjusted to pH4.5 and mixed overnight with stirring. The Chelex was removed by filtration and the pH was readjusted.

Stacking gel buffer

500 ml 0.5 M Tris-HCl pH 6.8 was prepared with 30g of Tris and made up with ultra-pure water and pH was adjusted with only HCl to pH 6.8.

2.1.2 Laboratory equipment

Name	Manufacturer
A1 inverted epifluorescent microscope	Carl Zeiss
Axioscope epifluorescent microscope	Carl Zeiss
Atomic absorption spectrophotometer (AAS)	Analytik Jena
BD Accuri TM C6 flow cytometer	BD Biosciences
Countess II FL automated cell counter	Invitrogen
Evos XL cell imaging system	ThermoFisher
Inductively coupled plasma mass spectrophotometer (ICP-MS)	ThermoFisher
Lambda-2 6-cell spectrophotometer	PerkinElmer
Lambda Bio 40 8-cell spectrophotometer	PerkinElmer
Lambda 35 8-cell spectrophotometer	PerkinElmer
Libra S22 UV/Vis Spectrophotometer	Biochrom
Optima TM XPN ultracentrifuge	Beckman Coulter
PerkinElmer 200 HPLC system	PerkinElmer
Agilent 1200 HPLC	Agilent
Speed Vac	ThermoFisher
U genius image capturing machine	Syngene
Seahorse XFp Analyzer	Agilent

2.2 **Preparation of LDL**

2.2.1 Ethics and blood collection

Permission was received from the University of Reading Research Ethics Committee to take blood from volunteers on 2 February 2012 (Project number 12/7; Isolating low density lipoprotein and monocytes from human blood for studies of atherosclerosis). Blood was usually taken from one to three healthy volunteers. Blood (250 ml) was collected from each donor into 50ml syringes containing (3 mM EDTA at pH 7.4 to inhibit the oxidation of LDL and prevent blood clotting).

2.2.2 Isolation of LDL

Native LDL (d=1.019-1.063g/ml) was isolated from normal human blood (after an overnight fast) according to the method previously used in our laboratory (Wilkins and Leake, 1994). LDL isolated by sequential ultracentrifugation of the plasma. Blood collected as described in section 2.2.1 were transferred into 50 ml tubes, the tubes were inverted to mix and placed on ice. The blood samples were centrifuged at 1500 g for 30 min at 4 °C. The plasma was removed from the upper layer and its volume measured. The density was adjusted to 1.019g/ml by adding HDS (prepared from 297 μ M EDTA, 2.62 M NaCl and 2.97 M KBr). The volume of HDS to be added was calculated as follows:

 $V_{HDS} = Vplasma (D_{REQ} - D_{CURR})/D_{ADD} - D_{REQ})$ where $V_{HDS} = Volume$ of high density solution HDS to be added to the pooled plasma, Vplasma =Volume of plasma obtained from blood collected, D_{REQ} = final density required which is 1.019 mg/l, D_{CURR} = the current density of pooled plasma which is 1.006 mg/l and D_{ADD} = the density of the HDS solution prepared.

Vol of HDS = 0.013 x volume of plasma

The density of HDS- 1.019

The calculated volume of HDS was added to the plasma transferred into the dialysis tubing and then placed in 2 litres of 1.019 g/ml NaCl/KBR/EDTA solution (prepared as described in

section 2.1.1), dialysed for 2 hours at 4 °C with stirring. The plasma was transferred into 35 ml ultracentrifuge tubes and centrifuged at 115,000 x g (40,000 rpm) at 4°C for 18 hours. The tubes were sliced at the upper region to obtain a coloured solution at the bottom. The clear solution in the upper region containing VLDL and IDL was discarded. The bottom fraction containing HDL, LDL and other plasma proteins was collected, the gelatinous material at the bottom was drawn with a Pasteur pipette and all gelatinous material were re-suspended. The volume was measured and the density was adjusted to approximately 1.063 g/ml by addition of HDS using the following equation to determine the amount required:

Vol of HDS = 0.044 x volume of plasma

The density of HDS - 1.063

To adjust the density to exactly 1.063g/ml the fraction was transferred into dialysis tubing and dialysed against 1 litre of 1.063 g/ml solution (prepared as described in section 2.1.1), for 2hours at 4° C with stirring. After dialysis, the plasma was transferred to ultracentrifuge tubes and centrifuged for 18 hours at 115,000 x g (40,000rpm) at 4°C. After 18 hours of centrifugation, the tubes where sliced below the upper golden layer containing the LDL, the bottom fraction containing HDL was discarded. The LDL fraction was transferred to a dialysis tube and dialysed for 2 hours against 1 litre of 1.063g/ml solution at 4°C with stirring. The solution was then centrifuged for another 18 hours at 115,000 x g (40,000rpm) at 4°C. After 18 hours the ultracentrifuge tubes were sliced below the LDL fraction at the upper layer and the lower layer was discarded.

The obtained LDL fraction was dialysed against NaCl/phosphate/EDTA (40 mM NaCl, 8.1 mM Na₂HPO₄, 1.9 mM NaH₂PO₄ and 100 μ M EDTA) (pH 7.4) overnight with several changes. The LDL was filter sterilised using a Millipore filter of pore size (0.45 μ m) and stored in the dark at 4 °C. The LDL protein was then assayed to determine the concentration. The LDL was stored at 4 $^{\circ}$ C in the dark room in the presence of 100 μ M EDTA and used within one month of isolation.

2.2.2 Enrichment of LDL with α-tocopherol

The α -tocopherol content of native LDL was increased as described by Esterbauer *et al* (Esterbauer *et al.*, 1991a). The plasma of blood collected from healthy volunteers was obtained by centrifugation at 1500 g for 30 min at 4 °C. The pooled plasma was then incubated for 3 h at 37 °C with 1% (v/v) DMSO containing 100 mM α -tocopherol (final concentration 1mM), to allow the α -tocopherol to diffuse into the LDL particles or 1% (v/v) DMSO as a control. LDL was then isolated by ultracentrifugation as described in section 2.2.1.

2.2.3 Lowry protein assay

The concentration of LDL protein was determined by a modified Lowry assay for protein determination previously described by Markwell *et al* (Markwell *et al.*, 1978). Standard masses (0, 20, 40, 60, 80, 100 μ g) of bovine serum albumin (BSA) were prepared from a stock solution of BSA (400 μ g/ml) in triplicates and made up to 500 μ l with ultra-pure water. Sample (125 μ l) was added to assay tubes in triplicates and also made up to 500 μ l with ultra-pure water. Freshly prepared Lowry A solution (1.5ml) was added to each assay tubes and placed at room temperature for ten minutes. Lowry B reagent (150 μ l) was added to each assay tube and incubated at 55°C for 5mins. The assay tubes were then allowed to cool to room temperature, the absorbance at 650nm was measured. The assay is based on the ability of the peptide bonds present in both BSA and apoB-100 of LDL to form a complex with copper ions in Lowry A reagent. The complex formed then reduces the Lowry B (phosphomolybdic-phosphotungstic Folin reagent) to a blue chromophore that absorbs light at 650nm. The LDL protein contained in the sample was then determined against the standards (BSA). Usually about 2.5mg to 4mg of LDL protein/ml was obtained.

2.2.4 Preparation of sphingomyelinase aggregated LDL

Native LDL was diluted to 2 mg protein/ml with LDL stabilising buffer containing (NaCl 150 mM, MgCl₂ 100 mM and HEPES 50 mM, pH 7.4). Sphingomyelinase from *Bacillus cereus* (Sigma-S9396-50UN) was added at 10 mU/ml and incubation at 37 °C was started in a water bath (Yabu et al., 2008, Walters and Wrenn, 2010). The attenuance (absorbance plus light scattering) was measured at 430 nm as well as 680 nm was measured at 0 h and every hour afterwards. At each time point LDL diluted to 100 µg protein/ml was measured in cuvettes containing PBS. Using PBS as reference the sample was read at 430 and 680 nm simultaneously. The process was stopped when the absorbance at 430 and 680 nm reached about 0.09 and 0.013 respectively. From previous confirmatory experiments using dynamic light scattering (Wen et al., submitted). It was indicated that the average LDL particle size at this stage would have increased from about 25 nm to about 200 nm. After suitable aggregation was achieved the incubation was stopped and sphingomyelinase aggregated LDL (SMase-LDL) was dialysed against 2 L of the buffer used for dialysing LDL (40 mM NaCl, 8.1 mM Na₂HPO₄, 1.9 mM NaH₂PO₄ and 100 µM EDTA) (pH 7.4) overnight in a cold room with stirring. The SMase-LDL was collected and sterilised with a 0.45 µm Minisart filter and SMase-LDL protein concentration was assessed by the Lowry method described in section 2.2.3. The aggregation was confirmed by comparing light scattering at 680 nm by comparing SMase-LDL to control LDL (Fig 2.1).



Figure 2.1 Measurement of aggregation in SMase-LDL compared to native LDL

The attenuance at 680 nm of native LDL and SMase-LDL (100 μ g protein /ml) out of 2 mg LDL protein/ml incubated with or without sphingomyelinase (10 mU/ml) at 37 °C was compared at different time intervals for up to 6 h.

2.2.4 Preparation of lipoprotein-deficient Serum (LPDS)

To prepare lipoprotein-deficient serum 1 ml was assayed for protein. To every 120 ml of serum 47.71 g potassium bromide was added to adjust the density to 1.25 g/ml stirred by hand then transferred to 35ml ultra clear centrifuge tubes and centrifuged at 40,000 rpm at 10 °C for 48 h. After 48 h the tubes were sliced at the upper layer to remove lipoproteins at the top. The remaining serum was then dialysed against 10 litres of dialysis buffer at 4 °C, the buffer was changed several times overnight. It was then dialysed against 10 litres of PBS containing calcium and magnesium with two changes for 3 h. The protein was assayed again and readjusted to the original protein concentration by adding PBS then filtered through a 0.22 μ M filter.

2.3 Spectrophotometric measurement of LDL oxidation by ferritin at lysosomal pH

Native LDL (50 μ g protein/ml) was oxidised with freshly prepared ferritin (0.05 μ M, 0.1 μ M and 0.2 μ M) in Chelex-100 treated NaCl/ sodium acetate buffer (150 mM NaCl, 10 mM acetic acid) (pH 4.5) at 37 °C. The Chelex-100, pre-washed in ultra-pure water was added to NaCl/sodium acetate buffer at 0.1% (w/v, 1 g/l), to remove soluble transition metal binding activity (Van Reyk *et al.*, 1995). The Chelex-100 treated NaCl/ sodium acetate buffer pH 4.5 left at 4 °C overnight with stirring. The Chelex-100 was then removed by filtration prior to readjusting pH to 4.5. The formation of conjugated diene was measured according to the method previously used by Esterbauer to continuously monitor LDL oxidation (Esterbauer *et al.*, 1989b). Freshly dissolved LDL (50 μ g protein/ml) and the different concentrations of freshly diluted ferritin was placed in 10 mm path length quartz cuvette in duplicates against reference cuvettes that lacked LDL. Ferritin was added last in all experiments. To measure the formation of conjugated dienes, the attenuance (absorbance plus UV scattering) at 234nm was measured at 37°C for 1200 minutes at a one-minute interval in a dual Lambda Bio 40 8-cell spectrophotometer with UV software. The attenuance at zero was subtracted from all time points.

2.3.1 Comparison of oxidation of LDL by ferritin at pH 4.5 and pH 7.4

The effect of changes in pH on the oxidation of LDL by ferritin was assessed. Native LDL (50 μ g protein/ml) was oxidised with 0.1 μ M or 0.2 μ M ferritin at 37°C using NaCl/ sodium acetate buffer (pH 4.5) and MOPS buffer (pH 7.4) both pre-treated with Chelex - 100. The formation of conjugated dienes was monitored.

2.3.2 Measurement of LDL aggregation in the presence of ferritin

Native LDL (50 μ g protein/ml) was oxidised with freshly prepared ferritin (0.1 μ M) in Chelex - 100 treated NaCl/ sodium acetate buffer (pH 4.5) at 37°C in capped quartz cuvettes. The method previously used by Khoo *et al* was adopted (Khoo *et al.*, 1988). Light scattering by LDL oxidised with freshly prepared ferritin was compared to LDL or ferritin by itself. The change in attenuance at 680 nm was monitored every minute at 37°C against reference cuvettes that lacked LDL. The attenuance was measured for 1200 minutes.

2.3.3 Atomic absorption spectrophotometric quantification of iron in ferritin and apoferritin

The amount of iron in horse spleen ferritin was determined using flame atomic absorption spectrophotometry (F-AAS). To digest the sample, 1 ml of 5 M nitric acid was added to ferritin (15 μ l of 53 mg/ml protein) or apoferritin (35 μ l of 25mg/ml) was prepared in triplicate in test tubes and heated for 2 h at 75° C. The samples were dissolved in 50 ml and 10 ml of water, respectively. Standard concentrations of iron (0, 1, 2, 3, 4, 5 mg/l) were prepared from a stock solution of iron (Fe³⁺) (10 mg/l) and treated the same way as the sample. Samples of the nitric acid digest were assayed directly in an atomic absorbance spectrophotometer (AAS) (Analytik JENA AAS NovAA) using an air/acetylene flame. Absorbance was measured at 248.3nm.

2.3.4 Comparison of iron released from ferritin at pH 4.5 and 7.4 using iron chelator

To determine the amount of iron released from ferritin at pH 4.5 and 7.4. Native LDL (50 μ g protein/ml) was oxidised with freshly prepared ferritin (0.1 μ M) in Chelex-treated NaCl/ sodium acetate buffer (pH 4.5) or MOPS buffer (pH 7.4) in 15ml tubes, in triplicate. These were incubated at 37°C. Aliquots of 1ml were taken at intervals up to 24 hours. Bathophenanthroline (BP) (30 μ l of 10mM) was added at different time points and absorbance was measured immediately at 535nm using the Biochrom Libra S22 spectrophotometer.

2.3.5 Comparison of iron released from ferritin by the ultrafiltration method

All solutions were prepared prior to use. Ferritin was incubated in triplicate at 0.1 μ M in Chelex-treated NaCl/ sodium acetate buffer (pH 4.5) in 15ml tubes at 37°C for 24 h, then

filtered with 30,000 Mr cut off microcentrifuge filter tubes at 9000g for 25 min. The quantity of iron in the filtrate was determined using AAS.

2.3.6 The effect of DTPA and EDTA on LDL oxidation by ferritin

The effect of iron chelators on LDL oxidation by ferritin was tested. Native LDL (50 μ g protein /ml) was oxidised with 0.1 μ M ferritin in the absence or presence of 100 μ M EDTA or DTPA. The formation of conjugated dienes was continuously monitored by measuring attenuance at 234nm for 1200 minutes at one minute interval.

2.3.7 Effect of antioxidants on LDL oxidation by ferritin

The effect of antioxidants (tempol, α -tocopherol, ascorbate, DPPD and cysteamine) on oxidation of LDL by ferritin was tested. Native LDL (50 µg protein/ml) was oxidised with 0.1µM of ferritin without or with enrichment with α -tocopherol or in the presence of tempol (10 µM) or different concentrations of ascorbate (10-100 µM), DPPD (5-10 µM) or cysteamine (5µM-10mM) in Chelex - 100 treated NaCl/sodium acetate buffer (pH 4.5) at 37°C. The samples were placed against appropriate reference cuvette that lacked LDL. The formation of conjugated dienes was monitored by measuring attenuance at 234nm for 1200 minutes at one-minute intervals.

2.3.8 Effect of cysteamine on iron released by ferritin

Iron released from ferritin in the presence of cysteamine was assessed. Native LDL 50 μ g protein/ml) was oxidised with freshly prepared ferritin (0.1 μ M) in Chelex-treated NaCl/acetate buffer (pH 4.5) in the presence or absence of cysteamine in 15ml tubes. A fraction of 1ml was taken at intervals up to 24 h. BP (30 μ l of 10mM) was added at different time points, absorbance was measured immediately at 535nm using the Biochrom Libra S22 spectrophotometer.

2.4 Measurement of LDL oxidation by reverse-phase HPLC analysis of lipid species

Reverse phase HPLC was used to analyse LDL oxidised by ferritin following a method previously used by Kritharides *et al.* (Kritharides *et al.*, 1993). The method of preparation of samples and analysis is described below. PerkinElmer 200 and Agilent 1200 system were used, data analysis was done using Totalchrom workstation and Chemstation software, respectively. The column used on both machines was a C18 reverse phase column, 250 mm x 4.6 mm, 5 μ m particle size with 5 μ m guard column. Reverse phase HPLC uses Stationary phase that is hydrophobic (non–polar) and an aqueous mobile phase (polar). The oxidised Lipid products were detected at 234 nm.

2.4.1 LDL oxidation with ferritin for HPLC analysis

Native LDL was incubated with or without freshly dissolved ferritin $(0.1\mu M)$ at 37°C using NaCl/Na acetate buffer (pH 4.5) in 15ml polypropylene tubes. At different time points for up to 48 h, the oxidation was stopped at each time point by adding BHT in ethanol stock concentration at 2 mM (final concentration of 80 μ M) and EDTA (final concentration 4mM). The lipids were then extracted as described in the next section.

2.4.2 Extraction of lipids from the oxidised sample for HPLC analysis

Methanol (1ml) was added to each oxidised sample (80 μ g LDL protein) and then vortexed for 10 seconds. Hexane (3ml) was added and the samples were vortexed for 30 seconds in order to transfer the hydrophobic fraction into hexane. The sample was then centrifuged at 1500 x g at room temperature for 15 min, 2ml of the upper hexane layer was then transferred to a 5 ml propylene tube and dried in SpeedVac (ThermoFisher). The dried samples were dissolved in 200 μ l of a relevant mobile phase and analysed.

2.4.3 HPLC Measurement

Each sample (20 μ l) was injected into a reverse-phase HPLC and C18 column (described in section 2.4). The Oxidised products 7-keto cholesterol (7-Keto) and cholesteryl linoleate hydroperoxide (CLOOH) were detected at 234 nm using a mobile phase as described in section 2.4.3.1. Example chromatograms are shown in figures 2.2 & 2.3. Oxidised products were quantified by comparing the peak area with those of known concentrations of commercially available standards.

2.4.3.1 Conditions for analysis of targeted lipid species

The targeted lipid species were 7-ketocholesterol and cholesteryl linoleate hydroperoxide. Mobile phase was prepared with 2% water, 44% acetonitrile, 54% isopropanol (by vol.) and oxidised lipids were detected at a wavelength of 234 nm at a flow rate of 1ml/min for 15 minutes.



Figure 2.2: Example chromatogram showing 7-ketocholesterol

Mobile phase as described in section 2.4.3.1 was used with a flow rate of 1 ml/min and detected at a wavelength of 234 nm in 7-ketocholesterol standard.



Figure 2.3: Example chromatogram showing cholesteryl linoleate hydroperoxide

Mobile phase as described in section 2.4.3.1 was used with a flow rate of 1 ml/min and detected at a wavelength of 234 nm in the cholesteryl linoleate hydroperoxide standard.

2.5 Measurement of LDL oxidation by determination of total lipid

hydroperoxide content by the tri-iodide assay

The method previously described by El-Saadani et al.(el-Saadani et al., 1989) was adapted to measure the total lipid hydroperoxides in LDL oxidised by ferritin at lysosomal pH. The samples of LDL used for assay had a final mass of 25 μ g LDL protein per tube. Standard masses of H₂O₂ (0, 5, 10, 15, 20, 25, 50 nmol) was prepared from a stock solution of H₂O₂ (100 μ M). LDL samples oxidised with ferritin as described in section 2.4.1 were made up in 15 ml polypropylene tubes (25 μ g protein/ml). Colour reagent (composition described in section 2.1.1) was then added (1 ml) to standards of H₂O₂ and LDL samples and vortexed. The samples were then placed in the dark for 1 hour at room temperature. The samples were read in a spectrophotometer at 365 nm. The total lipid hydroperoxides were determined in nmol/mg LDL protein. This assay is based on the ability of peroxides in standards and sample to oxidise iodide ions in the colour reagent to form molecular iodine which then reacts with excess iodide ions present in the colour reagent to form the tri-iodide chromophore (I₃⁻) which absorbs ultraviolet radiation at 365 nm.

2.6 Proteolytic degradation of ferritin

2.6.1 Enzyme digestions

2.6.1.1 Incubation of ferritin with cathepsin D

Enzyme digestions were carried out at 37°C, the protein solution was placed in NaCl/sodium acetate buffer (pH 4.5). Ferritin (2mg/ml) was incubated with or without 6.2 μ g/ml cathepsin D and incubated for 24 h.

2.6.1.2 Incubation of ferritin with cathepsin D and B

Ferritin (500 μ g protein/ml) was incubated with or without 5 μ g/ml cathepsin D and 5 μ g/ml cathepsin B in Nacl/sodium acetate buffer pH 4.5 at 37°C for 24 h.

2.6.1.3 Incubation of LDL with cathepsin D

The intact LDL (500 μ g protein/ml) was incubated in the presence or absence of cathepsin D (5 μ g/ml) in NaCl/sodium acetate buffer (pH 4.4) at 37°C for 24 h.

2.6.1.3 Co-Incubation of LDL with cholesteryl esterase.

LDL (50 μ g protein /ml) was co-incubated with 0.0625 unit/ml of cholesterol esterase from *Pseudomonas sp* which is active at pH 4.5. In another experiment, LDL (50 μ g protein /ml) was co-incubated with 1.25 μ g/ml cathepsin D, 1.25 μ g/ml cathepsin B and 0.0625 unit of cholesteryl esterase/ml.

2.6.2 Electrophoresis

The intact ferritin, LDL and digested proteins were evaluated by gel electrophoresis under reducing condition using 6X RSTB. The treatment with SDS and β -mercaptoethanol contained in the RSTB was used to produce a linear structure of the proteins. The SDS denatures the secondary and tertiary structures while the β -mercaptoethanol reduces and breaks the disulphide bonds. The SDS confers a uniform negative charge on the proteins by masking the native charge of the protein, hence enable them to migrate based on size. Ferritin contains heavy and light chains (21 kDa and 19 kDa). Samples (15 µg) were loaded per well in 15% polyacrylamide gels for ferritin and 5% polyacrylamide gel for LDL along with molecular weight markers and the gel was left to run at 150 volts for 1 h. The protein bands were located by staining with Coomassie blue for 1 h and then de-stained for several hours.

2.6.3 Effect of proteases and cholesterol esterase on LDL oxidation by ferritin

The effect of enzyme digestion on the oxidation of LDL by ferritin was tested. Native LDL (50 μ g protein/ml) was oxidised with 0.1 μ M ferritin only or ferritin that was pre-incubated

with cathepsin D only or with the addition of cathepsin B or co-incubated with cholesterol esterase from Pseudomonas sp. Oxidation of LDL degraded with cathepsin D by ferritin was also compared to intact LDL in Chelex - 100 treated NaCl/sodium acetate buffer (pH 4.5) at 37°C. Control samples not treated with enzymes were compared to samples treated with enzymes. The samples were placed against appropriate reference cuvettes that lacked LDL. The formation of conjugated dienes was monitored by measuring attenuance at 234nm for 1200 minutes at one-minute intervals.

2.7 Cell culture

THP-1 cells originally obtained from the blood of a one-year-old leukaemic patient were bought from European Collection of Authenticated Cell Cultures (EACC). They are monocytic cell line which can be differentiated into macrophages with phorbol 12-myristate 12-acetate (PMA). The THP-1 monocyte was maintained in RPMI 1640 medium modified with Lglutamine, phenol red, HEPES, sodium pyruvate with low sodium bicarbonate and high glucose. The medium obtained from Life Technologies, UK was supplemented with 10% (v/v) heat-inactivated foetal calf serum (FCS), antibiotics (50 UI penicillin, 50 µg streptomycin and 0.95 µg amphotericin/ml). Cells were cultured in T75 flasks with filter caps using 20 ml media. Cells were cultured under humidified air (95%) / 5% CO₂ at 37°C until confluent. Cells were subcultured usually every three to four days. The cell suspension was drawn out of a T75 flask into 50ml tubes and centrifuged at 500 x g for 5 min at room temperature. The supernatant was discarded and cells re-suspended in 5 to 7ml media, counted and re-seeded at 3 x 10⁵/ml in 20ml culture medium in the new flasks.

NB: All media used in the experimental procedures were supplemented with 10% FCS and antibiotics (50UI penicillin, 50 μ g streptomycin and 0.95 μ g amphotericin/ml), except stated otherwise.

2.7.1 Cryopreservation of THP-1 monocytes.

Cells were collected as described above and centrifuged at 500 x g for 5 min at room temperature. The supernatant was removed, and the pellet was re-suspended in an ice-cold cryopreservation medium containing RPMI/FCS/DMSO (5/4/1 by vol.) using 1 ml for each confluent T75 flask. The cells were then kept in cryovials packed in polystyrene containers to ensure slow cooling then transferred to a -80 °C freezer overnight. Vials were moved to liquid nitrogen to store for a longer period. When needed vials were taken out, rapidly thawed in a water bath at 37 °C. Each 1 ml content was transferred to 50 ml tubes and washed with RPMI by centrifuging twice at room temperature, then seeded in T75 flask with medium containing 20% FCS (volume in flask was 20 ml) then transferred to an incubator left in standing position for about 3 hours then placed in a normal position for five days after which they are split into media containing 10% FBS at 3 X 10⁵ cells/ml in 20ml culture medium.

2.7.2 Conversion of THP-1 cells to THP-1 derived macrophages

The THP-1 cells were cultured inT75 flask at 3 x 10^5 cells per ml and allowed to reach the density of 8 x 10^5 cells per ml before changing media. The viability of cells was maintained by not allowing the density to exceed 1 x 10^6 cells per ml. Depending on the experiment to be carried out, THP-1 cells were differentiated in tissue culture plates (6-, 12-, 24- or 96-well or) using phorbol 12-myristate 12-acetate (PMA) (25ng/ml). After 72 h cells were washed with sterile PBS and re-suspended in fresh media. Cells were then rested for 48 h before experiments were carried out.

2.8 Isolation of human monocyte-derived macrophages

Human monocytes were isolated from blood collected of healthy volunteers(as described in section2.2.1) by density gradient centrifugation with a LymphoprepTM solution (AXIS-SHIELD PoC AS, Oslo, Norway) (Firth *et al.*, 2007). Erythrocytes and granulocytes have a
density greater than 1.077g/ml, while monocytes have a density of less than 1.077g/ml. LymphoprepTM has a density of around 1.077g/ml and it, therefore, allows monocytes to be maintained at the interface between LymphoprepTM and the buffer in which the cells were suspended while erythrocytes and granulocytes sediments.

The blood collected from donors were transferred to 50ml centrifuge tubes and centrifuged at 1500g for 30 minutes at 4 °C. The plasma was carefully removed leaving about 1 cm of plasma above the buffy coat. Using a sterile Pasteur pipette the buffy coat was removed into new 50ml Falcon tubes. About 15ml buffy coat was obtained (containing red blood cells which were removed in the next step). Then 20 ml of PBS (at room temperature was added. The mixture of buffy coat and PBS was gently underlaid with 15 ml of LymphoprepTM. The solution was then centrifuged at 1000g at room temperature for 20 minutes with the brake off and soft start. The red blood cells were visible at the bottom and the white monocyte/lymphocyte floating half way down the tube was collected (approximately 15 ml per tube). PBS (at room temperature) was added to each tube to a final volume of 45 ml and tubes were inverted to mix and then centrifuged at 500g at room temperature for 15 minutes (with the break on at the end). The supernatant was taken off and the pellet re-suspended in 50 ml PBS and then centrifuged at 500g for 10 minutes. This process was repeated 3 times to ensure cells were fully washed. The cells concentration was adjusted to 5×10^6 / ml in serum-free RPMI (containing antibiotics) and then plated at 6 ml in 6 well suspension plates. Heat-inactivated human serum was added at 5 μ l /ml to each well and plates were incubated at 37 °C / 5 % CO₂ for 40 hours to allow T cell death and platelet adhesion while monocytes remain in suspension and viable.

The cell suspension from each well was collected into tubes pipetting thoroughly within the wells to collect resting monocytes whilst avoiding detachment of platelets. The cell suspension was centrifuged at 500 g for 15 minutes at room temperature. The supernatant was removed and re-suspended in RPMI media containing 10% heat-inactivated foetal calf serum and antibiotics (50 UI, 50 μ g streptomycin and 0.95 μ g amphotericin/ml). The cells were seeded at 3 x 10⁶ / ml in 12 well adherent plates and GM-CSF (25 ng/ml) was added to the 1ml of suspension in each well (to differentiate them into M1 (pro-inflammatory) macrophages) and then transferred to an incubator at 37 °C / 5 % CO₂. The cells' gross morphology was examined with a microscope to confirm they were monocytes. The media was changed every 3 days and monocytes were allowed to differentiate into macrophages after 7 days. The cells were plated according to the requirements of the particular experiments.

2.9 Determination of the effect of hepcidin on intracellular iron in

macrophages

2.9.1 Treatment of THP-1 derived macrophage cells with hepcidin.

THP-1 derived macrophages (1.2×10^5) differentiated with PMA (25ng/ml) and rested for 48 h were plated in 24 well tissue culture plates using three media to vary the media concentration of iron (RPMI-1640, DMEM and F10 supplemented with 10% FCS and antibiotics). For each media cells were treated with or without hepcidin (1000 nM) and media was replaced after 24 h with media containing 10% LPDS with or without hepcidin (100nM) every 2 days. The protein content and iron concentration were measured after seven days.

2.9.2 Determination of intracellular protein concentrations

2.9.2.1 Preparation of cell lysate

The medium was removed, the plates were then kept on ice. Cells were then washed with cold PBS. $500 \ \mu$ l RIPA lysis buffer with protease inhibitors was added to each well and then swirled to distribute buffer. The cells were gently scraped and transferred into tubes. The cells were then incubated on ice for 15 minutes. The lysate was then sonicated and centrifuged at 13,000 g for 5 minutes and the supernatant collected.

2.9.2.2.Determination of protein content with Bio-Rad DCTM protein assay kit.

BSA standards (0, 5, 25, 50, 100 μ g) was prepared and the protein concentration in samples was measured using a Bio-Rad DCTM protein assay kit. This assay is based on a similar principle with Lowry assay in which the protein content of the sample reacts with copper tartrate and then the copper treated protein subsequently reduces the Folin reagent. This reduction produces reduced species with characteristic blue colour which absorbs maximally at 750 nm. Reagent A was added to the reagent S ratio (200 μ l: 4 μ l). Reagent A (5000 μ l) and 100 μ l of reagent S. This mixture (200 μ l) was then added to each standard and samples. Reagent B (800 μ l) was then added, each tube was vortexed and left for 15 minutes at room temperature. The absorbance was measured at 750 nm. The protein contained in the cell lysates were then determined in comparison with the BSA standards.

2.9.3 Inductively coupled plasma mass spectrometry (ICP-MS) for measuring total iron in macrophages

THP-1 macrophage-like cells treated with or without hepcidin as described in section 2.9.1 were transferred to 15ml tubes and centrifuged at 500 g for 10 min. Cell pellets were washed with PBS. The cell pellets were then digested with nitric acid, equivalent to 2% of the final volume, and incubated at 75°C for 2 h. It was then diluted to 10ml with ultra-pure water. The concentration of iron in samples were measured with ICP-MS (Flores *et al.*, 2015). The use of

ICP-MS is an established method for the analysis of multiple elements, very sensitive and able to detect as low as the ultra-trace concentration of varying elements. The sample is introduced to the system true the liquid-phase nebulizer. The major analytical feature of the system is the production of ions in the inductively coupled plasma (ICP) section of the machine. The ICP mass spectra show the amount of ions measured in every second.

2.10 Measurement of lysosomal lipid peroxidation in THP-1 cells and human monocyte-derived macrophages.

The local lipid peroxidation (LPO) in lysosomes of macrophages was measured with the novel lysosome-targeted probe Foam-LPO which was recently synthesised by Zhang and colleagues (Zhang *et al.*, 2015). The Foam-LPO is a fluorescent probe, which possesses an alkaline tertiary amino group which enables it to localise in the acidic environment of lysosomes (becomes protonated and accumulates in the lysosome). Its fluorophore structure acts as a signal of lipid peroxidation and the conjugated diene group present in the BODIPY fluorophore are degraded in response to lipid peroxidation leading to a fluorescence shift from 586 nm to 512 nm which can be measured by two-way flow cytometry.

THP-1 macrophages prepared as described in section 2.7.2 were plated at 5 x 10^5 cells per well were treated with pre-warmed RPMI–1640 with 10% human serum (2ml per well) alone or with ferritin (100 µg/ml) for 24 h in an incubator at 37 °C / 5 % CO₂. The cells were then treated with or without LDL (200 µg protein/ml) for 24 h to have the following four conditions: no LDL, LDL only, and LDL with ferritin and ferritin only. After the total 48 h treatment for ferritin and LDL treatment, the cells were washed three times with PBS (pre-warmed to 37 °C) and then added fresh media. The adhered macrophages were scraped into a clear 96 well round bottom microplate (Greiner CellStar®) and centrifuged at 500 g at room temperature for 5 min. The cells were then re-suspended in 200 μ l media (no serum) containing 2 μ M Foam-LPO (a kind gift from Dr Xiao, Dalian University China) for 15 min at room temperature in the dark. The cells were then centrifuged at 500 g for 5 min at 4 °C and the supernatant was discarded and the cells re-suspended in PBS and centrifuged at 500 g for 5 min at 4 °C. Washing with PBS was repeated once and then the cells were re-suspended in flow cytometer buffer.

The cells were analysed with a Becton Dickinson flow cytometer (BD Biosciences C6 flow cytometer). FlowJo software was used to determine the mean fluorescence intensity of the two channels and the ratio of green channel (FL1) to the red channel (FL2) was used as a measure of LPO in lysosomes.

The experiment was also carried out using the same procedure with sphingomyelinaseaggregated LDL with THP-1 cells or human monocytes-derived macrophages (HMDM) isolated as described in section 2.8.

2.11 Measurement of intracellular lipid and ceroid levels after treatment with ferritin

2.11.1 Preparation of sterilised coverslips

Prior to the start of experiments sterile coverslips (18 x 18 mm) were cleaned in acid (65% nitric acid left overnight on a rocker at low rpm overnight well labelled. Coverslips were then washed in ultra-pure water several times. The coverslips were then dried in the safety cabinet to dry and then placed in a container and autoclaved. Prior to each experiment–requiring coverslips, the cover slips were exposed to UV light to further sterilise.

2.11.2 Measurement of intracellular lipids and intralysosomal ceroid

THP-1 cells differentiated with PMA (25ng/ml) were plated at 45,000 on sterilised coverslips in two 6-well tissue cultured plates. In each plate, three wells were pre-treated with 100 μ g ferritin for 24 h. Cells were washed with PBS and RPMI-1640 media (2ml) was added to two wells (RPMI only and RPMI + preloaded ferritin), native LDL (200 μ g protein/ml) was added to next two wells (native LDL only and native LDL + preloaded ferritin), the last two wells received 200 μ g protein/ml SMase-LDL (SMase-LDL only and SMase-LDL + preloaded ferritin) and the treatment was left for 24 h. The cells were then washed with PBS and 2ml of RPMI-1640 containing 10% (v/v) lipoprotein deficient serum and antibiotics (50UI, 50 μ g streptomycin and 0.95 μ g amphotericin/ml). This was replaced every two days for each well for seven d the cells in one plate were stained for intracellular lipids and the other for ceroid.

2.11.3 Oil Red O Staining for intracellular lipids

The tissue culture plate was transferred to the fume hood, the media was then replaced with 2ml of PBS. The PBS was then replaced with 2ml of 4% (w/v) paraformaldehyde and then incubated for 15 minutes. Paraformaldehyde was removed into a waste container. Cells were gently rinsed with distilled water. Cells were washed with PBS and after washing, 1.5 ml Oil Red O (0.5% w/v) working solution was added to each well and then incubated for 10 minutes. After 10 minutes. Oil Red O was removed and cells were gently rinsed with water until the distilled water rinsed clear. Coverslips were then mounted on glass slides with cells facing downwards on a drop of mounting media. The intracellular lipids were visualised using light microscopy with Axiovision software and analysed with Image J software.

2.11.4 Oil Red O Staining for Ceroid

At the end of treatment, cells were washed two times with PBS and then fixed by adding 2ml 4% paraformaldehyde for 15 minutes. The cells were gently rinsed with 2ml water and then washed with twice with 2ml PBS. The cells on coverslips were transferred to a glass staining rack and 2ml of xylene was added to each well and incubated for 5 minutes. Each well was then rinsed with 60% (v/v) isopropanol. 2ml ethanol was then added and incubated for another 5 minutes. Oil Red O (1.5 ml) was added and incubated for 15 minutes. The Oil Red O was removed and then rinsed with distilled water several times until it appeared clear. The ceroid content was then visualised with light microscopy (Axioscop 2, Carl Zeiss Ltd) and images were captured with Axiovision software and analysed with Image J software.

2.12 Determination of intracellular reactive oxygen species formation induced by ferritin-oxidised LDL

THP-1 cells (1.5 x 10^5 /ml) was added to PMA (25 ng/ml) with pre-warmed RPMI– 640 media containing 10% (v/v) heat-inactivated foetal calf serum and antibiotics (50UI, 50 µg streptomycin and 0.95 µg amphotericin/ml). The cell suspension (300 µl) (4.5 x 10^4) of THP-1 cells was placed in tissue culture-treated coverslips in 6-well tissue culture plates for 72 h with humidified air (95%) / 5% CO₂ at 37°C After 72 h the cells were washed with PBS and rested in pre-warmed RPMI – 1640 for 48 h. (coverslips were prepared as described in section 2.11.1)

Prior to treating cells, oxidised LDL was prepared by incubating (500 μ g LDL protein/ml) with 1 μ M ferritin at 37°C for 24 h NaCl/sodium acetate buffer (pH 4.5). LDL (500 μ g protein/ml),

ferritin (1 μ M) and buffer (equivalent volume), each in separate tubes were also incubated at 37°C for 24 h.

The THP-1 cells on coverslips were then incubated with pre-warmed RPMI–1640 media alone (2 ml per well) or containing either NaCl/ sodium acetate buffer or LDL (50 μ g protein/ml) or ferritin (0.2 μ M) or oxidised LDL (50 μ g protein/ml). After incubation for 24 h with humidified air (95%) / 5% CO₂. The cells were washed twice with PBS and incubated with 10 μ M dihydroethidium (DHE) in PBS (this working solution was prepared on the day of the experiment from 20mM (10 mg in 1585 μ l DMSO). Placed in a non-CO₂ incubator in the dark for 30 min. The cells were then washed twice with PBS and the cells were mounted with fluorescence mounting media containing DAPI. Images were captured using Axioimager. The intensity of the DHE stain was determined using ImageJ software

2.13 Determination of the effect of oxidation of LDL by ferritin on cellular respiration/metabolism using a Seahorse analyser.

The Seahorse XF analyser is an important tool in the measurement of cellular energy metabolism. The Seahorse can be used to determine the energy metabolism in live cells by monitoring the oxygen uptake and the pH in real time.

THP-1 cells were differentiated in T25 flasks using PMA (25ng/ml) at 3 x 10^5 cells/ ml for 72 h with humidified air (95%) / 5% CO₂ at 37°C with pre-warmed RPMI – 1640 media containing 10% (v/v) heat-inactivated foetal calf serum and antibiotics (50UI, 50 µg streptomycin and 0.95 µg amphotericin/ml). After 72 h the cells were washed and rested in pre-warmed RPMI-

1640 for 48 h prior to treatment with LDL. After a 24 h rest period, the cells were detached from the flask using Accutase and transferred to an Agilent Seahorse XFp cell culture mini plate to rest for another 24 h. The mini-plate has 8 wells labelled A to H. The wells A and H always served as blanks with no cells, just media, while wells 6 x 10^4 cells were plated in 200 µl for wells B to G.

On the day before the assay, the Agilent cartridge was hydrated with calibrant by adding 200 μ l calibrant in the wells A-H and 400 μ l PBS into the moat on the side. The sensor cartridge was then placed on to the plate submerging the sensors in the calibrant and then incubated in a non CO₂ incubator at 37°C for 24 h. The cells in Agilent plate was treated as follow, 3 wells E, F and G were pre-treated with ferritin (0.2 μ M) for 24 h. After treatment with ferritin, the cells in position B got RPMI media only, C and D got native LDL (100 μ g protein/ml), position E got only the pre-incubated ferritin while F and G got native LDL (100 μ g protein/ml) in addition to pre-incubated ferritin.

On the day of the assay, 10 ml of XF base media was prepared, glucose (10 mM), pyruvate (1 mM) and Glutamine (2 mM) were added. The pH of the media was then adjusted to pH 7.4 and warmed up to 37° C and filtered with a 0.2 µm Minisart filter. The cells in the Agilent mini plates were washed with XF base media freshly prepared by taking 180 µl of the RPMI-1640 from treated cells and replacing it with180 µl XF base media. This step was repeated 2 times using XF base media and wells A to H were then made up to a final volume of 180 µl. The cells in the Agilent plate was then placed in the non-CO₂ incubator at 37° C for 1 hour.

The Agilent Seahorse XFp analyser was turned on to allow it warm up to 37° C. A stressor mix was immediately prepared. This was prepared in accordance with Agilent Seahorse XFp cell energy phenotype kit. XF base media (120 µl) was added to 60 µl oligomycin (2 µM/well) and 120 µl FCCP (2 µM/well). The calibrated cartridge was brought out of the incubator after 24 h, 20 µl of the stressor mix was added to ports A of all the 8 positions of the hydrated cartridge (given 10 µM oligomycin and 20 µM FCCP in each of the port A). The energy phenotype test was selected on the Seahorse XFp analyser. The number of cycles for base line and after oligomycin injection was set at five cycles each. The cartridge plate already loaded with the stressor mix in each port A was then placed in the instrument for calibration. The calibration process takes approximately 20 minutes. The bottom plate of the cartridge was then replaced with the cells in the Agilent tissue culture plate that had stayed in the non CO₂ incubator for 1 h. The plate was placed without the lid. The energy phenotype test was then carried measuring the oxygen consumption rate (OCR) and the extracellular acid Rate (ECAR).

2.14 Determination of apoptosis in macrophages treated with ferritinoxidised LDL

Phosphatidylserine in healthy cells is mainly located on the cytosolic side of the plasma membrane. The externalisation of phosphatidylserine in cells to the surface of the cells is an early indication of apoptosis (Martin *et al.*, 1995). Annexin V binds to externalised phosphatidylserine and fluorescent annexin V can be used to detect apoptotic cells. Apoptosis was measured using the commercially available kit (FITC Annexin V Apoptosis Detection Kit with PI, BioLegend, San Diego, CA, USA) and flow cytometry. The kit contains fluorescein isothiocyanate (FITC) - annexin V conjugate which can be used to identify apoptotic cells. Propidium Iodide (PI) was added to separate the necrotic cells from apoptotic cells. PI has the

capacity to enter cells and stain the DNA of cells undergoing necrosis. Cells that possess increased levels of PI labelling are necrotic, cells with increased levels of fluorescent annexin V labelling are apoptotic and cells having high levels of both annexin V and PI are going through secondary necrosis.

Prior to treating cells, oxidised LDL was prepared by incubating (1 mg LDL protein/ml) with 2 μ M ferritin at 37°C for 24 h in NaCl/sodium acetate buffer (pH 4.5). LDL (1 mg protein/ml), ferritin (2 μ M) and buffer (equivalent volume), each in separate tubes were also incubated at 37°C for 24 h.

THP-1 cells were differentiated in 12-well tissue culture plates at 5 x 10^5 cells per well using PMA (25ng/ml) as described in section 2.7.1. After 72 h the cells were washed and rested in pre-warmed RPMI – 1640 for 48 h prior to treatment with LDL. The cells were then incubated with pre-warmed RPMI – 1640 media alone (2 ml per well) or containing either NaCl/sodium acetate buffer or LDL (100 µg protein/ml) or ferritin (0.2 µM) or oxidised LDL (100 µg protein/ml). After incubation for 48 h with humidified air (95%) / 5% CO₂ at 37 °C, the medium was removed from the cells and kept for each well. To detach cells, Accutase (750 µl) was added and then placed in an incubator for 10 min. The removed media was then added accordingly to each well (750 µl). Each well was then collected into an Eppendorf tube and centrifuged at 100g for 5min. The pellet was washed twice with cell staining buffer. The cells were re-suspended in annexin V binding buffer (800 µl), then 250 µl was transferred to brown Eppendorf tube for protection from light. PI (2.5 µg) and (FITC)-annexin V (0.5 µg) was added and then incubated for 15 min at room temperature in the dark. The volume was made up to 500 µl and analysed immediately using a Becton Dickinson FACScan flow cytometer (BD Biosciences C6 flow cytometer). The percentage of healthy cells, necrotic, apoptotic and

secondary necrotic cells was determined using the BD Biosciences C6 flow cytometer software.

2.15 Statistical analysis

The data are presented as the mean plus or minus standard error of the mean (SEM) of the number of independent experiments indicated. Data were statistically analysed using a Student's paired t test, one way analysis of variance (ANOVA) followed by Tukey's post hoc test but when comparing multiple parameters, we made use of two-way ANOVA and a Bonferroni post hoc test to determine the significance of the criterion tested. GraphPad prism 5 software was used for all statistical analysis carried out. For all hypothesis we tested, 0.05 was considered as the significance level. Hence, a P value < 0.05 was considered a statistical significance.

Chapter 3 - Mechanisms of oxidation of LDL by ferritin at lysosomal pH

3.0 Background and rational

Research over the last few decades has proposed oxidised LDL to be an important marker in the onset and progression of atherosclerosis. The Oxidised LDL hypothesis emphasised the fact that oxidative modification of LDL promotes its atherogenic nature. Oxidised LDL instigates many pro-atherogenic effects ranging from induction of formation of lipid-laden macrophages (foam cells) to the promotion of inflammatory responses hence activating endothelial cells and promoting smooth cells proliferation. The mechanism by which LDL is oxidised *in vivo*, the site of oxidation and when LDL is oxidised is continuously deliberated amongst researchers (Itabe *et al.*, 2011).

Many mechanisms have been proposed for oxidation of LDL ranging from oxidation by cells, which are dependent on the availability of free metal ions to free metal ion–independent oxidation mediated by enzymes such as lipoxygenase and myeloperoxidase (Yoshida and Kisugi, 2010). The oxidation of LDL by free metal ions is the most studied model of LDL oxidation *in vitro* and the oxidation of LDL by cells of the arterial walls indeed requires the presence of certain concentrations of copper or iron (Leake and Rankin, 1990, Leeuwenburgh *et al.*, 1997). Transition metal ions have been confirmed to catalyse the oxidation of DNA, protein, and lipids. Increased levels of transition metals have also been found in animal models and human atherosclerotic plaques (Stadler *et al.*, 2004, Stanley *et al.*, 2006). Hence, this supports the suggestion that metal ions contribute to the onset and progression of atherosclerosis.

The iron hypothesis of atherosclerosis proposed that men and post-menopausal women are more susceptible to heart diseases due to elevated levels of body iron compared to premenopausal women with lower levels of stored iron as a result of monthly bleeding (Sullivan, 1981). Over the last three decades, several researchers have measured the correlation between iron levels in the body and atherosclerosis. Despite the continuous research in this field, this hypothesis is continuously debated and remains controversial (Munoz-Bravo *et al.*, 2013, Aursulesei *et al.*, 2014). Sullivan, however, subsequently argued that this gender difference in the incidence of heart diseases is linked to body iron stores and not the oestrogen levels (Sullivan, 1989, Sullivan, 1992, Sullivan, 2003, sullivan, 2005, Sullivan, 2007). Further to this other studies have demonstrated an association between iron deposition and the progression of atherosclerosis (Lee *et al.*, 1998, Stadler *et al.*, 2004).

Iron is important in maintaining the integrity of cells and its function by mediating various metabolic systems through enzymes that are dependent on iron. Approximately all iron in the human body is bound to molecules. The existence of iron in bound form is highly advantageous to the body, as iron is capable of participating in redox reactions that are harmful to the body. It has the ability to exist in several oxidation states ranging from Fe^{2+} to Fe^{6+} . However, the transfer of one electron between ferric ion (Fe^{3+}) and ferrous iron (Fe^{2+}) has been implicated in iron–dependent free radical reactions, which have been linked to cell injury and several pathophysiological reactions (Ramakrishna *et al.*, 2003, Yuan and Li, 2008).

The redox-active iron pool, also referred to as labile iron pool (LIP), is present in small amounts in cells but it is an important part of certain metabolic pathways and Fenton reaction which may contribute to lipid peroxidation and pathological process. The LIP is not only found in the cytosol but also present in the nucleus, mitochondria and lysosomes (Kruszewski, 2003, Lv and Shang, 2018). Lysosomes possess a higher concentration of the LIP, as it was discovered to be about 16 μ M in endothelial cells of rat liver (Petrat *et al.*, 2001). LIP has been connected to atherogenesis in several studies. The iron storing and regulatory proteins such as transferrin, ferritin and hepcidin have been associated with atherosclerosis (Yuan and Li, 2008). Human atherosclerotic lesions have been shown to contain ferritin and lysosomal iron in the lipid laden cells present at this site (Yuan *et al.*, 1996, Lee *et al.*, 1998, Yuan, 1999). Lysosomes are one of the few sites in the body where redox-active iron is present; the source of iron in lysosomes could be from the decomposition of endocytosed or autophagocytosed iron containing proteins such as ferritin or mitochondrial metalloproteins. The iron present in these molecules can be released by the iron transporter DMT1 (Kurz *et al.*, 2007, Lv and Shang, 2018). Ferritin as a protein plays a key role in the storage of iron in the body. In recent times ferritin is now becoming an evidently important factor in the development of iron storage and transport diseases, cancer, diabetes, neuropsychiatric disorders and atherosclerosis (Goswami *et al.*, 2008).

The ferritin molecule is the main protein for the intracellular storage of iron present in almost all organisms. The hollow protein weighs about 500 kDa and contains 24 polypeptide chains made up of the light (L) subunits and heavy (H) subunits with molecular weights of 19kDa and 21kDa, respectively, with capacity for storing up to 4500 iron atoms. The ratio of these two species varies in different tissues, however, ferritin rich in L subunits is mainly found in pancreas, the heart, the liver and spleen (Bomford *et al.*, 1981, Rucker *et al.*, 1996, Goswami *et al.*, 2008, Arosio *et al.*, 2017). Although there are conflicting data from epidemiological studies on the role of ferritin in CVD, some studies have ascertained the association between serum ferritin levels and increased susceptibility to myocardial infarction and coronary artery diseases (Salonen *et al.*, 1992, Kiechl *et al.*, 1994, Olesnevich *et al.*, 2012).

Iron regulatory/binding proteins such as ferritin and hepcidin have been linked to the inflammatory response induced pathogenesis of atherosclerosis (Yuan and Li, 2008). Deposits of cholesterol crystals derived from LDL oxidation in lysosomes have been demonstrated to activate NLRP3 inflammasome and lead to these organelles rupturing in macrophages, hence

contributing to the destabilisation of lysosomes and promoting atherosclerosis (Duewell *et al.*, 2010). Our laboratory demonstrated and identified lysosomes in macrophages as a site for the oxidation of LDL and this is mainly mediated by iron (Wen and Leake, 2007). Oxidation of polyunsaturated fatty acids and cholesterol in low density lipoprotein gives rise to fatty acid hydroperoxides and derivatives of oxidised cholesterol. Cholesteryl esters and cholesterol are the major lipids in LDL particle. The core containing cholesteryl esters is fluid while the outer monolayer containing cholesterol is more rigid and this might help to explain why the formation of cholesteryl ester hydroperoxide is more rapid compared to phospholipid hydroperoxide in the outer monolayer (Noguchi *et al.*, 1998).

Decomposition of the hydroperoxides leads to the formation of aldehydes, apart from PUFA and cholesterol content of LDL the apoB content can also be chemically modified (Thomas *et al.*, 1994). The oxidation products and quantity obtained depends on the conditions used for oxidation. Oxidation by copper leads to fragmentation of apoB, loss of amino acids from apoB, generation of lipid hydroperoxides and aldehydes (Steinberg *et al.*, 1989, Esterbauer *et al.*, 1992). Oxidation by iron at lysosomal pH leads to formation of the oxidised product of cholesterol and hydroperoxides (Satchell and Leake, 2012, Ahmad, 2016) and loss of tryptophan from apoB (Ahmad, 2016, Ahmad and Leake, 2018). *In vitro* oxidation of LDL can be catalysed by iron and this oxidation has been proven to be much faster at acidic pH similar to the pH of lysosomes. The mechanisms by which iron contributes to LDL oxidation in lysosomes and at lysosomal pH has been previously explored in our laboratory (Morgan and Leake, 1993, Morgan and Leake, 1995, Wen and Leake, 2007, Satchell and Leake, 2012, Ahmad and Leake, 2018)

We have now postulated that ferritin could substitute for iron and oxidise LDL at lysosomal pH. This chapter aims to explore the mechanisms of LDL oxidation by ferritin at lysosomal pH while addressing the following objectives.

3.1 Objectives

i) To investigate the effect of ferritin on LDL oxidation at lysosomal pH (pH 4.5), compare the oxidation of LDL at pH 4.5 to pH 7.4 and to evaluate the effect of pH changes on LDL oxidation by ferritin.

ii) To investigate the possibility of ferritin as a source redox- iron at lysosomal pH 4.5

iii) To determine the effects of iron released from ferritin on oxidation of LDL by ferritin by testing the effects of iron chelators on this oxidation.

v) To explore the effect of apoferritin, the protein component of ferritin, on LDL oxidation at lysosomal pH.

Vii) To characterise the oxidised products of LDL oxidation by ferritin at lysosomal pH.

3.2 Methods

LDL was isolated as described in sections 2.2.1 and 2.2. Formation of conjugated dienes, kinetics of oxidation of LDL oxidation by ferritin and aggregation of LDL by ferritin was measured as described in (sections 2.3, 2.3.1, and 2.3.2). The amount of iron in ferritin and apoferritin was measured by AAS (section 2.3.2). Iron released from ferritin at varying pH was measured using the iron chelator bathophenanthroline disulphonic acid and by ultrafiltration followed by AAS (sections 2.3.4 and 2.3.5, respectively). Effect of iron chelators was also determined spectrometrically, as described in section 2.3.6. The formation of oxidised lipids,

mainly 7-ketocholesterol and cholesteryl linoleate hydroperoxide, in LDL oxidised by ferritin was measured by HPLC as described in sections 2.4, 2.4.1, 2.4.2, 2.4.3 and 2.4.3.1. The total hydroperoxides in the oxidised LDL were also measured colourimetrically using the tri-iodide assay (section 2.5).

3.3 Results

3.3.1 Spectrophotometric measurement of LDL oxidation by ferritin at lysosomal pH

In order to confirm the ability of ferritin to oxidize LDL at lysosomal pH, LDL (50 μ g protein/ml) was oxidised in NaCl/sodium acetate buffer (pH 4.5) at 37 °C with varying concentrations of ferritin (0.05 μ M, 0.1 μ M and 0.2 μ M). We monitored the kinetics of oxidation by measuring the formation of conjugated dienes at 234 nm.

All concentrations of ferritin (0.05 μ M - 0.2 μ M), used in this experiment oxidised LDL effectively (Fig 3.1A). The kinetics of oxidation was somewhat similar to what was previously observed for ferrous iron at pH 4.5 (lag, rapid, slow oxidation, aggregation and sedimentation phases) (Satchell, 2008). The lag phase of oxidation was not as pronounced in LDL oxidised by ferritin and was sometimes not present. The conjugated diene formation began at an almost constant rate for the three concentrations of ferritin. The slow phase was observed at the three concentrations, but this was observed more at 0.1 μ M and very little or no slow phase was seen at 0.2 μ M. At the highest concentration of ferritin, (0.2 μ M) aggregation (when the aggregated LDL scatters UV radiation) began as early as the first 300 minutes and the LDL began to sediment below the beam of UV in the spectrophotometer at about 580 minutes, while the presence of 0.1 μ M and 0.05 μ M caused aggregation later.

The attenuance at 200 min was compared with the control and to each other (Figure 3.1B). Statistical analysis showed the control (LDL only) was significantly different (p<0.001) as

compared to the addition of the three concentrations of ferritin (0.05 μ M, 0.1 μ M, and 0.2 μ M). The attenuance at 200 min was used for statistical analysis because at this time the LDL is in the middle of the oxidation phase and not yet in the aggregation phase. This time point should be accurate because there is quite a lot of attenuance but this is a long way from the aggregation phase as shown by fig. 3.2 and by previous research (Satchell and Leake, 2012). The maximum attenuance in experiments was not measured because at the stage there would be a "race" between aggregation and sedimentation of the LDL particles. The lag phases were also not considered, as this would be shortened by storage of the stock LDL at 4 °C. This present study was focused on the events at the oxidation phase, however, measuring for a longer time gives a picture of the events in the aggregation phase as this might provide evidence for further research.



Figure 3.1: The oxidation of LDL by different concentrations of ferritin

LDL (50µg protein/ml) in NaCl/ sodium acetate buffer (pH 4.5) was incubated in the absence (red line) or presence of varying concentrations of ferritin, 0.05 µM (yellow line), 0.1 µM (green line) and 0.2 µM (black line) at 37°C in capped cuvettes. The formation of conjugated dienes was monitored by measuring attenuance at 234 nm against appropriate reference cuvettes that lacked LDL (A). This is a representative of three independent experiments. The increase in attenuance at 200 min in the presence or absence of ferritin in four independent experiments. Mean and SEM were obtained for each concentration and compared to control (n=4). The means were compared with one-way ANOVA followed by a post hoc Tukey test (P < 0.001) (B).

3.3.2 Measurement of LDL aggregation during oxidation by ferritin

Aggregation of LDL has been shown to occur alongside its oxidation and also enhance its formation of lipid-laden macrophages (Maor *et al.*, 1997). It was of interest to investigate whether the oxidation of LDL by ferritin leads to its aggregation. To test the effect of ferritin on LDL aggregation. LDL ($50\mu g$ protein/ml) and ferritin ($0.1 \mu M$) was incubated in NaCl/sodium acetate buffer (pH 4.5) at 37°C in the presence or absence of each other. The aggregation was then continuously measured at 680 nm. The extent of LDL oxidation by ferritin was directly correlated with its rate of aggregation while the LDL and ferritin on their own were not aggregated

The increase in attenuance at 800 min, a time at which LDL was expected to have been fully oxidised, was compared and there was a significant difference when LDL was incubated with ferritin (P<0.001). There was very little aggregation during the oxidation phases.



Figure 3.2: Effect of LDL oxidation by ferritin on LDL aggregation

Native LDL (50µg protein/ml), ferritin only (0.1 µM) and LDL (50µg protein/ml) plus ferritin (0.1 µM) was incubated in NaCl/sodium acetate buffer (pH 4.5) at 37° C in capped cuvettes (A), the aggregation of samples was monitored by measuring attenuance at 680 nm against appropriate reference. This is a representative of three independent experiments. The means were compared with one-way ANOVA followed by a post hoc Tukey test (P < 0.0001). Attenuance at 800 min was significantly higher at in oxidised LDL compared to control (B).

3.3.3 Comparison of oxidation of LDL by Ferritin at pH 4.5 and pH 7.4

The effect of pH on the ability of ferritin to oxidise LDL was tested. LDL ($50\mu g$ protein/ml) was incubated with 0.1 and 0.2 μ M ferritin at pH 7.4 (MOPS buffer) was compared to pH 4.5 (NaCl/sodium acetate buffer) (Fig 3. 3). The formation of conjugated dienes was monitored continuously. The result showed that the oxidation of LDL by ferritin is slower at pH 7.4 for both concentrations as compared to pH 4.5 where LDL was effectively oxidised. The rate of oxidation observed at both pH values was similar to that of the oxidation of LDL by ferrous sulphate described by Satchell and Leake (Satchell and Leake, 2012): unlike the rapid oxidation observed at pH 4.5, the oxidation at pH 7.4 only rose to attenuance of about 0.2 throughout the course of the experiment. The increase in attenuance at 200 min was compared for each concentration (0.1 and 0.2 μ M) at the different pH (4.5 and 7.4) (Fig 3.3B). The statistical analysis of both concentrations showed a significant difference between incubation with ferritin at pH 4.5 and pH 7.4. The attenuance was higher at 0.2 μ M showing that the effect was also concentration dependent.



Figure 3.3: Comparison of oxidation of LDL by ferritin at pH 4.5 and pH 7.4

LDL (50µg protein/ml) was incubated with 0.1 µM or 0.2µM in NaCl/sodium acetate buffer (pH 4.5) and MOPS buffer (pH 7.4) at 37 $^{\circ}$ C in capped cuvettes (A), the formation of conjugated diene was monitored by measuring attenuance at 234 nm against appropriate reference cuvettes that lacked LDL. This is a representative of three independent experiments. The means were compared with two-way ANOVA followed by Bonferroni post-tests (P < 0.0001). Attenuance at 200 min at pH 4.5 was compared to pH 7.4 (B).

3.3.4 Determination of the quantity of iron contained in ferritin.

Ferritin has the capacity to bind up to 4500 atoms of iron per particle. In order to determine the level of saturation of the horse spleen ferritin used, the amount of iron present was estimated using FAAS at 248.3nm. The concentration of iron was obtained in comparison with known standards of iron (Fe³⁺⁾. Iron concentration in ferritin was estimated from the linear trendline of the standard plot (Fig 3.4). Ferritin particles were estimated to contain 1001 atoms of Fe /particles. (The molecular weight of ferritin obtained from Sigma is 440kDa.) The amount of iron contained in ferritin concentrations (0.05, 0.1, and 0.2 μ M) added was estimated to be 50 μ M, 100 μ M and 200 μ M (see appendix 2 for calculation of iron concentration).



Figure 3.4: Standard plot of iron determination in ferritin assay by atomic absorption spectroscopy

The blue line represents the standard plots obtained from the standards 0-5mg/L. The black line represents the trendline for the equation used to extrapolate the quantity of ferritin.

3.3.5 Measurement of iron released from ferritin at pH 4.5 versus pH 7.4 using the iron chelator bathophenanthroline (BP)

To examine whether more iron released from ferritin at acidic pH was responsible for the faster oxidation observed at pH 4.5 compared to pH 7.4, we measured the amount of iron released at lysosomal pH was compared to what was released at interstitial or plasma. Ferritin (0.1 μ M) was incubated at 37 °C in NaCl/sodium acetate buffer (pH 4.5) or MOPS buffer (pH 7.4) in 15 ml tubes, in triplicate. Bathophenanthroline (BP) (30 μ l of 10 mM) was added at different time points (up to 24 h). Absorbance was measured at 535 nm (Figure 3.5). The amount of ferrous complex released was higher at pH 4.5 compared to pH 7.4 at all the time points that are taken. At 0 hr, more ferrous complex was detected at pH 4.5 (5.49 \pm 0.17 μ M) compared to 0.68 \pm 0.07 μ M at pH 7.4. The initial rapid release is either due to iron that had already come out of ferritin during its storage or to iron still inside the ferritin which BP was able to bind. Over time, the concentration increased to about 15.1 \pm 0.1 μ M and 1.32 \pm 0.1 μ M respectively.



Figure 3.5: Ferrous complex formation at pH 7.4 and pH 4.5

Ferritin (0.1 μ M) was incubated at 37°C in NaCl/Na acetate buffer (pH 4.5) or MOPS buffer (pH 7.4) in 15ml tubes, in triplicate. One millilitre was taken out and BP was added at different time points (0, 1.5, 3, 6, and 24hrs), left for 5mins and absorbance was measured at 535nm. Mean ± SEM of iron concentration released at pH4.5 was compared to pH 7.4 with two-way ANOVA followed by Bonferroni post tests. *** indicates P <0.001.

3.3.6 Ultrafiltration method for determination of iron released

In order to confirm the iron released from ferritin was not due to the effect of the iron chelator BP, a confirmatory experiment was carried out by the method previously used to determine the amount of iron released from transferrin (Lamb and Leake, 1994). Ferritin (0.1 μ M) was incubated at 37 °C in NaCl/sodium acetate buffer (pH 4.5) or MOPS buffer (pH 7.4) for 24hrs, then filtered with 30,000Mr cut off microcentrifuge filter tubes and then assayed for iron using AAS. The amount of iron was determined from the standard plot of iron was 12. ± 2.3 μ M at pH 4.5 as compared to 3.29 ± 0.67 μ M released at pH7.4 (Fig 3.6).

Although the AAS measured total iron concentration while the iron chelator measured ferrous iron, it was suspected that the lower concentration observed with AAS might be as a result of some of the iron retained in the microcentrifuge. The iron released, as measured AAS after 24 h incubation and ultrafiltration were also significantly higher at pH 4.5 (P < 0.05).



Figure 3.6: Iron released from ferritin at pH 4.5 measured by ultrafiltration and AAS

Iron released from 0.1 μ M ferritin incubated in pH 4.5 and pH 7.4 is shown. Mean and SEM were obtained for each pH. A paired student's t-test comparism (n=3) showed there was a significant difference (P< 0.05).

3.3.6 Effect of iron chelators on LDL oxidation by ferritin at pH 4.5

In order to assess whether iron released by ferritin contributed to LDL oxidation by ferritin, we tested the effects of the iron chelators DTPA and EDTA on the formation of lipid peroxidation products (conjugated dienes). LDL ($50\mu g$ LDL protein/ml) was oxidised with $0.1\mu M$ ferritin in NaCl/sodium acetate buffer (pH 4.5) in the presence or absence of EDTA or DTPA (100 μM) (Fig. 3.6). Formation of conjugated dienes was monitored. It was expected that the presence of iron chelators would inhibit entirely the oxidation of LDL at all time points as it did with ferrous iron (Satchell and Leake, 2012). The presence of iron chelators did not completely inhibit the initial oxidation of LDL by ferritin but only appeared to reduce the formation of conjugated dienes at a later stage.

The attenuance at 200 min was compared for control oxidised LDL and LDL oxidised by ferritin in the presence of EDTA and DTPA using one-way ANOVA followed by Tukey post hoc test. The oxidation of LDL by ferritin in the presence of iron chelators were statistically different from the control.



Figure 3.6: Effects of EDTA and DTPA on oxidation of LDL by ferritin at lysosomal pH

LDL (50µg protein/ml) was oxidized with ferritin (0.1µM) in NaCl/sodium acetate buffer (pH 4.5) at 37 °C in the absence (red line) or presence of iron chelators, 100 µM EDTA (black line) and 100 µM DTPA (green line). The formation of conjugated dienes was monitored by measuring attenuance at 234 nm against appropriate reference cuvettes that lacked LDL. This is a representative of three independent experiments (A). Mean and SEM of increase in attenuance at 200 min in the presence or absence of iron chelators were obtained for each concentration and compared to control with one-way ANOVA (n=3) followed by a post hoc Tukey test. *** indicates P < 0.001 (B).

3.3.7 Oxidation of LDL in the presence of apoferritin at lysosomal pH

The effect of Apoferritin, the protein component of the iron storage protein ferritin, on LDL oxidation at lysosomal pH was explored. LDL (50 μ g protein/ml) was oxidised in NaCl/sodium acetate buffer (pH 4.5) at 37°C with varying concentrations of apoferritin (0.05 μ M, 0.1 μ M and 0.2 μ M). The course of oxidation was monitored by measuring the formation of conjugated dienes at 234 nm (Fig 3.7). There was little or no oxidation of LDL in the presence of 0.05 μ M apoferritin. At 0.1 μ M, there was a slow increase of attenuance, but the kinetics of oxidation were somewhat different from what was observed with ferritin. At 0.2 μ M, the kinetics were somewhat similar to what was described for FeS0₄ at pH 4.5 (lag, rapid, slow oxidation and aggregation phases).

The increase in attenuance at 200 min was compared (Fig 3.7B). Statistical analysis showed the control (LDL only) was not significantly different as compared to 0.05 μ M and 0.1 μ M but significantly different from 0.2 μ M apoferritin. Due to the effect observed at higher concentrations of apoferritin we thought it would be necessary to quantify the amount of residual iron present in apoferritin. The amount of iron was determined by AAS (assuming a Mr of 443,000) and was 17 atoms of Fe per apoferritin particle which is only 1.7% of the total Fe per ferritin particle.



Figure 3.7: Oxidation of LDL by apoferritin at lysosomal pH

LDL (50µg protein/ml) in NaCl/sodium acetate buffer (pH 4.5) was incubated in the absence (red line) or presence of varying concentrations of apoferritin, 0.05 µM (green line), 0.1 µM (blue line) and 0.2 µM (orange line) at 37 °C in capped cuvettes. The formation of conjugated diene was monitored by measuring attenuance at 234 nm against appropriate reference cuvettes that lacked LDL. This is a representative of three independent experiments (A). The mean and SEM of attenuance at 200 min were obtained for each concentration and compared (n=3) by one-way ANOVA followed by a Tukey post hoc test. Increase in attenuance for 0.2 µM was compared control. ** indicates P<0.01(B).

3.3.8 Formation of oxidised lipids by LDL oxidised by ferritin at lysosomal pH

In order to confirm the atherogenic nature of LDL oxidised by ferritin, it was necessary to determine the formation of oxidised lipid products after LDL ($50\mu g$ protein/ml) has been incubated in the presence or absence of ferritin ($0.1\mu M$) in NaCl/sodium acetate buffer (pH 4.5) at 37°C. The oxidised lipids present mainly 7-ketocholesterol and cholesteryl linoleate hydroperoxides were measured by reverse HPLC. The total hydroperoxides formed were measured by a tri-iodide assay. These parameters were measured after the reaction was stopped at varying time points up to 48 h using EDTA and BHT. Longer time points were taken for measurement of oxidised lipids due to the much slower formation of 7-ketocholesterol compared to hydroperoxides formed as shown by this present study and other previous research from our group (Satchell, 2012, Ahmad, 2018).

The levels of 7-ketocholesterol and cholesteryl linoleate hydroperoxide (CLOOH) increased in the presence of ferritin while very little or none was formed in the control LDL. After 48 h of incubation, 7-ketocholesterol increased from 0 to 3.9 ± 1.0 nmol/mg of LDL protein; however, the 7-ketocholesterol formed was highest at 12 h with the value 4.0 ± 2.2 nmol/mg of LDL protein. The level of CLOOH formed in LDL oxidised by ferritin increased to 237 ± 118 nmol/mg of LDL protein and none was observed in control LDL at all time points. Total hydroperoxide formed increased with time in the presence of ferritin up to 1513 ± 204 nmol/mg of LDL protein. The comparison between native LDL only and LDL plus ferritin were determined by two-way ANOVA followed by Bonferroni's post-test.


Figure 3.8: Oxidised lipids formed from oxidation of LDL by ferritin at pH 4.5

LDL (50 µg protein/ml) was incubated in the presence or absence of ferritin (0.1 µM) in sodium acetate buffer (pH 4.5) at 37 °C. At varying time points up to 48 h, the oxidation was stopped by addition of EDTA (4 mM) and BHT (80 µM). The samples were analysed for (A) 7-ketocholesterol or (B) cholesteryl linoleate hydroperoxide (CLOOH) by reverse-phase HPLC. Graphs are representative of three independent experiments. Difference between control LDL and LDL oxidised by ferritin at each time point were determined by two-way ANOVA followed by Bonferroni's post-test (* indicates p < 0.05. ** indicates P < 0.01.



Figure 3.9: Total hydroperoxides formed by LDL oxidised by ferritin at pH 4.5

LDL (50µg protein/ml) was incubated in the presence or absence of ferritin (0.1µM) in sodium acetate buffer (pH 4.5) at 37°C. At varying time points up to 48 h, the oxidation was stopped by addition of EDTA (4mM) and BHT (80 µM). The samples were then analysed for total hydroperoxides by a triiodide assay. The graph represents three independent experiments. Differences between control LDL and LDL oxidised by ferritin at each time point were determined by two-way ANOVA followed by Bonferroni post-tests. ** indicates P<0.01,*** indicates p<0.001.

3.4 Discussion

The mechanisms by which LDL is oxidised by iron in the lysosomes is still in its infancy (Wen and Leake, 2007). Upregulation of synthesis of the H and L subunits of ferritin was found in macrophages and endothelial cells of atherosclerotic lesions (Pang *et al.*, 1996, You *et al.*, 2003). The presence of redox-active iron in lysosomes has been previously demonstrated in some studies (Yuan *et al.*, 1996, Petrat *et al.*, 2001) and this has been associated with ferritin degradation (Sibille *et al.*, 1989, Radisky and Kaplan, 1998, Yu *et al.*, 2003, Lv and Shang, 2018). Increased levels of iron (Casey *et al.*, 1988) and cytokines (Miller *et al.*, 1991) are present in atherosclerotic lesions and both of this can upregulate the synthesis of ferritin. There are however suggestions that iron containing proteins might have a role to play as a source of iron in lysosomal oxidation (Leake, 1997, Satchell, 2008).

In this present study, it was demonstrated that ferritin can catalyse the oxidation of LDL at pH 4.5 which is an approximate pH of lysosomes. All concentrations of ferritin used (0.05 μ M, 0.1 μ M and 0.2 μ M) were effective in oxidising LDL. The kinetics of oxidation by ferritin, however, differs from that of ferrous sulphate or ferric chloride earlier described (Satchell and Leake, 2012) as the lag phase was often times not observed at the early stage of oxidation by ferritin. However, the aggregation and sedimentation phases were also observed. As earlier discussed by Satchell and Leake, aggregation of LDL at acidic pH might occur as a result of the presence of net positive charge and repels each other at pH4.5, oxidation of LDL might reduce the net positive charge and leads to aggregation of LDL particles (Satchell and Leake, 2012).

Aggregation of LDL occurring alongside its oxidation and often times promotes its proatherogenic effects (Maor *et al.*, 1997, Jayaraman *et al.*, 2011). Our data in the present study show that LDL oxidised by ferritin aggregates at lysosomal pH, which further implies that LDL oxidation by ferritin may promote fusion or binding together of these particles and enhance its atherogenic effects.

A previous study by Kidane and his colleagues has demonstrated that the acidity of lysosomes and the activity of proteases are necessary for the release of iron from ferritin (Kidane *et al.*, 2006). Although proteases would not have been present in our system, the increased oxidation by ferritin at lysosomal pH might be due to this pH enhancing the release of iron from the core of ferritin. Previous work from our laboratory has shown the release of iron from other iron binding proteins such as transferrin increased oxidation of LDL (Lamb and Leake, 1994c). It was thought that iron released from ferritin at this pH might be responsible for the higher oxidative effect observed at lysosomal pH (pH 4.5) compared to normal interstitial fluid or plasma (pH 7.4).

In this present study the oxidation of LDL in the presence of 0.1 μ M and 0.2 μ M ferritin was faster at the pH approximately similar to that of lysosomes pH 4.5, Fully saturated ferritin has the capacity to store up to 4500 atoms of iron. The determination of the amount of iron showed that ferritin from equine spleen used in this experiment contained about 1001 atoms of iron hence (0.05 μ M, 0.1 μ M and 0.2 μ M) used in this present study contained 50 μ M, 100 μ M and 200 μ M of iron respectively (Section 3.3.4). The presence of 5 μ M ferrous sulphate has been shown to catalyse the oxidation of LDL and oxidation of LDL by iron was previously demonstrated to be much slower at pH 7.4, the pH of normal interstitial fluid or plasma compared to acidic pH (Wen and Leake, 2007, Satchell and Leake, 2012). The link between rapid oxidation at lysosomal pH and the release of iron at this pH was investigated by measuring the amount of ferrous iron (Fe²⁺) that could be released at lysosomal pH (pH 4.5)

compared to normal interstitial fluid or plasma (pH 7.4). The levels of ferrous complex formed were significantly higher at lysosomal pH at all-time points as measured using the iron chelator bathophenathroline (BP). This iron chelator does not detect ferric iron (Fe³⁺), but only measures ferrous iron (Fe²⁺) (Lynch and Frei, 1995). It should be noted that ferritin was incubated in the absence of BP and samples were taken at various times and BP added to them and the absorbance quickly measured. The iron chelator was therefore not binding iron and sequestering it out of the ferritin during the 24 h incubation. About 5% of the iron contained in ferritin was released at lysosomal pH at 0 h compared to 0.7% released at pH 7.4. The immediate formation of ferrous complex may be due to loosely bound iron present in the ferritin core which had been released during the storage of the ferritin or iron still associated with the ferritin particles but loosely bound which the iron chelator BP might be able to quickly complex with. It is unclear why more iron is released at acidic pH, but it might be speculated that this is due to the solubility of iron at acidic pH and the fact that acidic pH might enhance the opening of the pores of ferritin through which iron can be released (Jin *et al.*, 2001). The amino acids present in the pores of ferritin are highly conserved to tightly regulate the opening and closing of the pores. The localised unfolding of these sites can increase the amount of iron that exits the pores (Takagi et al., 1998).

An iron chelator independent technique (ultrafiltration followed by AAS) also revealed that more iron was released from ferritin at lysosomal pH as this was significantly higher compared to pH 7.4. The form of the oxidation state of iron in lysosomes has been previously debated (Collins *et al.*, 1991) and some researchers have argued that both forms ferric (Fe³⁺) and ferrous (Fe²⁺) might be present (Meguro *et al.*, 2005). However, a study by Terman and Kurz (Terman and Kurz, 2013) suggested that the ferrous form (Fe²⁺) might be favoured in lysosomal conditions. We have shown in this present study that ferrous iron is spontaneously released from ferritin at acidic pH, a similar condition to lysosomal conditions.

EDTA has a good chelating capacity for iron (Rizvi *et al.*, 2011). DTPA has the capacity to bind all the coordination sites of both ferric and ferrous iron and has a much higher binding affinity for iron than does EDTA (Tang *et al.*, 1997), hence it might be able to inhibit ironmediated lipid peroxidation. The deleterious effect of lysosomal iron can be prevented by iron chelators present within lysosomes or entering the lysosomes through endocytosis, autophagy or permeating the membrane (Terman and Kurz, 2013). Wen and Leake also demonstrated the importance of iron in the lysosomal oxidation in macrophages, as LDL oxidation was inhibited in cells by desferrioxamine (an iron chelator) (Wen and Leake, 2007). Addition of iron chelators EDTA and DTPA showed inhibition of LDL oxidation by ferritin. It was expected that DTPA would prevent the formation of conjugated dienes completely as previously observed with ferrous iron (Satchell and Leake, 2012) but this was not the case for either DTPA or EDTA. It is proposed that, as well as iron release from the ferritin core, there might be formation of LDL by ferritin. The nature of these reactive oxygen species and the mechanism of their formation requires a lot more research.

Ferrous iron released from ferritin might generate superoxide radical (O_2^{-}) (eqn 1) by reacting with molecular oxygen. The superoxide radical (O_2^{-}) produced can be protonated at acidic pH to form the hydroperoxyl radical (HO₂⁻) (eqn 2). Hydroperoxyl radicals are much more reactive than superoxide radicals in oxidising LDL (Bedwell *et al.*, 1989). They are also less hydrophilic and might enter the LDL particles and abstract hydrogen atoms from bisallylic methylene groups in lipids of LDL (eqn 3 and 4).

$Fe^{2+} + O_2 \rightarrow Fe^{3+} + O_2$	eqn 1
O_{2} ·- + $H^+ \rightarrow HO_2$ ·	pKa 4.8 eqn 2
$HO_2 + LH \rightarrow H_2O_2 + L^{\cdot}$	eqn 3
$L^{\cdot} + O_2 \longrightarrow LOO^{\cdot}$	eqn 4

Apoferritin, the protein component of ferritin, oxidised LDL at a concentration of 0.2 μ M but there was no significant difference at 0.1 μ M and 0.05 μ M. The kinetics of oxidation were different from what was observed with ferritin. The formation of conjugated dienes was prolonged at the highest concentration. It is unclear what was responsible for this observation. A lag phase characteristic of FeSO₄ oxidation was observed although the kinetics of oxidation was not what was observed with FeSO₄. Quantification of residual iron contained in apoferritin revealed the presence of 3.4 μ M iron in 0.2 μ M. Ferrous or ferric iron at 3.4 μ M would give faster oxidation of LDL (Satchell and Leake, 2012) than that shown by apoferritin (0.2 μ M), suggesting that not all iron present in apoferritin was catalytically active.

Cholesterol oxidation products oxysterols (27-carbon products) have been demonstrated to possess several biological activities that can lead to initiation and progression of atherosclerosis (Brown and Jessup, 1999). 7-keto cholesterol is the main oxysterol found in atherosclerotic lesion (Lyons and Brown, 1999). The animal study by Lyons and Brown (1999) reported that 7-ketocholesterol is more atherogenic compared to cholesterol. It is cytotoxic and can induce apoptotic vascular cells (Lyons and Brown, 1999). Cholesteryl linoleate is found in abundance in LDL and quantification of its hydroperoxide is a good indicator of LDL oxidation. This present study indicated that 7-keto cholesterol, cholesteryl linoleate and total hydroperoxides were generated in LDL oxidised by ferritin. The high levels of these products may promote atherogenic properties to the oxidised LDL. Previously reported levels of lipid hydroperoxides formed by ferrous iron oxidised tend to decrease slowly between 24 h and 48 (Satchell and

Leake, 2012) but this was not the case for LDL oxidised by ferritin. The reason for this difference is unclear. Total lipid hydroperoxides continued to increase after 24 h, whereas CLOOH did not either because phospholipids oxidised later than CL or because CLOOH became oxidised further and eluted from the HPLC column at a different time and were not detected.

Work presented in this chapter indicates that ferritin could oxidise LDL at lysosomal pH. This oxidation was mediated mainly by redox-active iron, as iron was released from ferritin at acidic pH and iron chelators slowed down this oxidation. The fact that LDL oxidised by ferritin aggregates suggest that it may enhance lipid accumulation in macrophage lysosomes and promote lysosomal engorgement which may lead to lysosomal dysfunction. The atherogenicity of LDL oxidised in this manner was also suggested by its ability to form oxidised lipids products which have been shown to confer atherogenic properties on LDL. Further research into the role of lysosomal LDL oxidation by ferritin in atherosclerosis is required and the work presented here might form the basis for further research in this area.

Chapter 4: Degradation of ferritin and LDL in lysosomes: implications for lysosomal LDL oxidation by ferritin.

4.0 Background and rational

The majority of extracellular and intracellular proteins are degraded partly or fully in the lysosomes (Jackson and Hewitt, 2016). The classification of lysosomes as the main location for intracellular proteolysis has been established as far back as the early 1960s (De Duve, 1963). Autophagy is the main process through which long lived or damaged proteins and organelles are turned over and reutilised. Proteasomes (multicatalytic proteinases), which also play a key role in protein turnover, are also degraded through autophagy. Many long lived ferruginous proteins are turned over by autophagy which makes lysosome a source of low mass redox-active iron (Kurz *et al.*, 2008).

The most effective way to prevent the deleterious effects of increased labile iron cations is through storage in ferritin (Bou-Abdallah, 2010). Evidence of decrease in lysosomal degradation of intracellular ferritin and iron release has been found in cells after treatments with compounds that inhibit lysosomal activity but not with compounds that inhibit proteasome (Kidane *et al.*, 2006, Arosio *et al.*, 2009). This implies that lysosomal autophagy and degradation by proteases play a key role in the degradation of ferritin. Radisky and Kaplan demonstrated that degradation of ferritin in lysosomes is a means of mobilising iron from the core of ferritin (Radisky and Kaplan, 1998). However, De Domenico and colleagues have suggested that mobilisation of iron is not dependent on the degradation of ferritin as iron can be mobilized in the absence of degradation (De Domenico *et al.*, 2006). Other studies have shown the degradation low saturated ferritin with iron-binding capacity may temporarily protect against iron-mediated oxidative stress. Saturated ferritin, on the other hand, can be auto phagocytosed and degraded in the lysosomes leading to the presence of more redox active iron and susceptibility to oxidative stress (Kurz *et al.*, 2011).

The lysosomal compartment contains a wide range of hydrolytic enzymes which are very important mediators of recycling of biomolecules such as proteins, phospholipids and polysaccharides. Endopeptidases are mainly believed to initialise the cleavage of proteins and they act in synergy with exopeptidases to break down the resulting peptides to amino acids (van der Westhuyzen *et al.*, 1980). The cathepsins are evidently the main categories of the hydrolytic enzymes. They are classified as aspartic proteases (e.g. cathepsins D and E), cysteine proteases (e.g. cathepsins B and C) and serine carboxypeptidases (e.g. cathepsin G) with an optimum pH of around 5 (Turk *et al.*, 2002, Kuester *et al.*, 2008, Kurz *et al.*, 2011). However, the participation of cathepsin D (Dean, 1975) and cathepsin B (Hopgood *et al.*, 1977, Libby and Goldberg, 1978) in general degradation of intracellular proteins has been shown previously, using inhibitors that are specific to these enzymes, such as pepstatin and leupeptin, respectively.

Low density lipoprotein is internalized by receptor-mediated endocytosis, after which it is delivered to the lysosomes. The apoB-100 component of LDL is degraded in lysosomes and inhibitors of lysosomal function, such as chloroquine, have been demonstrated to prevent the degradation of apoB-100 (Stein *et al.*, 1977). Van der Westhuyzen and colleagues demonstrated that the degradation of apoB-100 component of LDL in cells cultured from arterial smooth muscle cells or human fibroblast and adult bovine aortic smooth muscle cells was initiated by cathepsin D. The enzyme then acted in synergy with other enzymes which included cathepsin B to complete the degradation process (van der Westhuyzen et al., 1980). This was supported by work by Leake and Peters which also suggested that the synergistic action of both cathepsins D and B might have degraded the apoB-100 component of LDL, but the majority of the degradation of LDL protein smooth muscle cells cultured from aorta of was carried out by cathepsin D (Leake and Peters, 1981). Recent *in vitro* study by Linke and others

have shown that cathepsins V, S, L, K, F and B degraded the protein component of LDL under acidic pH condition (pH 5.5) with cathepsin S having activity at pH 7.4. (Linke *et al.*, 2006). The core of LDL consists mostly of cholesteryl esters and cholesteryl esterase has the capacity to hydrolyse cholesteryl esters. Chao and colleagues hydrolysed the cholesteryl ester component of LDL using cholesteryl esterase and also suggested that the particle derived from this hydrolysis were chemically similar to the unesterified cholesterol-rich LDL found in atherosclerotic lesions (Chao *et al.*, 1992). Many forms of enzymatic modification of LDL such as apoB-100 proteolysis, lipolysis, which includes phospholipid hydrolysis, hydrolysis of cholesteryl esters and hydrolysis of sphingomyelin, can lead to aggregation and fusion of LDL into lipid droplets which have been linked to the development of atherosclerosis (Lu and Gursky, 2013).

Although the autophagic process is rapid and effective, it can still give room for some peroxidation catalysed by redox active iron to occur within the lysosomes. These can lead to oxidative modification of some materials autophagocytosed (Brunk and Terman, 2002). In view of the recent emergence of the lysosomal oxidation theory, it was hypothesised that during the process of degradation of both LDL and ferritin in lysosomes there might be conformational changes in ferritin and/or LDL that could enhance ferritin mediated LDL oxidation. The overall aim of this chapter was to modify ferritin and LDL with enzymes that possess activity at lysosomal pH and explore the effects of these modifications might have on oxidation of LDL by ferritin at lysosomal pH. The effects of major lysosomal endoproteases cathepsin D, cathepsin B and the lipolytic enzyme cholesteryl esterase were explored.

4.1 Objectives

- i) To proteolytically modify ferritin with cathepsins at lysosomal pH
- ii) To proteolytically modify of LDL with cathepsins at lysosomal pH

iii) To compare the oxidation of LDL by intact ferritin and enzyme-treated ferritiniv) To test the effects of modification of apoB content of LDL by cathepsins on oxidation ofLDL by ferritin at lysosomal pH.

v) To investigate the effect of cholesteryl esters degradation by cholesteryl esterase on LDL oxidation by ferritin.

4.2 Methods:

The method for this session is described in section 2.6. The digestion of ferritin and LDL with cathepsin D was described in section 2.6.1.1 and digestion of ferritin with cathepsins B and D was described in section 2.6.1.2. The co-incubation with cholesteryl esterase and proteases (cathepsins B and D) was described in section 2.6.1.3. The method for evaluation by SDS-PAGE after digestion was described in section 2.6.2 while the effects of the proteases and cholesteryl esterase on LDL oxidation was described in section 2.6.3.

4.3 Results

4.3.1 Proteolytic degradation of ferritin by cathepsins in vitro (pH 4.5)

As mentioned earlier, there is evidence from previous researchers suggesting the degradation of ferritin occurs in lysosomes. To test the impact of the major lysosomal endoprotease on degradation of ferritin at lysosomal pH, intact ferritin (2 mg/ml) was treated with or without cathepsin D ($6.2 \mu g/ml$) and incubated in Chelex - 100 treated NaCl/sodium acetate buffer (pH 4.5) at 37 °C for 24 h. After incubation, the ferritin samples were evaluated by polyacrylamide gel electrophoresis under reducing conditions. Figure 4.1A shows the pattern obtained with or without treatment of ferritin protein with cathepsin D. The heavy and light chains of 21 kDa and 19 kDa were observed without treatment, but a less intense band was observed for the light

chain. No lower Mr degradation products were observed after treatment with cathepsin D. Analysis of the density of the heavy chain bands was carried out using image J software to compare the band intensity with or without enzymes. The result showed band intensity of 96.5 \pm 17.0 in untreated ferritin to 136.9 \pm 14.9 band intensity of ferritin pre-treated with cathepsin D, there was no significant different between the ferritin treated with cathepsin and the control (P > 0.05). Ferritin modification in the presence of cathepsin B, another main endoprotease present in lysosome was tested with increased the enzymes to protein ratio. Ferritin (500 µg) was treated with or without both cathepsin D (5 µg) and cathepsin B (5 µg) and incubated in Chelex-100 treated NaCl/sodium acetate buffer (pH 4.5) at 37°C for 24 h. After incubation the ferritin samples were evaluated by polyacrylamide gel electrophoresis under reducing condition (Figure 4.1B). No lower molecular weight products were again observed.



Figure 4.1: SDS-PAGE of ferritin treated with cathepsins

Ferritin (2 mg/ml) incubated with or without cathepsin D (6.2 μ g/ml) at 37 °C for 24 h, lane 1 contains molecular weight markers, lane 2 contains ferritin only, lane 3 contains ferritin plus cathepsin D (A). Ferritin incubated with or without cathepsin B and D at 37 °C for 24 h, lane 1 contains molecular weight markers, lane 2 contains ferritin only, lane 3 contains ferritin pre-incubated with cathepsin B and cathepsin D at weight ratio 100:1:1 (Ferritin (500 μ g/ml), Cathepsin D (5 μ g/ml) and cathepsin B (5 μ g/ml) and lane 5 contains cathepsin B only (B). All samples were treated with reducing sample treatment buffer (RSTB) and about 15 μ g in terms of ferritin protein was loaded onto gels with stacking gel of 4% and resolving gel gradient ratio of 15%. It was run for one hour at 150 V before staining with Coomassie blue to detect bands and de-stained. The pictures of the gels were captured using a U Genius image capturing machine.

4.3.2 Effects of treatment of ferritin with proteases on LDL oxidation by ferritin at lysosomal pH

Previous work with caeruloplasmin which contains the majority of copper ions in the plasma has shown that proteolytically modified caeruloplasmin was not able to enhance oxidation of LDL by SMC compared to the intact caeruloplasmin molecule (Mukhopadhyay *et al.*, 1996). It was of interest to test whether proteolytic modification of ferritin affects its ability to oxidise LDL at lysosomal pH (Figure 4.2). Ferritin (2 mg/ml) was incubated with or without cathepsin D ($6.2 \mu g/ml$) for 24 h in NaCl/sodium acetate buffer (pH 4.5) at 37 °C and then used the ferritin ($0.1 \mu M$) to oxidise native LDL ($50 \mu g$ protein/ml). The formation of conjugated dienes every minute interval at 234 nm for up to 1200 minutes were measured. The results show that treatment of ferritin with cathepsin D, did not affect its ability to catalyse LDL oxidation. The oxidation followed a similar pattern at the early phase of oxidation. In ferritin pre-incubated with cathepsin D, the aggregation phase began later and the peak attenuance was higher compared to untreated ferritin. The attenuance at 200 min was compared to control. The statistical analysis using Student's paired t test showed there was no statistically significant difference to control (P > 0.05).

Following up with this result, the effects of a decreased protein enzyme ratio and addition of cathepsin B, another lysosomal protease to the system were tested. We incubated ferritin (500 μ g) with or without cathepsin D (5 μ g) and cathepsin B (5 μ g) for 24 h in NaCl/sodium acetate buffer (pH 4.5) at 37°C. After incubation, the ferritin (0.1 μ M) with or without enzymes were used to oxidise native LDL (50 μ g protein/ml) in capped cuvettes. Cathepsin D ad B (0.44 μ g/ml each) were added to freshly prepared to ferritin (0.1 μ M) and native LDL (50 μ g protein/ml, as an additional control (Fig. 4.3A). The attenuance at 200 min was compared

using one way ANOVA followed by Tukey post hoc test. The attenuance of the control at this time was 0.070 ± 0.021 compared to 0.100 ± 0.038 observed in ferritin pre-treated with cathepsin D and B (Fig. 4.3B). However, the treatments were not statistically significantly different when compared to control or each other (P > 0.05), but results suggest that ferritin is able to catalyse LDL oxidation even after treatment with enzymes.



Figure 4.2: Effects of pre-treatment of ferritin with cathepsin D on LDL oxidation by ferritin

Native LDL (50 μ g protein/ml) was oxidised with ferritin (pre-incubated alone or with cathepsin D) (0.1 μ M), in NaCl/ sodium acetate buffer (pH 4.5) at 37 $^{\circ}$ C in capped cuvettes. The formation of conjugated dienes was monitored by attenuance at 234 nm against appropriate reference cuvettes that lacked LDL (A). This is a representative of three independent experiments. The increase in attenuance at 200 min compared with the controls using Student's paired t test (n=3) (P > 0.05). ns indicates not significant (A).



Figure 4.3: Effect of treatment of ferritin with cathepsin D and B on LDL oxidation by ferritin

Native LDL (50 μ g protein/ml) was oxidised with ferritin pre-incubated alone or with cathepsin D and B (0.1 μ M), and freshly prepared cathepsin D and B (0.44 μ g) and ferritin

(0.1 μ M) in NaCl/ sodium acetate buffer (pH 4.5) at 37 $^{\circ}$ C in capped cuvettes. The formation of conjugated dienes was monitored at attenuance of 234 nm against appropriate reference cuvettes that lacked LDL (A). This is a representative of three independent experiments. Mean \pm SEM increase in attenuance at 200 min was compared with one way ANOVA (n = 3) followed by Tukey post hoc test (P > 0.05). ns indicates not significant (B).

4.3.3 Proteolytic degradation of LDL by cathepsin D

The intact LDL (500 μ g protein/ml) was incubated in Chelex - 100 treated NaCl/sodium acetate buffer (pH 4.4) at 37°C for 24 h in the presence or absence of cathepsin D (5 μ g/ml). After incubation, the extent of degradation of the samples was evaluated using polyacrylamide gel electrophoresis under reducing condition. Figure 4.4 shows the pattern obtained with or without treatment of LDL with cathepsin D. The Mr weight of around 250 kDa was observed in LDL without cathepsin D, whereas the M_r of apoB-100 is 513 kDa. This discrepancy might be explained by the migration of very large proteins into SDS-PAGE gels might not always be consistent. There were many degradation products in the presence of cathepsin D. This *in vitro* result was as expected, as previous work has shown the importance of cathepsin D in LDL degradation by arterial smooth muscle cells (Leake and Peters, 1981).



Figure 4.4: SDS-PAGE of LDL treated with cathepsin D

LDL (500 μ g/ml) incubated with or without cathepsin D 5 μ g/ml) at 37 °C for 24hrs, lane 1 contains molecular weight marker, lane 2 and 3 contains LDL only, lane 4 and 5 contains LDL plus cathepsin D. All samples were diluted to 15 μ g, treated with reducing sample treatment buffer (RSTB) and loaded onto gels with stacking gel of 4% to resolving gel of 5%. The gel was run for one hour at 150 V before staining with Coomassie blue to detect bands and de-stained. The picture of the gel was captured using a U Genius image capturing machine.

4.3.4 Effect of treatment of LDL with proteases on LDL oxidation by ferritin at lysosomal pH

It was of interest to test whether proteolytic modification of LDL can affect its oxidation by ferritin at lysosomal pH, as LDL will be degraded in lysosomes at the same time as it is being oxidised (Figure 4.5). LDL (500 μ g protein/ml) was incubated with or without cathepsin D and B (5 μ g/ml each) for 24 h in NaCl/sodium acetate buffer (pH 4.5) at 37°C. The pre-treated LDL (50 μ g protein/ml) was oxidised with ferritin (0.1 μ M). LDL not pre-treated with enzymes (50 μ g protein/ml) was also oxidised with ferritin (0.1 μ M) in the presence or absence of freshly added cathepsin D and B (0.5 μ g/ml each) to the cuvettes; this was equivalent to the concentration of enzymes that would have present in the pre incubated LDL. This was done because the cathepsins would have been active in the cuvettes at pH 4.5. The formation of conjugated dienes was measured at every one minute interval at 234 nm for 1200 minutes. The result shows that proteolytic degradation of LDL increased the attenuance at 234 nm compared to the control. The rise in attenuance was more prominent in proteolysed LDL with the aggregation phase observed earlier compared to the untreated LDL and the LDL freshly treated with the enzymes in the test cuvettes.

The increase in attenuance at 200 min was compared with one-way ANOVA followed by a Tukey post hoc test (n = 4) (Fig 4.5B). Oxidation of previously proteolysed LDL was significantly different from control LDL (P < 0.01). However, the freshly treated LDL was not statistically different from control.



Figure 4.5: Effects of treatment of LDL with cathepsin D and B on LDL oxidation by ferritin

LDL (500 µg protein/ml) was incubated with or without cathepsin D and B (5 µg/ml each) for 24 h in NaCl/sodium acetate buffer (pH 4.5) at 37°C. LDL pre-treated with cathepsin D and B (50 µg protein/ml) was then oxidised with ferritin (0.1 µM), in NaCl/ sodium acetate buffer (pH 4.5) at 37 °C. LDL with freshly added cathepsin D and B to the cuvettes (0.5 µg/ml), the concentrations they would have been present in the cuvettes due to carry over, was also included. The formation of conjugated dienes was monitored at attenuance of 234 nm against appropriate reference cuvettes that lacked LDL. This is a representative of four independent experiments (A). The increase in attenuance at 200 min compared with one-way ANOVA followed by Tukey post hoc test (n = 4). ** indicates P < 0.01 and ns indicates not significant (B).

4.3.5 Effect of co-incubation with cholesteryl esterase on LDL oxidation by ferritin at lysosomal pH.

It was of interest to test the effect of modifying the lipid component of LDL with a lipolytic enzyme on its oxidation by ferritin at lysosomal pH, as this will mimic what might happen in lysosomes. Previous work by Kyger et al. has demonstrated that bovine and human pancreatic cholesteryl esterase had optimum catalytic pH from pH 5.5 to 6.5 in the presence of taurocholate. Their data also showed some taurocholate independent activity of the enzyme in the synthetic activities (Kyger et al., 1990). Work by Chao et al. revealed the dependence of hydrolysis of lipoprotein cholesteryl esters by cholesteryl esterase from Candida cylindracea on the treatment of LDL with trypsin as the cholesteryl ester hydrolysis was only possible after treatment with trypsin (Chao et al., 1992). In this present study, cholesteryl esterase from Pseudomonas sp was used as the lipolytic enzyme. Preliminary data showed increased cholesterol content in LDL from the degradation of cholesteryl esters into fatty acids and cholesterol, in the presence of cholesteryl esterase with or without trypsin. LDL (50 µg protein/ml) treated with cholesteryl esterase (0.0625 unit/ml) in the presence or absence of trypsin (1.25 μ g/ml) was incubated in NaCl/sodium acetate buffer (pH 4.5) at 37°C for 2 h. The cholesterol content was measured by reverse phase HPLC at 210 nm. The data showed increased cholesterol content from 775 \pm 91.5 nmol/mg LDL protein to 1454 \pm 65.0 nmol/mg LDL protein in LDL plus cholesteryl esterase and 2360 ± 220.5 nmol/mg LDL protein in LDL plus cholesteryl esterase and trypsin. The result showed that cholesteryl esterase from Pseudomonas sp can hydrolyse cholesteryl esters in the presence or absence of trypsin (unpublished data).

To test the effect of hydrolysis with cholesteryl esterase on LDL oxidation by ferritin at lysosomal pH, we incubated LDL (50 µg protein/ml) with or without cholesteryl esterase (CE)

from *Pseudomonas sp* (0.0625 unit/ml) in NaCl/sodium acetate buffer (pH 4.5) at 37°C with ferritin (0.1 μ M). The results show that co-incubation with cholesteryl esterase increased the formation of conjugated dienes by LDL compared to control (Fig 4.6B). The early phase of oxidation was increased, and the aggregation phase occurred earlier compared to the control. The increase in attenuance at 200 min was compared with a paired Student's t test (n = 3). The oxidation of LDL was significantly increased (P<0.001) (Fig. 4.6B)



Figure 4.6: Effects co-incubation with cholesteryl esterase on LDL oxidation by

ferritin

LDL (50 µg protein/ml) was incubated with or without cholesteryl esterase (0.0625 unit/ml) and was oxidised with ferritin (0.1 µM) in NaCl/ sodium acetate buffer (pH 4.5) at 37 °C. The formation of conjugated dienes was monitored at attenuance of 234 nm against appropriate reference cuvettes that lacked LDL (A). This is a representative of three independent experiments. The increase in attenuance at 200 min compared with paired Student's t test (n = 3). *** indicates P < 0.001 (B).

4.3.6 Effects of co-incubation with cholesteryl esterase and proteases on LDL oxidation by ferritin at lysosomal pH.

The lysosomal compartment contains both proteolytic and lipolytic enzymes along with other hydrolases. The multienzynme composition of the lysosomal compartment allows its rapid, efficient and coordinated degradation of targetted substrates. Degradation of macromolecules often involves multiple enzymes working together in a stepwise manner (Bonten *et al.*, 2014). It was of interest to test the effect of modifying a part of lipid component (cholesteryl esters) and the protein component (apoB-100) of LDL concurrently on its oxidation by ferritin at lysosomal pH (Figure 4.7). LDL (50 µg protein/ml) was co-incubated with or without cholesteryl esterase (0.0625 unit/ml) and cathepsin D and B (1.25 µg/ml) in NaCl/sodium acetate buffer (pH 4.5) at 37°C, with ferritin (0.1 µM). The formation of conjugated dienes was measured for every one minute interval at 234 nm for up to 1200 minutes. The results show that co-incubation with cholesteryl esterase and proteases cathepsin D and B increased the formation of conjugated dienes by LDL compared to control.

The increase in attenuance at 200 min was compared with a paired Student's t test (n = 3). The oxidation of LDL was significantly increased by the presence of cholesteryl esterase and the proteases P < 0.01 (Fig. 4.7B).



Figure 4.7: Effects of co-incubation with cholesteryl esterase and proteases on LDL oxidation by ferritin

LDL (50 µg protein/ml) was incubated with or without cholesteryl esterase (0.0625 unit/ml), cathepsin D and B (1.25 µg/ml each) with ferritin (0.1 µM) in NaCl/ sodium acetate buffer (pH 4.5) at 37 $^{\circ}$ C. The formation of conjugated dienes was monitored by measuring attenuance of 234 nm against appropriate reference cuvettes that lacked LDL (A). This is a representative of three independent experiments. Figure 4.7B shows the increase in attenuance at 200 min compared with paired Student's t test (n = 3). ** indicates P < 0.01 (B).

4.4 Discussion

Existing evidence regarding the mechanisms for degradation of ferritin and the link to iron mobilisation is controversial. However, some have suggested that lysosomal enrichment with ferritin stabilise cells after autophagocytosis by binding low mass iron in the lysosomes (Garner et al., 1997, Persson et al., 2001a). Others have argued that there is possibility for the release of iron from ferritin in lysosomes during ferritin degradation (Roberts and Bomford, 1988, Radisky and Kaplan, 1998, Zhang et al., 2010). Lysosomal proteases such as cathepsins D and B, have been demonstrated to be involved in the degradation of the apoB-100 component of LDL (van der Westhuyzen et al., 1980, Leake and Peters, 1981, Linke et al., 2006) and ferritin (Richter, 1986, Kidane et al., 2006, Laskar et al., 2012). This evidence put together suggests the degradation of ferritin and LDL under lysosomal conditions using enzymes in vitro and within lysosomes in cells. The recent lysosomal LDL oxidation theory has provided data suggesting the lysosomal oxidation of LDL mediated by iron in macrophages in vitro to the onset and progression of atherosclerosis (Wen and Leake, 2007, Satchell and Leake, 2012, Ahmad and Leake, 2018, Ahmad and Leake, 2019). Work from the previous chapter of this thesis demonstrated that ferritin can catalyse LDL oxidation at lysosomal pH and also showed that there is enhanced release of iron from ferritin under the lysosomal acidic condition in the absence of proteases. This study explored the effects of proteases and lipolytic enzymes on the oxidation of LDL by ferritin at lysosomal pH.

The present study showed that intact ferritin and ferritin pre-treated with cathepsin D can catalyse the oxidation of LDL in a similar manner, not surprisingly. However, pre-incubation of ferritin with cathepsin had no effect on the rate of oxidation compared to control. It was tested whether pre-incubating ferritin with more proteolytic enzyme (with lower protein to enzyme ratio) would catalyse the oxidation of LDL by ferritin more effectively, but there was no significant effect. The results of SDS-PAGE analysis of ferritin after incubation with proteolytic enzymes showed that ferritin is somewhat resistant to degradation by cathepsin D and B at pH 4.5, although exposure of apoB-100 in LDL to similar condition with ferritin caused extensive degradation of apoB-100. Chrichton has previously demonstrated the synergistic digestion of ferritin and apoferritin using cathepsin D and pepsin at pH 3.0 (Crichton, 1971), which is a more acidic pH compared to what was used in this study. However, their findings indicated that ferritin was less susceptible to digestion compared to apoferritin. The capacity of lysosomes to degrade endogenous and extracellular biomolecules is of great importance in handling the excess lipid and cytotoxic materials that are present in atherosclerotic plaque. The paramount role of lysosomes in degradation has been well acknowledged (Sergin *et al.*, 2015). From our results, it is suspected that ferritin degradation may take a longer time and require synergistic actions of several proteolytic enzymes which were not included in this present study but can be explored in the future.

These experiments have shown that ferritin showed no observable degradation products after treatment with cathepsins D and B but it catalysed LDL oxidation after treatment with enzymes. This implies that it may stay longer in the lysosomes to oxidise LDL. The fact that ferritin was still able to catalyse LDL oxidation without complete degradation supports the spontaneous release of iron at lysosomal pH which catalyses LDL (demonstrated in chapter 3). The question of introducing more lysosomal enzymes is, however, relevant because autophagocytosed ferritin would be eventually be degraded in lysosomes. It would be of interest to know if it would be more or less effective in oxidising LDL during its life time in lysosomes before it is degraded entirely.

Enzymatically modified LDL has been produced *in vitro* with the use of both proteases and cholesteryl ester hydrolase (Bhakdi *et al.*, 1995, Chellan *et al.*, 2016). Preliminary data from our laboratory showed that LDL pre-incubated at pH 7.4 and 4.5 was degraded by trypsin and cathepsin D, respectively (unpublished data). The proteolytic modification of LDL in this present study was carried out with cathepsin D at lysosomal pH (pH 4.5). The enzyme cathepsin D has been identified as the key player in proteolytic degradation of LDL (Leake and Peters, 1981). The result from this present *in vitro* experiment was consistent with our previous work (unpublished data) and the result from others which showed that lysosomal proteases degraded apoB-100 at pH 5.5 (Linke *et al.*, 2006). Although cathepsin D is not a cysteine protease and was not among the cathepsins used by Link *et al.*, however, low molecular weight fragments were observed in LDL pre-treated with cathepsin D at pH 4.5. Previous studies have demonstrated the importance of cathepsin D and B in the degradation of LDL in arterial smooth muscle cells (van der Westhuyzen *et al.*, 1980, Leake and Peters, 1981).

Enzymatically modified LDL has been demonstrated to be present in atherosclerotic lesions (Torzewski *et al.*, 1998, Hakala *et al.*, 2003) and cathepsins are highly synthesised in atherosclerotic plaques (Hakala *et al.*, 2003, Sukhova *et al.*, 2003, Oorni *et al.*, 2004) which could act on LDL in lysosomes. Cathepsin B, D and X has been suggested to play a key role in the development of atherosclerosis (Zhao and Herrington, 2016). It was reported here that treatment of LDL with cathepsin D and B prior to oxidising with ferritin, increased modification of LDL at lysosomal pH but also its aggregation. Further work will be required to see if this increased modification is due to lipid peroxidation forming conjugated dienes or it is aggregation causing UV scattering. Measuring oxidised lipids by HPLC should clarify this question. Previous studies suggest that degrading apoB-100 in LDL leads to aggregation and fusion of lipoprotein in complex lipid aggregates that increase as the atherosclerotic lesions

progress (Piha *et al.*, 1995, Camejo, 2003). This present finding might imply that pre-treatment with cathepsins promote the formation of aggregated lipids and enhance atherogenicity in lysosomal LDL oxidised by ferritin. The important role of cathepsins, mainly the cysteine proteases, in atherosclerosis was earlier suggested by Liu *et al.* (Liu *et al.*, 2004). This was later supported with evidence from reviews by Lutgens and others, whose work suggested that potential inhibitors of cathepsins might be beneficial in the treatment of atherosclerosis (Lutgens *et al.*, 2007, Sjöberg and Shi, 2011).

High concentrations of cholesteryl ester hydrolase have been detected in atherosclerotic lesions by immunostaining (Sakurada *et al.*, 1976, Hakala *et al.*, 2003). The purified enzyme had an optimum pH of 4.5 -5.0 and 7.0 -7.5 (Sakurada *et al.*, 1976) which includes the pH of lysosomes and that of plasma, respectively. The co-incubation of LDL with cholesteryl esterase from *Pseudomonas sp* and ferritin (Figure 4.6) showed a significant (P < 0.001) increase in the formation of conjugated dienes compared to incubation with ferritin alone. This breakdown of cholesteryl esters may contribute to more lipid peroxidation in lysosomal LDL oxidation, possibly by giving ferritin iron more access to the PUFA formed than to cholesteryl esters.

Hydrolysing cholesteryl esters and degrading the apoB-100 component of LDL at the same time produced LDL that had a similar chemical structure and biological activity to LDL derived from atherosclerotic lesions (Bhakdi *et al.*, 1995). The lysosomal compartment is highly dynamic with multiple enzymes acting in synergy to degrade biomolecules. These enzymes working together may have positive and negative consequence in relation to disease development (Bonten *et al.*, 2014). The effect of the joint action of proteases (cathepsins B and D) and cholesteryl esterase (figure 4.7) on LDL oxidation by ferritin showed increased formation of conjugated dienes (P < 0.01). This suggests that the oxidation of LDL in lysosomes might speed up as the LDL is degraded in these organelles. The findings in this chapter put together show that LDL oxidation by ferritin can proceed in the presence of lysosomal enzymes and might actually be increased by them. The degradation of the protein and lipid component of LDL particles enhanced its oxidation by ferritin at lysosomal pH. The results suggest that, as LDL is degraded in lysosomes its oxidation by ferritin might be speeded up. The results also show the remarkable resistance of ferritin (unlike apoB-100 in LDL) to proteolysis by major lysosomal endoproteases cathepsin B and D. The next chapter explores the role of antioxidants in the prevention of LDL oxidation by ferritin at lysosomal pH.

Chapter 5: The role of antioxidants in preventing LDL oxidation by ferritin

5.0 Background and rational

Atherosclerosis is a complex disease involving many factors. The oxidative modification of LDL and a chronic inflammatory response has been suggested as key events in the development of atherosclerotic lesions. LDL oxidation is believed by many to be closely linked with atherosclerosis with relevance in all stages of atherosclerosis (Nilsson *et al.*, 1992, Itabe *et al.*, 2011). The fact that low concentrations of antioxidant compounds can delay or inhibit the oxidation of oxidisable substrates (Halliwell, 1990) makes antioxidants one of the promising candidates for prevention and treatment of atherosclerosis. Nutrients, such as and phenolic component of red wine (Teissedre *et al.*, 1996), vitamins C (Retsky et al., 1993) and E (Jessup *et al.*, 1990) can reduce the susceptibility of LDL to oxidation *in vitro*. Plasma thiols such as homocysteine, cysteine and reduced glutathione have been demonstrated to inhibit LDL oxidation *in vitro* (Lynch *et al.*, 2000).

The LDL isolated from humans contains a number of lipid-soluble antioxidants which includes several carotenoids and oxycarotenoids, ubiquinol-10, γ -tocopherol and α -tocopherol (Esterbauer *et al.*, 1992). Out of these antioxidants, α -tocopherol (Figure 5.1A) is believed to be the most important and abundant antioxidant present in human LDL (Esterbauer *et al.*, 1990a, Meydani, 2001). It possesses the ability to limit the production of lipid peroxidation products via its chain breaking action by scavenging alkoxyl and peroxyl radicals (Sies *et al.*, 1992, Liebler, 1993). The α -tocopherol radical produced is stabilised by delocalisation of unpaired electron around its ring. LDL isolated from people who had previously received vitamin E as a supplement was less susceptible to oxidation induced by macrophages or copper (Jessup *et al.*, 1990, Dieber-Rotheneder *et al.*, 1991) and lipid peroxidation increases after α tocopherol consumed during LDL oxidation (Esterbauer *et al.*, 1987, Jessup *et al.*, 1990). However, this was contradicted by arguments from other researchers who reported that the decrease in susceptibility of LDL to oxidation does not correspond to the α -tocopherol content of LDL (Stocker *et al.*, 1991, Bowry *et al.*, 1992) and the pro-oxidant activity of vitamin E *in vivo* and *in vitro* especially in conditions of low oxidative stress, was demonstrated by Bowry and others (Bowry and Stocker, 1993, Neuzil *et al.*, 1997). Other studies have suggested that the activity of α -tocopherol as an antioxidant *in vivo* can be upheld by activities of other antioxidants, such as ascorbate, found within atherosclerotic lesion (Suarna *et al.*, 1995) and ubiquinol-10 which can both convert the pro-oxidant radical form of α -tocopherol back to the antioxidant α -tocopherol (Stocker *et al.*, 1991, Neuzil *et al.*, 1997, Stocker, 1999). Ubiquinol-10 and vitamin C (Retsky et al., 1993) prevents the formation of lipid peroxidation products in LDL containing vitamin E (Bowry *et al.*, 1992).



Figure 5.1: The structures of α-tocopherol, ascorbate and dehydroascorbate

Vitamin C (ascorbate) (Figure 5.1B) is considered to be a main water-soluble antioxidant which also possesses a chain breaking activity like the lipid-soluble vitamin E. Ascorbic acid possess a strong reducing ability and has been demonstrated to act as antioxidant *in vitro* and *in vivo* (Bendich *et al.*, 1986, Vinson and Jang, 2001). Vitamin C preserved LDL and human plasma lipids from oxidant-induced peroxidative damage and was suggested to be more potent in preventing peroxidative events than endogenous antioxidants present in LDL and plasma (Frei
et al., 1989, Frei, 1991) and Jialal *et al.* previously reported that vitamin C is a more potent antioxidant than vitamin E, in terms of inhibition of oxidative modification of LDL (Jialal et al., 1990). Vitamin C protected against homocysteine and ferric iron induced LDL oxidation (Alul *et al.*, 2003). Horsley *et al* reported the pro-oxidant and antioxidant behaviour of oxidised form of vitamin C, dehydroascorbate (Fig. 5.1C) (Horsley *et al.*, 2007). They showed that dehydroascorbate can act as an antioxidant with fresh LDL or a pro-oxidant with mildly-oxidised LDL in the presence of copper. This finding corroborates previous findings by Stait and Leake that dehydroascorbate and ascorbate increase the oxidation of partially oxidised LDL and enhance its uptake by macrophages (Stait and Leake, 1994, Stait and Leake, 1996).

The role of vitamin E and C in protecting LDL against oxidation remains controversial. The role of antioxidants in preventing atherosclerosis is continuously challenged by the lack of protection against CVD in large human clinical trials with vitamin E, vitamin C or β -carotene (Jialal *et al.*, 1999, Yusuf *et al.*, 2000, de Gaetano, 2001, Lonn *et al.*, 2002, Collins *et al.*, 2002) and (Lee *et al.*, 2004, Cook *et al.*, 2007, Sesso *et al.*, 2008). Although some small trials with vitamin E (Stephens *et al.*, 1996, Boaz *et al.*, 2000) and vitamin C (Khaw *et al.*, 2001, Osganian *et al.*, 2003) showed protection against CVD but there are no substantial data to support a protective role for vitamin E and C from human trials. A recent review by Al-Khudairy and colleagues suggested that there is no evidence for the reduction of risk of CVD by vitamin C in healthy subjects (Al-Khudairy *et al.*, 2017). The lack of strong evidence of protection from the clinical trials and the usually positive evidence on protection of fresh LDL against oxidation makes the role of vitamin E and C in lysosomal LDL oxidation a subject of interest in this present study.

Tempol (4-hydroxy-2,2,6,6-tetramethylpiperidine-N-oxyl) (Fig 5.2) is a potent antioxidant with super oxide (O_2^-) radical scavenging, superoxide dismutase mimetic property and has the ability to scavenge other reactive oxygen species (Wilcox and Pearlman, 2008). Tempol has been previously reported to reduce lipid peroxidation and blood pressure in animal models (Schnackenberg *et al.*, 1998, Schnackenberg and Wilcox, 1999) with the antihypertensive effect been related to its *in vitro* ability to dismutate superoxide (Patel *et al.*, 2006). Tempol is considered to have the most powerful effects amongst nitroxides in preserving cells from the deleterious effects of reactive oxygen species (Krishna *et al.*, 1998, Li *et al.*, 2006). The superoxide mimetic property and the free radical scavenging capacity might be of value in protecting LDL from oxidation by ferritin.



(4-Hydroxy-2,2,6,6-tetramethylpiperidine-1-oxyl)

Figure 5.2: Structure of tempol

N, N'- Diphenyl 1, 4-phenylenediamine (DPPD) (Figure 5.3A) is an orally active compound with antioxidant properties. Previous studies have revealed that DPPD reduced the progression of atherosclerosis in rabbits (Sparrow *et al.*, 1992) and apoE deficient mice (Tangirala *et al.*, 1995) fed high cholesterol diets without affecting the cholesterol concentration in plasma.

DPPD has previously been shown to protect LDL from oxidation by endothelial cells (Sparrow *et al.*, 1992) and copper-mediated oxidation (Tangirala *et al.*, 1995).

Chloroquine a weak base with the ability to increase the pH of lysosome and concentrate in lysosomes inhibited the oxidation of LDL (Wen and Leake, 2007). Antioxidants that possess the ability to concentrate in the lysosomes might be protective against lysosomal LDL oxidation. Cysteamine (Figure 5.3B) is present normally in low concentrations in plasma because it is produced endogenously from the degradation of coenzyme A. It can upregulate the synthesis of glutathione, protecting cells from oxidative damage (Wilmer *et al.*, 2011). Cysteamine is currently used in the treatment of cystinosis, a disease characterised by the accumulation of cystine in lysosomes due to mutation in the genes for the lysosomal cystine transporter (cystinosin). The disease leads to progressive dysfunction of multiple organs due to accumulation of cystine in all the cells in the body. Cysteamine exports cysteine out of the lysosome via the lysine transport system by the formation of the mixed disulphide of cysteamine and cysteine. Cysteamine, which was introduced in 1976, remains the only drug for the treatment of cystinosis (Gahl *et al.*, 2002, Napolitano *et al.*, 2015).



Figure 5.3: Structures of DPPD and cysteamine

The usual view of the oxidative modification process of LDL is challenged by the mostly negative effects seen with large clinical trials against CVD (Steinberg and Witztum, 2002) and the fact that low concentrations of serum or interstitial fluid inhibit LDL oxidation (Leake and Rankin, 1990, Dabbagh and Frei, 1995). These challenges can be accounted for by the recently proposed lysosomal theory which suggests that non-oxidatively modified LDL can be rapidly taken up by macrophages and then extensively oxidised within the lysosomes (Wen and Leake, 2007). A recent study from our laboratory showed the ability of DPPD and cysteamine to inhibit LDL oxidation mediated by iron at lysosomal pH (Ahmad and Leake, 2018).

In view of all the controversy surrounding the use of antioxidants in prevention and treatment of atherosclerotic cardiovascular disease, the identification of lysosomal LDL oxidation is a candidate for reinvigorating the LDL oxidation hypothesis. There is a need for identification of appropriate antioxidants that can be beneficial for the treatment of atherosclerosis. Chapter three and four of this thesis have shown that ferritin can mediate LDL oxidation at lysosomal pH. It was hypothesised that antioxidants might play a role in protecting LDL from oxidation by ferritin. Hence, this chapter explores the role of antioxidants in preventing LDL oxidation by ferritin using the major physiological lipid and water soluble vitamins (vitamin E and C) with antioxidant properties, the amphipathic compound tempol, the hydrophobic compound DPPD and the water soluble lysosomotropic compound cysteamine.

5.1 Objectives

i) To investigate the role of vitamin E (α -tocopherol) and vitamin C(ascorbate and dehydroascorbate) in LDL oxidation mediated by ferritin at lysosomal pH.

ii) To evaluate the effects of existing lipid hydroperoxides and pH on antioxidant effects of ascorbate

iii) To explore the ability of tempol to scavenge superoxide radical during LDL oxidation by ferritin and prevent oxidation.

iv) To evaluate the effects of DPPD and cysteamine on LDL oxidation by ferritin at lysosomal pH.

5.2 Methods

LDL was isolated as described in sections 2.2.1 and 2.2. Formation of conjugated dienes, kinetics of oxidation of LDL oxidation by ferritin in the presence or absence of antioxidants was monitored as described in section 2.3.7. The release of iron from ferritin in the presence of cysteamine at varying time points was measured using the iron chelator bathophenanthroline disulphonic acid (sections 2.3.8).

5.3 Results

5.3.1 Effect of α-tocopherol in LDL oxidation mediated by ferritin

The enrichment of LDL with α -tocopherol inhibited LDL oxidation by copper sulphate (5 μ M) at pH 7.4 (Dieber-Rotheneder *et al.*, 1991, Satchell and Leake, 2012, Alboaklah, 2018b). In order to test the effect of vitamin E on LDL oxidation by ferritin, we enriched LDL with α -tocopherol by incubating human plasma with α -tocopherol dissolved in DMSO for 3 h and then isolating LDL by ultracentrifugation. The α -tocopherol content was measured as previously described by Satchell (Satchell and Leake, 2012). The α -tocopherol content in LDL increased from 15 ± 0.4 to 26 ± 0.8 nmol/mg protein (Alboaklah, 2018b), increasing the average number of α -tocopherol molecules per LDL particle from about 8 to 13. Control LDL or α -tocopherol enriched LDL (50 μ g protein/ml) was oxidised with ferritin (0.1 μ M) at pH 7.4 (MOPS buffer) and pH 4.5 (NaCl/sodium acetate buffer) (Figure 5.4) at 37 ^oC. The formation of conjugated

dienes was monitored continuously at 234 nm. The mean attenuance at 200 min at acidic pH was slightly increased from 0.204 ± 0.12 in control compared to 0.260 ± 0.12 in α -tocopherol enriched LDL. The increase in formation of conjugated dienes by α -tocopherol was not statistically significant compared to control LDL at either pH 4.5 or pH 7.4.



Figure 5.4. The effect of α-tocopherol on LDL oxidation by ferritin at pH 7.4 and pH 4.5

Control LDL or LDL enriched with α -tocopherol (50 µg protein/ml) was incubated with ferritin (0.1 µM) at pH 4.5 or pH 7.4 and the formation of conjugated dienes was monitored at 234 nm against appropriate reference cuvettes that lacked LDL. This is a representative of three independent experiments (A). The mean and SEM of attenuance at 200 min were obtained for each concentration and compared by one-way ANOVA followed by a Tukey post hoc test. (n=4) (p>0.05), ns indicates not significant (B).

5.3.2 Effects of vitamin C (ascorbate and dehydroascorbate) on LDL oxidation by ferritin at lysosomal pH.

According to a 2015 study, the recommended vitamin C intake is an amount that can make up for the metabolic loss of vitamin C and guarantees a fasting plasma level of 50 μ M of ascorbate (German Nutrition, 2015). The effect of varying concentrations of vitamin C (ascorbate and the oxidised form dehydroascorbate) on LDL oxidation mediated by ferritin were evaluated. LDL (50 μ g protein/ml) was incubated with ferritin (0.1 μ M) in the presence or absence of varying concentrations (10 - 100 μ M) of ascorbate or dehydroascorbate in NaCl/sodium acetate buffer at 37 °C. The formation of conjugated dienes was monitored at 234 nm at one minute intervals for about 1200 minutes. Ascorbate inhibited the initial phase of oxidation of LDL in a concentration-dependent manner, however, ascorbate exhibited a pro-oxidant effect at a later stage (Fig 5.5). The oxidised form (dehydroascorbate) does not offer any protection towards LDL oxidised by ferritin; it rather acted as a pro-oxidant increasing the oxidation of LDL by ferritin in a concentration-dependent manner. The pro-oxidant effect was seen from early phase with rapid oxidation and early observation of aggregation and sedimentation phase (Fig. 5.6)

Statistical analysis of the increase in attenuance at 200 min with one way ANOVA (n = 4) followed by Tukey post hoc test showed that the two higher concentrations (30 μ M and 100 μ M) of ascorbate was significantly different from control with 100 μ M exhibiting a stronger statistical effect (Figure 5.5). The pro-oxidant effects of higher concentrations of dehydroascorbate were significantly different from control (p < 0.001) (Fig. 5.5). The effects of varying concentrations dehydroascorbate on the early oxidation of LDL was the opposite of that for ascorbate.



Figure 5.5: The effect of ascorbate on LDL oxidation by ferritin at lysosomal pH

LDL (50µg protein/ml) in NaCl/sodium acetate buffer (pH 4.5) was incubated with ferritin(0.1 μ M) in the absence (red line) or presence of varying concentrations of ascorbate 10 μ M (grey line), 30 μ M (blue line) and 100 μ M (green line) at 37 °C in capped cuvettes. The formation of conjugated diene was monitored by measuring attenuance at 234 nm against appropriate reference cuvettes that lacked LDL. This is a representative of four independent experiments (Fig. 5.5A). The mean and SEM of attenuance at 200 min were obtained for each concentration and compared (n=4) by one-way ANOVA followed by a Tukey post hoc test. * indicates P < 0.05 and ** indicates P < 0.01 (Fig. 5.5B).





Figure 5.6: The effect of dehydroascorbate on LDL oxidation by ferritin at lysosomal pH

LDL (50µg protein/ml) in NaCl/sodium acetate buffer (pH 4.5) was incubated with ferritin (0.1 µM) in the absence (red line) or presence of varying concentrations of dehydroascorbate 10 µM (black line), 30 µM (green line) and 100 µM (blue line) at 37° C in capped cuvettes. The formation of conjugated diene was monitored by measuring attenuance at 234 nm against appropriate reference cuvettes that lacked LDL. This is a representative of four independent experiments (A). The mean and SEM of attenuance at 200 min were obtained for each concentration and compared to the control by one-way ANOVA (n = 4) followed by a Tukey post hoc test. *** indicates P < 0.001 (B).

5.3.3 Effects of existing oxidised lipids and pH on antioxidant effects of ascorbate

A Previous study has demonstrated that the antioxidant activity of ascorbate and dehydroascorbate on copper-mediated oxidation of LDL was lost when LDL is partially oxidised (Stait and Leake, 1996). The effects of the presence of existing oxidised lipids on antioxidant/pro-oxidant activities of ascorbate on LDL oxidised by ferritin was tested. LDL (50 μ g protein/ml) was incubated with ferritin (0.1 μ M) in the presence or absence of 30 μ M ascorbate added at three different time points (0, 100 and 200 minutes) in NaCl/sodium acetate buffer at 37 °C. The formation of conjugated dienes was monitored continuously at 234 nm at one minute intervals for about 1200 minutes (Figure 5.7A). The results showed that the presence of lipid hydroperoxides or other products of oxidation led to ascorbate exhibiting an immediate and rapid pro-oxidant effect on LDL. The increase in attenuance at 200 min were compared with one way ANOVA (n = 3) followed by Tukey post hoc test showed that ascorbate significantly decreased the attenuance at pH 7.4 compared to control (p< 0.001) (Fig 5.7B).

Ascorbate has been shown to inhibit oxidation of fresh LDL by macrophages (Stait and Leake, 1994) and copper (Stait and Leake, 1996). The effect of pH on the antioxidant activity of ascorbate towards fresh LDL oxidised by 5 μ M CuSO₄ was tested. LDL (50 μ g protein/ml) was incubated with CuSO₄ (5 μ M) in in NaCl/sodium acetate buffer (pH4.5) or MOPS buffer (pH 7.4) in the presence or absence of 30 μ M ascorbate at 37 °C. The formation of conjugated dienes was monitored at 234 nm at one minute intervals for about 1200 minutes (Figure 5.8A). Ascorbate effectively delayed the formation of conjugated dienes in LDL oxidised with CuSO₄ at pH 7.4, but after a lag time of about 100 min increased the rate of oxidation of LDL oxidised with one

way ANOVA (n = 3) followed by Tukey post hoc test showed that ascorbate significantly decreased the attenuance at pH 7.4 compared to control (p < 0.001) (Fig 5.8B).



Figure 5.7. The effect of existing oxidised lipids on the effect of ascorbate on LDL oxidation by ferritin at lysosomal pH

LDL (50µg protein/ml) in NaCl/sodium acetate buffer (pH 4.5) was incubated with ferritin (0.1 µM) in the absence (red line) or presence of 30 µM ascorbate added at different time points, namely 0 min (black line), 100 min (green line) and 200min (blue line) at 37 $^{\circ}$ C in capped cuvettes. The formation of conjugated diene was monitored by measuring attenuance at 234 nm against appropriate reference cuvettes that lacked LDL. This is a representative of three independent experiments (A). The mean and SEM of attenuance at 200 min were obtained for each concentration and compared by one-way ANOVA (n = 3) followed by a Tukey post hoc test. *** indicates P < 0.001. ### indicates P < 0.001 to shown comparison (B).



Figure 5.8. The effect pH on ascorbate protecting LDL from copper-mediated LDL oxidation

LDL (50µg protein/ml) in NaCl/sodium acetate buffer (pH 4.5) or MOPS buffer (pH 7.4) was incubated with 5 µM CuSO₄ in the absence or presence 30 µM ascorbate at 37^oC in capped cuvettes. The formation of conjugated diene was monitored by measuring attenuance at 234 nm against appropriate reference cuvettes that lacked LDL. This is a representative of three independent experiments (A). The mean and SEM of attenuance at 200 min were obtained for each experiment and compared by one-way ANOVA (n = 3) followed by a Tukey post hoc test. *** indicates P < 0.001, ns indicates not significant (B).

5.3.4 Effects of tempol on LDL oxidation by ferritin at lysosomal pH.

Work from chapter three of this thesis suggested that oxidation of LDL by ferritin might be mediated by highly reactive hydroperoxyl radicals which are formed by protonation of superoxide radicals and this reaction is highly favoured at acidic pH. Hence, the superoxide dismutase mimetic property of tempol is of interest as this could prevent the formation of hydroperoxyl radicals and prevent LDL oxidation by ferritin. Previous work from our laboratory (unpublished data) demonstrated that tempol inhibited all phases of oxidation completely in LDL oxidised by 5 μ M FeSO₄ but not the lag phase. This effect was concentration dependent with concentration as low as 10 μ M exhibiting this effect.

LDL (50 µg protein/ml) was incubated with ferritin (0.1 µM) in the presence or absence of 10 µM tempol in NaCl/sodium acetate buffer (pH 4.5) at 37° C. The formation of conjugated dienes was monitored at 234 nm at one minute intervals for about 1200 minutes (Figure 5.9A). The reactive oxygen species scavenging compound, tempol (10 µM) exhibited no effect on the early stage of oxidation of LDL by ferritin but inhibited the formation of conjugated dienes later on (Fig 5.9). The inhibition observed at later times was sometimes a partial inhibition, as shown in Fig. 5.9A, or sometimes more pronounced as shown in Fig 5.9B. Statistical analysis of mean \pm SEM increase in attenuance at 200 min with student t test (n = 6) showed the effect of tempol on LDL oxidation by ferritin was not statistically significant (P > 0.05) (Figure 5.9C).



Fig. 5.9: The effect of tempol on LDL oxidation by ferritin at lysosomal pH

LDL (50µg protein/ml) in NaCl/sodium acetate buffer (pH 4.5) was incubated with ferritin (0.1 µM) at 37 $^{\circ}$ C in the absence or presence of tempol (10 µM). These results are representative of six independent experiments (A and B). The mean and SEM of attenuance at 200 min were obtained for each experiment and compared by paired Students t test (n = 6). ns indicates not significant (C).

5.3.5 Effects of DPPD on LDL oxidation by ferritin at lysosomal pH.

DPPD has been shown to promote LDL resistance to copper-mediated oxidation (Tangirala *et al.*, 1995). We tested the effect of DPPD on LDL oxidised by ferritin at lysosomal pH. Native LDL (50 μ g protein/ml) was oxidised with 0.1 μ M ferritin in NaCl/sodium acetate buffer (pH 4.5) in the presence or absence of DPPD (5 μ M or 10 μ M). DPPD was dissolved in ethanol hence ethanol was added (1% v/v) in the absence of DPPD to check if it can influence LDL oxidation. Formation of conjugated dienes was monitored (Fig. 5.10). The presence of DPPD greatly inhibited the formation of conjugated dienes. Ethanol had little effect on the oxidation of LDL by ferritin.

The rise in attenuance at 200 min for three independent experiments were compared using oneway ANOVA followed by Tukey post hoc test (Fig. 5.10B). The oxidation of LDL by ferritin in the presence of 5 μ M and 10 μ M are statistically different from the control (P<0.001). However, the addition of 1% (v/v) ethanol exhibited a little effect on LDL oxidation.





LDL (50µg protein/ml) was oxidized with ferritin 0.1µM in NaCl/sodium acetate buffer (pH 4.5) at 37°C in the absence or presence of DPPD 5 µM, 10 µM and ethanol (1% v/v). The formation of conjugated diene was monitored by measuring attenuance at 234 nm against appropriate reference that lacked LDL. This is a representative of three independent experiments (A). The rise in attenuance at 200 min in the presence or absence of cysteamine (5, 10µM) and ethanol (1% v/v) in three independent experiments was determined. Mean and SEM were obtained for each concentration and compared with one-way ANOVA (n=3) followed by Tukey's post hoc test. *** indicates P <0.001 and * indicates P <0.05 (B).

5.3.6 Effect of cysteamine on LDL oxidation mediated by ferritin at lysosomal pH

5.3.6.1 Effect of lower concentrations of cysteamine on LDL oxidation by ferritin at lysosomal pH

It has been proposed that the ability of cysteamine to concentrate in the lysosomes might be of benefit in inhibition of LDL oxidation. It was of interest to see if cysteamine will exert on inhibitory effect on the oxidation of LDL by ferritin.

LDL (50 μ g LDL protein/ml) was oxidised with 0.1 μ M ferritin in NaCl/sodium acetate buffer (pH 4.5) in the presence or absence of cysteamine using varying concentrations of cysteamine (5 μ M – 1000 μ M). Formation of conjugated dienes was monitored. Cysteamine had a complex, but consistent, the effect on the rate of LDL oxidation. The presence of cysteamine slowed down the early oxidation by ferritin in a concentration-dependent manner (Fig. 5.11A). At later time points, cysteamine increased the rate of oxidation of LDL compared to the control LDL with the greatest effect seen with 25-100 μ M cysteamine. All cysteamine concentrations decreased the time to maximum aggregation, except 1,000 μ M. The rise in attenuance at 200 min for three independent experiments were compared using one-way ANOVA followed by Tukey post hoc test. The oxidation of LDL by ferritin in the presence of 250 μ M and 1000 μ M were statistically different from the control (Fig. 5.11B).







LDL (50µg protein/ml) was oxidised with ferritin (0.1µM) in NaCl/sodium acetate buffer (pH 4.5) at 37°C in the absence or presence of cysteamine (at various concentration (5 - 1000 µM). The formation of conjugated dienes was monitored by measuring attenuance at 234 nm against appropriate references that lacked LDL. This is representative of three independent experiments (A). The rise in attenuance at 200 min in the presence or absence of cysteamine (5 - 1000 µM) in three independent experiments was determined. Mean and SEM were obtained for each concentration and compared (n=3) ANOVA followed by Tukey's post hoc test. * indicates P <0.05 compared to control. $\neq \neq$ indicates P <0.01 to shown comparison (B).

5.3.6.2 Effect of higher concentrations of cysteamine on LDL oxidation by ferritin at lysosomal pH

Up to $1.1.95 \text{g/m}^2/\text{day}$ of cysteamine is recommended to patients with nephropathic cystinosis over the age of 12 and weight above 50kg divided over a period of four times (Besouw *et al.*, 2011, Besouw and Levtchenko, 2014). The concentration of cysteamine in plasma was estimated to be about 40 μ M. It should accumulate up in several orders of magnitude in lysosomes (Pisoni *et al.*, 1995), hence millimolar concentrations would be expected to accumulate in the lysosomes.

LDL (50µg LDL protein/ml) was oxidised with 0.1µM ferritin in NaCl/sodium acetate buffer (pH 4.5) in the presence or absence of cysteamine (1, 3 and 10mM). Formation of conjugated dienes was monitored (Fig. 5.12). The presence of higher concentration of cysteamine inhibited the formation of conjugated dienes to a large extent. The rise in attenuance at 200 min for three independent experiments were compared using one-way ANOVA followed by Tukey post hoc test. The oxidation of LDL by ferritin in the presence of 1mM, 3mM and 10 mM are statistically different from the control (Fig. 5.11B). However, there was no significant different between the two higher concentrations as compared to each.



Fig 5.12: The effect of higher concentrations of cysteamine on LDL oxidation by ferritin at lysosomal pH.

LDL (50µg protein/ml) was oxidised with ferritin 0.1μ M in NaCl/sodium acetate buffer (pH 4.5) at 37°C in the absence, 0μ M (red line) or presence of cysteamine 1mM (black line), 3mM (orange line) and 10 mM (green line). The formation of conjugated dienes was monitored by measuring attenuance at 234nm against appropriate reference that lacked LDL. This is a representative of three independent experiments (A). The rise in attenuance at 200 min in the presence or absence of cysteamine (1, 3 and 10mM) in three independent experiments was determined. Mean and SEM were obtained for each concentration and compared by ANOVA followed by Tukeys post hoc test. *** indicates P <0.001 (B).

5.3.7 Effect of cysteamine on ferrous complex formation by ferritin

Iron released from ferritin in the presence of cysteamine was assessed to explore the prooxidant mechanism of cysteamine. Ferritin (0.1 μ M) was incubated in Falcon tubes (15ml) in Chelex-treated NaCl/sodium acetate buffer (pH 4.5) at 37 °C in the presence or absence of 25 μ M or 1mM cysteamine. A fraction of 1ml was taken at intervals up to 24 h. BP (30 μ l of 10mM) was added at different time points and absorbance was measured at 535nm. More ferrous complex was formed in the presence of cysteamine (Fig. 5.13).

The ferrous complex formed were compared for each time point, using two-way ANOVA followed by Bonferoni post-test. The mean ferrous complex formed by ferritin in the presence of cysteamine 25 μ M and 1 mM (A and B) were significantly higher over time when compared to the controls.



Fig. 5.13: The effect of cysteamine on ferrous complex formation by ferritin at pH 4.5.

Ferritin (0.1 μ M) was incubated at 37 °C in NaCl/Na acetate buffer (pH 4.5) in the absence or presence of cysteamine (25 μ M) (A) or 1 mM (B). At different times, a sample (1 ml) was taken and the ferrous chelator bathophenanthroline was added, left for 5 min and the absorbance was measured at 535 nm. These results were obtained from mean \pm SEM of three independent experiments. The means were compared with two-way ANOVA followed by Bonferoni post-test. *** indicates P <0.001 and * indicates P <0.05.

5.4 Discussion

The Oxidative modification hypothesis argues that oxidised LDL is an early event in atherosclerosis and oxidised LDL promotes progression of atherosclerosis. Oxidised LDL incites many atherogenic events (Steinberg *et al.*, 1989). This hypothesis was supported by the protective effects observed with some antioxidants in animal models reviewed by Steinberg in 2009 (Steinberg, 2009). However, the relevence of this hypothesis is continually questioned due to emerging evidence of lack of antioxidant protection against CVD in human trials (Steinberg and Witztum, 2002). These suggest that there is a need for more understanding of the mechanism of LDL oxidation, identification of more appropriate antioxidants. Our laboratory has shown that lysosomal oxidation of LDL might account for this challenge (Wen and Leake, 2007) and demonstrated that some antioxidants inhibit the oxidation of LDL by FeSO4 less at lysosomal pH (Ahmad and Leake, 2018) than they do at interstitial fluid or plasma pH. This present study examined the effects of antioxidants on LDL oxidation by ferritin.

Our laboratory has shown previously that α -tocopherol enrichment of LDL inhibits LDL oxidation by copper ions at pH 7.4, but it did not effectively inhibit LDL oxidation by ferrous iron at lysosomal pH and had an initial pro-oxidative effect (Satchell, 2008, Satchell and Leake, 2012, Alboaklah, 2018a). In this present study, enrichment of LDL with α -tocopherol did not decrease the rate of oxidation of LDL by ferritin at pH 4.5 or pH 7.4. There was a tendency of α -tocopherol enrichment of LDL to increase the oxidation of LDL by ferritin at pH 4.5, but this was not statistically significant. The lack of antioxidant effect might be due to ferric ions released from ferritin (or ferrous ions released from ferritin subsequently oxidised to ferric ions) converting α -tocopherol to α -tocopheroxyl radicals, which are not entirely stable and can

abstract a hydrogen atom from the bisallylic groups of polyunsaturated fatty acids present in lipids in LDL increasing lipid peroxidation (equation 1 and 2 (Bowry *et al.*, 1992).

 $\begin{array}{ll} \label{eq:a-toc} a\text{-toc}~OH + Fe^{3+} \rightarrow a\text{-toc}~O^{\cdot} + H^+ + Fe^{2+} & eqn~1\\ \\ a\text{-toc}~O^{\cdot} + LH \rightarrow a\text{-toc}~OH + L^{\cdot} & eqn~2 \end{array}$

This pro-oxidant effect of α -tocopherol might negate the antioxidant effect of α -tocopherol (equation 3) leading to little net effect of this compound.

α -toc OH + LOO[.] $\rightarrow \alpha$ -toc OH + LOOH eqn 3

This increased oxidation effects observed in ferritin oxidation of LDL enriched with α -tocopherol supports previous evidence from our laboratory that vitamin E loses its antioxidant potential against Cu²⁺ and Fe³⁺ at lysosomal pH (Alboaklah, 2018a).

The role of ascorbate in atherosclerosis remains controversial, whilst some studies have demonstrated an anti-atherogenic effect for ascorbate supplementation in cholesterol fedanimals (Verlangieri *et al.*, 1977, Beetens *et al.*, 1984, Lynch *et al.*, 1996). Some other studies have found no beneficial effect against atherosclerosis in animal models (Morel *et al.*, 1994). Antioxidant and pro-oxidant effects of ascorbate against cell-mediated and cell-free LDL oxidation systems has been previously demonstrated (Stait and Leake, 1994, Stait and Leake, 1996). It was demonstrated that ascorbate exhibited a concentration-dependent antioxidant and pro-oxidant effects on LDL oxidation by ferritin at pH 4.5. It can be proposed that the antioxidant effect observed may be due to ascorbate regenerating α -tocopherol from α -tocopheroxyl radicals (equation 4). The ascorbyl radical formed during this process can be converted back to its stable form by dismutation, hence ascorbate may be acting as a co-antioxidant for α -tocopherol (Carr *et al.*, 2000b). Ascorbate might possibly be binding the ferrous iron released from ferritin and preventing it from mediating lipid peroxidation reaction.

α -toc O[·] + ASC H₂ $\rightarrow \alpha$ -toc OH + ASCH[·] eqn 4

However, antioxidants that regenerate α -tocopherol may not be the most appropriate antioxidants for preventing lysosomal LDL oxidation. They may not be efficient in protecting LDL as their antioxidant capacity might be slowed down or be lost completely when endogenous α -tocopherol in LDL is depleted.

The antioxidant activity of ascorbate was lost over time or in the presence of midly oxidised LDL. It can be suggested that the pro-oxidant effect of ascorbate may be due to (1) reduction of ferric iron to ferrous iron, the ferrous iron may then mediate the formation of hydroperoxyl radical as discussed in chapter 3. (2) The pro-oxidant effect in the presence of midly oxidised LDL may be due to the released ferrous iron rapidly reacting with the pre-existing lipid hydroperoxides (ferric iron reacts slowly with lipid hydroperoxides equation 5 to 7).

 $\begin{aligned} & Ferritin-Fe^{3+} + ASC \ H_2 \rightarrow Ferritin- \ Fe^{2+} \ + \ ASCH^{\cdot} + H^+ \ eqn \ 5 \\ & Fe^{2+} + LOOH \rightarrow LO^{\cdot} + OH^{\cdot} + Fe^{3+} \ fast \ eqn \ 6 \\ & Fe^{3+} + LOOH \rightarrow LOO^{\cdot} + OH^{\cdot} + Fe^{2+} \ slow \ eqn \ 7 \end{aligned}$

Dehydroascorbate did not inhibit LDL oxidation by ferritin at lysosomal pH. A previous study has demonstrated that dehydroascorbate increased the oxidation of copper mediated LDL oxidation in fresh LDL and midly oxidly oxidisded LDL in a concentration-dependent manner (Stait and Leake, 1996). Similarly, in this present study dehydroascorbate promotes the oxidation of LDL by ferritin in a concentration-depentent manner. Varying plasma concentration of dehydroascorbate has been reported with 29 μ M siting at the top of the list (Sinclair *et al.*, 1991). Dehydroascorbate at 30 μ M concentration significantly increased LDL oxidation by ferritin at pH 4.5 (P<0.001). As previously discussed by Horsley *et al*, this prooxidative effect of dehydroascorbate might be as a result of an indirect effect of its irreversible delactonisation product (2,3-diketo-L-gulonic acid) and reversible reduction products (ascorbic acid and erythroascorbate) (Horsley *et al.*, 2007). Dehydroascorbate is capable of producing a wide range of degradation product which may, in turn, produce free radicals that can mediate lipid peroxidation in the lipids of LDL (Takagi *et al.*, 1988, Deutsch *et al.*, 1994). As expected, ascorbate effectively inhibited LDL oxidation by 5 μ M CuSO₄ at pH 7.4 but had no antioxidant effect on oxidation of copper at lysosomal pH. The effect of ascorbate on fresh LDL oxidation by CuSO₄ at pH 7.4 was similar to a previous observation by Jialal *et al.* and others (Jialal *et al.*, 1990, Stait and Leake, 1996). Similarly to result observed with α -tocoperol, ascorbate loses its antioxidant activity at lysosomal pH and this might be as a result of reducton of Cu²⁺ to Cu⁺ by ascorbate (equation 7 and 8). The Cu⁺ formed can rapidly react with lipid hydroperoxides (Stait and Leake, 1996).

ASC H₂ + Cu²⁺ \rightarrow ASCH[·] + Cu⁺ + H⁺ eqn 8 Cu⁺ + LOOH \rightarrow LO[·] + OH⁻ + Cu²⁺ eqn 9

The balance between the antioxidant and pro-oxidant effects of ascorbate (and dehydroascorbate) appears to be altered as the pH changes. These findings put together may account for the failure of vitamin E and C to decrease cardiovascular disease in large human clinical trials (Yusuf *et al.*, 2000, Collins *et al.*, 2002, Sesso *et al.*, 2008).

Tempol is suggested to be the most potent nitroxide compound in scavenging reactive oxygen species and possesses superoxide dismutase activity (Li *et al.*, 2006, Wilcox and Pearlman, 2008). Tempol completely decreases the later phases of LDL oxidation by ferrous and ferric iron in a concentration-dependent manner but could not inhibit the core oxidation of the cholesteryl ester rich core during the lag phase of ferrous/ferric oxidation of LDL (unpublished data). Surprisingly, tempol did not prevent the initial phase of LDL oxidation by ferritin at lysosomal pH but inhibited the later phase of oxidation partially and sometimes totally.

The amphipathic property of tempol might enhance its accumulation in the outer monolayer of LDL containing phospholipid, rather than in the core of LDL containing mainly cholesteryl

ester. It was suspected that LDL oxidation by iron at lysosomal pH begins in the core of LDL and then later proceeds in its monolayer (Ahmad and Leake, 2018). Tempol might lack the capacity to quickly scavenge highly reactive hydroperoxyl radicals effectively enough to prevent them from mediating oxidation of the hydrophobic core of LDL, but it might be able to prevent the oxidation of the phospholipids surface monolayer by scavenging alkoxyl or peroxyl radicals formed by the phospholipids in the monolayer. Recent work from our laboratory demonstrated that tempol completely inhibited LDL oxidation by copper at pH 7.4 but loses its antioxidant capacity on the initial phase of LDL oxidation in the presence of iron at pH 4.5 (Alboaklah, 2018a).

DPPD has been demonstrated to effectively scavenge the peroxyl radical (Tangirala *et al.*, 1995). In this present study, DPPD concentrations as low as 5μ M and 10 μ M were able to reduce the formation of conjugated dienes in LDL oxidised by ferritin to a large extent (P<0.05). Others have shown DPPD to effectively inhibit oxidation of LDL by copper and endothelial cells (Sparrow *et al.*, 1992). Its hydrophobic nature is of advantage, as it can possibly scavenge radicals formed within the core of LDL. The ability of DPPD to inhibit LDL oxidation mediated by ferrous iron (Ahmad and Leake, 2018) and ferritin at lysosomal pH, might explain while it protected animal models against atherosclerosis. The use of DPPD in humans is, however, prevented by its mutagenic nature (Sofuni *et al.*, 1990).

A recent study from our laboratory has shown that cysteamine effectively inhibited LDL oxidation by ferrous iron at pH 4.5 (Ahmad and Leake, 2018). This present study showed that $5 \,\mu$ M -1000 μ M cysteamine protected LDL oxidation but not completely, as a later pro-oxidant effect was observed. All concentrations slowed down the oxidation for about 200 min. The

antioxidant effect increased with the concentration of cysteamine, the two upper concentrations were significantly different from others and each other (P<0.05).

Cysteamine contains a thiol group and thiols have previously been seen to act as promoters (Heinecke *et al.*, 1993) and inhibitors of LDL oxidation (Patterson *et al.*, 2003b). Pro-oxidant and antioxidant effects might depend on cysteamine concentrations in differing ways and this may explain the complex kinetics. From low to moderate concentrations of cysteamine, the pro-oxidant effect observed at later times increased, but with higher concentrations (3 mM and over) an antioxidant effect was seen at all times. The complexity of the effect of cysteamine was concentration and time dependent. The antioxidant effect observed in the presence of cysteamine might be due to the lower availability of iron, as cysteamine might possibly be able to bind to and inactivate released iron. It might also be scavenging the peroxyl radical (HO₂⁻) and superoxide radical (O₂⁻) formed as oxidation products (equation 10 and 11 below).

Proposed mechanisms for antioxidants effects of cysteamine:

$HO_{2} + NH_{2}CH_{2}CH_{2}SH \rightarrow H_{2}O_{2} + NH_{2}CH_{2}CH_{2}S^{\cdot} \text{ eqn 10}$ $O_{2} + NH_{2}CH_{2}CH_{2}SH + H^{+} \rightarrow H_{2}O_{2} + NH_{2}CH_{2}CH_{2}S^{\cdot} \text{ eqn 11}$

The pro-oxidant effect observed might be due to cysteamine removing iron from ferritin, as it might reduce ferric ion (Fe³⁺) in ferritin to ferrous ion (Fe²⁺) which might diffuse out of the pores in the surface of the ferritin particle. The Fe²⁺ might mediate the generation of lipid peroxides, peroxyl radicals (HO₂⁻) and superoxide radicals (O₂⁻⁻) (as described in chapter 3, section 3.4).

Proposed mechanisms for prooxidants effects of cysteamine:

$NH_2CH_2CH_2SH + Fe^{3+} \rightarrow NH_2CH_2CH_2S^{\cdot} + H^+ + Fe^{2+} \quad eqn \ 12$

It was thought that the concentration-dependent inhibition observed might mean further increases in cysteamine concentration will exert a near complete inhibitory effect. Plasma concentration of about 40µM are found in cystinosis patients (Dohil et al., 2006). Cysteamine is expected to accumulate up to two or three order of magnitude in lysosomes due to proton trapping and so lysosomal concentration might be at least several millimolar. Higher concentration of cysteamine (1 mM, 3 mM and 10 mM) inhibited the oxidation of LDL to a large extent (P < 0.05). Both low and high concentrations of cysteamine were confirmed to release iron from ferritin at lysosomal pH. The antioxidant capacity of cysteamine is more pronounced as more cysteamine is available to scavenge free radicals or possibly bind and inactivate released iron. The concentrations at which cysteamine inhibited LDL oxidation by ferritin is consistent with the concentrations already used in the treatment of the rare lysosomal storage disorder cystinosis. It is interesting to note that there was an inverse correlation between the years of cysteamine intake by patients with cystinosis and arterial calcification (Ueda et al., 2006). The potency of cysteamine to reduce atherosclerosis might be supported by the ability of cysteamine to reduce ceroid formation within lysosomes (Ahmad and Leake, 2018, Wen et *al.*, submitted) and significantly reduced atherosclerotic lesions in mice (Wen *et al.*, submitted).

Work presented in this chapter suggests that the failure of some antioxidants to show protection against cardiovascular disease in human trials does not disprove the LDL oxidation hypothesis of atherosclerosis. The study, however, implies that the lysosomotropic antioxidant, cysteamine might be a more appropriate antioxidant in inhibiting LDL and reducing events of atherosclerosis. The role of ferritin and antioxidants in lysosomal oxidation has been explored so far at lysosomal pH, there is however need to further explore the impact of LDL oxidised by ferritin within lysosomes on macrophages and atherosclerosis. This is addressed in the next chapter of this thesis. Chapter 6: The effects of lysosomal oxidation of LDL by ferritin in macrophages

6.0 Background and rational

Macrophages play a key role in the onset and progression of atherosclerosis and they regulate cellular immune response and metabolism of lipids (Ross, 1999, Lusis, 2000). As one of the major types of cells found in atherosclerotic plaque, they have great influence on inflammation, accumulation of lipids, formation of the necrotic core and degradation of extracellular matrix (Legein *et al.*, 2013). The accumulation of lipid-laden macrophages in the walls of the artery is considered an early indication of an atherosclerotic lesion. Lipid-loaded macrophages (foam cells) triggers the mediators of inflammation such as cytokines and are able to recruit other cell types to contribute to the advancement of the lesions (Shibata and Glass, 2009).

The uptake of VLDL, LDL and oxidised LDL by macrophages occur through phagocytosis, micropinocytosis, receptor-mediated endocytosis and pathways mediated by scavenger receptors. Lipids ingested by macrophages are digested in lysosomes (Tabas and Bornfeldt, 2016, Remmerie and Scott, 2018). The prevailing concept of oxidative modification is that LDL is modified in the arterial wall and then recognised by pattern recognition receptors on immune cells. The type of pattern recognition receptors on macrophages are scavenger receptors, they were suggested to recognise and take up modified LDL. Different family members of this receptor have been identified, such as scavenger receptor A (SRA), scavenger receptor class B member 1(SRB1), CD36, lectin-type oxidised LDL receptor 1 (LOX1) and others which can all bind oxidised LDL and promote foam cell formation (Moore and Freeman, 2006). However, SRA and CD36 have been identified as the major receptors responsible for mediating uptake and degradation of acetylated and oxidised LDL in macrophages (Kunjathoor *et al.*, 2002). The foam cell formation in apoE knockout mice deficient in CD36 and SRA was reduced, but not completely prevented which suggests there are other mechanisms for the formation of foam cells *in vivo* (Manning-Tobin *et al.*, 2009), suggesting there might be an

additional mechanism for macrophage transformation to foam cells. The *in vivo* relevance of this oxidation paradigm is challenged by the failure of antioxidants to be effective against CVD in large clinical trials (Steinberg and Witztum, 2002). Many types of LDL oxidation strongly inhibited by serum or interstitial fluid (Leake and Rankin, 1990, Dabbagh and Frei, 1995, Patterson *et al.*, 2003a). These findings have led to the consideration of alternate hypotheses for foam cell formation.

Modification of LDL by various proteases and lipases present in the arterial intima can promote its uptake by macrophages. Phospholipase A2 (Oorni and Kovanen, 2009) and sphingomyelinases (Xu and Tabas, 1991) have been shown to modify LDL to enhance its uptake. These events may contribute to foam cell formation, initiation and progression of atherosclerosis. Our laboratory proposed an alternate mechanism for oxidation of LDL, demonstrated that LDL is oxidised within lysosomes of macrophages (Wen and Leake, 2007) and suggested that LDL is aggregated by sphingomyelinase (SMase) or other means and rapidly taken up by macrophages and oxidised within lysosomes (Wen et al., 2015). Sphingomyelinase has been shown to be present extracellularly in atherosclerotic lesion (Marathe *et al.*, 1999). Lysosomal oxidation of SMase-LDL mediated by iron has been shown to increase secretion of pro-inflammatory cytokines (Ahmad and Leake, 2019). The proinflammatory cytokines produced by foam cells lead to vascular inflammation and lipoprotein accumulation (Libby, 2002). The link between inflammation and plaque formation has been supported by numerous studies (Libby, 2002, Libby et al., 2009, Libby, 2012). Inflammation promotes atherosclerosis and eventually causes complications linked to thrombosis, which eventually leads to the clinical events, thrombotic stroke, myocardial infarction (MI) and death arising from CVD (Libby, 2002, Libby et al., 2009).

Hepcidin, the systemic iron regulatory hormone, has been demonstrated to be elevated during inflammation (Nemeth *et al.*, 2004a, Wrighting and Andrews, 2006). Following up on the iron hypothesis of atherosclerosis discussed in chapter one of this thesis, hepcidin has been proposed to be linked to increased risk of atherosclerotic cardiovascular disease by reducing the exit of iron from macrophages and increasing their transformation to foam cells (Sullivan, 2007). Hepcidin levels and hepcidin/ferritin ratio has been shown to be associated with increased atherosclerosis in post-menopausal women (Galesloot *et al.*, 2014). The intracellular amount of ferritin is controlled by the intracellular labile iron pool and chelatable iron-induced translation of heavy and light chain mRNAs (Hentze *et al.*, 1987, Casey *et al.*, 1988, Gray *et al.*, 1993). The pathways of ferritin regulation have been linked with inflammation, as the heavy chain is transcriptionally induced by cytokines. Interleukin (IL-1), IL-6 and TNF- α upregulate the post-transcriptional synthesis of ferritin (Muntane-Relat *et al.*, 1995, Tran *et al.*, 1997). Thus, during the inflammatory process and when iron is trapped in macrophages by hepcidin, ferritin levels might also be increased. It will be interesting to understand how this is linked to lysosomal LDL oxidation by ferritin.

Lipid peroxidation has been proposed to be a major contributor to the pathogenesis of atherosclerosis. Lipid peroxidation, a free radical-mediated chain reaction propagation of oxidation of unsaturated fatty acids, might promote atherosclerosis by inducing transformation of macrophages into foam cells (Esterbauer *et al.*, 1993, Chisolm and Steinberg, 2000, Young and McEneny, 2001). Ferritin has been demonstrated in a previous chapter to oxidise LDL at lysosomal pH, hence it is relevant to explore the effect of ferritin on lipid peroxidation within lysosomes of macrophages. Reactive oxygen species (ROS) have been shown to contribute to oxidation of LDL. ROS mediation of lipid oxidation is highly dependent on superoxide anion formation which can form other reactive species such as hydroperoxyl radicals, lipid peroxides
and peroxynitrites that can further contribute to the pathogenesis of atherosclerosis (Madamanchi *et al.*, 2005, Leopold and Loscalzo, 2008). Oxidative stress has been proposed to be important in the onset of atherosclerosis (Witztum and Berliner, 1998) and ROS derived from macrophages are capable of regulating matrix metalloproteinases and might contribute to plaque instability (Rajagopalan *et al.*, 1996).

Macrophages undergo strong metabolic changes after activation which can, in turn, control the inflammatory responses. In atherosclerosis alterations in macrophage metabolism is an important factor that can dictate the function of macrophages and the progression of the disease. The mechanism of metabolism in macrophages can be modified to meet immediate cellular demands, such as proliferation, production of cytokines and phagocytosis (Liu et al., 2016, Lachmandas et al., 2016). The transformation of macrophages to classically activated phenotypes or the alternatively activated phenotypes requires a change in metabolic processes from anabolic to the catabolic mechanism, respectively. The inflammatory phenotype M1 macrophages, which are classically activated, utilises anabolic metabolism to make up for the higher rate of glycolysis and the increased need for macromolecular building blocks. The ATP quickly generated from glycolysis serves as a source of main intermediates utilised in pentose phosphate pathway (PPP), which is essential for nucleotide and NADPH synthesis and other events. The change in metabolic pathways continues to promote ongoing and future inflammatory responses (Koelwyn et al., 2018). Understanding the immunometabolic pathways in atherosclerosis and how they can be regulated might represent a novel therapeutic target in treating atherosclerosis. It was of interest to understand the effect oxidation of LDL by ferritin within macrophages can have on their energy phenotype.

Macrophages that accumulate in atherosclerotic plaques become less able to migrate, which further contributes to inflammation and progression to a more advanced plaque. In the complex plaque, macrophages continue the production of proteases that can degrade the matrix and the macrophages eventually die by necrosis or apoptosis (Randolph, 2008, Moore et al., 2013). Apoptotic cell death occurs in all stages of atherosclerotic lesion development, however, apoptosis might have a negative impact on atherosclerotic plaque stability (Kockx, 1998, Kockx et al., 1998). LDL oxidised by copper at pH 7.4 has been shown to induce apoptosis in HMDM (Wintergerst et al., 2000). Oxidised LDL has also been shown to induce apoptosis in cultured cardiomyocytes from neonatal rats (Wang et al., 2016). Apoptosis mainly occurs in the region of the lesion where the large lipid core is found and the majority of cells that die by apoptosis are macrophages (Hegyi et al., 2001). The cytotoxicity of oxidised LDL towards HMDM has been attributed to its ability to induce apoptosis in these cells (Hardwick et al., 1996). Necrotic cells are characterised by loss of integrity of the cell membrane that allows an influx of water, Ca²⁺ and Na²⁺, which leads to swelling of cytoplasm followed by the release of the cellular content to extracellular space and result in inflammation (Roos et al., 2004, Tabas, 2005).

Studies by Jerome and others have previously shown that different forms of modified lipids can disrupt the function of lysosomes and result in intralysosomal accumulation of lipids (Griffin *et al.*, 2005, Cox *et al.*, 2007, Jerome, 2010). Following up on this Emanuel *et al.* later demonstrated that exposure of macrophages to cholesterol crystals and oxLDL led to lysosomal dysfunction which is characterised by increased pH, decreased function of lysosomal enzymes and loss of lysosomal membrane integrity (Emanuel *et al.*, 2014). The oxidised LDL used previously to demonstrate pro-atherogenic effects of oxidised LDL has mostly been produced by incubating with copper ions at pH 7.4. Work presented in this thesis and previously from our laboratory have demonstrated that iron is important in mediating LDL oxidation within macrophage lysosomes (Wen and Leake, 2007, Satchell and Leake, 2012, Ahmad and Leake, 2018), lysosomal LDL oxidation promotes formation of ceroid in lysosomes of macrophages (Wen and Leake, 2007, Ahmad and Leake, 2018), promote cellular senescence and disrupts lysosomal function by altering their pH (Ahmad and Leake, 2019). If lysosomal dysfunction occurs in macrophages, the lysosomes might release their content into the cytosol, which might induce apoptosis and once cells die it contents might be released extracellularly to affect neighbouring macrophages. Further elucidation of the role of ferritin in oxidising LDL within lysosomes of macrophage might provide a novel insight into the role of ferritin-oxidised LDL in atherosclerosis. It was hypothesised that lysosomal oxidation of LDL in macrophages can modulate macrophage function and promote atherosclerosis. Hence, possible effects of lysosomal oxidation of LDL by ferritin on macrophage function as regards atherosclerosis was explored.

6.1 Objectives

i) To examine the role of hepcidin in influencing iron/ferritin levels in macrophages might play a role in lysosomal LDL oxidation.

ii) To Test the effects of ferritin on lysosomal lipid peroxidation in macrophages.

iii) To evaluate the effect of lysosomal ferritin-oxidised LDL on intracellular lipids levels and ceroid formation macrophages.

iv) To test if ferritin-oxidised LDL can mediate oxidative stress by upregulation of intracellular levels of reactive oxygen species

v) To examine the effects of ferritin-oxidised LDL on macrophages cellular respiration/ metabolism.

vi) To test the effects of ferritin-oxidised LDL on apoptotic cell death of macrophages.

6.2 Methods

The methods used for culturing THP-1 cells was described in section 2.7 and the isolation of HMDM was described in section 2.8. The effect of hepcidin on intracellular iron levels was tested as described in section 2.9. Lysosomal lipid peroxidation was measured with the novel lysosomal targettable probe (Foam-LPO) as described in section 2.10. The intracellular lipids and ceroid were measured using Oil Red O staining, as described in section 2.11. Intracellular reactive oxygen species were measured with the fluorescent probe dihydroethidium (DHE), as described in section 2.12. The effect of LDL oxidised by ferritin on respiration/metabolism was determined with Seahorse analyser, as described in section 2.13. The induction of apoptosis in macrophages was measured using propidium iodide and annexin V, as described in section 2.14.

6.3 Results

6.3.1 Effect of hepcidin on intracellular iron in macrophages

Hepcidin has been associated with decreasing body iron stores and promoting iron sequestration in macrophages during inflammation and infection (Ganz and Nemeth, 2009). The link between inflammation and sequestration in macrophages of iron on oxidation of LDL in macrophages was therefore explored. THP-1 macrophage-like cells (1.2×105) derived from THP-1 monocytes (Fig 6.1) were treated for 24 h with or without hepcidin (1000 nM) with RPMI-1640 (which does not contain added iron), DMEM (which contains 2.5 μ M ferric iron) and F10 (which contains 3 μ M added ferrous iron) supplemented with 10% (v/v) FCS. The media was replaced after 24 h with media supplemented with 10% (v/v) LPDS containing hepcidin (100 nM) or media only. This was added every 2 days. The protein content per well and iron concentration was measured after seven days.

The protein content of cells treated with hepcidin was not significantly different from control cells for all media treatments (Fig. 6.2). Surprisingly, the iron content of THP-1 macrophages was not increased by hepcidin treatment (Fig. 6.3). Work by Agoro and Mura (Agoro and Mura, 2016) suggested that the link between inflammation and hepcidin is dependent on macrophage polarization. Hence polarisation of THP-1 cells might make a difference in the experiments designed to increase macrophage iron levels by hepcidin. It would be interesting to address this in future work.

A

THP-1 monocytes



B THP1 macrophages



Figure 6.1: Untreated THP1 monocytes and macrophages

Thp1 monocytic cell line (A) and THP1 converted to macrophages after treatment with 25ng/ml PMA for 72 h (B). The images were captured with an Evos XL cell imaging system using a 20x objective.



Figure 6.2: Intracellular protein content of THP1 macrophages treated with hepcidin

THP-1 cells (1.2×10^5) were treated with or without hepcidin (1000 nM) in RPMI-1640 (A), DMEM (B) or F10 (C) with 10% (v/v) FCS. The media was replaced after 24 h with media containing 10% (v/v) LPDS with or without hepcidin (100 nM), added every 2 days. The protein content of cells per well was measured with a Bio-Rad DCTM protein assay kit. The protein content of cells treated with hepcidin was not different compared to control cells using paired Student's t test (n=3). ns indicates not significant.



Figure 6.3: Intracellular iron content of THP-1 macrophages treated with hepcidin

THP-1 cells (1.2×10^5) were treated with or without hepcidin (1000 nM) in RPMI-1640 (A), DMEM (B) and F10 (C) with 10% (v/v) FCS. The media was replaced after 24 h with media containing 10% (v/v) LPDS with or without hepcidin (100 nM), added every 2 days. The iron content of cells was measured with ICP-MS. There was no significant different between iron levels in cells treated with hepcidin compared to control cells, as compared with paired Student's t test (n=3). ns indicates not significant.

6.3.2 Effects of ferritin on lipid peroxidation within lysosomes of macrophages

Lipid peroxidation mediated by free radicals has been linked to in vivo oxidative stress and associated with the lipid modification hypothesis of atherosclerosis (Niki *et al.*, 2005). In a previous chapter, it was demonstrated that ferritin increased the formation of conjugated dienes and total lipid hydroperoxides, both of which are products of lipid peroxidation in LDL, at lysosomal pH. It was of interest to test the effect of ferritin on lipid peroxidation within the lysosomes of cultured macrophages in order to determine if ferritin contributes to increasing lysosomal LDL oxidation.

THP-1 macrophages (5 x 10^5 cells/well) treated RPMI–1640 with 10% human serum (2ml per well) alone or with ferritin (100 µg protein/ml in terms of total protein and this is equivalent to 0.2 µM) for 24 h. The cells were then treated with or without LDL (200 µg protein/ml) for 24 h to give the following condition (no LDL, native LDL only, LDL with ferritin and ferritin only). After the 48 h treatment with ferritin and LDL, the intralysosomal lipid peroxidation was measured by two-way flow cytometry (using the green (FL1) and the red (FL2 channel), following treatment with Foam-LPO (2 µM) for 15 min in the dark. The lysosomal targettable probe is flexibly linked to a diene-fluorophore, the Foam-LPO is intercalated in the lipid accumulation and can detect the local lipid peroxidation within lysosomes. Foam-LPO detects the peroxidation by spectral shifting. Hence, lysosomal lipid peroxidation was quantified by the ratio of the intensity of the fluorescent green channel to that of the red channel (FL1/FL2). The analysis was carried using flow Jo software version 10.

The results showed that ferritin with or without native LDL significantly increased lipid peroxidation (P <0.001) (Figs. 6.4 & 6.5). A significant difference was seen with ferritin alone as well as ferritin plus LDL. Ferritin plus LDL was significantly different (P< 0.001) from

native LDL only and control (p <0.05). While treatment with LDL only was not significantly different compared to control.

LDL aggregated with sphingomyelinase has been demonstrated to be present in atherosclerotic lesions (Marathe *et al.*, 1999) and rapidly taken up by macrophages and oxidised in lysosomes (Wen *et al.*, 2015, Ahmad and Leake, 2019). The mechanism of SMase aggregation of LDL was described in detail in section 1.5.5 and measured in fig. 2.1. THP1- cells and HMDM cells were treated with or without ferritin (0.2 μ M) for 24 h. We then incubated with or without SMase-LDL (200 μ g protein/ml), respectively. We analysed for lipid peroxidation using the lysosomal targeted probe. The results showed that lysosomal lipid peroxidation increased in the presence of ferritin in THP-1 macrophages (P < 0.001) and HMDM (P<0.01) (Fig. 6.6). The results show that ferritin with or without SMase-LDL can mediate lysosomal lipid peroxidation in macrophages.



Figure 6.4: Two-way flow cytometry analysis of lipid peroxidation in THP-1 macrophages

THP1-cells were treated with or without ferritin (0.2 μ M) for 24 h and then incubated in the presence or absence of native LDL (200 μ g protein/ml) for 24 h. THP1-cells were harvested and incubated with the lysosomal lipid peroxidation probe, Foam-LPO (2 μ M) for 15 min and assayed by flow cytometry. The ratio of FL1/FL2 corresponds to the level of lipid peroxidation as detected by the Foam-LPO (A) Flow cytometry results for untreated macrophages (control cells) and macrophages treated with native LDL. (B) Flow cytometry results for control cells and macrophages pre-treated with ferritin and then treated with native LDL. (C) Flow cytometry results for control cells and macrophages pre-treated with ferritin. The image shown is a representative of three independent experiments.



Figure 6.5: Measurement of lipid peroxiation in THP-1 macrophages in the presence of native LDL and ferritin.

THP1-cells were treated with or without ferritin for 24 h and then treated with or without native LDL (200 µg protein/ml) for 24 h. THP1-cells were then incubated with the lysosomal lipid perorixidation probe, Foam-LPO (2 µM) for 15 min. The lysosomal lipid peroxidation was determined by two-way flow cytometry and quantified by the ratio of intensity of fluorescence of the green channel compared to that of the red channel (FL1/FL2) using Flow Jo software. The ratio of intensity was compared for the treated cells and control cells using one-way ANOVA followed by a Tukey's post hoc test (n=3). * indicates P < 0.05 and *** indicates P < 0.001 compared to control cells. $\neq\neq\neq$ indicates P < 0.001 for the indicated comparison.



Figure 6.6: Measurement of lipid peroxiation in THP-1 macrophages and HMDM in the presence of SMase-LDL and ferritin.

THP1-cells (A) or HMDM cells (B) were treated with or without ferritin for 24 h and then treated with or without SMase-LDL (200 µg protein/ml) 24 h. THP1-cells were then incubated with the lysosomal lipid perorixidation probe, Foam-LPO (2 µM) for 15 min. The lysosomal lipid peroxidation was determined with a two-way flow cytometry and quantified by the ratio of intensity of fluorescence of green channel to that of red channel (FL1/FL2) using Flow Jo software. The ratio of intensity was compared for the treated cells and control cells using one-way ANOVA followed by a Tukey's post hoc test (n=3). * indicates P < 0.05 and *** indicates P<0.001 compared to control cells. $\neq\neq$ indicates P < 0.001 for the shown comparison.

6.3.3 Effects of ferritin on intracellular lipids and ceroid

Ceroid is an advanced product of oxidation of lipids which consists of complex polymerised insoluble lipids and proteins and has been demonstrated to be present both intracellularly and extracellularly in atherosclerotic lesions (Mitchinson *et al.*, 1985, Haka *et al.*, 2011). Both intracellular lipids and ceroid can be detected using Oil Red O, however, detection of ceroid requires removal of other lipids using organic solvents using ethanol and xylene as previously described by Ball *et al.* (Ball *et al.*, 1988). Wen and Leake (2007) have previously demonstrated ceroid formation in lysosomes of HMDM and J774 cells treated with aggregated LDL (Wen and Leake, 2007). Having established that ferritin might upregulate lipid peroxidation in lysosomes of human THP1 macrophages and HMDM, it was of interest to examine the effects of ferritin on intracellular lipids and the formation of the advanced product of oxidation, ceroid.

THP-1 cells (45,000) were plated on sterilised coverslips in two 6-well tissue cultured plates. In each plate, three wells were pre-treated with 100 μ g protein/ml (0.2 μ M) ferritin for 24 h. Cells were washed with PBS and incubated for 24 h with RPMI only (2 ml), native LDL (200 μ g protein/ml) or SMase-LDL (200 μ g protein/ml). After the 24 h treatment, the cells were then washed with PBS and the medium was replaced with medium containing 10% (v/v) lipoprotein deficient serum. This was replaced every two days for each plate for seven days and the cells in one plate were stained for intracellular lipids and the other for ceroid. The intracellular lipids and ceroid were measured by calculating the mean integrated density of at least 50 cells expressed as the percentage decrease compared to the cells treated with SMase-LDL and ferritin.

The levels of intracellular lipids with native LDL or SMase-LDL were significantly increased in the presence of ferritin (Fig. 6.7 & 6.8). The intracellular lipid significantly increased with

native LDL to 80 ± 5 % (of that with SMase-LDL and ferritin) in the presence of ferritin compared $63 \pm 3\%$ with native LDL only. The intracellular lipid levels with SMase-LDL addition was also increased in the presence of ferritin compared to when cells were treated with SMase-LDL only (P < 0.001). Similarly, the levels of ceroid in the cells were significantly higher in the presence of ferritin (Fig. 6.9 & 6.10). SMase-LDL treatment alone showed significantly lower amount of ceroid compared to SMase-LDL with ferritin (P < 0.001). The ceroid levels in cells treated with SMase-LDL was significantly greater compared to native LDL (P< 0.05). The percentage of ceroid in cells treated with or without native LDL was also increased with the presence of ferritin (P < 0.05). The increase in the ceroid compared to the controls (no LDL) was much greater than the increase in the intracellular lipids compared to controls (no LDL). Overall these results show that ferritin can be a source of iron in lysosomal LDL oxidation in macrophages and might contribute to the significant increase in the advance lipid oxidation product ceroid found in atherosclerotic lesions. Experiments were normalised with cell numbers.



Figure 6.7: Detection of intracellular lipids in THP-1 macrophages

THP-1 cells (4.5 X 10^4) cultured on sterilised coverslips were incubated with or without 100 µg protein/ml (0.2 µM) ferritin for 24 h. Cells were washed with PBS and treated with RPMI only (2 ml) or with native LDL (200 µg protein/ml) or with or without SMase-LDL (200 µg protein/ml) and the treatment was left for another 24 h. After seven days incubation in RPMI medium containing LPDS, the treated cells were stained for intracellular lipids using Oil Red O and images was taken with light microscopy. Scale bar is 100 µm. This is a representative of three independent experiment.



Figure 6.8: Measurement of percentage intracellular lipids in THP-1 macrophages

THP-1 cells (4.5 X 10⁴) cultured on sterilised coverslips were incubated with or without 100 μ g protein/ml (0.2 μ M) ferritin for 24 h. Cells were washed with PBS and treated with RPMI only (2 ml) or with native LDL (200 μ g protein/ml) or with SMase-LDL (200 μ g protein/ml) and the treatment was left for another 24 h. After seven days incubation in RPMI medium containing LPDS, the treated cells were stained for intracellular lipids using Oil Red O. The levels of intracellular lipids was quantified with ImageJ by measuring the mean intensity of at least 50 cells for each treatment and presented as percentage intracellular lipids in relation to cells treated with ferritin and SMase-LDL. Treated cell were compared by one way ANOVA (n=3 independent experiments). ** indicates P < 0.01) and *** indicates P<0.001 compared to SMase-LDL plus ferritin. \neq indicates P < 0.05 for the indicated comparison.



Figure 6.9: Detection of ceroid in lysosomes of THP-1 macrophages

THP-1 cells (4.5 X 10^4) cultured on sterilised coverslips were incubated with or without 100 µg protein /ml (0.2 µM) ferritin for 24 h. Cells were washed with PBS and treated with RPMI only (2 ml) or with native LDL (200 µg protein/ml) or with SMase-LDL (200 µg protein/ml) and the treatment was left for another 24 h. After seven days incubation in RPMI medium containing LPDS, soluble lipids were removed with organic solvents (ethanol and xylene) and the treated cells were stained for ceroid using Oil Red O and images were taken by light microscopy. Scale bar is 100 µm. This is representative of three independent experiments.



Figure 6.10: Measurement of percentage ceroid in THP-1 macrophages

THP-1 cells (4.5 X 10⁴) cultured on sterilised coverslips were pre-treated with or without 100 µg protein /ml (0.2 µM) ferritin for 24 h. Cells were washed with PBS and treated with RPMI only (2 ml) or with native LDL (200 µg protein/ml) or with SMase-LDL (200 µg protein/ml) and the treatment was left for another 24 h. After seven days incubation in RPMI medium containing LPDS, soluble lipids were removed with organic solvents and the treated cells were stained for ceroid using Oil Red O. The levels of ceroid was quantified with ImageJ by measuring the mean intensity of at least 50 cells for each treatment and presented as percentage ceroid in relation to cells treated with ferritin and SMase-LDL. Treated cell were compared by one way ANOVA (n=3 independent experiments). * indicates P < 0.05) and *** indicates P<0.001. \neq indicates P < 0.05 and $\neq \pm$ indicates P < 0.01 for the shown comparison.

6.3.4 Effects of ferritin-oxidised LDL on intracellular ROS in macrophages

Intracellular reactive oxygen species generated in macrophages have been suggested to play a key role in atherogenesis by mediating signalling pathways that can contribute to the onset and progression of atherosclerotic lesion (Singh and Jialal, 2006). We measured the intracellular levels of reactive oxygen species in macrophages using the fluorescent probe, DHE which has been frequently used as the probe to measure the levels of intracellular superoxide (Wardman, 2007). THP-1 cells were incubated with pre-warmed RPMI – 1640 media alone (2 ml per well) or containing either NaCl/sodium acetate buffer (10% as a vehicle control for the oxidised LDL) or ferritin (0.2 μ M) or LDL (50 μ g protein/ml) or ferritin-oxidised LDL (50 μ g protein/ml). The cells were also treated with these conditions in the presence or absence of cysteamine (40 μ M). The cells were then washed twice with PBS and mounted with fluorescence mounting media containing DAPI. Images were captured using fluorescent microscopy and intensity of DHE staining was quantified using ImageJ software and compared with one-way ANOVA (n=3) followed by Tukey post hoc test.

The results showed that the intracellular level of reactive oxygen species was increased significantly in cells treated with oxidised LDL (Fig. 6.11), the cells treated with LDL only and ferritin only was not significantly different from the controls. This indicates that ferritin-oxidised LDL can promote the generation of reactive oxygen species in macrophages. The effect of the antioxidant cysteamine on intracellular reactive oxygen species was tested. Cysteamine (40 μ M) significantly reduced intracellular levels of ROS in cells (Fig. 6.12). Cells treated with cysteamine had significantly less amount of reactive oxygen species compared to the controls.



Figure 6.11: Effect of ferritin-oxidised LDL on THP-1 cells intracellular ROS

THP-1 macrophages (4.5 x10⁴) were plated on cover slips in 6 well sterile tissue culture plates. The cells were incubated in RPMI medium (containing 10% (v/v) FCS) (A), with sodium acetate buffer pH4.5 (10% (v/v) as a vehicle control) (B), with native LDL (50 µg protein/ml) (C), ferritin (0.1 µM) (D) or ferritin-oxidised LDL (50 µg protein /ml) for 24 h (E). The cells were then washed with PBS and incubated with 10 µM DHE for 30 min in the dark at 37 °C in a non CO₂ incubator, washed with PBS and mounted using fluorescence mounting medium containing DAPI. The images were captured using an Axioimager epifluorescent microscope using a 20x objective. (F) Shows the increase in fluorescence intensity of DHE after treatments (n=3 independent experiments). *** indicates P<0.001 compared to the control cells. $\neq\neq\neq$ indicates P < 0.001 for the shown comparison

Control Control+cysteamine Buffer pH4.5 Buffer pH 4.5 +cysteamine Ferritin Ferritin + cysteamine OxLDL OxLDL +cysteamine B Control DHE fluorescence intensity (a.u) 5 40 μM Cysteamine LDL ferritin Control Vehicle control OxLDL

A

Figure 6.12: Effect of cysteamine on ROS formation induced by ferritin-oxidised LDL

THP-1 macrophages (4.5 x10⁴) were plated on cover slips in 6 well sterile tissue culture plates. The cells were incubated in RPMI medium (containing 10% (v/v) FCS), with sodium acetate buffer pH 4.5 (10% (v/v) as a vehicle control), with native LDL (50 μ g protein/ml), ferritin (0.1 μ M) or ferritin-oxidised LDL (50 μ g protein/ml) for 24 h. All conditions were incubated with or without cysteamine (40 μ M). The cells were then washed with PBS and incubated with 10 μ M DHE for 30 min in the dark at 37 °C in a non CO₂ incubator, washed with PBS and mounted using fluorescence mounting medium containing DAPI. The images were captured using an Axioimager epifluorescent microscope using a 20x objective. The fluorescence intensity of DHE was compared by one way ANOVA (n=3 independent experiments). ** indicates P < 0.01) and *** indicates P<0.001 compared to the cells without cysteamine.

6.3.5 Effect of ferritin oxidation of LDL on cellular respiration/metabolism in THP1-cells The metabolic energy phenotype was measured using Agilent Seahorse XF technology. Glycolysis has been suggested to be important in fuelling inflammation and the progression of atherosclerosis (Groh *et al.*, 2018). The effects of LDL oxidation by ferritin on the metabolic pathways in macrophages were tested. THP-1 cells (6×10^4) in Seahorse tissue culture wells were incubated with or without ferritin (0.2μ M) for 24 h. They were then incubated for 24 h in RPMI medium alone or with native LDL (100 µg protein/ml). The metabolic and energy phenotype was then determined by measuring the oxygen consumption rate (OCR) and extracellular acidification rate (ECAR). Difference between control and treated cells were determined by two-way ANOVA followed by Bonferroni's post-test.

The oxygen consumption rate in cells treated with LDL and ferritin showed a significant increase (P <0.001), but no significant increase was observed with LDL only (Fig. 6.13). The treatment with LDL and ferritin caused significant increase in the OCR (P < 0.001) in all measurements made after the addition of the stressor mix oligomycin and FCCP compared to control cells and a decreased oxygen uptake over time. The ECAR followed a similar pattern, the extracellular acidification rate significantly increased in the presence of ferritin and LDL. The results suggest that the cells were more energetic and metabolically activated after treatment hence becoming more glycolytic (Fig. 6.14).



Figure 6.13: Time course of effect of ferritin and LDL on metabolism of macrophages

THP-1 macrophages (6×10^4 per well) were cultured in sterile Seahorse tissue culture plates with ferritin (0.2μ M) for 24 h. They were then incubated in RPMI medium alone or native LDL (100 µg protein/ml) for 24 h. The cells were then washed with XF base media containing glucose, glutamine and pyruvate as fuels and analysed to determine the metabolic phenotype measuring the OCR and ECAR. The OCR and ECAR over time were compared for the treated cells and control using two-way ANOVA followed by a Bonferroni's post-test (n=4). *** indicates P < 0.001.



XF Cell Energy Phenotype

Figure 6.14: Effects of LDL and ferritin on metabolism of macrophages

THP-1 macrophages (6×10^4 per well) were grown in 8 well Seahorse sterile tissue culture plates. The control cells were incubated with or without ferritin (0.2μ M) for 24 h. The cells were the incubated in RPMI medium or with native LDL (100μ g protein/ml) for 24 h. The cells were then washed and incubated with XF base media containing glucose, glutamine and pyruvate as fuels. The cells were then analysed to determine their metabolic phenotype. The cells were stressed with oligomycin, FCCP, and the energy phenotype before (unfilled squares) and after (filled in squares) the stressor mix were measured. The result was analysed with Seahorse bioscience Wave software. This is a representative of four independent experiments. OCR is oxygen consumption rate and ECAR is extracellular acidification rate.

6.3.6 Effect of ferritin-oxidised LDL on macrophages cell death

The ability of LDL oxidised by ferritin to induce cell death in THP-1 macrophages was tested. THP-1 cells (5 x 10^5 cells per well) were cultured in RPMI-1640 containing 10% FCS (v/v) either alone or in the presence of NaCl/sodium acetate buffer (pH 4.5) (10%) or native LDL (100 µg protein/ml) or ferritin alone (0.2 µM) or ferritin-oxidised LDL (100 µg protein/ml). After 48 hours of incubation the proportion of cells that were live or undergoing apoptosis, necrosis or secondary necrosis were assayed using flow cytometry.

The externalisation of phosphatidylserine, which is a measure of apoptosis, was determined by the binding of FITC-labelled annexin V. The permeability of the plasma membrane to propidium iodide was used as a measure of necrosis. Significantly lower levels of live cells were detected in cells treated with ferritin and ferritin-oxidised LDL. The percentage of cells that were alive, apoptotic, necrotic or secondary necrotic was determined by the measurement of the levels of annexin V and PI they contained (Figures 6.15 & 6.16) using flow cytometer. Higher levels of apoptotic cells were observed in cells treated with ferritin and oxidised LDL. The levels of apoptotic cells present in ferritin and ferritin-oxidised LDL treated cells were significantly higher compared to control cells and cells treated with LDL (P <0.001). However, the level of necrotic cells observed was very low in all treatments and there was no statistically significant difference compared to control cells. The levels of secondary necrotic cells (apoptosis followed by permeabilisation of the plasma membrane) was significantly higher in cells treated with ferritin-oxidised LDL compared to control (P < 0.01) and cells treated with LDL alone (P < 0.01). A proportion of the toxic effects observed was due to the presence of ferritin itself. However, the toxic effect was enhanced in the presence of ferritin-oxidised LDL (which would have contained ferritin).



Figure 6.15: Flow cytometry analysis of ThP1 macrophages cell death

Ferritin-oxidised LDL prepared by incubating (1 mg LDL protein/ml) with 2 μ M ferritin at 37°C for 24 h in NaCl/sodium acetate buffer (pH 4.5). THP-1 cells (5 x 10⁵ cells per well) were then cultured in RPMI-1640 containing 10% FCS (v/v) either alone or in the presence of NaCl/sodium acetate buffer (pH 4.5) (10%) or native LDL (100 μ g protein/ml) or ferritin (0.2 μ M) alone or ferritin-oxidised LDL (100 μ g protein/ml). After incubation for 48 h the cells were harvested and assayed by flow cytometry. The results were analysed using the BD Biosciences C6 flow cytometer software. The abscissa shows annexin V binding and the ordinate shows the PI content. The data shown is a representative of three independent experiments.



Figure 6.16: Effects of ferritin-oxidised LDL on apoptosis in THP-1 macrophages

Ferritin-oxidised LDL prepared by incubating (1 mg LDL protein/ml) with 2 μ M ferritin at 37°C for 24 h in NaCl/sodium acetate buffer (pH 4.5). THP-1 ells (5 x 10⁵ cells per well) were then cultured in RPMI-1640 containing 10% FCS (v/v) either alone or in the presence of NaCl/sodium acetate buffer (pH 4.5) (10%) or native LDL (100 μ g protein/ml) or ferritin (0.2 μ M) alone or ferritin-oxidised LDL (100 μ g protein/ml). After incubation for 48 h the cells were harvested and assayed by flow cytometry. The apoptotc cells were measured by externalisation of phosphatidyl serine determined by the FITC-labelled annexin V binding and necrosis was measured by membrane permeability to propidium iodide (PI). The percentage of live cells (A), necrotic cells (B), apoptotic cells (C) and secondary necrotic cells (D) was compared by a one-way ANOVA followed by a Tukey's post hoc test (n=3). * indicates P < 0.05, ** indicates P < 0.01 and *** indicates P < 0.001. $\neq \neq$ indicates P < 0.001 for the shown comparison.

6.4 Discussion

Macrophages are well recognised to play an important role in all stages of atherosclerosis, as they constitute major cellular components of early and advanced atherosclerotic lesions (Glass and Witztum, 2001). The sustained recruitment of macrophages and continuous proinflammatory mediators in the lesion supports the growth of the lesion and instability of the plaque. The balance between macrophages maintaining their function, survival, apoptosis and clerance from the lesion environment is a major determining factor of the progression and the fate of atherosclerotic lesions (Tano *et al.*, 2012). This thesis reported in previous chapters of that ferritin is a possible candidate for lysosomal LDL oxidation at lysosomal pH. In this present chapter, we investigated the role of ferritin in promoting the oxidation of LDL in the lysosomes of macrophages and the possible effects of this oxidation on macrophage function and atherosclerosis.

Studies investigating the role of body iron stores in CVD have remained inconclusive (Moore *et al.*, 1995, Meyers *et al.*, 2002, Ganz, 2005, Peffer *et al.*, 2013). This inconclusive evidence might because the association is not driven by plasma iron levels but driven by the distribution of iron in macrophages, hepatocytes and enterocytes, as determined by serum hepcidin levels (Galesloot *et al.*, 2015). An attempt was made to upregulate intracellular iron levels and ferritin levels through the addition of hepcidin up to 1000 nM. The result showed that hepcidin did not significantly increase the iron levels in THP-1 macrophages. The reason for this observation is unclear, as hepcidin has been previously demonstrated to decrease iron concentration in blood but increase intracellular levels of iron in macrophages by binding to and downregulating the iron transporter ferroportin 1 (FPN1). FPN1 is known as the sole iron transporter responsible for the outflow of iron from cells. The decrease in the quantity of FPN1 in hepatic and splenic

macrophages by hepcidin decreases the ability of macrophage iron efflux (Zhao *et al.*, 2013). Contrary to our result, a previous study by Chung *et al.* showed that exposure of THP-1 macrophages to 300 and 1000 nM hepcidin decreased FPN1 expression and caused a decrease in duodenal iron and serum iron. However, they did not measure the intracellular iron levels in THP-1 macrophages. They reported that the efflux of iron from Caco-2 cells was significantly reduced by addition of hepcidin, although in addition to hepcidin the Caco-2 cells received 10 μ M iron in form of FeCl₃ and the THP-1 macrophages used was transformed with twice the amount of PMA used in our experiment (Chung *et al.*, 2009). RPMI 1640 (which does not contain added iron) was used, but also F10 and DMEM media (which contain iron) could not increase the iron levels inside the macrophages. Work by Agoro and Mura also suggested that macrophage iron levels (Agoro and Mura, 2016). Hence in the future experiment, the introduction of an external source of iron and tranformation of THP-1 macrophages to the M1 phenotype would be considered.

Macrophages treated with SMase-LDL have recently been demonstrated to exhibit increased lysosomal lipid peroxidation (Ahmad and Leake, 2019). Lipid peroxidation in LDL increases the atherogenicity of LDL (Steinbrecher, 1991), hence the role of ferritin in lysosomal lipid peroxidation were determined. The cells were pre-incubated with ferritin to allow it to be be endocytosed and delivered to the lysosomes and then followed up with treatment with either native LDL or SMase-LDL. Cells that were pretreated with ferritin had a significant increase in levels of lysosomal lipid peroxidation, as shown by the lysosomally targetted lipid peroxidation probe Foam-LPO. The increase in lysosomal lipid peroxdation suggests that ferritin has the potential to promote the oxidation of LDL inside the inflammatory cells, macrophages. Our laboratory has shown that SMase-LDL is rapidly endocytosed by

macrophages and it was mainly deposted in the lysosomes where it was oxidised by iron (Wen *et al.*, 2015, Wen *et al.*, submitted). Aggregated forms of LDL are notable features of atherosclerotic lesions and they are powerful inducers of foam cell formation in atherosclerotic lesions (Hoff and Morton, 1985, Guyton and Klemp, 1996, Tabas, 1999). LDL aggregated by sphingomyelinase is one of the most feasible mechanisms for LDL aggregation *in vivo* (Pentikiiinen *et al.*, 1996). The addition of SMase-LDL and ferritin also promoted increased lipid peroxidation in the same pattern in both THP-1 macrophages and HMDM. The Foam-LPO itself consists of a conjugated diene group which can undergo oxidation in lysosomes and can cause spectral shifting (Zhang *et al.*, 2015). The increased green to red ratio also signifies more potential for lipid peroxidation in the presence of ferritin with or without SMase-LDL. The competition between the probe and Smase-LDL for oxidation by ferritin might be responsible for the slightly lower peroxidation when SMase-LDL was present. These experiments support the possibility that ferritin might catalyse lipid peroxidation in lysosomes.

Our laboratory has previously shown that cultured THP-1 (Ahmad and Leake, 2019) J774 cells and HMDM (Wen and Leake, 2007) treated with aggregated LDL formed increased ceroid in their lysosomes and this can be attributed to iron-catalysed oxidation. In this present study, our results showed that ceroid was increased considerably in the presence of ferritin and and was increased further when SMase-LDL was added, much more than the intracellular lipids were increased. Potentially importantly, this suggests that ferritin may be a key factor in releasing catalytically active iron that can mediate LDL oxidation in lysosomes of macrophages.

Studies have shown that oxLDL stimulates the intracellular production of ROS in macrophages, VSMCs and endothelial cells (Hsieh *et al.*, 2001, van Aalst *et al.*, 2004,

Zmijewski *et al.*, 2005). The superoxide radical is one of the major reactive oxygen species, and work presented in chapters three and five of this thesis suggested that the superoxide radical and its protonated form, hydroperoxyl radicals, might be involved in lysososomal LDL oxidation by ferritin. Here it was shown that ferritin-oxidised LDL can induce increased formation of reactive oxygen species in THP-1 macrophages. Increased levels of reactive oxygen species can be prevented by antioxidants that can scavenge free radicals. Cysteamine demonstrated strong ability to reduce the formation of conjugated dienes in LDL oxidised with ferritin. Cysteamine significantly reduced the formation of reactive oxygen species in THP-1 cells treated with ferritin-oxidised LDL. In this context, it is interesting that Vendrov *et al* showed that limiting the production of superoxide in macrophages/monocytes or other cells of the vessel wall caused decreased atherogenesis (Vendrov *et al.*, 2007).

Oxidised LDL has been demonstrated to induce an inflammatory response in macrophages (Li *et al.*, 2010, Lara-Guzmán *et al.*, 2018). Induction of a pro-inflammatory reasponse in macrophages is characterised by a shift in metabolic pathways to glycolysis leading to decreased uptake of oxygen (Fukuzumi *et al.*, 1996, Rodriguez-Prados *et al.*, 2010) and reduction in oxidative phosphorylation (Haschemi *et al.*, 2012, Freemerman *et al.*, 2014). Metabolic pathways have a central role in the function of immune cells and many studies have utilised lipolysaccharide (LPS) to understand the effects of inflammation on metabolic pathways in macrophages. Shirai *et al.* explored the effects of stimulating monocytes on their metabolic pathways. LPS and IFN γ -stimulated monocytes from healthy individuals had a lower oxygen consumption rate (OCR) and extracellular acidificaion rate (ECAR) compared to monocytes isolated from atherosclerotic patients which demonstrated increased glycolytic flux, (Shirai *et al.*, 2016). Treatment of macrophages with LPS and IFN γ is well known to produce the pro- inflammatory macrophages (M1 macrophages) and treatment of bone marrow-derived

macrophages with LPS and IFN γ significantly upregulated glycolytic pathways significantly (Liu *et al.*, 2016). This present study showed that treatment of THP-1 macrophages with ferittin, especially in combination with LDL, which can promote lysosomal oxidation of LDL, significantly increased their oxygen uptake (P <0.001) and also the ECAR was signifuicantly increased (P <0.001) Overall these results showed that there was increased glycolysis and aerobic capacity in the macrophages and the macrophages became more metabolically activated. This might enhance the macrophages' pro-inflammatory response. In addition, it might acidify the extracellular space and further promote atherosclerosis (Leake, 1997).

Cell death caused by oxidised lipoprotein is believed to be important in lesion progression. Different oxidised lipid products of LDL have been demonstrated to be cytotoxic to cells, lipid hydroperoxides (Siow et al., 1999a) as well as oxysterols (Colles et al., 2001a), have been demonstrated to be cytotoxic to cells. Pevious work by Gerry and Leake (Gerry and Leake, 2008) has demonstrated that oxysterol-rich LDL was more cytotoxic to macrophages than LDL with high levels of lipid hydroperoxides. The most potent cytotoxic substance out of these two products of LDL oxidation remains controversial as previous work by Siow et al showed that lipid hydroperoxide-rich LDL was more toxic than hydroperoxide-rich LDL to smooth muscle cells (Siow et al., 1999a). Other products of LDL oxidation, such as aldehydes, have also been shown to be cytotoxic (Esterbauer et al., 1991c). Oxidised phospholipids have also been shown to be cytotoxic to macrophages (Stemmer et al., 2012) and fibroblasts (Colles and Chisolm, 2000). Work presented in this chapter has demostrated that ferritin-oxidised LDL can induce apoptosis, but not primary necrosis, in cultered THP-1 macrophages. The toxicity of ferritinoxidised LDL was not unexpected. Ferritin oxidation of LDL promoted the formation of 7ketocholesterol, CLOOH and total hydroperoxides and all these oxidised lipid products have been demostrated to be cytotoxic. Ferritin-oxidised LDL also increased the level of ROS in

cells and this might possibly mediate induction of apoptosis in cells. The role of ROS in prommoting apoptosis was previously reviewed by Simon *et al.* (Simon *et al.*, 2000).

Data presented in this chapter demonstrated that ferritin can also mediate some of the effects observed with ferritin-oxidised LDL, such as apoptosis and lipid peroxidation. It was speculated that the iron-rich protein, ferritin, can be endocytosed by macrophages and delivered to the lysosomes where it can release its iron to catalyse redox reactions and mediate lysosomal lipid peroxidation. The continuous oxidation process in the lysosomes mediated by iron released from ferritin might lead to destabilisation of these organnelles and cell death. Recent work from our laboratory has demonstrated that lysosomal LDL oxidation mediated by iron can lead to increased lysosomal pH (Ahmad and Leake, 2019), which might cause loss of lysosomal function (Ohkuma and Poole, 1978). There is also evidence that lysosomal destabilisation and loss of function mediated by iron-catalysed redox reactions less can cause enzymes to leak and trigger apoptotic cell death (Guicciardi *et al.*, 2004, Aits and Jäättelä, 2013).

Work presented in this chapter lends support to the ability of ferritin to oxidise LDL within the lysosomes of macrophages and possible promotion of atherogenesis. Further experiments are required to further explore the upregulation of iron and ferritin by hepcidin within macrophages in relation to lysosomal LDL oxidation. However, there is evidence that ferritin promotes lysosomal lipid peroxidation, causes increased formation of the advanced lipid oxidation products ceroid and activates macrophages making them more metabolically active and glycolytic. Ferritin-oxidised LDL was also shown to increase the intracellular generation of reactive oxygen species in macropohages and promote apototic cell death. Overall, all these events have pro-atherogenic consequences.

Chapter 7: General Discussion
7.0 General Discussion

Modification of LDL is important in the onset and advancement of atherosclerotic lesions (Jialal and Devaraj, 1996). Oxidised LDL exerts effects on many in vitro activities of vascular cells, which includes proliferation, inflammatory response, migration and cell death. The majority of these effects are consistent with its enhanced pro-atherogenic nature (Berliner and Heinecke, 1996, Steinberg, 1997b, Mehta and Li, 2005). Several mechanisms have been postulated for the oxidation of LDL in vivo (Yoshida and Kisugi, 2010, Maiolino et al., 2013). The precise mechanisms and site of modification of LDL in vivo remain debatable. Jerome suggested that accumulation of lipids in the lysosomes is an important component in atherosclerosis (Jerome, 2006). Some studies have suggested a link between interference of autophagy and increased atherosclerosis (Ouimet et al., 2011, Razani et al., 2012) and the enhancement of autophagic-lysosomal biogenesis has been suggested as a plausible mechanism for reducing atherosclerotic cardiovascular diseases (Sergin et al., 2017). The lysosomes are indispensable intracellular organelles responsible for the degradation of both intracellular and extracellular cargo including lipoproteins (Sergin et al., 2015). The lysosomal compartment is suggested to be rich in redox active iron since many macromolecules derived from heterophagy or autophagy might be degraded to release iron into lysosomes, hence it is suggested to have the highest amount of redox active iron (Petrat et al., 2001, Kurz et al., 2007, Lv and Shang, 2018). These events make lysosomes a possible site for LDL oxidation.

As mentioned earlier in this thesis, our laboratory (Wen and Leake, 2007) was first to identify lysosomes as a site for LDL oxidation and proposed iron as the possible mediator of the oxidation LDL in lysosomes. Studies have suggested the presence of transition metals (Lamb *et al.*, 1995) and iron in particular in atherosclerotic lesions (Yuan *et al.*, 1996, Lee *et al.*, 1998, Stadler *et al.*, 2004). Since most of the transition metals *in vivo* are tightly bound to prosthetic

groups or sequestered by proteins, the question of how the free copper or iron ions can be available in significant enough amounts to mediate LDL oxidation is a debated topic (Halliwell and Gutteridge, 1990, Mukhopadhyay and Fox, 1998). Our laboratory (Satchell and Leake, 2012, Ahmad and Leake, 2018) later explored the mechanisms by which Fe^{2+} and Fe^{3+} can mediate lysosomal LDL oxidation with respect to the possible treatment of the disease with antioxidants. Previous work from Balla *et al.* and others (Balla *et al.*, 1991, Abdalla *et al.*, 1993, Lamb and Leake, 1994c, Lamb and Leake, 1994b, Mukhopadhyay *et al.*, 1996, Rodriguez-Malaver *et al.*, 1997, Grinshtein *et al.*, 2003) have identified some metal-ion binding proteins that can possibly contribute to oxidation of LDL. However, the exact role of iron binding proteins in lysosomal LDL oxidation is yet to be explored in great detail and if there is a pathophysiological role for the main iron storage protein ferritin in lysosomal LDL oxidation *in vivo*.

Work presented in this thesis has demonstrated that ferritin is a candidate for oxidation of LDL. It was shown for the first time the novel contributions ferritin might have as a major ironstorage protein to the oxidation of LDL at lysosomal pH and within lysosomes of macrophages. The present study utilised different independent methods to assess the ability of ferritin to promote formation of products of oxidation in human LDL. The method for continuously measuring conjugated dienes formation in LDL established by Esterbauer and colleagues three decades ago (Esterbauer *et al.*, 1989b), is still widely accepted and applied in the research community to access the kinetics of LDL oxidation. The experiments were conducted with LDL were incubated with 0.1 μ M ferritin in terms of total protein content (which was measured to contain iron centration of 100 μ M). The rate of oxidation as measured by the formation of conjugated dienes over time was different from the FeSO₄-induced oxidation, in the sense that there were less distinctive rapid and slow phases of oxidation. There was extensive oxidation of LDL by ferritin at pH 4.5, but very little or no oxidation was observed at normal interstitial fluid or plasma pH (pH 7.4). It was expected that ferritin oxidation may progress more in the lysosomes rather than interstitial fluid or plasma because of substantial antioxidant protection that is present in arterial intima (Leake and Rankin, 1990, Dabbagh and Frei, 1995). Our data showed that ferritin spontaneously released iron at lysosomal pH without requiring the activity of lysosomal protease, in contrast to what was earlier suggested by Kidane and colleagues (Kidane *et al.*, 2006). As much as 15 μ M iron was demonstrated to be released form ferritin over 24 h which is similar to the 16 μ M concentration that Petrat *et al* found in lysosomes of endothelial cells from rat Liver (Petrat *et al.*, 2001). The iron chelators EDTA and DTPA slowed down the oxidation of LDL by ferritin at lysosomal pH significantly, thus the ferritin-mediated LDL oxidation is mainly due to the iron released from ferritin at pH 4.5.

The data obtained from HPLC and tri-iodide assays lend support to the ability of ferritin to oxidise LDL at lysosomal pH. It revealed that ferritin promotes formation of oxidised lipids in LDL (7-ketocholesterol and cholesteryl linoleate hydroperoxide). Elevated levels of 7-ketocholesterol were observed in patients with CAD was compared to patients with normal arteries. The increase in 7-ketocholesterol correlated with the events of myocardial infarction and increased C-reactive proteins providing a link between 7-ketocholesterol levels and inflammation (Hitsumoto *et al.*, 2009) and increased atherosclerotic cardiovascular events (Hitsumoto *et al.*, 2009, Song *et al.*, 2017). Walter *et al.*, also showed that elevated lipid hydroperoxides were a predictive factor for cardiovascular events and this prediction was independent of inflammatory markers and other risk factors of CVD (Walter *et al.*, 2008). Hence oxidised products from LDL oxidation by ferritin at lysosomal could potentially promote atherosclerotic cardiovascular events.

Lysosomes with a pH between 4.5 and 5.0 (Mindell, 2012) consist of over sixty types of hydrolytic enzymes (Xu and Ren, 2015) and lysosomal enzymes such as cathepsins might possibly influence the progression of CVD by contributing to inflammation and apoptosis (Lutgens *et al.*, 2007). The effects certain lysosomal proteases might have on lysosomal LDL oxidation by ferritin were demonstrated. The results obtained showed that oxidation of LDL by ferritin occurred in the presence of the proteolytic enzymes cathepsins B and D. However, there was no significant difference between oxidation mediated by ferritin pre-treated with cathepsins. This observation might be attributed to the lack of degradation products observed in ferritin after treatments with cathepsins B and D when analysed with SDS-PAGE. The ApoB-100 component of LDL was evidently susceptible to degradation by cathepsin D, showing that the cathepsin D was active. Ferritin is highly resistant to proteolysis by the major endoproteases cathepsins D and B and thus might survive in lysosomes for long periods of time. Treatment with cathepsins B and D increased LDL oxidation by ferritin, which suggests that the lysosomal oxidation of LDL by ferritin might accelerate during the lysosomal degradation of LDL.

We described the impact of treating LDL with a lipolytic enzyme on its rate of oxidation. Coincubation of LDL with cholesteryl esterase from *Pseudomonas sp* increased the rate of the initial phase of oxidation. Cholesteryl esterase might degrade the core cholesteryl esters into fatty acids and cholesterol allowing free radicals generated from ferritin-iron more access to PUFA such as Linoleic acid and arachidonic acid and hence increased conjugated diene formation. Lysosomal enzymes released by human macrophages modified LDL in vitro (Hakala *et al.*, 2003). The modification of LDL by processes such as proteolysis, lipolysis and oxidation carried out *in vitro* has helped in producing LDL particles similar to those found in atherosclerotic lesions and thus contributed to more understanding of the disease. The results presented here demonstrated that ferritin might continually oxidise LDL even after modification by enzymes in lysosomes.

Due to the novel suggested lysosomal LDL oxidation mechanisms involving redox active iron and ferritin-iron, there is a need to reconsider the choice of appropriate antioxidants in the treatment of atherosclerosis. Antioxidants that can be targeted to the lysosomes and able to prevent the oxidation in the core of LDL appear to be more appropriate. The failure of human antioxidant clinical trials that mainly used α -tocopherol (Yusuf *et al.*, 2000, de Gaetano, 2001, Collins et al., 2002, Cook et al., 2007, Sesso et al., 2008) did not mean that modified LDL lacks pro-atherogenic properties. It rather suggests the need for a better understanding of the mechanisms involved and a better approach with selection of appropriate antioxidants to reduce atherogenic events caused by oxidised LDL. Work presented in this thesis and previously (Ahmad and Leake, 2018) suggested that the free radicals superoxide and its protonated form, HO₂⁻ are involved in ferritin and FeSO₄ oxidation of LDL. Hence the antioxidant of choice must also possess strong ability to scavenge superoxide and the highly reactive HO_2 . The question of why the antioxidants used have failed in large clinical trials of CVD persists. It should be noted that we showed the inability of enrichment of LDL with α -tocopherol to prevent LDL oxidation by ferritin at lysosomal pH. This supported previous findings from our laboratory have shown that α -tocopherol has both pro-oxidant and antioxidant effects on LDL oxidation by FeSO₄ (Satchell and Leake, 2012, Alboaklah, 2018b). This might help to explain why the large antioxidant trials using α -tocopherol did not succeed. It was also shown that vitamin C did not effectively inhibit the oxidation of LDL by ferritin at lysosomal pH. The data showed that the antioxidant properties of ascorbate were lost over time in the presence of lipid hydroperoxides. Ascorbate was not a strong inhibitor of oxidation at pH 4.5 compared to pH 7.4, where it completely inhibited oxidation by CuSO₄. We found that the reduced form of vitamin C, dehydroascorbate was a pro-oxidant and increased oxidation of LDL by ferritin. Hence these vitamins might not be the most efficient choice of antioxidants in preventing lysosomal LDL oxidation by ferritin.

Tempol has superoxide dismutase mimetic properties (Luo *et al.*, 2009) and reduced the later oxidation phase of LDL, but notably had no effect on the early phase of LDL oxidation. HO_2^- is a strong oxidant which can initiate lipid peroxidation even in the esterified cholesterol core of LDL, due to its ability to pass through the phospholipid monolayer of LDL. The amphipathic nature of tempol might prevent it from entering the hydrophobic core of LDL. This might allow the initial oxidation of the esterified cholesterol rich core of LDL by hydroperoxyl radicals to proceed uninhibited. Tempol, being amphiphilic, might enter the phospholipid monolayer of LDL and inhibit the later oxidation of LDL here. Like ascorbate, tempol was able to completely inhibit LDL oxidation mediated by copper at pH 7.4 but loses this ability at lysosomal pH (Alboaklah, 2018a). Hence antioxidants that are able to maintain their activities at lysosomal pH might be important in reducing the development of atherosclerosis.

DPPD and cysteamine showed more potential in preventing LDL oxidation. However, the use of DPPD in humans is prevented due to its mutagenic nature (Sofuni *et al.*, 1990). But drugs with analogous antioxidant properties to DPPD and not mutagenic can be synthesised, which might be a plausible treatment for atherosclerosis. Antioxidants that can concentrate in reasonable concentrations in lysosomes would, however, be more beneficial. Cysteamine is drug well tolerated by humans and it is currently in use for treatment of cystinosis (Gahl *et al.*, 2002). Although cysteamine had complex effects on LDL oxidation by ferritin at lower concentrations, the concentrations equivalent to what would be present in lysosomes were highly effective in inhibiting LDL oxidation by ferritin at lysosomal pH. Cysteamine has the

ability to scavenge superoxide radical (Sunman *et al.*, 1993). Cysteamine has now been demonstrated as a possible treatment for atherosclerosis. The study revealed the inhibitory effects of cysteamine on LDL oxidation in the lysosomal environment. Cysteamine also reduced atherosclerotic lesions in mice lacking LDL receptors fed a high fat diet by 33% (Wen *et al.*, submitted). Recent work from our laboratory showed that cysteamine decreased ceroid formation in macrophages incubated with sphingomyelinase-aggregated LDL (Ahmad and Leake, 2018).

It is noteworthy that lysosomes are of significant importance in biomedicine as lysosomal enzymes and lysosomal alterations are linked to a number of human diseases (Lübke *et al.*, 2009). The overexpression of lysosomal proteins and lysosomal acid lipase was demonstrated to promote atherosclerosis (Zschenker *et al.*, 2006). The animal models and human atherosclerotic lesions contain many macrophage foam cells with lipid-engorged lysosomes (Jerome and Yancey, 2003, Jerome *et al.*, 2008). It was important to answer the question of the role lysosomal ferritin-mediated LDL oxidation might play in the lysosomes of macrophages and the progression of atherosclerosis. We attempted to investigate how existing inflammation can increase the iron and ferritin levels in macrophages can influence lysosomal LDL oxidation. Nevertheless, this investigation was unsuccessful, as surprisingly we were unable to upregulate intracellular iron with hepcidin treatment.

Atherosclerosis is a direct example of a disease amplified by a lipid peroxidation process and the transformation of macrophages to lipid-laden cells is well acknowledged as the starting phase of atherosclerosis (Esterbauer *et al.*, 1993). Elevated lipid peroxides were found in atherosclerotic tissue compared to the normal human aorta (Piotrowski *et al.*, 1990). It was established that the presence of ferritin increased intracellular lipid levels, lysosomal lipid peroxidation and percentage ceroid formation in lysosomes of macrophages. These three events have major implications for the onset and progression of atherosclerotic lesions. Our laboratory has previously demonstrated that J774 mouse macrophages and HMDM treated with additional iron or ferritin generated more 7-ketocholesterol (Wen and Leake, 2007). Sphingomyelinase aggregation of LDL produces ceramide, which has been suggested to enhance uptake through endocytosis by causing invagination of the cellular membrane (Holopainen *et al.*, 2000). SMase-LDL is increasingly taken up rapidly by macrophages (Wen *et al.*, 2015). Wen and Leake treated macrophages with acetylated LDL and ferritin. It is interesting, to see ferritin also promotes lipid oxidation and ceroid in human THP-1 macrophages treated with the more pathophysiological SMase-LDL and native LDL.

White *et al* earlier suggested that development of atherosclerosis involves peroxynitrite formation from nitric oxide and superoxide (White *et al.*, 1994). Superoxide or its protonated form the hydroperoxyl radical is proposed to oxidise LDL in lysosomes (Ahmad and Leake, 2019). Measuring intracellular levels of superoxide was therefore relevant to atherosclerosis. We used the fluorescent probe DHE, as it has high affinity for superoxide (Wardman, 2007) (but also has a non-specific affinity for peroxynitrite and hydroxyl radicals) (Gomes *et al.*, 2005, Kalyanaraman *et al.*, 2012). We reported that LDL oxidised by ferritin at lysosomal pH enhances the formation of reactive oxygen species in THP1- macrophages. The potency of cysteamine in protecting the cells from increased reactive oxygen species was demonstrated, as there was significant reduction of ROS in the presence of cysteamine.

It was also reported here that ferritin and ferritin-oxidised LDL produced *in vitro* induced apoptosis and secondary necrosis in cultured macrophages. These findings imply that, if cells have extensively oxidised LDL in lysosomes and the oxidised LDL is released, possibly due to

lysosomal destabilisation or cell death, the released oxidised LDL might increase oxidative stress or apoptosis in those cells or neighbouring cells. There is evidence that intralysosomal redox reactions can cause leakage from lysosomes and trigger apoptosis and necrosis of cells (Brunk *et al.*, 2001, Guicciardi *et al.*, 2004). Hence, it would be appropriate to suggest that ferritin-mediated oxidation of LDL is a possible redox-iron driven reaction that can occur in lysosomes and trigger apoptosis.

The role of apoptosis in atherosclerosis depends on the disease stage. The occurrence of apoptosis in early lesions is likely to be beneficial, as phagocytic events may still occur efficiently to clear foam cells before they induce an inflammatory response. However in advanced lesions, the clearing of apoptotic cells is less efficient and they might contribute to inflammation and necrotic core formation (Tabas, 2005, Schrijvers *et al.*, 2005). Cells might release their content, such as prothrombotic molecules and matrix metalloproteinases, which can cause thrombogenic events and degradation of ECM (Mallat *et al.*, 1999, Mallat and Tedgui, 2000, Kenagy *et al.*, 2011). Consequently, they can promote plaque vulnerability and make them more prone to rupture (Tabas, 2005). Thus, apoptosis induced by ferritin-oxidised LDL might be detrimental to atherosclerosis. It was suspected that the lysosomal oxidation of LDL partly by ferritin might contribute to atherosclerosis and in advanced lesions to apoptosis would contribute to plaque instability.

Previous studies have shown that activated macrophages show enhanced glycolysis along with decreased oxygen uptake. Monocytes stimulated with LPS switched towards a more glycolytic pathway which was suggested to be due to an increase in hypoxia inducible factor (HIF- 1 α) enhanced by NF- κ B. HIF-1 α causes an increase in uptake of glucose by upregulating the synthesis of GLUT-1 glucose transporter (Tawakol *et al.*, 2015, Groh *et al.*, 2018). The study

reports that the extracellular acidification rate and oxygen consumption rate were significantly increased in cells treated with ferritin and LDL. The treatment of THP1- macrophages with LDL and ferritin appears to be driving cells towards a more energetic state (with greater rather than less oxygen consumption). This metabolic switch is attributed to the pro-inflammatory phenotype (Koelwyn *et al.*, 2018). It would be relevant in the future to observe the effects lipopolysaccharide (LPS) might have on the effects observed. Ahmad and Leake (Ahmad and Leake, 2019) have recently shown that native LDL and SMase-aggregated LDL increased LPS-induced secretion of pro-inflammatory cytokines. Hence LDL oxidation by ferritin in lysosomes of macrophages might also elicit a pro-inflammatory effect and contribute to acidifying the extracellular space. Inflammation (Libby, 2012) can promote atherogenesis and acidification might also do so (Leake, 1997). Figure 7.1 shows a schematic diagram of the possible events involving lysosomal LDL oxidation by ferritin can contribute to atherogenesis.



Figure 7.1: The role of Lysosomal LDL oxidation by ferritin in macrophages and atherosclerosis.

Lysosomal LDL oxidation induced by iron released from ferritin might play an important role in atherosclerosis. LDL or aggregated LDL (such as SMase-LDL) are taken up by macrophages and targetted to lysosomes for degradation. Extracellular and intracellular ferritin is also delivered to the cells through endocytosis and autophagy. The acidic environment of lysosomes can enhance the release of redox active iron, which can oxidise LDL. This process leads to increased intracellular lipids, increased lysosomal lipid peroxidation and lysosomal ceroid accumulation that transforms the macrophages to foam cells. The accumulation of lipids in the lysosomes can cause lysosomal dysfunction and cause the contents to leak, which can then release oxidised LDL into the extracellular space. The lysosomal LDL oxidation can promote release of pro-inflammatory cytokines such as TNF- α , IL-6 and MCP-1 causing more macrophages to be recruited which can take up oxidised LDL. The oxidised LDL can promote ROS generation and increased macrophage cell death. The lysosomal dysfunction can also lead to the inability cells to clear dead cells by phagocytosis and promote the activation of the inflammasome upregulating IL-1 β ; these can contribute to the pro inflammatory response. All these events might promote atherogenesis.

7.1 Critical evaluation of the present study

7.1.1 Summary of main findings

Atherosclerosis is a multifactorial disease, with complex events involved in the pathogenesis. This thesis provides answers to some important questions in relation to the onset, progression and considerations for treatment of atherosclerosis. The main findings of the present research work can be summarised as follows:

- Ferritin effectively promotes LDL oxidation at lysosomal pH as shown by increased conjugated diene formation and higher amount of oxidised lipids in the presence of ferritin.
- The oxidation was much faster at lysosomal compared to interstitial fluid or plasma pH, which was demonstrated to be as a result of more iron release from ferritin at lysosomal pH. The contribution of ferritin-iron to the oxidation of LDL was further demonstrated by the ability of the iron chelators EDTA and DTPA to inhibit this oxidation.
- Oxidation of LDL by ferritin proceeded in the presence of lysosomal cathepsins B and D. The proteolysis and lipolysis of LDL enhanced its susceptibility to oxidation by ferritin. Ferritin showed remarkable resistance to proteolytic degradation by cathepsin B and D, hence it may take longer to be degraded in the lysosome and have a longer time to oxidise LDL.
- LDL oxidation by ferritin was not inhibited by α-tocopherol enrichment of LDL or dehydroascorbate. Ascorbate or had both pro-oxidant and antioxidant effects on LDL oxidation by ferritin.

- Tempol slowed down the later, but not the early phase of oxidation of LDL by ferritin, possibly by scavenging lipid radicals in the phospholipid monolayer of LDL. The lysosomotropic antioxidant, cysteamine was shown to be the most appropriate antioxidant in preventing oxidative modifications of LDL by ferritin at lysosomal pH.
- Incubation of macrophages with ferritin promoted intralysosomal lipid peroxidation in human macrophages. The presence of ferritin increased intracellular lipids and intralysosomal ceroid formation in human THP-1 macrophages.
- Ferritin-oxidised LDL induced increased intracellular formation of reactive oxygen species that might further contribute to the pathogenesis of atherosclerosis and cysteamine significantly reduced the generation of reactive oxygen species in human THP-1 macrophages.
- Macrophages treated with ferritin and LDL became activated and glycolytic compared to untreated THP1-macrophages. Ferritin–oxidised LDL induced apoptosis in human THP1-macrophages.

7.1.2 Limitations of the study

Although it was established that ferritin caused increased lipid peroxidation in THP-1 macrophages and human monocytes derived macrophages in a similar way, it would be of advantage to also conduct all other experiments in HMDM as well. However, this could not be achieved due to time and financial constraints. We could not increase iron and ferritin levels in macrophages to try to increase lysosomal LDL oxidation and did not do *in vivo* work (there is no mouse knockout model for ferritin, as it is embryonically lethal if homozygous).

7.2 Possible future work

Building on previous research from our Laboratory, the findings presented in this thesis offers some new insights into the lysosomal LDL oxidation theory. The role of ferritin in human diseases is currently generating interest within the cardiovascular research community. This presents the opportunity to undertake further work within this topical area, which would add more value to this field of research. Further investigation could include:

- Characterise phospholipid oxidation products in LDL oxidised by ferritin by liquid chromatography-electrospray ionisation mass spectrometry.
- Characterise the oxidised lipids in LDL pre-treated with enzymes (cathepsin D and B) prior to oxidation by ferritin and test the possible effects of LDL oxidation by ferritin on LDL degradation and explore the effects of other lysosomal proteases on oxidation of LDL by ferritin.
- LPS and IFN-γ stimulated macrophages could be treated with hepcidin in the presence or absence of iron. The upregulation of iron could be measured using ICP-MS and increase in ferritin levels determined with western blot for ferritin heavy and light chains. The effect of these on lysosomal LDL oxidation could be monitored by ceroid detection.
- As part of another study, the plasma concentrations of ferritin and iron could be compared to those in macrophages of the LDL receptor knock out mice and also stain for ferritin in the atherosclerotic lesions with or without the novel anti-atherosclerotic drug cysteamine.

The results of these studies might further demonstrate the link between ferritin and CVD and provide evidence for increased ferritin levels as a well-defined biomarker in the incidence of atherosclerosis and development of CVD. References

- ABDALLA, D. S. P., CAMPA, A. & MONTEIRO, H. P. 1993. Low-Density-Lipoprotein Oxidation by Stimulated Neutrophils and Ferritin (Atherosclerosis, Vol 97, Pg 149, 1992). Atherosclerosis, 98, 257-257.
- AGORO, R. & MURA, C. 2016. Inflammation-induced up-regulation of hepcidin and downregulation of ferroportin transcription are dependent on macrophage polarization. *Blood Cells, Molecules, and Diseases,* 61, 16-25.
- AHMAD, F. 2016. *Lysosomal oxidation of Low Density Lipoproteins*. Doctor of philosophy, University of Reading.
- AHMAD, F. & LEAKE, D. S. 2018. Antioxidants inhibit low density lipoprotein oxidation less at lysosomal pH: A possible explanation as to why the clinical trials of antioxidants might have failed. *Chem Phys Lipids*, 213, 13-24.
- AHMAD, F. & LEAKE, D. S. 2019. Lysosomal oxidation of LDL alters lysosomal pH, induces senescence, and increases secretion of pro-inflammatory cytokines in human macrophages. J Lipid Res, 60, 98-110.
- AITS, S. & JÄÄTTELÄ, M. 2013. Lysosomal cell death at a glance. *Journal of Cell Science*, 126, 1905.
- AKISHIMA, Y., AKASAKA, Y., ISHIKAWA, Y., LIJUN, Z., KIGUCHI, H., ITO, K., ITABE, H. & ISHII, T. 2005. Role of macrophage and smooth muscle cell apoptosis in association with oxidised low-density lipoprotein in the atherosclerotic development. *Mod Pathol*, 18, 365-73.
- AL-KHUDAIRY, L., FLOWERS, N., WHEELHOUSE, R., GHANNAM, O., HARTLEY, L., STRANGES, S. & REES, K. 2017. Vitamin C supplementation for the primary prevention of cardiovascular disease. *Cochrane Database Syst Rev*, 3, Cd011114.
- ALBOAKLAH, H. 2018a. Effect of vitamin e and tempol on low density lipoprotein oxidation at lysosomal pH. *Cardiovascular Research*, 114, S46.
- ALBOAKLAH, H. K. 2018b. Vitamin E can increase, rather than decrease, low density lipoprotein oxidation at lysosomal pH. *Atherosclerosis Supplements*, 32, 123.
- ALUL, R. H., WOOD, M., LONGO, J., MARCOTTE, A. L., CAMPIONE, A. L., MOORE, M. K. & LYNCH, S. M. 2003. Vitamin C protects low-density lipoprotein from homocysteine-mediated oxidation. *Free Radical Biology and Medicine*, 34, 881-891.
- APPELQVIST, H., WASTER, P., KAGEDAL, K. & OLLINGER, K. 2013. The lysosome: from waste bag to potential therapeutic target. *J Mol Cell Biol*, 5, 214-26.
- ARAI, H., KOKUBO, Y., WATANABE, M., SAWAMURA, T., ITO, Y., MINAGAWA, A., OKAMURA, T. & MIYAMATO, Y. 2013. Small dense low-density lipoproteins

cholesterol can predict incident cardiovascular disease in an urban Japanese cohort: the Suita study. *J Atheroscler Thromb*, 20, 195-203.

- ARMITAGE, J. M., BOWMAN, L., CLARKE, R. J., WALLENDSZUS, K., BULBULIA, R., RAHIMI, K., HAYNES, R., PARISH, S., SLEIGHT, P., PETO, R. & COLLINS, R. 2010. Effects of homocysteine-lowering with folic acid plus vitamin B12 vs placebo on mortality and major morbidity in myocardial infarction survivors: a randomized trial. *Jama*, 303, 2486-94.
- AROSIO, P., ELIA, L. & POLI, M. 2017. Ferritin, Cellular Iron Storage and Regulation. *IUBMB Life*, 69, 414-422.
- AROSIO, P., INGRASSIA, R. & CAVADINI, P. 2009. Ferritins: a family of molecules for iron storage, antioxidation and more. *Biochim Biophys Acta*, 1790, 589-99.
- ARROYO, L. H. & LEE, R. T. 1999. Mechanisms of plaque rupture mechanical and biologic interactions. *Cardiovascular Research*, 41, 369-375.
- ARUOMA, O. I. 1999. Free radical antioxidants and international nutrition. *Asia Pacific J Clin Nutr*, 8, 53-63.
- ASSMANN, G., CULLEN, P., JOSSA, F., LEWIS, B. & MANCINI, M. 1999. Coronary heart disease: reducing the risk. *Arterioscl. Thromb. Vasc. Biol.*, 19, 1819-1824.
- AURSULESEI, V., COZMA, A. & KRASNIQI, A. 2014. Iron hypothesis of cardiovascular disease: still controversial. *Rev Med Chir Soc Med Nat Iasi*, 118, 901-9.
- AUSTIN, M. A., BRESLOW, J. L., HENNEKENS, C. H., BURING, J. E., WILLETT, W. C. & KRAUSS, R. M. 1988. Low-density lipoprotein subclass patterns and risk of myocardial infarction. *Jama*, 260, 1917-21.
- BAIGENT, C., BLACKWELL, L., EMBERSON, J., HOLLAND, L. E., REITH, C., BHALA, N., PETO, R., BARNES, E. H., KEECH, A., SIMES, J. & COLLINS, R. 2010. Efficacy and safety of more intensive lowering of LDL cholesterol: a meta-analysis of data from 170,000 participants in 26 randomised trials. *Lancet*, 376, 1670-81.
- BALL, R. Y., CARPENTER, K. L. & MITCHINSON, M. J. 1988. Ceroid accumulation by murine peritoneal macrophages exposed to artificial lipoproteins: ultrastructural observations. *Br J Exp Pathol*, 69, 43-56.
- BALLA, G., JACOB, H. S., EATON, J. W., BELCHER, J. D. & VERCELLOTTI, G. M. 1991. Hemin: a possible physiological mediator of low density lipoprotein oxidation and endothelial injury. *Arteriosclerosis and Thrombosis: A Journal of Vascular Biology*, 11, 1700-1711.
- BANCELLS, C., VILLEGAS, S., BLANCO, F. J., BENÍTEZ, S., GÁLLEGO, I., BELOKI, L., PÉREZ-CUELLAR, M., ORDÓÑEZ-LLANOS, J. & SÁNCHEZ-QUESADA, J. L. 2010. Aggregated electronegative low density lipoprotein in human plasma shows a high tendency toward phospholipolysis and particle fusion. *The Journal of biological chemistry*, 285, 32425-32435.

- BEAGLEHOLE, R. & BONITA, R. 2008. Global public health: a scorecard. *Lancet*, 372, 1988-96.
- BECKMAN, J. S., YE, Y. Z., P.G., A., CHEN, J., ACCAVITTI, M., TARPEY, M. M. & WHITE, C. R. 1994. Extensive nitration of protein tyrosines in human atherosclerosis detected by immunohistochemistry. *Biol Chem Hoppe-Seyler*, 375, 81.
- BEDWELL, S., DEAN, R. T. & JESSUP, W. 1989. The action of defined oxygen-centered free radicals on human low-density lipoprotein. *Biochemical Journal*, 262, 707-712.
- BEETENS, J. R., COENE, M. C., VERHEYEN, A., ZONNEKEYN, L. & HERMAN, A. G. 1984. Influence of vitamin C on the metabolism of arachidonic acid and the development of aortic lesions during experimental atherosclerosis in rabbits. *Biomed Biochim Acta*, 43, S273-6.
- BEKRI, S., GUAL, P., ANTY, R., LUCIANI, N., DAHMAN, M., RAMESH, B., IANNELLI, A., STACCINI-MYX, A., CASANOVA, D., BEN AMOR, I., SAINT-PAUL, M. C., HUET, P. M., SADOUL, J. L., GUGENHEIM, J., SRAI, S. K., TRAN, A. & LE MARCHAND-BRUSTEL, Y. 2006. Increased adipose tissue expression of hepcidin in severe obesity is independent from diabetes and NASH. *Gastroenterology*, 131, 788-96.
- BELLETTATO, C. M. & SCARPA, M. 2010. Pathophysiology of neuropathic lysosomal storage disorders. *J Inherit Metab Dis*, 33, 347-62.
- BENDICH, A., MACHLIN, L. J., SCANDURRA, O., BURTON, G. W. & WAYNER, D. D. M. 1986. The antioxidant role of vitamin C. Advances in Free Radical Biology & Medicine, 2, 419-444.
- BERLINER, J. A. & HEINECKE, J. W. 1996. The role of oxidized lipoproteins in atherogenesis. *Free Radic Biol Med*, 20, 707-27.
- BERLINER, J. A., TERRITO, M. C., SEVANIAN, A. S., RAMIN, S., KIM, J. A., BAMSHAD, B., ESTERSON, M. & FOGELMAN, A. M. 1990. Minimally modified low density lipoprotein stimulates monocyte endothelial interactions. *J Clin Invest*, 85, 1260-1266.
- BESOUW, M. T., BOWKER, R., DUTERTRE, J. P., EMMA, F., GAHL, W. A., GRECO, M., LILIEN, M. R., MCKIERNAN, J., NOBILI, F., SCHNEIDER, J. A., SKOVBY, F., VAN DEN HEUVEL, L. P., VAN'T HOFF, W. G. & LEVTCHENKO, E. N. 2011. Cysteamine toxicity in patients with cystinosis. *J Pediatr*, 159, 1004-11.
- BESOUW, M. T. & LEVTCHENKO, E. N. 2014. Improving the prognosis of nephropathic cystinosis. *International journal of nephrology and renovascular disease*, 7, 297-302.
- BHAKDI, S., DORWEILER, B., KIRCHMANN, R., TORZEWSKI, J., WEISE, E., TRANUM-JENSEN, J., WALEV, I. & WIELAND, E. 1995. On the pathogenesis of atherosclerosis: enzymatic transformation of human low density lipoprotein to an atherogenic moiety. J Exp Med, 182, 1959-71.

- BHATNAGAR, P., WICKRAMASINGHE, K., WILLIAMS, J., RAYNER, M. & TOWNSEND, N. 2015. The epidemiology of cardiovascular disease in the UK 2014. *Heart (British Cardiac Society)*, 101, 1182-1189.
- BIERMAN, E. L. 1992. George Lyman Duff Memorial Lecture. Atherogenesis in diabetes. *Arterioscler Thromb Vasc Biol*, 12, 647-656.
- BOAZ, M., SMETANA, S., WEINSTEIN, T., MATAS, Z., GAFTER, U., IAINA, A., KNECHT, A., WEISSGARTEN, Y., BRUNNER, D., FAINARU, M. & GREEN, M. S. 2000. Secondary prevention with antioxidants of cardiovascular disease in endstage renal disease (SPACE): randomised placebo-controlled trial. *Lancet*, 356, 1213-8.
- BOBRYSHEV, Y. V., SHCHELKUNOVA, T. A., MOROZOV, I. A., RUBTSOV, P. M., SOBENIN, I. A., OREKHOV, A. N. & SMIRNOV, A. N. 2013. Changes of lysosomes in the earliest stages of the development of atherosclerosis. *J Cell Mol Med*, 17, 626-35.
- BOMFORD, A., CONLON-HOLLINGSHEAD, C. & MUNRO, H. N. 1981. Adaptive responses of rat tissue isoferritins to iron administration. Changes in subunit synthesis, isoferritin abundance, and capacity for iron storage. *J Biol Chem*, 256.
- BONTEN, E. J., ANNUNZIATA, I. & D'AZZO, A. 2014. Lysosomal multienzyme complex: pros and cons of working together. *Cellular and molecular life sciences : CMLS*, 71, 2017-2032.
- BOU-ABDALLAH, F. 2010. The iron redox and hydrolysis chemistry of the ferritins. *Biochim Biophys Acta*, 1800, 719-31.
- BOULLIER, A., BIRD, D. A., CHANG, M. K., DENNIS, E. A., FRIEDMAN, P., GILLOTRE-TAYLOR, K., HORKKO, S., PALINSKI, W., QUEHENBERGER, O., SHAW, P., STEINBERG, D., TERPSTRA, V. & WITZTUM, J. L. 2001. Scavenger receptors, oxidized LDL, and atherosclerosis. *Ann N Y Acad Sci*, 947, 214-22; discussion 222-3.
- BOUTON, C., RAVEAU, M. & DRAPIER, J. C. 1996. Modulation of iron regulatory protein functions. Further insights into the role of nitrogen- and oxygen-derived reactive species. *J Biol Chem*, 271, 2300-6.
- BOWLES, D. K. & LAUGHLIN, M. H. 2011. Mechanism of beneficial effects of physical activity on atherosclerosis and coronary heart disease. *Journal of applied physiology* (*Bethesda, Md. : 1985*), 111, 308-310.
- BOWRY, V. W., INGOLD, K. U. & STOCKER, R. 1992. Vitamin E in human low-density lipoprotein. When and how this antioxidant becomes a pro-oxidant. *Biochemical Journal*, 288, 341-344.
- BOWRY, V. W. & STOCKER, R. 1993. Tocopherol-mediated peroxidation. The prooxidant effect of vitamin E on the radical-initiated oxidation of human low-density lipoprotein. *Journal of the American Chemical Society*, 115, 6029-6044.

- BOYD, H. C., GOWN, A. M., WOLFBAUER, G. & CHAIT, A. 1989. Direct evidence for a protein recognized by a monoclonal antibody against oxidatively modified LDL in atherosclerotic lesions from a Watanabe heritable hyperlipidemic rabbit. *The American journal of pathology*, 135, 815-825.
- BOYLE, J. J. 1999. Vascular smooth muscle cell apoptosis in atherosclerosis. *International Journal of Experimental Pathology*, 80, 197-203.
- BROWN, A. J., DEAN, R. T. & JESSUP, W. 1996. Free and esterified oxysterol: formation during copper-oxidation of low density lipoprotein and uptake by macrophages. *J Lipid Res*, 37, 320-35.
- BROWN, A. J. & JESSUP, W. 1999. Oxysterols and atherosclerosis. *Atherosclerosis*, 142, 1-28.
- BROWN, M. S., DANA, S. E. & GOLDSTEIN, J. L. 1973. Regulation of 3-hydroxy-3methylglutaryl coenzyme a reductase-activity in human fibroblasts by lipoproteins. *Proceedings of the National Academy of Sciences of the United States of America*, 70, 2162-2166.
- BROWN, M. S. & GOLDSTEIN, J. L. 1983. Lipoprotein metabolism in the macrophage: implications for cholesterol deposition in atherosclerosis. *Annu Rev Biochem*, 52, 223-61.
- BRUNK, U. T., NEUZIL, J. & EATON, J. W. 2001. Lysosomal involvement in apoptosis. *Redox Rep*, 6, 91-7.
- BRUNK, U. T. & TERMAN, A. 2002. Lipofuscin: Mechanisms of age-related accumulation and influence on cell functions. *Free Radic Biol Med*, 33, 611-9.
- BURTON, G. W. & INGOLD, K. U. 1984. beta-Carotene: an unusual type of lipid antioxidant. *Science*, 224, 569-73.
- CAIRO, G., RECALCATI, S., MANTOVANI, A. & LOCATI, M. 2011. Iron trafficking and metabolism in macrophages: contribution to the polarized phenotype. *Trends Immunol*, 32, 241-7.
- CAMEJO, G. 2003. Hydrolytic enzymes released from resident macrophages and located in the intima extracellular matrix as agents that modify retained apolipoprotein B lipoproteins. *Arterioscler Thromb Vasc Biol*, 23, 1312-3.
- CAREW, T. E., PITTMAN, R. C., MARCHAND, E. R. & STEINBERG, D. 1984. Measurement in vivo of irreversible degradation of low density lipoprotein in the rabbit aorta. Predominance of intimal degradation *Arteriosclerosis*, 4, 214-24.
- CARR, A. C., MYZAK, M. C., STOCKER, R., MCCALL, M. R. & FREI, B. 2000a. Myeloperoxidase binds to low-density lipoprotein: potential implications for atherosclerosis. *FEBS Lett*, 487, 176-80.

- CARR, A. C., ZHU, B. Z. & FREI, B. 2000b. Potential antiatherogenic mechanisms of ascorbate (vitamin C) and alpha-tocopherol (vitamin E). *Circ Res*, 87, 349-54.
- CASEY, J. L., HENTZE, M. W., KOELLER, D. M., CAUGHMAN, S. W., ROUAULT, T. A., KLAUSNER, R. D. & HARFORD, J. B. 1988. Iron-responsive elements: regulatory RNA sequences that control mRNA levels and translation. *Science*, 240, 924-8.
- CASIDAY, R. & FREY, R. online. Iron use and storage in the body: ferritin and molecular representations.<u>http://www.chemistry.wustl.edu/~edudev/LabTutorials/Ferritin/Ferritin.html</u> [Accessed 31st October 2018].
- CATHCART, M. K., MCNALLY, A. K., MOREL, D. W. & CHISOLM, G. M., 3RD 1989. Superoxide anion participation in human monocyte-mediated oxidation of low-density lipoprotein and conversion of low-density lipoprotein to a cytotoxin. *J Immunol*, 142, 1963-9.
- CESENA, F. H. Y., LAURINAVICIUS, A. G., VALENTE, V. A., CONCEIÇÃO, R. D., SANTOS, R. D. & BITTENCOURT, M. S. 2017. The Expected Cardiovascular Benefit of Plasma Cholesterol Lowering with or Without LDL-C Targets in Healthy Individuals at Higher Cardiovascular Risk. *Arquivos brasileiros de cardiologia*, 108, 518-525.
- CHAIT, A. 1987. Progression of atherosclerosis: the cell biology. *Eur Heart J*, 8 Suppl E, 15-22.
- CHAIT, A. & BORNFELDT, K. E. 2009. Diabetes and atherosclerosis: is there a role for hyperglycemia? *Journal of Lipid Research*, 50, S335-S339.
- CHAJEK, T. & FIELDING, C. J. 1978. Isolation and Characterization of a Human Serum Cholesteryl Ester Transfer Protein. *Proceedings of the National Academy of Sciences of the United States of America*, 75, 3445-3449.
- CHAO, F. F., BLANCHETTE-MACKIE, E. J., TERTOV, V. V., SKARLATOS, S. I., CHEN, Y. J. & KRUTH, H. S. 1992. Hydrolysis of cholesteryl ester in low density lipoprotein converts this lipoprotein to a liposome. *J Biol Chem*, 267, 4992-8.
- CHELLAN, B., REARDON, C. A., GETZ, G. S. & HOFMANN BOWMAN, M. A. 2016. Enzymatically Modified Low-Density Lipoprotein Promotes Foam Cell Formation in Smooth Muscle Cells via Macropinocytosis and Enhances Receptor-Mediated Uptake of Oxidized Low-Density Lipoprotein. Arterioscler Thromb Vasc Biol, 36, 1101-13.
- CHEN, M. A., KAWAKUBO, M., COLLETTI, P. M., XU, D., LABREE DUSTIN, L., DETRANO, R., AZEN, S. P., WONG, N. D. & ZHAO, X.-Q. 2013. Effect of age on aortic atherosclerosis. *Journal of Geriatric Cardiology : JGC*, 10, 135-140.
- CHISOLM, G. M. & STEINBERG, D. 2000. The oxidative modification hypothesis of atherogenesis: an overview. *Free Radic Biol Med*, 28, 1815-26.
- CHOY, K., BECK, K., PNG, F. Y., WU, B. J., LEICHTWEIS, S. B., THOMAS, S. R., HOU, J. Y., CROFT, K. D., MORI, T. A. & STOCKER, R. 2005. Processes involved in the

site-specific effect of probucol on atherosclerosis in apolipoprotein E gene knockout mice. *Arterioscler Thromb Vasc Biol*, 25, 1684-90.

- CHUNG, B., CHASTON, T., MARKS, J., SRAI, S. K. & SHARP, P. A. 2009. Hepcidin decreases iron transporter expression in vivo in mouse duodenum and spleen and in vitro in THP-1 macrophages and intestinal Caco-2 cells. *J Nutr*, 139, 1457-62.
- CLARKE, R., PEDEN, J. F., HOPEWELL, J. C., KYRIAKOU, T., GOEL, A., HEATH, S. C., PARISH, S., B., S , FRANZOSI, M. G., RUST, S., BENNETT, D., SILVEIRA, A., MALARSTIG, A., GREEN, F. R., MARK LATHROP, M., GIGANTE, B., LEANDER, K., FAIRE, U., SEEDORF, U., HAMSTEN, A., COLLINS, R., WATKINS, H. & FARRALL, M. 2009. Genetic Variants Associated with Lp(a) Lipoprotein Level and Coronary Disease . PROCARDIS Consortium. N Engl J Med, 2518-28.
- COLAS, R., PRUNETA-DELOCHE, V., GUICHARDANT, M., LUQUAIN-COSTAZ, C., CUGNET-ANCEAU, C., MORET, M., VIDAL, H., MOULIN, P., LAGARDE, M. & CALZADA, C. 2010. Increased lipid peroxidation in LDL from type-2 diabetic patients. *Lipids*, 45, 723-731.
- COLLES, S. M. & CHISOLM, G. M. 2000. Lysophosphatidylcholine-induced cellular injury in cultured fibroblasts involves oxidative events. *J Lipid Res*, 41, 1188-98.
- COLLES, S. M., MAXSON, J. M., CARLSON, S. G. & CHISOLM, G. M. 2001a. Oxidized LDL-induced injury and apoptosis in atherosclerosis. Potential roles for oxysterols. *Trends Cardiovasc Med*, 11, 131-8.
- COLLES, S. M., MAXSON, J. M., CARLSON, S. G. & CHISOLM, G. M. 2001b. Oxidized LDL-Induced Injury and Apoptosis in Atherosclerosis: Potential Roles for Oxysterols. *Trends in Cardiovascular Medicine*, 11, 131-138.
- COLLINS, D. S., UNANUE, E. R. & HARDING, C. V. 1991. Reduction of disulfide bonds within lysosomes is a key step in antigen processing. *J Immunol*, 147, 4054-9.
- COLLINS, J. F., WESSLING-RESNICK, M. & KNUTSON, M. D. 2008. Hepcidin regulation of iron transport. *The Journal of nutrition*, 138, 2284-2288.
- COLLINS, R., ARMITAGE, J., PARISH, S., SLEIGHT, P. & PETO, R. 2002. MRC/BHF Heart Protection Study of antioxidant vitamin supplementation in 20536 high-risk individuals: A randomised placebo-controlled trial. *Lancet*, 360, 23–33.
- COLLINS, R. G. 2000. P-selectin or intercellular adhesion molecule (ICAM-1) deficiency substantially protects against atherosclerosis in apolipoprotein E-deficient mice. *J. Exp. Med.*, 191, 189-194.
- COLLOT-TEIXEIRA, S., MARTIN, J., MCDERMOTT-ROE, C., POSTON, R. & MCGREGOR, J. L. 2007. CD36 and macrophages in atherosclerosis. *Cardiovasc Res* 75, 468–77.

- COOK, N. R., ALBERT, C. M., GAZIANO, J. M., ZAHARRIS, E., MACFADYEN, J., DANIELSON, E., BURING, J. E. & MANSON, J. E. 2007. A randomized factorial trial of vitamins C and E and beta carotene in the secondary prevention of cardiovascular events in women: results from the Women's Antioxidant Cardiovascular Study. Arch Intern Med, 167, 1610-8.
- COX, B. E., GRIFFIN, E. E., ULLERY, J. C. & JEROME, W. G. 2007. Effects of cellular cholesterol loading on macrophage foam cell lysosome acidification. *J Lipid Res*, 48, 1012-21.
- COZZI, A., SANTAMBROGIO, P., CORSI, B., CAMPANELLA, A., AROSIO, P. & LEVI, S. 2006. Characterization of the 1-ferritin variant 460InsA responsible of a hereditary ferritinopathy disorder. *Neurobiol Dis*, 23, 644-52.
- CRICHTON, R. R. 1971. Studies on the structure of ferritin and apoferritin from horse spleen II. chymotrypsin, subtilisin, cathepsin D and pepsin digestion of ferritin and apoferritin. *Biochimica et biophysica Acta (BBA)*, 229, 75-82.
- CUSHING, S. D., BERLINER, J. A., VALENTE, A. J., TERRITO, M. C., NAVAB, M., PARHAMI, F., GERRITY, R., SCHWARTZ, C. J. & FOGELMAN, A. M. 1990. Minimally modified low density lipoprotein induces monocyte chemotactic protein 1 in human endothelial cells and smooth muscle cells. *Proc Natl Acad Sci U S A*, 87, 5134-8.
- CYRUS, T. 1999. Disruption of 12/15-lipoxygenase diminishes atherosclerosis in apoEdeficient mice. J. Clin. Invest., 103, 1597-1604.
- DABBAGH, A. J. & FREI, B. 1995. Human suction blister interstitial fluid prevents metal iondependent oxidation of low density lipoprotein by macrophages and in cell-free systems. *J Clin Invest*, 96, 1958-66.
- DARLEY, V. S., MAR, V. M., HOGG, H., O'LEARY, V. J. & MONCADA, S. 1992. The simultaneous generation of superoxide and nitric oxide can initiate lipid peroxidation in human LDL. *Free Radic. Res. Commun.*, 17, 19-26.
- DAVIES, M. J. 2016. Protein oxidation and peroxidation. *The Biochemical journal*, 473, 805-825.
- DAVIES, M. J. & THOMAS, A. C. 1985. Plaque fissuring--the cause of acute myocardial infarction, sudden ischaemic death, and crescendo angina. *Br Heart J.*, 53, 363-73.
- DE DOMENICO, I., MCVEY WARD, D. & KAPLAN, J. 2008. Regulation of iron acquisition and storage: consequences for iron-linked disorders. *Nat Rev Mol Cell Biol*, 9, 72-81.
- DE DOMENICO, I., VAUGHN, M. B., LI, L., BAGLEY, D., MUSCI, G., WARD, D. M. & KAPLAN, J. 2006. Ferroportin-mediated mobilization of ferritin iron precedes ferritin degradation by the proteasome. *Embo j*, 25, 5396-404.

DE DUVE, C. 1963. The lysosome. Sci Am 208, 64-72.

- DE DUVE, C., PRESSMAN, B. C., GIANETTO, R., WATTIAUX, R. & APPELMANS, F. 1955. Tissue fractionation studies. 6. Intracellular distribution patterns of enzymes in rat-liver tissue. *Biochem J*, 60, 604-17.
- DE GAETANO, G. 2001. Low-dose aspirin and vitamin E in people at cardiovascular risk: a randomised trial in general practice. Collaborative Group of the Primary Prevention Project. *Lancet*, 357, 89-95.
- DE GRAAF, J., HAK-LEMMERS, H. L., HECTORS, M. P., DEMACKER, P. N., HENDRIKS, J. C. & STALENHOEF, A. F. 1991. Enhanced susceptibility to in vitro oxidation of the dense low density lipoprotein subfraction in healthy subjects. *Arterioscler Thromb*, 11, 298-306.
- DEAN, R. T. 1975. Direct evidence of importance of lysosomes in degradation of intracellular proteins. *Nature*, 257, 414-6.
- DELPORTE, C., VAN ANTWERPEN, P., VANHAMME, L., ROUMEGUERE, T. & ZOUAOUI BOUDJELTIA, K. 2013. Low-density lipoprotein modified by myeloperoxidase in inflammatory pathways and clinical studies. *Mediators Inflamm*, 2013, 971579.
- DEUTSCH, J. C., SANTHOSH-KUMAR, C. R., HASSELL, K. L. & KOLHOUSE, J. F. 1994. Variation in ascorbic acid oxidation routes in hydrogen peroxide and cupric ion solution as determined by GC/MS. *Analytical Chemistry*, 66, 345-350.
- DI ANGELANTONIO, E., SARWAR, N., PERRY, P., KAPTOGE, S., RAY, K. K., THOMPSON, A., WOOD, A. M., LEWINGTON, S., SATTAR, N., PACKARD, C. J., COLLINS, R., THOMPSON, S. G. & DANESH, J. 2009. Major lipids, apolipoproteins, and risk of vascular disease. *Jama*, 302, 1993-2000.
- DIEBER-ROTHENEDER, M., PUHL, H., WAEG, G., STRIEGL, G. & ESTERBAUER, H. 1991. Effect of oral supplementation with D-alpha-tocopherol on the vitamin E content of human low density lipoproteins and resistance to oxidation. *J Lipid Res*, 32, 1325-32.
- DOHIL, R., FIDLER, M., BARSHOP, B. A., GANGOITI, J., DEUTSCH, R., MARTIN, M. & SCHNEIDER, J. A. 2006. Understanding intestinal cysteamine bitartrate absorption. *J Pediatr*, 148, 764-9.
- DOLL, R. & HILL, A. B. 1956. Lung cancer and other causes of death in relation to smoking: a second report on the mortality of British doctors. *Br Med J*, 2, 1071–1081.
- DONG, Z. M. 1998. The combined role of P- and E-selectins in atherosclerosis. J. Clin. Invest., 102, 145-152.
- DUBOIS-RANDE, J. L., ARTIGOUT, J. Y., DARMONJ, J. Y., HABBAL, R., MANUEL, C., TAYARANIJ, I., CASTAIGNE, A. & GROSGOGEATF, Y. 1994. Oxidative stress in patients with unstable angina. *European Heart Journal* 15, 179-183.
- DUEWELL, P., KONO, H., RAYNER, K. J., SIROIS, C. M., VLADIMER, G., BAUERNFEIND, F. G., ABELA, G. S., FRANCHI, L., NUÑEZ, G., SCHNURR, M.,

ESPEVIK, T., LIEN, E., FITZGERALD, K. A., ROCK, K. L., MOORE, K. J., WRIGHT, S. D., HORNUNG, V. & LATZ, E. 2010. NLRP3 inflammasomes are required for atherogenesis and activated by cholesterol crystals. *Nature*, 464, 1357.

- DURAND, P., S., L.-C. & D., B. 1997. Acute methionine load-induced hyperhomocysteinemia enhances platelet aggregation, thromboxane biosynthesis, and macrophage-derived tissue factor activity in rats. *FASEB J.*, 11, 1157-1168.
- EICHNER, J. E., QI, H., MOORE, W. E. & SCHECHTER, E. 1998. Iron measures in coronary angiography patients. *Atherosclerosis Supplements*, 136.
- EL-SAADANI, M., ESTERBAUER, H., EL-SAYED, M., GOHER, M., NASSAR, A. Y. & JURGENS, G. 1989. A spectrophotometric assay for lipid peroxides in serum lipoproteins using a commercially available reagent. *J Lipid Res*, 30, 627-30.
- EMANUEL, R., SERGIN, I., BHATTACHARYA, S., TURNER, J. N., EPELMAN, S., SETTEMBRE, C., DIWAN, A., BALLABIO, A. & RAZANI, B. 2014. Induction of lysosomal biogenesis in atherosclerotic macrophages can rescue lipid-induced lysosomal dysfunction and downstream sequelae. *Arterioscler Thromb Vasc Biol*, 34, 1942-52.
- ERKKILA, A. T., NARVANEN, O., LEHTO, S., UUSITUPA, M. I. & YLA-HERTTUALA, S. 2000. Autoantibodies against oxidized low-density lipoprotein and cardiolipin in patients with coronary heart disease. *Arterioscler Thromb Vasc Biol*, 20, 204-9.
- ESTERBAUER, H., DIEBER-ROTHENEDER, M., STRIEGL, G. & WAEG, G. 1991a. Role of vitamin E in preventing the oxidation of low-density lipoprotein. *Am J Clin Nutr*, 53, 314s-321s.
- ESTERBAUER, H., DIEBER-ROTHENEDER, M., WAEG, G., PUHL, H. & TATZBER, F. 1990a. Endogenous antioxidants and lipoprotein oxidation. *Biochem Soc Trans*, 18, 1059-61.
- ESTERBAUER, H., DIEBER-ROTHENEDER, M., WAEG, G., STRIEGL, G. & JURGENS, G. 1990b. Biochemical, structural, and functional properties of oxidized low-density lipoprotein. *Chem Res Toxicol*, *3*, 77-92.
- ESTERBAUER, H., GEBICKI, J., PUHL, H. & JÜRGENS, G. 1992. The role of lipid peroxidation and antioxidants in oxidative modification of LDL. *Free Radical Biology and Medicine*, 13, 341-390.
- ESTERBAUER, H., JURGENS, G., QUEHENBERGER, O. & KOLLER, E. 1987. Autoxidation of human low density lipoprotein: loss of polyunsaturated fatty acids and vitamin E and generation of aldehydes. *J Lipid Res*, 28, 495-509.
- ESTERBAUER, H., PUHL, H., DIEBER-ROTHENEDER, M., WAEG, G. & RABL, H. 1991b. Effect of antioxidants on oxidative modification of LDL. *Ann Med*, 23, 573-81.

- ESTERBAUER, H., SCHAUR, R. J. & ZOLLNER, H. 1991c. Chemistry and biochemistry of 4-hydroxynonenal, malonaldehyde and related aldehydes. *Free Radic Biol Med*, 11, 81-128.
- ESTERBAUER, H., STRIEGL, G., PUBL, H. & ROTHENDER, M. 1989a. Continuous monitoring of in vitro oxidation of human low density lipoprotein. *Free Radic Res Commun*, 6, 67–75.
- ESTERBAUER, H., STRIEGL, G., PUHL, H. & ROTHENEDER, M. 1989b. Continuous monitoring of in vitro oxidation of human Low Density Lipoprotein. *Free Radic Res Commun*, 6, 67-75.
- ESTERBAUER, H., WAG, G. & PUHL, H. 1993. Lipid peroxidation and its role in atherosclerosis. *Br Med Bull*, 49, 566-76.
- FACCHINI, F. S. & SAYLOR, K. L. 2002. Effect of iron depletion on cardiovascular risk factors: studies in carbohydrate-intolerant patients. *Ann N Y Acad Sci*, 967, 342-51.
- FARNIER, M. 2014. PCSK9: From discovery to therapeutic applications. Archives of Cardiovascular Diseases, 107, 58-66.
- FEINGOLD, K. R. & GRUNFELD, C. 2000. Introduction to Lipids and Lipoproteins. *In:* DE GROOT, L. J., BECK-PECCOZ, P., CHROUSOS, G., DUNGAN, K., GROSSMAN, A., HERSHMAN, J. M., KOCH, C., MCLACHLAN, R., NEW, M., REBAR, R., SINGER, F., VINIK, A. & WEICKERT, M. O. (eds.) *Endotext*. South Dartmouth (MA): MDText.com, Inc.
- FIRTH, C. A., YANG, Y.-T. & GIESEG, S. P. 2007. Lipid oxidation predominates over protein hydroperoxide formation in human monocyte-derived macrophages exposed to aqueous peroxyl radicals. *Free Radical Research*, 41, 839-848.
- FLORES, S. E., DAY, A. S. & KEENAN, J. I. 2015. Measurement of total iron in Helicobacter pylori-infected gastric epithelial cells. *BioMetals*, 28, 143-150.
- FOGELMAN, A. M., SHECHTER, I., SEAGER, J., HOKOM, M., CHILD, J. S. & EDWARDS, P. A. 1980. Malondialdehyde alteration of low density lipoproteins leads to cholesteryl ester accumulation in human monocyte-macrophages. *Proc Natl Acad Sci U S A*, 77, 2214-8.
- FREEMERMAN, A. J., JOHNSON, A. R., SACKS, G. N., MILNER, J. J., KIRK, E. L., TROESTER, M. A., MACINTYRE, A. N., GORAKSHA-HICKS, P., RATHMELL, J. C. & MAKOWSKI, L. 2014. Metabolic reprogramming of macrophages: glucose transporter 1 (GLUT1)-mediated glucose metabolism drives a proinflammatory phenotype. J Biol Chem, 289, 7884-96.
- FREI, B. 1991. Ascorbic acid protects lipids in human plasma and low-density lipoprotein against oxidative damage. *Am J Clin Nutr*, 54, 1113s-1118s.

- FREI, B., ENGLAND, L. & AMES, B. N. 1989. Ascorbate is an outstanding antioxidant in human blood plasma. Proceedings of the National Academy of Sciences of the United States of America, 86, 6377-6381.
- FUKUZUMI, M., SHINOMIYA, H., SHIMIZU, Y., OHISHI, K. & UTSUMI, S. 1996. Endotoxin-induced enhancement of glucose influx into murine peritoneal macrophages via GLUT1. *Infect Immun*, 64, 108-12.
- FUSTER, V. & KELLY, B. B. 2010. Epidemiology of Cardiovascular Disease [Online]. Washington (DC): National Academies Press (US). Available: https://www.ncbi.nlm.nih.gov/books/NBK45688/ [Accessed 31st October 2018].
- GABAY, C. & KUSHNER, I. 1999. Acute-phase proteins and other systemic responses to inflammation. *N Engl J Med*, 340, 448-54.
- GAHL, W. A., THOENE, J. G. & SCHNEIDER, J. A. 2002. Cystinosis. N Engl J Med, 347, 111-21.
- GALESLOOT, T. E., HOLEWIJN, S., KIEMENEY, L. A., GRAAF, J., VERMEULEN, S. H. & SWINKELS, D. W. 2014. Serum hepcidin is associated with presence of plaque in postmenopausal women of a general population. *Arterioscler Thromb Vasc Biol.*, 34.
- GALESLOOT, T. E., JANSS, L. L., BURGESS, S., KIEMENEY, L. A. L. M., DEN HEIJER,
 M., DE GRAAF, J., HOLEWIJN, S., BENYAMIN, B., WHITFIELD, J. B.,
 SWINKELS, D. W. & VERMEULEN, S. H. 2015. Iron and hepcidin as risk factors in atherosclerosis: what do the genes say? *BMC Genetics*, 16, 79.
- GANZ, T. 2005. Hepcidin--a regulator of intestinal iron absorption and iron recycling by macrophages. *Best Pract Res Clin Haematol*, 18, 171-82.
- GANZ, T. & NEMETH, E. 2009. Iron sequestration and anemia of inflammation. Semin *Hematol*, 46, 387-93.
- GARNER, B., LI, W., ROBERG, K. & BRUNK, U. T. 1997. On the cytoprotective role of ferritin in macrophages and its ability to enhance lysosomal stability. *Free Radic Res*, 27, 487-500.
- GAZIANO, J. M. 1996. Epidemiology of risk factor reduction. In: Vascular Medicine, Boston, MA, Little Brown.
- GAZIANO, J. M., HATTA, A., FLYNN, M., JOHNSON, E. J., KRINSKY, N. I., RIDKER, P. M., HENNEKENS, C. H. & FREI, B. 1995. Supplementation with beta-carotene in vivo and in vitro does not inhibit low density lipoprotein oxidation. *Atherosclerosis*, 112, 187-95.
- GE, L., HONG-MEI, G. & DA-WEI, Z. 2013. ATP-binding cassette transporters and cholesterol translocation. *IUBMB Life*, 65, 505-512.
- GEBICKI, J. M., DU, J., COLLINS, J. & TWEEDDALE, H. 2000. Peroxidation of proteins and lipids in suspensions of liposomes, in blood serum, and in mouse myeloma cells. *Acta Biochim Pol*, 47, 901-11.

- GERALD, H. T. & DAPHNE, O. 2012. LDL as a Cause of Atherosclerosis. *The Open Atherosclerosis & Thrombosis Journal*, 5.
- GERHARD, G. T. & DUELL, P. B. 1999. Homocysteine and atherosclerosis. *Curr. Opin. Lipidol.*, 10, 417-429.

GERMAN NUTRITION, S. 2015. New Reference Values for Vitamin C Intake. *Annals of Nutrition and Metabolism*, 67, 13-20.

- GERRY, A. B. & LEAKE, D. S. 2008. A moderate reduction in extracellular pH protects macrophages against apoptosis induced by oxidized low density lipoprotein. *Journal of Lipid Research*, 49, 782-789.
- GERRY, A. B., SATCHELL, L. & LEAKE, D. S. 2008. A novel method for production of lipid hydroperoxide- or oxysterol-rich low-density lipoprotein. *Atherosclerosis*, 197, 579-87.
- GIESEG, S., DUGGAN, S. & GEBICKI, J. M. 2000. Peroxidation of proteins before lipids in U937 cells exposed to peroxyl radicals. *Biochem J*, 350 Pt 1, 215-8.
- GLASS, C. K. & WITZTUM, J. L. 2001. Atherosclerosis. the road ahead. Cell, 104, 503-16.
- GOFMAN, J. W., JONES, H. B., LINDGREN, F. T., LYON, T. P., ELLIOTT, H. A. & STRISOWER, B. 1950. Blood lipids and human atherosclerosis. *Circulation*, 2, 161-78.
- GOLDBOURT, U. & NEUFELD, H. N. 1988. Genetic aspects of arteriosclerosis. Arteriosclerosis, 6, 357-377.
- GOLDSTEIN, J. L., ANDERSON, R. G. W. & BROWN, M. S. 1979a. Coated Pits, Coated Vesicles, and Receptor-Mediated Endocytosis. *Nature*, 279, 679-685.
- GOLDSTEIN, J. L. & BROWN, M. S. 1973. Familial hypercholesterolemia identification of a defect in regulation of 3-hydroxy-3-methylglutaryl coenzyme-a reductase-activity associated with overproduction of cholesterol. *Proceedings of the National Academy of Sciences of the United States of America*, 70, 2804-2808.
- GOLDSTEIN, J. L. & BROWN, M. S. 1977. The low-density lipoprotein pathway and its relation to atherosclerosis. *Annu Rev Biochem*, 46, 897-930.
- GOLDSTEIN, J. L. & BROWN, M. S. 1979. LDL Receptor Locus and the Genetics of Familial Hypercholesterolemia. *Annual Review of Genetics*, 13, 259-289.
- GOLDSTEIN, J. L. & BROWN, M. S. 1986. Hyperlipidemia in Coronary Heart-Disease a Biochemical Genetic Approach. *Journal of Laboratory and Clinical Medicine*, 108, 174-181.
- GOLDSTEIN, J. L., HO, Y. K., BASU, S. K. & BROWN, M. S. 1979b. Binding sites on macrophages that mediate uptake and degradation of acetylated low density lipoprotein, producing massive cholesterol deposition. *Proc. Natl Acad. Sci. USA*, 76, 333-337.

- GOLDSTEIN, L. J. A. B., S M. 1977. The Low-Density Lipoprotein Pathway and its Relation to Atherosclerosis. July 1977 ed.: Annual reviews Inc.
- GOMES, A., FERNANDES, E. & LIMA, J. L. F. C. 2005. Fluorescence probes used for detection of reactive oxygen species. *Journal of Biochemical and Biophysical Methods*, 65, 45-80.
- GOSLING, J., SLAYMAKER, S., GU, L., TSENG, S., ZLOT, C. H., YOUNG, S. G., ROLLINS, B. J. & CHARO, I. F. 1999. MCP-1 deficiency reduces susceptibility to atherosclerosis in mice that overexpress human apolipoprotein B. *J Clin Invest*, 103, 773-8.
- GOSWAMI, B., TAYAL, D. & MALLIKA, V. 2008. Ferritin: A multidimensional bio marker. *The Internet Journal of Laboratory Medicine*, 3.
- GOZIN, A., FRANZINI, E., ANDRIEU, V., DA COSTA, L., ROLLET-LABELLE, E. & PASQUIER, C. 1998. Reactive oxygen species activate focal adhesion kinase, paxillin and p130cas tyrosine phosphorylation in endothelial cells. *Free Radic Biol Med*, 25, 1021-32.
- GRAHAM, A. N., HOGG, N., KALYANARAMAN, B., O'LEARY, V., DARLEY-USMAR, V. & MONCADE, S. 1993. Peroxynitrite modifications of LDL leads to recognition by the macrophage scavenger receptor. *FEBS Lett*, 330, 181.
- GRAY, N. K., QUICK, S., GOOSSEN, B., CONSTABLE, A., HIRLING, H., KUHN, L. C. & HENTZE, M. W. 1993. Recombinant iron-regulatory factor functions as an ironresponsive-element-binding protein, a translational repressor and an aconitase. A functional assay for translational repression and direct demonstration of the iron switch. *Eur J Biochem*, 218, 657-67.
- GRIFFIN, E. E., ULLERY, J. C., COX, B. E. & JEROME, W. G. 2005. Aggregated LDL and lipid dispersions induce lysosomal cholesteryl ester accumulation in macrophage foam cells. *J Lipid Res*, 46, 2052-60.
- GRINSHTEIN, N., BAMM, V. V., TSEMAKHOVICH, V. A. & SHAKLAI, N. 2003. Mechanism of low-density lipoprotein oxidation by hemoglobin-derived iron. *Biochemistry*, 42, 6977-6985.
- GROH, L., KEATING, S. T., JOOSTEN, L. A. B., NETEA, M. G. & RIKSEN, N. P. 2018. Monocyte and macrophage immunometabolism in atherosclerosis. *Semin Immunopathol*, 40, 203-214.
- GRUNE, T., REINHECKEL, T. & DAVIES, K. J. 1997. Degradation of oxidized proteins in mammalian cells. *Faseb j*, 11, 526-34.
- GU, L. 1998. Absence of monocyte chemoattractant protein-1 reduces atherosclerosis in low density lipoprotein-deficient mice. *Mol. Cell*, 2, 275-281.
- GUICCIARDI, M. E., LEIST, M. & GORES, G. J. 2004. Lysosomes in cell death. *Oncogene*, 23, 2881-90.

- GUYTON, J. R. & KLEMP, K. F. 1996. Development of the lipid-rich core in human atherosclerosis. *Arterioscler Thromb Vasc Biol*, 16, 4-11.
- HAJJAR, K. A. & NACHMAN, R. L. 1996. The role of lipoprotein(a) in atherogenesis and thrombosis. *Annu Rev Med*, 47, 423-442.
- HAKA, A. S., KRAMER, J. R., DASARI, R. R. & FITZMAURICE, M. 2011. Mechanism of ceroid formation in atherosclerotic plaque: in situ studies using a combination of Raman and fluorescence spectroscopy. *Journal of biomedical optics*, 16, 011011-011011.
- HAKALA, J. K., OKSJOKI, R., LAINE, P., DU, H., GRABOWSKI, G. A., KOVANEN, P. T. & PENTIKAINEN, M. O. 2003. Lysosomal enzymes are released from cultured human macrophages, hydrolyze LDL in vitro, and are present extracellularly in human atherosclerotic lesions. *Arterioscler Thromb Vasc Biol*, 23, 1430-6.
- HALLIWELL, B. 1990. How to characterise a biological antioxidant. Free Radic Res Commun, 9, 1-32.
- HALLIWELL, B. & GUTTERIDGE, J. M. 1990. Role of free radicals and catalytic metal ions in human disease: an overview. *Methods Enzymol*, 186, 1-85.
- HALVORSEN, B., OTTERDAL, K., DAHL, T. B., SKJELLAND, M., GULLESTAD, L., ØIE, E. & AUKRUST, P. 2008. Atherosclerotic Plaque Stability-What Determines the Fate of a Plaque? *Progress in Cardiovascular Diseases*, 51, 183-194.
- HAMMOND, E. C. & HORN, D. 1958. Smoking and death rates: report on forty-four months of follow-up of 187,783 men. 2. Death rates by cause. *J Am Med Assoc*, 166, 1294-1308.
- HANSSON, L., ZANCHETTI, A., CARRUTHERS, S. G., DAHLOF, B., ELMFELDT, D., JULIUS, S., MENARD, J., RAHN, K. H., WEDEL, H. & WESTERLING, S. 1998. Effects of intensive blood pressure lowering and low dose aspirin in patients with hypertension: principal results of the hypertension optimal treatment (HOT) randomised trial. *Lancet*, 351, 1755–1762.
- HARATS, D., SHAISH, A., GEORGE, J., MULKINS, M., KURIHARA, H., LEVKOVITZ, H. & SIGAL, E. 2000. Overexpression of 15-lipoxygenase in vascular endothelium accelerates early atherosclerosis in LDL receptor-deficient mice. *Arterioscler Thromb Vasc Biol*, 20, 2100-5.
- HARDWICK, S. J., HEGYI, L., CLARE, K., LAW, N. S., CARPENTER, K. L., MITCHINSON, M. J. & SKEPPER, J. N. 1996. Apoptosis in human monocytemacrophages exposed to oxidized low density lipoprotein. *J Pathol*, 179, 294-302.
- HARKEWICZ, R., HARTVIGSEN, K., ALMAZAN, F., DENNIS, E. A., WITZTUM, J. L. & MILLER, Y. I. 2008. Cholesteryl ester hydroperoxides are biologically active components of minimally oxidized low density lipoprotein. J Biol Chem, 283, 10241-51.

- HASCHEMI, A., KOSMA, P., GILLE, L., EVANS, C. R., BURANT, C. F., STARKL, P., KNAPP, B., HAAS, R., SCHMID, J. A., JANDL, C., AMIR, S., LUBEC, G., PARK, J., ESTERBAUER, H., BILBAN, M., BRIZUELA, L., POSPISILIK, J. A., OTTERBEIN, L. E. & WAGNER, O. 2012. The sedoheptulose kinase CARKL directs macrophage polarization through control of glucose metabolism. *Cell Metab*, 15, 813-26.
- HAYASHI, T., FUKUTO, J. M., IGNARRO, L. J. & CHAUDHURI, G. 1995. Gender differences in atherosclerosis: possible role of nitric oxide. *J Cardiovasc Pharmacol*, 26, 792-802.
- HEGYI, L., HARDWICK, S. J., SIOW, R. C. & SKEPPER, J. N. 2001. Macrophage death and the role of apoptosis in human atherosclerosis. *J Hematother Stem Cell Res*, 10, 27-42.
- HEINECKE, J. W. 1998. Oxidants and antioxidants in the pathogenesis of atherosclerosis: implications for the oxidized low density lipoprotein hypothesis. *Atherosclerosis*, 141, 1-15.
- HEINECKE, J. W., BAKER, L., ROSEN, H. & CHAIT, A. 1986. Superoxide-mediated modification of low density lipoprotein by arterial smooth muscle cells. *Journal of Clinical Investigation*, 77, 757-761.
- HEINECKE, J. W., KAWAMURA, M., SUZUKI, L. & CHAIT, A. 1993. Oxidation of low density lipoprotein by thiols: superoxide-dependent and -independent mechanisms. *J Lipid Res.*, 34, 2051-61.
- HEINECKE, J. W., ROSEN, H. & CHAIT, A. 1984. Iron and copper promote modification of low density lipoprotein by human arterial smooth muscle cells in culture. *J Clin Invest*, 74, 1890–4.
- HENRIKSEN, T., MAHONEY, E. M. & STEINBERG, D. 1981. Enhanced macrophage degradation of low density lipoprotein previously incubated with cultured endothelial cells: recognition by receptors for acetylated low density lipoproteins. *Proc Natl Acad Sci U S A*, 78, 6499-503.
- HENRIKSEN, T., MAHONEY, E. M. & STEINBERG, D. 1983. Enhanced macrophage degradation of biologically modified low density lipoprotein. *Arteriosclerosis*, *3*, 149-59.
- HENTZE, M. W., ROUAULT, T. A., CAUGHMAN, S. W., DANCIS, A., HARFORD, J. B. & KLAUSNER, R. D. 1987. A cis-acting element is necessary and sufficient for translational regulation of human ferritin expression in response to iron. *Proc Natl Acad Sci U S A*, 84, 6730-4.
- HERRINGTON, W., LACEY, B., SHERLIKER, P., ARMITAGE, J. & LEWINGTON, S. 2016. Epidemiology of Atherosclerosis and the Potential to Reduce the Global Burden of Atherothrombotic Disease. *Circ Res*, 118, 535-46.
- HESSLER, J. R., MOREL, D. W., LEWIS, L. J. & CHISOLM, G. M. 1983. Lipoprotein oxidation and lipoprotein-induced cytotoxicity. *Arteriosclerosis*, 3, 215-22.

- HESSLER, J. R., ROBERTSON, A. L., JR. & CHISOLM, G. M., 3RD 1979. LDL-induced cytotoxicity and its inhibition by HDL in human vascular smooth muscle and endothelial cells in culture. *Atherosclerosis*, 32, 213-29.
- HEYDECK, D., UPSTON, J. M., VIITA, H., YLÄ-HERTTUALA, S. & STOCKER, R. 2001. Oxidation of LDL by rabbit and human 15-lipoxygenase: Prevalence of nonenzymatic reactions. *Journal of Lipid Research*, 42, 1082-1088.
- HIMMELFARB, J., MCMENAMIN, M. E., LOSETO, G. & HEINECKE, J. W. 2001. Myeloperoxidase-catalyzed 3-chlorotyrosine formation in dialysis patients. *Free Radical Biology and Medicine*, 31, 1163-1169.
- HITSUMOTO, T., TAKAHASHI, M., LIZUKA, T. & SHIRAI, K. 2009. Clinical significance of serum 7-ketocholesterol concentrations on the progression of coronary atherosclerosis *Journal of atherosclerosis and thrombosis*, 16, 363-370.
- HOFF, H. F. & MORTON, R. E. 1985. Lipoproteins containing apo B extracted from human aortas. Structure and function. *Ann N Y Acad Sci*, 454, 183-94.
- HOFF, H. F. & O'NEIL, J. 1991. Lesion-derived low density lipoprotein and oxidized low density lipoprotein share a lability for aggregation, leading to enhanced macrophage degradation. *Arterioscler Thromb*, 11, 1209-22.
- HOFF, H. F., O'NEIL, J., WU, Z., HOPPE, G. & SALOMON, R. L. 2003. Phospholipid hydroxyalkenals: biological and chemical properties of specific oxidized lipids present in atherosclerotic lesions. *Arterioscler Thromb Vasc Biol*, 23, 275-82.
- HOLOPAINEN, J. M., ANGELOVA, M. I. & KINNUNEN, P. K. 2000. Vectorial budding of vesicles by asymmetrical enzymatic formation of ceramide in giant liposomes. *Biophys J*, 78, 830-8.
- HOMMA, Y. 2004. Predictors of atherosclerosis. J Atheroscler Thromb, 11, 265-70.
- HONARMAND EBRAHIMI, K., HAGEDOORN, P. L. & HAGEN, W. R. 2015. Unity in the biochemistry of the iron-storage proteins ferritin and bacterioferritin. *Chem Rev*, 115, 295-326.
- HOPGOOD, M. F., CLARK, M. G. & BALLARD, F. J. 1977. Inhibition of protein degradation in isolated rat hepatocytes. *Biochem J*, 164, 399-407.
- HORSLEY, E. T. M., BURKITT, M. J., JONES, C. M., PATTERSON, R. A., HARRIS, L. K., MOSS, N. J., DEL RIO, J. D. & LEAKE, D. S. 2007. Mechanism of the antioxidant to pro-oxidant switch in the behavior of dehydroascorbate during LDL oxidation by copper(II) ions. Archives of Biochemistry and Biophysics, 465, 303-314.
- HSIEH, C. C., YEN, M. H., YEN, C. H. & LAU, Y. T. 2001. Oxidized low density lipoprotein induces apoptosis via generation of reactive oxygen species in vascular smooth muscle cells. *Cardiovasc Res*, 49, 135-45.

- HULLEY, S., GRADY, D., BUSH, T., FURBERG, C., HERRINGTON, D., RIGGS, B. & VITTINGHOFF, E. 1998. Randomized trial of estrogen plus progestin for secondary prevention of coronary heart disease in postmenopausal women. Heart and Estrogen/progestin Replacement Study (HERS) Research Group. Jama, 280, 605-13.
- HUSSAIN, M. M. 2014. Intestinal lipid absorption and lipoprotein formation. *Curr Opin Lipidol*, 25, 200-6.
- HUUSKONEN, J., OLKKONEN, V. M., JAUHIAINEN, M. & EHNHOLM, C. 2001. The impact of phospholipid transfer protein (PLTP) on HDL metabolism. *Atherosclerosis*, 155, 269-81.
- ILHAN, F. & KALKANLI, S. T. 2015. Atherosclerosis and the role of immune cells. *World Journal of Clinical Cases : WJCC*, 3, 345-352.
- INDOLFI, C. 2002. Genetic factors in atherosclerosis: status and perspectives. *Europian Heart Journal supplemments*, 4 (supplement B), B14-B16.
- INOUE, T., UCHIDA, T., KAMISHIRADO, H., TAKAYANAGI, K., HAYASHI, T. & MOROOKA, S. 2001. Clinical significance of antibody against oxidized low density lipoprotein in patients with atherosclerotic coronary artery disease. *J Am Coll Cardiol*, 37, 775-9.
- INVESTIGATORS, T. H. A. H.-T. T. 2005. Effects of long-term vitamin E supplementation on cardiovascular events and cancer: a randomized controlled trial. *JAMA*, 293, 1338-1347.
- ISHIDA, Y., NAYAK, S., MINDELL, J. A. & GRABE, M. 2013. A model of lysosomal pH regulation. *The Journal of general physiology*, 141, 705-720.
- ITABE, H. 2003. Oxidized low-density lipoproteins: what is understood and what remains to be clarified. *Biol Pharm Bull*, 26, 1-9.
- ITABE, H., OBAMA, T. & KATO, R. 2011. The dynamics of oxidized LDL during atherogenesis. *Journal of lipids*.
- ITABE, H., TAKESHIMA E FAU IWASAKI, H., IWASAKI H FAU KIMURA, J., KIMURA J FAU - YOSHIDA, Y., YOSHIDA Y FAU - IMANAKA, T., IMANAKA T FAU - TAKANO, T. & TAKANO, T. 1994. A monoclonal antibody against oxidized lipoprotein recognizes foam cells in atherosclerotic lesions. Complex formation of oxidized phosphatidylcholines and polypeptides. J Biol Chem, 269, 15274-9.
- JACKSON, M. P. & HEWITT, E. W. 2016. Cellular proteostasis: degradation of misfolded proteins by lysosomes. *Essays in biochemistry*, 60, 173-180.
- JAYARAMAN, S., GANTZ, D. L. & GURSKY, O. 2011. Effects of phospholipase A(2) and its products on structural stability of human LDL: relevance to formation of LDL-derived lipid droplets. *Journal of lipid research*, 52, 549-557.

- JEROME, W. G. 2006. Advanced Atherosclerotic Foam Cell Formation Has Features of an Acquired Lysosomal Storage Disorder. *Rejuvenation Research*, 9 245-255.
- JEROME, W. G. 2010. Lysosomes, cholesterol and atherosclerosis. *Clinical lipidology*, 5, 853-865.
- JEROME, W. G., COX, B. E., GRIFFIN, E. E. & ULLERY, J. C. 2008. Lysosomal cholesterol accumulation inhibits subsequent hydrolysis of lipoprotein cholesteryl ester. *Microscopy and microanalysis : the official journal of Microscopy Society of America, Microbeam Analysis Society, Microscopical Society of Canada*, 14, 138-149.
- JEROME, W. G. & YANCEY, P. G. 2003. The role of microscopy in understanding atherosclerotic lysosomal lipid metabolism. *Microsc Microanal*, 9, 54-67.
- JESSUP, W., KRITHARIDES, L. & STOCKER, R. 2004. Lipid oxidation in atherogenesis: an overview. *Biochem Soc Trans*, 32, 134-8.
- JESSUP, W., RANKIN, S. M., DE WHALLEY, C. V., HOULT, J. R., SCOTT, J. & LEAKE, D. S. 1990. Alpha-tocopherol consumption during low-density-lipoprotein oxidation. *Biochem J*, 265, 399-405.
- JIALAL, I. & DEVARAJ, S. 1996. The role of oxidized low density lipoprotein in atherogenesis. *J Nutr*, 126, 1053S-7S.
- JIALAL, I., DEVARAJ, S., HUET, B. A. & TRABER, M. 1999. GISSI-Prevenzione trial. *The Lancet*, 354, 447-455.
- JIALAL, I., NORKUS, E. P., CRISTOL, L. & GRUNDY, S. M. 1991. beta-Carotene inhibits the oxidative modification of low-density lipoprotein. *Biochim Biophys Acta*, 1086, 134-8.
- JIALAL, I., VEGA, G. L. & GRUNDY, S. M. 1990. Physiologic levels of ascorbate inhibit the oxidative modification of low density lipoprotein. *Atherosclerosis*, 82, 185-91.
- JIN, W., TAKAGI, H., PANCORBO, B. & THEIL, E. C. 2001. "Opening" the ferritin pore for iron release by mutation of conserved amino acids at interhelix and loop sites. *Biochemistry*, 40, 7525-32.
- JONASSON, L., HOLM, J., SKALLI, O., BONDJERS, G. & HANSSON, G. K. 1986. Regional accumulations of T cells, macrophages, and smooth muscle cells in the human atherosclerotic plaque. *Arteriosclerosis*, 6, 131-8.
- KALYANARAMAN, B., DARLEY-USMAR, V., DAVIES, K. J. A., DENNERY, P. A., FORMAN, H. J., GRISHAM, M. B., MANN, G. E., MOORE, K., ROBERTS, L. J., 2ND & ISCHIROPOULOS, H. 2012. Measuring reactive oxygen and nitrogen species with fluorescent probes: challenges and limitations. *Free radical biology & medicine*, 52, 1-6.

- KAMANNA, V. S., GANJI, S. H., SHELKOVNIKOV, S., NORRIS, K. & VAZIRI, N. D. 2012. Iron Sucrose Promotes Endothelial Injury and Dysfunction and Monocyte Adhesion/Infiltration. *American Journal of Nephrology*, 35, 114-119.
- KANNEL, W. B., DOYLE, J. T., OSTFELD, A. M., JENKINS, C. D., KULLER, L., PODELL,
 R. N. & STAMLER, J. 1984. Optimal resources for primary prevention of artherosclerotic diseases. Atherosclerosis Study Group. *Circulation*, 70, 155a-205a.
- KANNEL, W. B., MARTHANA C, H., MCNAMARA, P. M. & GORDON TAVIA 1976. Menopause and Risk of Cardiovascular Disease: The Framingham Study. *Annals of Internal Medicine*, 85, 447-452.
- KASHIBA-IWATSUKI, M., YAMAGUCHI, M. & INOUE, M. 1996. Role of ascorbic acid in the metabolism of S-nitroso-glutathione. *FEBS Letters*, 389, 149-152.
- KATOUAH, H., CHEN, A., OTHMAN, I. & GIESEG, S. P. 2015. Oxidised low density lipoprotein causes human macrophage cell death through oxidant generation and inhibition of key catabolic enzymes. *Int J Biochem Cell Biol*, 67, 34-42.
- KAWAMURA, M., J.W, H. & CHAIT, A. 1994. Pathophysiological concentrations of glucose promote oxidative modification of low density lipoprotein by a superoxide-dependent pathway. *J Clin Invest*, 94, 771–8.
- KEANEY JR, J. F. 2000. Atherosclerosis: from lesion formation to plaque activation and endothelial dysfunction. *Molecular Aspects of Medicine*, 21, 99-166.
- KENAGY, R. D., MIN, S.-K., MULVIHILL, E. & CLOWES, A. W. 2011. A link between smooth muscle cell death and extracellular matrix degradation during vascular atrophy. *Journal of vascular surgery*, 54, 182-191.e24.
- KHAW, K. T., BINGHAM, S., WELCH, A., LUBEN, R., WAREHAM, N., OAKES, S. & DAY, N. 2001. Relation between plasma ascorbic acid and mortality in men and women in EPIC-Norfolk prospective study: a prospective population study. European Prospective Investigation into Cancer and Nutrition. *Lancet*, 357, 657-63.
- KHOO, J. C., MILLER, E., MCLOUGHLIN, P. & STEINBERG, D. 1988. Enhanced macrophage uptake of low density lipoprotein after self-aggregation. *Arteriosclerosis*, 348-358., 348-358.
- KIDANE, T. Z., SAUBLE, E. & LINDER, M. C. 2006. Release of iron from ferritin requires lysosomal activity. *American Journal of Physiology* 291.
- KIECHL, S., AICHNER, F., GERSTENBRAND, F., EGGER, G., MAIR, A., RUNGGER, G., SPOGLER, F., JAROSCH, E., OBERHOLLENZER, F. & WILLEIT, J. 1994. Body iron stores and presence of carotid atherosclerosis. Results from the Bruneck Study. *Arterioscler Thromb*, 14, 1625-30.
- KISS, R. S., MCMANUS, D. C., FRANKLIN, V., TAN, W. L., MCKENZIE, A., CHIMINI, G. & Y.L. MARCEL 2003. The lipidation by hepatocytes of human apolipoprotein A-

I occurs by both ABCA1-dependent and independent pathways J Biol Chem, 278, 10119-10127.

- KNOTT, H. M., BROWN, B. E., DAVIES, M. J. & DEAN, R. T. 2003. Glycation and glycoxidation of low-density lipoproteins by glucose and low-molecular mass aldehydes formation of modified and oxidized particles. *Eur. J. Biochem.*, 270, 3572–3582.
- KNUTSON, M. D., OUKKA, M., KOSS, L. M., AYDEMIR, F. & WESSLING-RESNICK, M. 2005. Iron release from macrophages after erythrophagocytosis is up-regulated by ferroportin 1 overexpression and down-regulated by hepcidin. *Proc Natl Acad Sci U S* A, 102, 1324-8.
- KOCKX, M. M. 1998. Apoptosis in the atherosclerotic plaque: quantitative and qualitative aspects. *Arterioscler Thromb Vasc Biol*, 18, 1519-22.
- KOCKX, M. M., DE MEYER, G. R., MUHRING, J., JACOB, W., BULT, H. & HERMAN, A. G. 1998. Apoptosis and related proteins in different stages of human atherosclerotic plaques. *Circulation*, 97, 2307-15.
- KOCKX, M. M. & HERMAN, A. G. 2000. Apoptosis in atherosclerosis: beneficial or detrimental? *Cardiovascular Research*, 45, 736-746.
- KOELWYN, G. J., CORR, E. M., ERBAY, E. & MOORE, K. J. 2018. Regulation of macrophage immunometabolism in atherosclerosis. *Nature immunology*, 19, 526-537.
- KOENIG, W., SUND, M., FROHLICH, M., FISCHER, H. G., LOWEL, H., DORING, A., HUTCHINSON, W. L. & PEPYS, M. B. 1999. C-Reactive protein, a sensitive marker of inflammation, predicts future risk of coronary heart disease in initially healthy middle-aged men: results from the MONICA (Monitoring Trends and Determinants in Cardiovascular Disease) Augsburg Cohort Study, 1984 to 1992. *Circulation*, 99, 237-42.
- KOHGO, Y., YOKOTA, M. & DRYSDALE, J. W. 1980. Differential turnover of rat liver isoferritins. *J Biol Chem 1980*, 255, 5195 –200.
- KOLODGIE, F. D., NARULA, J., BURKE, A. P., HAIDER, N., FARB, A., HUI-LIANG, Y., SMIALEK, J. & VIRMANI, R. 2000. Localization of apoptotic macrophages at the site of plaque rupture in sudden coronary death. *Am J Pathol*, 157, 1259-68.
- KONTUSH, A., MEYER, S., FINCKHL, B., KOHLSCHÜTTER, A. & BEISIEGEL, U. 1996. Alpha;-tocopherol as a reductant for Cu(II) in hitman lipoprotins: Triggering role in the initiation of lipoprotein oxidation. *Journal of Biological Chemistry*, 271, 11106-11112.
- KOPELMAN, P. G. 2000. Obesity as a medical problem. Nature 404, 635-649.
- KRAUSS, R. M. 2010. Lipoprotein subfractions and cardiovascular disease risk. *Curr Opin Lipidol*, 21, 305-11.
- KREUZALER, P. A., STANISZEWSKA, A. D., LI, W., OMIDVAR, N., KEDJOUAR, B., TURKSON, J., POLI, V., FLAVELL, R. A., CLARKSON, R. W. & WATSON, C. J. 2011. Stat3 controls lysosomal-mediated cell death in vivo. *Nat Cell Biol*, 13, 303-9.
- KRINSKY, N. I. 1989. Antioxidant functions of carotenoids. Free Radic Biol Med, 7, 617-35.
- KRISHNA, M. C., DEGRAFF, W., HANKOVSZKY, O. H., SÁR, C. P., KÁLAI, T., JEKŐ, J., RUSSO, A., MITCHELL, J. B. & HIDEG, K. 1998. Studies of Structure–Activity Relationship of Nitroxide Free Radicals and Their Precursors as Modifiers Against Oxidative Damage. *Journal of Medicinal Chemistry*, 41, 3477-3492.
- KRITHARIDES, L., JESSUP, W., GIFFORD, J. & DEAN, R. T. 1993. A Method for Defining the Stages of Low-Density Lipoprotein Oxidation by the Separation of Cholesterol and Cholesteryl Ester-Oxidation Products Using HPLC. *Analytical Biochemistry*, 213, 79-89.
- KRITHARIDES, L., UPSTON, J., JESSUP, W. & DEAN, R. T. 1998. Accumulation and metabolism of low density lipoprotein-derived cholesteryl linoleate hydroperoxide and hydroxide by macrophages. *The Journal of Lipid Research*, 39, 2394-2405.
- KRUSZEWSKI, M. 2003. Labile iron pool: the main determinant of cellular response to oxidative stress. *Mutat Res*, 531, 81-92.
- KUESTER, D., LIPPERT, H., ROESSNER, A. & KRUEGER, S. 2008. The cathepsin family and their role in colorectal cancer. *Pathol Res Pract*, 204, 491-500.
- KULAKSIZ, H., THEILIG, F., BACHMANN, S., GEHRKE, S. G., ROST, D., JANETZKO, A., CETIN, Y. & STREMMEL, W. 2005. The iron-regulatory peptide hormone hepcidin: expression and cellular localization in the mammalian kidney. *J Endocrinol*, 184, 361-70.
- KUNJATHOOR, V. V., FEBBRAIO, M., PODREZ, E. A., MOORE, K. J., ANDERSSON, L., KOEHN, S., RHEE, J. S., SILVERSTEIN, R., HOFF, H. F. & FREEMAN, M. W. 2002. Scavenger receptors class A-I/II and CD36 are the principal receptors responsible for the uptake of modified low density lipoprotein leading to lipid loading in macrophages. *J Biol Chem*, 277, 49982-8.
- KURZ, T., EATON, J. W. & BRUNK, U. T. 2011. The role of lysosomes in iron metabolism and recycling. *Int J Biochem Cell Biol*, 43, 1686-97.
- KURZ, T., TERMAN, A. & BRUNK, U. T. 2007. Autophagy, ageing and apoptosis: The role of oxidative stress and lysosomal iron. *Archives of Biochemistry and Biophysics*, 462, 220-230.
- KURZ, T., TERMAN, A., GUSTAFSSON, B. & BRUNK, U. T. 2008. Lysosomes and oxidative stress in aging and apoptosis. *Biochim Biophys Acta* 1780, 1291-303.
- KYGER, E. M., RILEY, D. J., SPILBURG, C. A. & LANGE, L. G. 1990. Pancreatic cholesterol esterases. 3. Kinetic characterization of cholesterol ester resynthesis by the pancreatic cholesterol esterases. *Biochemistry*, 29, 3853-8.

- LACHMANDAS, E., BOUTENS, L., RATTER, J. M., HIJMANS, A. & HOOIVELD, G. J. 2016. Microbial stimulation of different Toll-like receptor signalling pathways induces diverse metabolic programmes in human monocytes. 2, 16246.
- LAMB, D. J. & LEAKE, D. S. 1994a. Acidic pH enables caeruloplasmin to catalyse the modification of low-density lipoprotein. *FEBS Letters*, 338 122-126.
- LAMB, D. J. & LEAKE, D. S. 1994b. Acidic pH enables caeruloplasmin to catalyse the modification of low-density lipoprotein. *FEBS Lett*, 338, 122-6.
- LAMB, D. J. & LEAKE, D. S. 1994c. Iron released from transferrin at acidic pH can catalyse the oxidation of low density lipoprotein. *FEBS Lett*, 352, 15-8.
- LAMB, D. J., MITCHINSON, M. J. & LEAKE, D. S. 1995. Transition metal ions within human atherosclerotic lesions can catalyse the oxidation of low density lipoprotein by macrophages. *FEBS Lett*, 374, 12-6.
- LARA-GUZMÁN, O. J., GIL-IZQUIERDO, Á., MEDINA, S., OSORIO, E., ÁLVAREZ-QUINTERO, R., ZULUAGA, N., OGER, C., GALANO, J.-M., DURAND, T. & MUÑOZ-DURANGO, K. 2018. Oxidized LDL triggers changes in oxidative stress and inflammatory biomarkers in human macrophages. *Redox Biology*, 15, 1-11.
- LASKAR, A., GHOSH, M., KHATTAK, S. I., LI, W. & YUAN, X. M. 2012. Degradation of superparamagnetic iron oxide nanoparticle-induced ferritin by lysosomal cathepsins and related immune response. *Nanomedicine (Lond)*, 7, 705-17.
- LEAKE, D. S. 1997. Does an acidic pH explain why low density lipoprotein is oxidised in atherosclerotic lesions? *Atherosclerosis*, 129, 149-57.
- LEAKE, D. S. & PETERS, T. J. 1981. Proteolytic degradation of low density lipoproteins by arterial smooth muscle cells: The role of individual cathepsins. *Biochimica et Biophysica Acta (BBA) Lipids and Lipid Metabolism*, 664, 108-116.
- LEAKE, D. S. & RANKIN, S. M. 1990. The oxidative modification of low-density lipoproteins by macrophages. *Biochemical Journal*, 270, 741-748.
- LEE, D. H., FOLSOM, A. R., HARNACK, L., HALLIWELL, B. & JACOBS, D. R., JR. 2004. Does supplemental vitamin C increase cardiovascular disease risk in women with diabetes? *Am J Clin Nutr*, 80, 1194-200.
- LEE, F. Y., LEE, T. S., PAN, C. C., HUANG, A. L. & CHAU, L. Y. 1998. Colocalization of iron and ceroid in human atherosclerotic lesions. *Atherosclerosis*, 138, 281-8.
- LEE, S., BIRUKOV, K. G., ROMANOSKI, C. E., SPRINGSTEAD, J. R., LUSIS, A. J. & BERLINER, J. A. 2012. Role of phospholipid oxidation products in atherosclerosis. *Circ Res*, 111, 778-99.
- LEE, T. S., SHIAO, M. S., PAN, C. C. & L.Y, C. 1999. Iron-deficient diet reduces atherosclerotic lesions in apoE-deficient mice. *Circulation* 99, 1222-9.

- LEEUWENBURGH, C., RASMUSSEN, J. E., HSU, F. F., MUELLER, D. M., PENNATHUR, S. & HEINECKE, J. W. 1997. Mass spectrometric quantification of markers for protein oxidation by tyrosyl radical, copper, and hydroxyl radical in low density lipoprotein isolated from human atherosclerotic plaques. *J Biol Chem*, 272, 3520-6.
- LEGEIN, B., TEMMERMAN, L., BIESSEN, E. A. & LUTGENS, E. 2013. Inflammation and immune system interactions in atherosclerosis. *Cell Mol Life Sci*, 70, 3847-69.
- LEITINGER, N. 2003. Cholesteryl ester oxidation products in atherosclerosis. *Molecular* Aspects of Medicine, 24, 239-250.
- LEIVA, A., VERDEJO, H., BENITEZ, M. L., MARTINEZ, A., BUSSO, D. & RIGOTTI, A. 2011. Mechanisms regulating hepatic SR-BI expression and their impact on HDL metabolism. *Atherosclerosis*, 217, 299-307.
- LEOPOLD, J. A. & LOSCALZO, J. 2008. Oxidative mechanisms and atherothrombotic cardiovascular disease. *Drug discovery today. Therapeutic strategies*, 5, 5-13.
- LEVITAN, I., VOLKOV, S. & SUBBAIAH, P. V. 2010. Oxidized LDL: diversity, patterns of recognition, and pathophysiology. *Antioxid Redox Signal*, 13, 39-75.
- LEWINGTON, S., CLARKE, R., QIZILBASH, N., PETO, R. & COLLINS, R. 2002. Agespecific relevance of usual blood pressure to vascular mortality: a meta-analysis of individual data for one million adults in 61 prospective studies. *Lancet*, 360, 1903-13.
- LEWINGTON, S., WHITLOCK, G., CLARKE, R., SHERLIKER, P., EMBERSON, J., HALSEY, J., QIZILBASH, N., PETO, R. & COLLINS, R. 2007. Blood cholesterol and vascular mortality by age, sex, and blood pressure: a meta-analysis of individual data from 61 prospective studies with 55,000 vascular deaths. *Lancet*, 370, 1829-39.
- LI, H., CYBULSKY, M. I., GIMBRONE, M. A., JR. & LIBBY, P. 1993. An atherogenic diet rapidly induces VCAM-1, a cytokine-regulatable mononuclear leukocyte adhesion molecule, in rabbit aortic endothelium. *Arterioscler Thromb*, 13, 197-204.
- LI, M., LIN, J., WANG, Z., HE, S., MA, X. & LI, D. 2010. Oxidized low-density lipoproteininduced proinflammatory cytokine response in macrophages are suppressed by CD4CD25(+)Foxp3(+) regulatory T cells through downregulating toll like receptor 2mediated activation of NF-kappaB. *Cell Physiol Biochem*, 25, 649-56.
- LI, W. G., ZHANG, X. Y., WU, Y. J., GAO, M. T. & ZHENG, R. L. 2006. The relationship between structure and antioxidative activity of piperidine nitroxides. *J Pharm Pharmacol*, 58, 941-9.
- LIAO, F., BERLINER, J. A., MEHRABIAN, M., NAVAB, M., DEMER, L. L., LUSIS, A. J. & FOGELMAN, A. M. 1991. Minimally modified LDL is biologically active in vivo in mice. J. Clin. Invest., 87 2253-2257.
- LIBBY, P. 2002. Inflammation in atherosclerosis. *Nature* 420, 19.

- LIBBY, P. 2012. Inflammation in atherosclerosis. *Arteriosclerosis, thrombosis, and vascular biology*, 32, 2045-2051.
- LIBBY, P. & GOLDBERG, A. L. 1978. Leupeptin, a protease inhibitor, decreases protein degradation in normal and diseased muscles. *Science*, 199, 534-6.
- LIBBY, P., RIDKER, P. M., HANSSON, G. K. & LEDUCQ TRANSATLANTIC NETWORK ON, A. 2009. Inflammation in atherosclerosis: from pathophysiology to practice. *Journal of the American College of Cardiology*, 54, 2129-2138.
- LIBBY, P., RIDKER, P. M. & MASERI, A. 2002. Inflammation and atherosclerosis. *Circulation*, 105, 1135-43.
- LIEBLER, D. C. 1993. The role of metabolism in the antioxidant function of vitamin E. *Crit Rev Toxicol*, 23, 147-69.
- LINKE, M., GORDON, R. E., BRILLARD, M., LECAILLE, F., LALMANACH, G. & BROMME, D. 2006. Degradation of apolipoprotein B-100 by lysosomal cysteine cathepsins. *Biol Chem*, 387, 1295-303.
- LIU, J., SUKHOVA, G. K., SUN, J. S., XU, W. H., LIBBY, P. & SHI, G. P. 2004. Lysosomal cysteine proteases in atherosclerosis. *Arterioscler Thromb Vasc Biol*, 24, 1359-66.
- LIU, J., THEWKE, D. P., SU, Y. R., LINTON, M. F., FAZIO, S. & SINENSKY, M. S. 2005. Reduced macrophage apoptosis is associated with accelerated atherosclerosis in lowdensity lipoprotein receptor-null mice. *Arterioscler Thromb Vasc Biol*, 25, 174-9.
- LIU, L., LU, Y., MARTINEZ, J., BI, Y., LIAN, G., WANG, T., MILASTA, S., WANG, J., YANG, M., LIU, G., GREEN, D. R. & WANG, R. 2016. Proinflammatory signal suppresses proliferation and shifts macrophage metabolism from Myc-dependent to HIF1α-dependent. *Proceedings of the National Academy of Sciences of the United States of America*, 113, 1564-1569.
- LONN, E., YUSUF, S., HOOGWERF, B., POGUE, J., YI, Q., ZINMAN, B., BOSCH, J., DAGENAIS, G., MANN, J. F. & GERSTEIN, H. C. 2002. Effects of vitamin E on cardiovascular and microvascular outcomes in high-risk patients with diabetes: results of the HOPE study and MICRO-HOPE substudy. *Diabetes Care*, 25, 1919-27.
- LOVREN, F., TEOH, H. & VERMA, S. 2015. Obesity and Atherosclerosis: Mechanistic Insights. *Canadian Journal of Cardiology*, 31, 177-183.
- LU, M. & GURSKY, O. 2013. Aggregation and fusion of low-density lipoproteins in vivo and in vitro. *Biomolecular concepts*, 4, 501-518.
- LÜBKE, T., LOBEL, P. & SLEAT, D. E. 2009. Proteomics of the lysosome. *Biochimica et biophysica acta*, 1793, 625-635.
- LUO, Z., CHEN, Y., CHEN, S., WELCH, W. J., ANDRESEN, B. T., JOSE, P. A. & WILCOX, C. S. 2009. Comparison of inhibitors of superoxide generation in vascular smooth muscle cells. *Br J Pharmacol*, 157, 935-43.

LUSIS, A. J. 2000. Atherosclerosis. Nature, 407, 233-241.

- LUTGENS, S. P., CLEUTJENS, K. B., DAEMEN, M. J. & HEENEMAN, S. 2007. Cathepsin cysteine proteases in cardiovascular disease. *Faseb j*, 21, 3029-41.
- LV, H. & SHANG, P. 2018. The significance, trafficking and determination of labile iron in cytosol, mitochondria and lysosomes. *Metallomics*, 10, 899-916.
- LYNCH, S. M., CAMPIONE, A. L. & MOORE, M. K. 2000. Plasma thiols inhibit hemindependent oxidation of human low-density lipoprotein. *Biochimica et Biophysica Acta* (*BBA*) - *Molecular and Cell Biology of Lipids*, 1485, 11-22.
- LYNCH, S. M. & FREI, B. 1995. Reduction of copper, but not iron, by human low density lipoprotein (LDL). Implications for metal ion-dependent oxidative modification of LDL. *J Biol Chem*, 270, 5158-63.
- LYNCH, S. M., GAZIANO, J. M. & FREI, B. 1996. Ascorbic acid and atherosclerotic cardiovascular disease. *Subcell Biochem*, 25, 331-67.
- LYONS, M. A. & BROWN, A. J. 1999. 7-Ketocholesterol. *The International Journal of Biochemistry & Cell Biology*, 31, 369-375.
- MADAMANCHI, N. R., VENDROV, A. & RUNGE, M. S. 2005. Oxidative stress and vascular disease. *Arterioscler Thromb Vasc Biol*, 25, 29-38.
- MAHLEY, R. W. & JI, Z. S. 1999. Remnant lipoprotein metabolism: key pathways involving cell-surface heparan sulfate proteoglycans and apolipoprotein E. *J Lipid Res*, 40, 1-16.
- MAIOLINO, G., ROSSITTO, G., CAIELLI, P., BISOGNI, V., ROSSI, G. P. & CALO, L. A. 2013. The role of oxidized low-density lipoproteins in atherosclerosis: the myths and the facts. *Mediators Inflamm*, 2013, 714653.
- MALLAT, Z., HUGEL, B., OHAN, J., LESECHE, G., FREYSSINET, J. M. & TEDGUI, A. 1999. Shed membrane microparticles with procoagulant potential in human atherosclerotic plaques: a role for apoptosis in plaque thrombogenicity. *Circulation*, 99, 348-53.
- MALLAT, Z. & TEDGUI, A. 2000. Apoptosis in the vasculature: mechanisms and functional importance. *British Journal of Pharmacology*, 130, 947-962.
- MANNING-TOBIN, J. J., MOORE, K. J., SEIMON, T. A., BELL, S. A., SHARUK, M., ALVAREZ-LEITE, J. I., DE WINTHER, M. P. J., TABAS, I. & FREEMAN, M. W. 2009. Loss of SR-A and CD36 activity reduces atherosclerotic lesion complexity without abrogating foam cell formation in hyperlipidemic mice. *Arteriosclerosis, thrombosis, and vascular biology*, 29, 19-26.
- MANSON, J. E., HSIA, J., JOHNSON, K. C., ROSSOUW, J. E., ASSAF, A. R., LASSER, N. L., TREVISAN, M., BLACK, H. R., HECKBERT, S. R., DETRANO, R., STRICKLAND, O. L., WONG, N. D., CROUSE, J. R., STEIN, E. & CUSHMAN, M.

2003. Estrogen plus progestin and the risk of coronary heart disease. *N Engl J Med*, 349, 523-34.

- MAOR, I., HAYEK, T., COLEMAN, R. & AVIRAM, M. 1997. Plasma LDL oxidation leads to its aggregation in the atherosclerotic apolipoprotein E-deficient mice. *Arterioscler Thromb Vasc Biol*, 17, 2995-3005.
- MARATHE, S., KURIAKOSE, G., WILLIAMS, K. J. & TABAS, I. 1999. Sphingomyelinase, an enzyme implicated in atherogenesis, is present in atherosclerotic lesions and binds to specific components of the subendothelial extracellular matrix. *Arterioscl. Thromb. Vasc. Biol.*, 19, 2648-2658.
- MARDUEL, M., CARRIE, A., SASSOLAS, A., DEVILLERS, M., CARREAU, V., DI FILIPPO, M., ERLICH, D., ABIFADEL, M., MARQUES-PINHEIRO, A., MUNNICH, A., JUNIEN, C., BOILEAU, C., VARRET, M. & RABES, J. P. 2010. Molecular spectrum of autosomal dominant hypercholesterolemia in France. *Hum Mutat*, 31, E1811-24.
- MARENBERG, M. E., RISCH, N., BERKMAN, L. F., FLODERUS, B. & DE FAIRE, U. 1994. Genetic susceptibility to death from coronary heart disease in a study of twins. *N Engl J Med*, 330, 1041-6.
- MARKWELL, M. A., HAAS, S. M., BIEBER, L. L. & TOLBERT, N. E. 1978. A modification of the Lowry procedure to simplify protein determination in membrane and lipoprotein samples. *Anal Biochem*, 87, 206-10.
- MARTIN, S. J., REUTELINGSPERGER, C. P., MCGAHON, A. J., RADER, J. A., VAN SCHIE, R. C., LAFACE, D. M. & GREEN, D. R. 1995. Early redistribution of plasma membrane phosphatidylserine is a general feature of apoptosis regardless of the initiating stimulus: inhibition by overexpression of Bcl-2 and Abl. *J Exp Med*, 182, 1545-56.
- MARTINET, W., SCHRIJVERS, D. M. & DE MEYER, G. R. 2011. Necrotic cell death in atherosclerosis. *Basic Res Cardiol*, 106, 749-60.
- MAZZONE, T., CHAIT, A. & PLUTZKY, J. 2008. Cardiovascular disease risk in type 2 diabetes mellitus: insights from mechanistic studies. *Lancet*, 371, 1800-9.
- MCGILL, H. C., JR., MCMAHAN, C. A., HERDERICK, E. E., ZIESKE, A. W., MALCOM, G. T., TRACY, R. E. & STRONG, J. P. 2002. Obesity accelerates the progression of coronary atherosclerosis in young men. *Circulation*, 105, 2712-8.
- MCLAREN, J. E., MICHAEL, D. R., ASHLIN, T. G. & RAMJI, D. P. 2011. Cytokines, macrophage lipid metabolism and foam cells: Implications for cardiovascular disease therapy. *Progress in Lipid Research*, 50, 331-347.
- MEGURO, R., ASANO, Y., ODAGIRI, S., LI, C., IWATSUKI, H. & SHOUMURA, K. 2005. The presence of ferric and ferrous iron in the nonheme iron store of resident macrophages in different tissues and organs: histochemical demonstrations by the perfusion-Perls and -Turnbull methods in the rat. *Arch Histol Cytol*, 68, 171-83.

- MEHTA, J. L. & LI, D. 2005. Oxidized LDL, a critical factor in atherogenesis. *Cardiovascular Research*, 68, 353-354.
- MERCHED, A. J., KO, K., K.H, G., C.N, S. & CHAN, L. 2008. Atherosclerosis: evidence for impairment of resolution of vascular inflammation governed by specific lipid mediators. *FASEB J* 22, 3595–606.
- MERLE, U., FEIN, E., GEHRKE, S. G., STREMMEL, W. & KULAKSIZ, H. 2007. The iron regulatory peptide hepcidin is expressed in the heart and regulated by hypoxia and inflammation. *Endocrinology*, 148, 2663-8.
- MEYDANI, M. 2001. Vitamin E and atherosclerosis: beyond prevention of LDL oxidation. J *Nutr*, 131, 366s-8s.
- MEYERS, D. G., JENSEN, K. C. & MENITOVE, J. E. 2002. A historical cohort study of the effect of lowering body iron through blood donation on incident cardiac events. *Transfusion.*, 42.
- MIHAYLOVA, B., EMBERSON, J., BLACKWELL, L., KEECH, A., SIMES, J., BARNES, E. H., VOYSEY, M., GRAY, A., COLLINS, R. & BAIGENT, C. 2012. The effects of lowering LDL cholesterol with statin therapy in people at low risk of vascular disease: meta-analysis of individual data from 27 randomised trials. *Lancet (London, England)*, 380, 581-590.
- MILLER, L. L., MILLER, S. C., TORTI, S. V., TSUJI, Y. & TORTI, F. M. 1991. Ironindependent induction of ferritin H chain by tumor necrosis factor. . *Proc Natl Acad Sci U S A.*, 88, 4946-50.

 MINDELL, J. A. 2012. Lysosomal acidification mechanisms. *Annu Rev Physiol*, 74, 69-86.
 MINOTTI, G. & AUST, S. D. 1987. The role of iron in the initiation of lipid peroxidation. *Chem Phys Lipids*, 44, 191-208.

- MITCHINSON, M. J. 1982. Insoluble lipids in human atherosclerotic plaques. *Atherosclerosis*, 45, 11-15.
- MITCHINSON, M. J., HOTHERSALL, D. C., BROOKS, P. N. & DE BURBURE, C. Y. 1985. The distribution of ceroid in human atherosclerosis. *J Pathol*, 145, 177-83.
- MONS, U., MÜEZZINLER, A., GELLERT, C., SCHÖTTKER, B., ABNET, C. C., BOBAK, M., DE GROOT, L., FREEDMAN, N. D., JANSEN, E., KEE, F., KROMHOUT, D., KUULASMAA, K., LAATIKAINEN, T., O'DOHERTY, M. G., BUENO-DE-MESQUITA, B., ORFANOS, P., PETERS, A., VAN DER SCHOUW, Y. T., WILSGAARD, T., WOLK, A., TRICHOPOULOU, A., BOFFETTA, P. & BRENNER, H. 2015. Impact of smoking and smoking cessation on cardiovascular events and mortality among older adults: meta-analysis of individual participant data from prospective cohort studies of the CHANCES consortium. *BMJ : British Medical Journal*, 350.
- MOORE, K. J. & FREEMAN, M. W. 2006. Scavenger receptors in atherosclerosis: beyond lipid uptake. *Arterioscler Thromb Vasc Biol*, 26, 1702-11.

- MOORE, K. J., SHEEDY, F. J. & FISHER, E. A. 2013. Macrophages in atherosclerosis: a dynamic balance. *Nature reviews. Immunology*, 13, 709-721.
- MOORE, K. J. & TABAS, I. 2011. Macrophages in the pathogenesis of atherosclerosis. *Cell*, 145, 341-55.
- MOORE, M., FOLSOM, A. R., BARNES, R. W. & ECKFELDT, J. H. 1995. No Association between Serum Ferritin and Asymptomatic Carotid Atherosclerosis - the Atherosclerosis Risk in Communities (Aric) Study. *American Journal of Epidemiology*, 141, 719-723.
- MOREL, D. W., DE LA LLERA-MOYA, M. & FRIDAY, K. E. 1994. Treatment of cholesterol-fed rabbits with dietary vitamins E and C inhibits lipoprotein oxidation but not development of atherosclerosis. *J Nutr*, 124, 2123-30.
- MOREL, D. W., HESDER, J. R. & CHISOLM, G. M. 1983. Low density lipoprotein cytotoxicity induced by free radical peroxidation of lipid. *J. Lipid Res*, 24, 1070-1076.
- MORGAN, J. & LEAKE, D. S. 1993. Acidic pH increases the oxidation of LDL by macrophages. *FEBS Lett*, 333, 275-9.
- MORGAN, J. & LEAKE, D. S. 1995. Oxidation of low density lipoprotein by iron or copper at acidic pH. *J Lipid Res*, 36, 2504-12.
- MORRISETT, J. D. 2000. The role of lipoprotein[a] in atherosclerosis. *Curr Atheroscler Rep*, 2, 243-50.
- MOSSER, D. M. & EDWARDS, J. P. 2008. Exploring the full spectrum of macrophage activation. *Nature Reviews Immunology*, 8, 958-969.
- MOZURAITYTE, R., RUSTAD, T. & STORRØ, I. 2008. The Role of Iron in Peroxidation of Polyunsaturated Fatty Acids in Liposomes. *Journal of Agricultural and Food Chemistry*, 56, 537-543.
- MUKHOPADHYAY, C. K., EHRENWALD, E. & FOX, P. L. 1996. Ceruloplasmin enhances smooth muscle cell- and endothelial cell-mediated low density lipoprotein oxidation by a superoxide-dependent mechanism. *J Biol Chem*, 271, 14773-8.
- MUKHOPADHYAY, C. K. & FOX, P. L. 1998. Ceruloplasmin copper induces oxidant damage by a redox process utilizing cell-derived superoxide as reductant. *Biochemistry*, 37, 14222-9.
- MULLER, W. A. 2014. How endothelial cells regulate transmigration of leukocytes in the inflammatory response. *The American journal of pathology*, 184, 886-896.
- MUNOZ-BRAVO, C., GUTIERREZ-BEDMAR, M., GOMEZ-ARACENA, J., GARCIA-RODRIGUEZ, A. & NAVAJAS, J. F. 2013. Iron: protector or risk factor for cardiovascular disease? Still controversial. *Nutrients*, 5, 2384-404.
- MUNOZ, M., VILLAR, I. & GARCIA-ERCE, J. A. 2009. An update on iron physiology. *World J Gastroenterol*, 15, 4617-26.

- MUNTANE-RELAT, J., OURLIN, J. C., DOMERGUE, J. & MAUREL, P. 1995. Differential effects of cytokines on the inducible expression of CYP1A1, CYP1A2, and CYP3A4 in human hepatocytes in primary culture. *Hepatology*, 22, 1143-53.
- MURDOCH, S. J. & BRECKENRIDGE, W. C. 1996. Effect of lipid transfer proteins on lipoprotein lipase induced transformation of VLDL and HDL. *Biochim Biophys Acta* 1303 222-232.
- NAMAZI, M. R. 2009. Hypothesis: does propranolol afford protection against atherosclerosis? *Niger J Med*, 18, 339.
- NAPOLITANO, G., JOHNSON, J. L., HE, J., ROCCA, C. J., MONFREGOLA, J., PESTONJAMASP, K., CHERQUI, S. & CATZ, S. D. 2015. Impairment of chaperonemediated autophagy leads to selective lysosomal degradation defects in the lysosomal storage disease cystinosis. *EMBO Mol Med*, 7, 158-74.
- NEMETH, E. & GANZ, T. 2006. Regulation of Iron Metabolism by Hepcidin. *Annual Review* of Nutrition, 26, 323-342.
- NEMETH, E., RIVERA, S., GABAYAN, V., KELLER, C., TAUDORF, S., PEDERSEN, B. K. & GANZ, T. 2004a. IL-6 mediates hypoferremia of inflammation by inducing the synthesis of the iron regulatory hormone hepcidin. *J Clin Invest*, 113, 1271-6.
- NEMETH, E., TUTTLE, M. S., POWELSON, J., VAUGHN, M. B., DONOVAN, A., WARD, D. M., GANZ, T. & KAPLAN, J. 2004b. Hepcidin regulates cellular iron efflux by binding to ferroportin and inducing its internalization. *Science*, 306, 2090-3.
- NEUZIL, J., THOMAS, S. R. & STOCKER, R. 1997. Requirement for, promotion, or inhibition by alpha-tocopherol of radical-induced initiation of plasma lipoprotein lipid peroxidation. *Free Radic Biol Med*, 22, 57-71.
- NICHOLS, M., TOWNSEND, N., LUENGO-FERNANDEZ, R., LEAL, J., GRAY, A., SCARBOROUGH, P. & RAYNER, M. 2012. European Cardiovascular Disease Statistics 2012. Brussells: BHF.
- NIKI, E., YOSHIDA, Y., SAITO, Y. & NOGUCHI, N. 2005. Lipid peroxidation: Mechanisms, inhibition, and biological effects. *Biochemical and Biophysical Research Communications*, 338, 668-676.
- NILSSON, J., REGNSTROM, J., FROSTEGARD, J. & STIKO, A. 1992. Lipid oxidation and atherosclerosis. *Herz*, 17, 263-9.
- NOGUCHI, N., NUMANO, R., KANEDA, H. & NIKI, E. 1998. Oxidation of lipids in low density lipoprotein particles. *Free Radic Res*, 29, 43-52.
- OHARA, Y., PETERSON, T. E. & HARRISON, D. G. 1993. Hypercholesterolemia increases endothelial superoxide anion production. *J Clin Invest*, 91, 2546-51.

- OHKUMA, S., MORIYAMA, Y. & TAKANO, T. 1982. Identification and characterization of a proton pump on lysosomes by fluorescein-isothiocyanate-dextran fluorescence. *Proc Natl Acad Sci U S A*, 79, 2758-62.
- OHKUMA, S. & POOLE, B. 1978. Fluorescence probe measurement of the intralysosomal pH in living cells and the perturbation of pH by various agents. *Proceedings of the National Academy of Sciences of the United States of America*, 75, 3327-3331.
- OHMAN, M. K., WRIGHT, A. P., WICKENHEISER, K. J., LUO, W., RUSSO, H. M. & EITZMAN, D. T. 2010. Monocyte chemoattractant protein-1 deficiency protects against visceral fat-induced atherosclerosis. *Arterioscler Thromb Vasc Biol*, 30, 1151-8.
- OLESNEVICH, M. E., FANELLI KUCZMARSKI, M., MASON, M., FANG, C., ZONDERMAN, A. B. & EVANS, M. K. 2012. Serum ferritin levels associated with increased risk for developing CHD in a low-income urban population. *Public health nutrition*, 15, 1291-1298.
- ÖLLINGER, K. & BRUNK, U. T. 1995 Cellular injury induced by oxidative stress is mediated through lysosomal damage. *Free Radic. Biol. Med.*, 19, 565-574.
- OLSEN, M. H., ANGELL, S. Y., ASMA, S., BOUTOUYRIE, P., BURGER, D., CHIRINOS, J. A., DAMASCENO, A., DELLES, C., GIMENEZ-ROQUEPLO, A. P., HERING, D., LOPEZ-JARAMILLO, P., MARTINEZ, F., PERKOVIC, V., RIETZSCHEL, E. R., SCHILLACI, G., SCHUTTE, A. E., SCUTERI, A., SHARMAN, J. E., WACHTELL, K. & WANG, J. G. 2016. A call to action and a lifecourse strategy to address the global burden of raised blood pressure on current and future generations: the Lancet Commission on hypertension. *Lancet*, 388, 2665-2712.
- OORNI, K. & KOVANEN, P. T. 2009. Lipoprotein modification by secretory phospholipase A(2) enzymes contributes to the initiation and progression of atherosclerosis. *Curr Opin Lipidol*, 20, 421-7.
- OORNI, K., PENTIKAINEN, M. O., ALA-KORPELA, M. & KOVANEN, P. T. 2000. Aggregation, fusion, and vesicle formation of modified low density lipoprotein particles: molecular mechanisms and effects on matrix interactions. *J Lipid Res*, 41, 1703-14.
- OORNI, K., SNECK, M., BROMME, D., PENTIKAINEN, M. O., LINDSTEDT, K. A., MAYRANPAA, M., AITIO, H. & KOVANEN, P. T. 2004. Cysteine protease cathepsin F is expressed in human atherosclerotic lesions, is secreted by cultured macrophages, and modifies low density lipoprotein particles in vitro. *J Biol Chem*, 279, 34776-84.
- OSGANIAN, S. K., STAMPFER, M. J., RIMM, E., SPIEGELMAN, D., HU, F. B., MANSON, J. E. & WILLETT, W. C. 2003. Vitamin C and risk of coronary heart disease in women. *J Am Coll Cardiol*, 42, 246-52.
- OSKUI, P. M., FRENCH, W. J., HERRING, M. J., MAYEDA, G. S., BURSTEIN, S. & KLONER, R. A. 2013. Testosterone and the cardiovascular system: a comprehensive review of the clinical literature. *J Am Heart Assoc*, 2, e000272.

- OUIMET, M., FRANKLIN, V., MAK, E., LIAO, X., TABAS, I. & MARCEL, Y. L. 2011. Autophagy regulates cholesterol efflux from macrophage foam cells via lysosomal acid lipase. *Cell Metab*, 13, 655-67.
- PAFFEN, E. & DEMAAT, M. P. M. 2006. C-reactive protein in atherosclerosis: A causal factor? *Cardiovascular Research*, 71, 30-39.
- PAFFENBARGER, R. S., JR., HYDE, R. T., WING, A. L., LEE, I. M., JUNG, D. L. & KAMPERT, J. B. 1993. The association of changes in physical-activity level and other lifestyle characteristics with mortality among men. *N Engl J Med*, 328, 538-45.
- PAGANGA, G., RICE-EVANS, C., RULE, R. & LEAKE, D. 1992. The interaction between ruptured erythrocytes and low-density lipoproteins. *FEBS Lett*, 303, 154-8.
- PANG, J. H., JIANG, M. J., CHEN, Y. L., WANG, F. W., WANG, D. L., CHU, S. H. & CHAU, L. Y. 1996. Increased ferritin gene expression in atherosclerotic lesions. *J Clin Invest*, 97, 2204-12.
- PARTHASARATHY, S., RAGHAVAMENON, A., GARELNABI, M. O., AND & SANTANAM, N. 2010. Oxidized low-density lipoprotein. *Methods Mol Biol*, 610, 403–417.
- PARTHASARATHY, S., YOUNG, S. G., WITZTUM, J. L., PITTMAN, R. C. & STEINBERG, D. 1986. Probucol inhibits oxidative modification of low density lipoprotein. *J Clin Invest*, 77, 641-4.
- PATEL, K., CHEN, Y., DENNEHY, K., BLAU, J., CONNORS, S., MENDONCA, M., TARPEY, M., KRISHNA, M., MITCHELL, J. B., WELCH, W. J. & WILCOX, C. S. 2006. Acute antihypertensive action of nitroxides in the spontaneously hypertensive rat. *Am J Physiol Regul Integr Comp Physiol*, 290, R37-43.
- PATTERSON, R. A., HORSLEY, E. T. & LEAKE, D. S. 2003a. Prooxidant and antioxidant properties of human serum ultrafiltrates toward LDL: important role of uric acid. *J Lipid Res*, 44, 512-21.
- PATTERSON, R. A., LAMB, D. J. & LEAKE, D. S. 2003b. Mechanisms by which cysteine can inhibit or promote the oxidation of low density lipoprotein by copper. *Atherosclerosis*, 169, 87-94.
- PATTERSON, R. A. & LEAKE, D. S. 1998. Human serum, cysteine and histidine inhibit the oxidation of low density lipoprotein less at acidic pH. *FEBS Letters*, 434, 317-321.
- PECH, M. A., MYARA, I., VEDIE, B. & MOATTI, N. 1992. [Modified LDL and atherosclerosis. Nature of modifications. Physicochemical and biological properties]. *Ann Biol Clin (Paris)*, 50, 213-27.
- PEFFER, K., DEN HEIJER, M., HOLEWIJN, S., DE GRAAF, J., SWINKELS, D. W., VERBEEK, A. L. M. & ATSMA, F. 2013. The effect of frequent whole blood donation on ferritin, hepcidin, and subclinical atherosclerosis. *Transfusion*, 53, 1468-1474.

- PELUZIO, M. C., HOMEM, A. P., CESAR, G. C., AZEVEDO, G. S., AMORIM, R., CARA, D. C., SALIBA, H., VIEIRA, E. C., ARANTES, R. E. & ALVAREZ-LEITE, J. 2001. Influences of alpha-tocopherol on cholesterol metabolism and fatty streak development in apolipoprotein E-deficient mice fed an atherogenic diet. *Braz J Med Biol Res*, 34, 1539-45.
- PENTIKAINEN, M. O., LEHTONEN, E. M. & KOVANEN, P. T. 1996. Aggregation and fusion of modified low density lipoprotein. *J Lipid Res*, 37, 2638-49.
- PENTIKIIINEN, M., LEHTONEN, E. M. P. & KOVANE, P. T. 1996. Aggregation and fusion of modified low density lipoprotein. *Journal of Lipid Research*, 37, 2638-2649.
- PERSSON, H. L., NILSSON, K. J. & BRUNK, U. T. 2001a. Novel cellular defenses against iron and oxidation: ferritin and autophagocytosis preserve lysosomal stability in airway epithelium. *Redox Rep.*, 6, 57-63.
- PERSSON, H. L., SVENSSON, A. I. & BRUNK, U. T. 2001b. α-Lipoic acid and α-lipoamide prevent oxidant-induced lysosomal rupture and apoptosis. *Redox Rep*, 6, 327-334.
- PERSSON, H. L., YU, Z., TIROSH, O., EATON, J. W. & BRUNK, U. T. 2003. Prevention of oxidant-induced cell death by lysosomotropic iron chelators. *Free Radic Biol Med*, 34, 1295-305.
- PETRAT, F., DE GROOT, H. & RAUEN, U. 2001. Subcellular distribution of chelatable iron: a laser scanning microscopic study in isolated hepatocytes and liver endothelial cells. *The Biochemical journal*, 356, 61-69.
- PEYSSONNAUX, C., ZINKERNAGEL, A. S., DATTA, V., LAUTH, X., JOHNSON, R. S. & NIZET, V. 2006. TLR4-dependent hepcidin expression by myeloid cells in response to bacterial pathogens. *Blood*, 107, 3727-32.
- PIHA, M., LINDSTEDT, L. & KOVANEN, P. T. 1995. Fusion of proteolyzed low-density lipoprotein in the fluid phase: a novel mechanism generating atherogenic lipoprotein particles. *Biochemistry*, 34, 10120-9.
- PIOTROWSKI, J. J., HUNTER, G. C., ESKELSON, C. D., DUBICK, M. A. & BERNHARD, V. M. 1990. Evidence for lipid peroxidation in atherosclerosis. *Life Sciences*, 46, 715-721.
- PISONI, R. L., PARK, G. Y., VELILLA, V. Q. & THOENE, J. G. 1995. Detection and characterization of a transport system mediating cysteamine entry into human fibroblast lysosomes. Specificity for aminoethylthiol and aminoethylsulfide derivatives. *J Biol Chem*, 270, 1179-84.
- POWELL, K. E., THOMPSON, P. D., CASPERSEN, C. J. & KENDRICK, J. S. 1987. Physical activity and the incidence of coronary heart disease. *Annu Rev Public Health*, 8, 253-87.
- PU, J., GUARDIA, C. M., KEREN-KAPLAN, T. & BONIFACINO, J. S. 2016. Mechanisms and functions of lysosome positioning. *Journal of Cell Science*, 129, 4329.

- PUIG, S., RAMOS-ALONSO, L., ROMERO, A. M. & MARTINEZ-PASTOR, M. T. 2017. The elemental role of iron in DNA synthesis and repair. *Metallomics*, 9, 1483-1500.
- QIAN, Z. M., CHANG, Y. Z., LEUNG, G., DU, J. R., ZHU, L., WANG, Q., NIU, L., XU, Y. J., YANG, L., HO, K. P. & KE, Y. 2007. Expression of ferroportin1, hephaestin and ceruloplasmin in rat heart. *Biochim Biophys Acta*, 1772, 527-32.
- QUINN, M. T., PARTHASARATHY, S., FONG, L. G. & STEINBERG, D. 1987. Oxidatively modified low density lipoproteins: a potential role in recruitment and retention of monocyte/macrophages during atherogenesis. *Proc Natl Acad Sci U S A.*, 84, 2995– 2998.
- RADISKY, D. C. & KAPLAN, J. 1998. Iron in cytosolic ferritin can be recycled through lysosomal degradation in human fibroblasts. *Biochem J*, 336 (Pt 1), 201-5.
- RAJAGOPALAN, S., MENG, X. P., RAMASAMY, S., HARRISON, D. G. & GALIS, Z. S. 1996. Reactive oxygen species produced by macrophage-derived foam cells regulate the activity of vascular matrix metalloproteinases in vitro. Implications for atherosclerotic plaque stability. *Journal of Clinical Investigation*, 98, 2572-2579.
- RAJAVASHISTH, T. B., LIAO, J. K., GALIS, Z. S., TRIPATHI, S., LAUFS, U., TRIPATHI, J., CHAI, N. N., XU, X. P., JOVINGE, S., SHAH, P. K. & LIBBY, P. 1999. Inflammatory cytokines and oxidized low density lipoproteins increase endothelial cell expression of membrane type 1-matrix metalloproteinase. *J Biol Chem*, 274, 11924-9.
- RAMAKRISHNA, G., ROOKE, T. W. & COOPER, L. T. 2003. Iron and peripheral arterial disease: revisiting the iron hypothesis in a different light. *Vascular Medicine*, 8, 203-210.
- RAMASAMY, I. 2013. Recent advances in physiological lipoprotein metabolism. *Clin Chem Lab Med* 1-33.
- RANDOLPH, G. J. 2008. Emigration of monocyte-derived cells to lymph nodes during resolution of inflammation and its failure in atherosclerosis. *Current opinion in lipidology*, 19, 462-468.
- RAUSCH, A. & KORTLEEVER, C. 2011. Human physiology. https://humanphysiology2011.wikispace.com/ [Accesed 21st Jan, 2016].
- RAZANI, B., FENG, C., COLEMAN, T., EMANUEL, R., WEN, H., HWANG, S., TING, J. P., VIRGIN, H. W., KASTAN, M. B. & SEMENKOVICH, C. F. 2012. Autophagy links inflammasomes to atherosclerotic progression. *Cell Metab*, 15, 534-44.
- REIS, A. & SPICKETT, C. M. 2012. Chemistry of phospholipid oxidation. *Biochimica et Biophysica Acta (BBA) Biomembranes*, 1818, 2374-2387.
- REMMERIE, A. & SCOTT, C. L. 2018. Macrophages and lipid metabolism. *Cellular Immunology*, 330, 27-42.

- RETSKY, K. L., FREEMAN, M. W. & FREI, B. 1993. Ascorbic acid oxidation product(s) protect human low density lipoprotein against atherogenic modification anti- rather than prooxidant activity of vitamin c in the presence of transition metal ions. *The Journal Of Biological Chemistry*, 268, 1304-1309.
- RICE-EVANS, C., GREEN, E., PAGANGA, G., COOPER, C. & WRIGGLESWORTH, J. 1993. Oxidised low density lipoproteins induce iron release from activated myoglobin. *FEBS Lett*, 326, 177-82.
- RICHTER, G. W. 1986. Studies of iron overload. Lysosomal proteolysis of rat liver ferritin. *Pathol Res Pract*, 181, 159-67.
- RIDKER, P. M., BURING, J. E., SHIH, J., MATIAS, M. & HENNEKENS, C. H. 1998. Prospective study of C-reactive protein and the risk of future cardiovascular events among apparently healthy women. *Circulation*, 98, 731-3.
- RIDKER, P. M., EVERETT, B. M., THUREN, T., MACFADYEN, J. G., CHANG, W. H., BALLANTYNE, C., FONSECA, F., NICOLAU, J., KOENIG, W., ANKER, S. D., KASTELEIN, J. J. P., CORNEL, J. H., PAIS, P., PELLA, D., GENEST, J., CIFKOVA, R., LORENZATTI, A., FORSTER, T., KOBALAVA, Z., VIDA-SIMITI, L., FLATHER, M., SHIMOKAWA, H., OGAWA, H., DELLBORG, M., ROSSI, P. R. F., TROQUAY, R. P. T., LIBBY, P. & GLYNN, R. J. 2017. Antiinflammatory Therapy with Canakinumab for Atherosclerotic Disease. N Engl J Med, 377, 1119-1131.
- RIDKER, P. M., STAMPFER, M. J. & RIFAI, N. 2001. Novel risk factors for systemic atherosclerosis: A comparison of c-reactive protein, fibrinogen, homocysteine, lipoprotein(a), and standard cholesterol screening as predictors of peripheral arterial disease. *JAMA*, 285, 2481-2485.
- RIZVI, M. A., SYED, R. M. & KHAN, B. 2011. Complexation Effect on Redox Potential of Iron(III)–Iron(II) Couple: A Simple Potentiometric Experiment. *Journal of Chemical Education*, 88, 220-222.
- ROBASZKIEWICZ, A., BARTOSZ, G., PITT, A. R., THAKKER, A., ARMSTRONG, R. A., SPICKETT, C. M. & SOSZYNSKI, M. 2014. HOCl-modified phosphatidylcholines induce apoptosis and redox imbalance in HUVEC-ST cells. *Arch Biochem Biophys*, 548, 1-10.
- ROBERTS, S. & BOMFORD, A. 1988. Ferritin iron kinetics and protein turnover in K562 cells. *J Biol Chem*, 263, 19181-7.
- RODRIGUEZ-MALAVER, A. J., LEAKE, D. S. & RICE-EVANS, C. A. 1997. The effects of pH on the oxidation of low-density lipoprotein by copper and metmyoglobin are different. *FEBS Lett*, 406, 37-41.
- RODRIGUEZ-PRADOS, J. C., TRAVES, P. G., CUENCA, J., RICO, D., ARAGONES, J., MARTIN-SANZ, P., CASCANTE, M. & BOSCA, L. 2010. Substrate fate in activated macrophages: a comparison between innate, classic, and alternative activation. J Immunol, 185, 605-14.

- RODRIGUEZ, G., MAGO, N. & ROSA, F. 2009. [Role of inflammation in atherogenesis]. *Invest Clin*, 50, 109-29.
- ROKITANSKY, K. 1849. A manual of pathological anatomy, London, Sydenham Society.
- ROOS, A., XU, W., CASTELLANO, G., NAUTA, A. J., GARRED, P., DAHA, M. R. & VAN KOOTEN, C. 2004. Mini-review: A pivotal role for innate immunity in the clearance of apoptotic cells. *European Journal of Immunology*, 34, 921-929.
- ROOYAKKERS, T. M., STROES, E. S., KOOISTRA, M. P., VAN FAASSEN, E. E., HIDER, R. C., RABELINK, T. J. & MARX, J. J. 2002. Ferric saccharate induces oxygen radical stress and endothelial dysfunction in vivo. *Eur J Clin Invest*, 32 Suppl 1, 9-16.
- ROSS, R. 1999. Atherosclerosis--an inflammatory disease. N Engl J Med, 340, 115-2.
- ROSS, R. & GLOMSET, J. A. 1973. Atherosclerosis and the arterial smooth muscle cell: proliferation of smooth muscle is a key event in the genesis of the lesions of atherosclerosis. *Science*, 180, 1332–1339.
- RUCKER, P., TORTI, F. M. & TORTI, S. V. 1996. Role of H and L subunits in mouse ferritin. *J Biol Chem*, 271, 33352-7.
- RUDECK, M., VOLK, T., SITTE, N. & GRUNE, T. 2000. Ferritin oxidation in vitro: implication of iron release and degradation by the 20S proteasome. *IUBMB Life*, 49, 451-6.
- SAFTIG, P. & KLUMPERMAN, J. 2009. Lysosome biogenesis and lysosomal membrane proteins: trafficking meets function. *Nat Rev Mol Cell Biol*, 10, 623-35.
- SAKURADA, T., ORIMO, H., OKABE, H., NOMA, A. & MURAKAMI, M. 1976. Purification and properties of cholesterol ester hydrolase from human aortic intima and media. *Biochim Biophys Acta*, 424, 204-12.
- SALONEN, J. T., KORPELA, H., NYYSSONEN, K., PORKKALA, E., TUOMAINEN, T. P., BELCHER, J. D., JACOBS, D. R., JR. & SALONEN, R. 1995. Lowering of body iron stores by blood letting and oxidation resistance of serum lipoproteins: a randomized cross-over trial in male smokers. *J Intern Med*, 237, 161-8.
- SALONEN, J. T., NYYSSONEN, K., KORPELA, H., TUOMILEHTO, J., SEPPANEN, R. & SALONEN, R. 1992. High stored iron levels are associated with excess risk of myocardial infarction in eastern Finnish men. *Circulation*, 86, 803-11.
- SANCHEZ-QUESADA, J. L., VILLEGAS, S. & ORDONEZ-LLANOS, J. 2012. Electronegative low-density lipoprotein. A link between apolipoprotein B misfolding, lipoprotein aggregation and proteoglycan binding. *Curr Opin Lipidol*, 23, 479-86.
- SANDOO, A., VAN ZANTEN, J. J. C. S. V., METSIOS, G. S., CARROLL, D. & KITAS, G. D. 2010. The endothelium and its role in regulating vascular tone. *The open cardiovascular medicine journal*, 4, 302-312.

- SATCHELL, L. 2008. The oxidation of low density lypoprotein at lysosomal pH with respect to atherosclerosis. Doctor of philosophy, University of Reading.
- SATCHELL, L. & LEAKE, D. S. 2012. Oxidation of low-density lipoprotein by iron at lysosomal pH: implications for atherosclerosis. *Biochemistry*, 51, 3767-75.
- SCHAFER, F. Q., QIAN, S. Y. & BUETTNER, G. R. 2000. Iron and free radical oxidations in cell membranes. *Cell Mol Biol (Noisy-le-grand)*, 46, 657-62.
- SCHINDHELM, R. K., VAN DER ZWAN, L. P., TEERLINK, T. & SCHEFFER, P. G. 2009. Myeloperoxidase: a useful biomarker for cardiovascular disease risk stratification? *Clin Chem*, 55, 1462–70.
- SCHISSEL, S. L., TWEEDIE-HARDMAN, J., RAPP, J. H., GRAHAM, G., WILLIAMS, K. J. & TABAS, I. 1996. Rabbit aorta and human atherosclerotic lesions hydrolyze the sphingomyelin of retained low-density lipoprotein. Proposed role for arterial-wall sphingomyelinase in subendothelial retention and aggregation of atherogenic lipoproteins. *The Journal of clinical investigation*, 98, 1455-1464.
- SCHMIDT, P. J. 2015. Regulation of Iron Metabolism by Hepcidin under Conditions of Inflammation. *The Journal of Biological Chemistry*, 290, 18975-18983.
- SCHNACKENBERG, C. G., WELCH, W. J. & WILCOX, C. S. 1998. Normalization of blood pressure and renal vascular resistance in SHR with a membrane-permeable superoxide dismutase mimetic: role of nitric oxide. *Hypertension*, 32, 59-64.
- SCHNACKENBERG, C. G. & WILCOX, C. S. 1999. Two-week administration of tempol attenuates both hypertension and renal excretion of 8-Iso prostaglandin f2alpha. *Hypertension*, 33, 424-8.
- SCHNEIDER, C. 2009. An update on products and mechanisms of lipid peroxidation. *Molecular nutrition & food research*, 53, 315-321.
- SCHRIJVERS, D. M., MARTINET, W., DE MEYER, G. R. Y., KOCKX, M. M. & HERMAN, A. G. 2005. Abstract no.: 3 Impaired clearance of apoptotic cells in atherosclerosis. *Fundamental & Clinical Pharmacology*, 19, 401-401.
- SCHROEDER, T. E., HAWKINS, S. A., HYSLOP, D., VALLEJO, A. F., JENSKY, N. E. & WISWELL, R. A. 2007. Longitudinal change in coronary heart disease risk factors in older runners. *Age Ageing*, 36, 57-62.
- SCHULTZ, M. L., TECEDOR, L., CHANG, M. & DAVIDSON, B. L. 2011. Clarifying lysosomal storage diseases. *Trends Neurosci*, 34, 401-10.
- SCHWARZ, A., BONATERRA, G. A., SCHWARZBACH, H. & KINSCHERF, R. 2017. Oxidized LDL-induced JAB1 influences NF-κB independent inflammatory signaling in human macrophages during foam cell formation. *Journal of Biomedical Science*, 24, 12.

- SCHWENKE, D. C. & CAREW, T. E. 1989. Initiation of atherosclerotic lesions in cholesterolfed rabbits. II. Selective retention of LDL vs. selective increases in LDL permeability in susceptible sites of arteries. *Arteriosclerosis*, 9, 908-18.
- SCHWENKE, D. C., RUDEL, L. L., SORCI-THOMAS, M. G. & THOMAS, M. J. 2002. Alpha-tocopherol protects against diet induced atherosclerosis in New Zealand white rabbits. *J Lipid Res*, 43, 1927-38.
- SCHWENKE, D. C. & ZILVERSMIT, D. B. 1989. The arterial barrier to lipoprotein influx in the hypercholesterolemic rabbit. 1. Studies during the first two days after mild aortic injury. *Atherosclerosis*, 77, 91-103.
- SEMPOS, C. T., LOOKER, A. C., GILLUM, R. F. & MAKUC, D. M. 1994. Body iron stores and the risk of coronary heart disease. *N Engl J Med*, 330, 1119-24.
- SERGIN, I., EVANS, T. D. & RAZANI, B. 2015. Degradation and beyond: the macrophage lysosome as a nexus for nutrient sensing and processing in atherosclerosis. *Curr Opin Lipidol*, 26, 394-404.
- SERGIN, I., EVANS, T. D., ZHANG, X., BHATTACHARYA, S., STOKES, C. J., SONG, E., ALI, S., DEHESTANI, B., HOLLOWAY, K. B., MICEVYCH, P. S., JAVAHERI, A., CROWLEY, J. R., BALLABIO, A., SCHILLING, J. D., EPELMAN, S., WEIHL, C. C., DIWAN, A., FAN, D., ZAYED, M. A. & RAZANI, B. 2017. Exploiting macrophage autophagy-lysosomal biogenesis as a therapy for atherosclerosis. *Nature Communications*, 8, 15750.
- SERRUYS, P. W., DE FEYTER, P., MACAYA, C., KOKOTT, N., PUEL, J., VROLIX, M., BRANZI, A., BERTOLAMI, M. C., JACKSON, G., STRAUSS, B. & MEIER, B. 2002. Fluvastatin for prevention of cardiac events following successful first percutaneous coronary intervention: a randomized controlled trial. *Jama*, 287, 3215-22.
- SESSO, H. D., BURING, J. E., CHRISTEN, W. G., KURTH, T., BELANGER, C., MACFADYEN, J., BUBES, V., MANSON, J. E., GLYNN, R. J. & GAZIANO, J. M. 2008. Vitamins E and C in the prevention of cardiovascular disease in men: the Physicians' Health Study II randomized controlled trial. *Jama*, 300, 2123-33.
- SHAH ANOOP, S. V., STELZLE, D., LEE KUAN, K., BECK EDUARD, J., ALAM, S., CLIFFORD, S., LONGENECKER CHRIS, T., STRACHAN, F., BAGCHI, S., WHITELEY, W., RAJAGOPALAN, S., KOTTILIL, S., NAIR, H., NEWBY DAVID, E., MCALLISTER DAVID, A. & MILLS NICHOLAS, L. 2018. Global Burden of Atherosclerotic Cardiovascular Disease in People Living With HIV. *Circulation*, 138, 1100-1112.
- SHAPIRO, M. D. & FAZIO, S. 2017. PCSK9 and Atherosclerosis Lipids and Beyond. Journal of atherosclerosis and thrombosis, 24, 462-472.
- SHARMA, N., SHARMA, P., JASUJA, N. D. & JOSHI, S. C. 2013. Hypocholesterolemic and Antioxidant Potentials of Some Plants and Herbs: A Review. *Research and Reviews: Journal of Zoological Sciences*, 1, 26-42.

- SHEN, J., HERDERICK, E., CORNHILL, J. F., ZSIGMOND, E., KIM, H. S., KÜHN, H., GUEVARA, N. V. & CHAN, L. 1996. Macrophage-mediated 15-lipoxygenase expression protects against atherosclerosis development. *Journal of Clinical Investigation*, 98, 2201-2208.
- SHEPHERD, J., COBBE, S. M., FORD, I., ISLES, C. G., LORIMER, A. R., MACFARLANE, P. W., MCKILLOP, J. H. & PACKARD, C. J. 1995. Prevention of coronary heart disease with pravastatin in men with hypercholesterolemia. *N Engl J Med*, 333.
- SHIBATA, N. & GLASS, C. K. 2009. Regulation of macrophage function in inflammation and atherosclerosis. *J Lipid Res*, 50 Suppl, S277-81.
- SHIMANO, H. 2009. [Obesity and atherosclerosis]. *Nihon rinsho. Japanese journal of clinical medicine*, 67, 333-337.
- SHIRAI, T., NAZAREWICZ, R. R., WALLIS, B. B., YANES, R. E., WATANABE, R., HILHORST, M., TIAN, L., HARRISON, D. G., GIACOMINI, J. C., ASSIMES, T. L., GORONZY, J. J. & WEYAND, C. M. 2016. The glycolytic enzyme PKM2 bridges metabolic and inflammatory dysfunction in coronary artery disease. *J Exp Med*, 213, 337-54.
- SIBILLE, J. C., CIRIOLO, M., KONDO, H., CRICHTON, R. R. & AISEN, P. 1989. Subcellular localization of ferritin and iron taken up by rat hepatocytes. *Biochem J*, 262, 685-8.
- SIES, H., STAHL, W. & SUNDQUIST, A. R. 1992. Antioxidant functions of vitamins. Vitamins E and C, beta-carotene, and other carotenoids. *Ann N Y Acad Sci*, 669, 7-20.
- SIESS, W. 2006. Platelet interaction with bioactive lipids formed by mild oxidation of lowdensity lipoprotein. *Pathophysiol Haemost Thromb*, 35, 292-304.
- SIMON, H.-U., HAJ-YEHIA, A. & LEVI-SCHAFFER, F. 2000. Role of reactive oxygen species (ROS) in apoptosis induction. *Apoptosis*, 5, 415-418.
- SINCLAIR, A. J., GIRLING, A. J., GRAY, L., LE GUEN, C., LUNEC, J. & BARNETT, A. H. 1991. Disturbed handling of ascorbic acid in diabetic patients with and without microangiopathy during high dose ascorbate supplementation. *Diabetologia*, 34, 171-5.
- SINGH, U. & JIALAL, I. 2006. Oxidative stress and atherosclerosis. *Pathophysiology*, 13, 129-142.
- SIOW, R. C., RICHARDS, J. P., PEDLEY, K. C., LEAKE, D. S. & MANN, G. E. 1999a. Vitamin C protects human vascular smooth muscle cells against apoptosis induced by moderately oxidized LDL containing high levels of lipid hydroperoxides. *Arterioscler Thromb Vasc Biol*, 19, 2387-94.
- SIOW, R. C., SATO, H., LEAKE, D. S., ISHII, T., BANNAI, S. & MANN, G. E. 1999b. Induction of antioxidant stress proteins in vascular endothelial and smooth muscle cells: protective action of vitamin C against atherogenic lipoproteins. *Free Radic Res*, 31, 309-18.

- SJÖBERG, S. & SHI, G.-P. 2011. Cysteine Protease Cathepsins in Atherosclerosis and Abdominal Aortic Aneurysm. *Clinical reviews in bone and mineral metabolism*, 9, 138-147.
- SMITH, C., MITCHINSON, M. J., ARUOMA, O. I. & HALLIWELL, B. 1992. Stimulation of lipid peroxidation and hydroxyl-radical generation by the contents of human atherosclerotic lesions. *Biochem J*, 286 (Pt 3), 901-5.
- SOBAL, G., MENZEL, J. & SINZINGER, H. 2000. Why is glycated LDL more sensitive to oxidation than native LDL? A comparative study. *Prostaglandins Leukot Essent Fatty Acids*, 63, 177-86.
- SOFUNI, T., MATSUOKA, A., SAWADA, M., ISHIDATE, M., JR., ZEIGER, E. & SHELBY, M. D. 1990. A comparison of chromosome aberration induction by 25 compounds tested by two Chinese hamster cell (CHL and CHO) systems in culture. *Mutat Res*, 241, 175-213.
- SONG, J., WANG, D., CHEN, H., HUANG, X., ZHONG, Y., JIANG, N., CHEN, C. & XIA, M. 2017. Association of Plasma 7-Ketocholesterol With Cardiovascular Outcomes and Total Mortality in Patients With Coronary Artery Disease. *Circulation Research*, 120, 1622-1631.
- SPARROW, C. P., DOEBBER, T. W., OLSZEWSKI, J., WU, M. S., VENTRE, J., STEVENS, K. A. & CHAO, Y. S. 1992. Low density lipoprotein is protected from oxidation and the progression of atherosclerosis is slowed in cholesterol-fed rabbits by the antioxidant N,N'-diphenyl-phenylenediamine. J Clin Invest, 89, 1885-91.
- SPENCE, J. D. & PILOTE, L. 2015. Importance of sex and gender in atherosclerosis and cardiovascular disease. *Atherosclerosis*, 241, 208-10.
- SPICKETT, C. M., JERLICH, A., PANASENKO, O. M., ARNHOLD, J., PITT, A. R., STELMASZYNSKA, T. & SCHAUR, R. J. 2000. The reactions of hypochlorous acid, the reactive oxygen species produced by myeloperoxidase, with lipids. *Acta Biochim Pol*, 47, 889-99.
- SPICKETT, C. M., REIS, A. & PITT, A. R. 2011. Identification of oxidized phospholipids by electrospray ionization mass spectrometry and LC–MS using a QQLIT instrument. *Free Radical Biology and Medicine*, 51, 2133-2149.
- STADLER, N., LINDNER, R. A. & DAVIES, M. J. 2004. Direct Detection and Quantification of Transition Metal Ions in Human Atherosclerotic Plaques: Evidence for the Presence of Elevated Levels of Iron and Copper. *Arterioscler Thromb Vasc Biol*, 24, 949-954.
- STAIT, S. E. & LEAKE, D. S. 1994. Ascorbic acid can either increase or decrease low density lipoprotein modification. *FEBS Lett*, 341, 263-7.
- STAIT, S. E. & LEAKE, D. S. 1996. The effects of ascorbate and dehydroascorbate on the oxidation of low-density lipoprotein. *Biochem J*, 320 (Pt 2), 373-81.

- STANLEY, N., STADLER, N., WOODS, A. A., BANNON, P. G. & DAVIES, M. J. 2006. Concentrations of iron correlate with the extent of protein, but not lipid, oxidation in advanced human atherosclerotic lesions. *Free radical biology & medicine*, 40, 1636-1643.
- STARKEBAUM, G. & HARLAN, J. M. 1986. Endothelial cell injury due to copper-catalyzed hydrogen peroxide generation from homocysteine. *Journal of Clinical Investigation*, 77, 1370-1376.
- STARY, H. C. 1983. Evolution of atherosclerotic plaques in the coronary arteries of young adults. *Arteriosclerosis*, 3, 417A–421A.
- STARY, H. C., CHANDLER, A. B., DINSMORE, R. E., FUSTER, V., GLAGOV, S., INSULL, W., JR., ROSENFELD, M. E., SCHWARTZ, C. J., WAGNER, W. D. & WISSLER, R. W. 1995. A definition of advanced types of atherosclerotic lesions and a histological classification of atherosclerosis. A report from the Committee on Vascular Lesions of the Council on Arteriosclerosis, American Heart Association. *Circulation*, 92, 1355-74.
- STEIN, Y., EBIN, V., BAR-ON, H. & STEIN, O. 1977. Chloroquine-induced interference with degradation of serum lipoproteins in rat liver, studied in vivo and in vitro. *Biochim Biophys Acta*, 486, 286-97.
- STEINBERG, D. 1997a. Lewis A. Conner Memorial Lecture. Oxidative modification of LDL and atherogenesis. *Circulation*, 95, 1062-71.
- STEINBERG, D. 1997b. Low density lipoprotein oxidation and its pathobiological significance. *J Biol Chem*, 272, 20963-6.
- STEINBERG, D. 2002. Atherogenesis in perspective :hypercholesterolemia and inflammation as partners in crime. *Nat Med*, 8, 1211-7.
- STEINBERG, D. 2009. The LDL modification hypothesis of atherogenesis: an update. *J. Lipid Res*, 50, S376-S381.
- STEINBERG, D., PARTHASARATHY, S., CAREW, T. E., KHOO, J. C. & WITZTUM, J. L. 1989. Beyond cholesterol: Modifications of low-density lipoprotein that increase its atherogenicity. *New England Journal of Medicine*, 320, 915-924.
- STEINBERG, D. & WITZTUM, J. L. 2002. Is the oxidative modification hypothesis relevant to human atherosclerosis? Do the antioxidant trials conducted to date refute the hypothesis? *Circulation*, 105, 2107-11.
- STEINBRECHER, U. P. 1991. Role of lipoprotein peroxidation in the pathogenesis of atherosclerosis. *Clin Cardiol*, 14, 865-7.
- STEINBRECHER, U. P. & LOUGHEED, M. 1992. Scavenger receptor-independent stimulation of cholesterol esterification in macrophages by low density lipoprotein extracted from human aortic intima. *Arterioscler Thromb*, 12, 608-25.

- STEINBRECHER, U. P., PARTHSARATHY, S., LEAKE, D. S., WITZTUM, J. L. & STEINBERG, D. 1984. Modifications of low density lipoprotein by endothelial cells involves lipid peroxidation and degradation of low density lipoprotein phospholipids. *Proc Natl Acad Sci USA*, 81, 3883-7.
- STEMMER, U., DUNAI, Z. A., KOLLER, D., PÜRSTINGER, G., ZENZMAIER, E., DEIGNER, H. P., AFLAKI, E., KRATKY, D. & HERMETTER, A. 2012. Toxicity of oxidized phospholipids in cultured macrophages. *Lipids in Health and Disease*, 11, 110.
- STEPHENS, N. G., PARSONS, A., SCHOFIELD, P. M., KELLY, F., CHEESEMAN, K. & MITCHINSON, M. J. 1996. Randomised controlled trial of vitamin E in patients with coronary disease: Cambridge Heart Antioxidant Study (CHAOS). *Lancet*, 347, 781-6.
- STOCKER, R. 1999. The ambivalence of vitamin E in atherogenesis. *Trends in Biochemical Sciences*, 24, 219-223.
- STOCKER, R., BOWRY, V. W. & FREI, B. 1991. Ubiquinol-10 protects human low density lipoprotein more efficiently against lipid peroxidation than does alpha-tocopherol. *Proc Natl Acad Sci U S A*, 88, 1646-50.
- STOCKER, R. & KEANEY, J. F., JR. 2004. Role of oxidative modifications in atherosclerosis. *Physiol Rev*, 84, 1381-478.
- STONE, J. R. 2012. Pathology of myocardial infarction, coronary artery disease, plaque disruption, and the vulnerable atherosclerotic plaque. *Diagnostic Histopathology*, 18, 478-483.
- STOUT, R. W. 1987. Ageing and athrosclerosis. Age and Ageing, 16, 65-75.
- SUARNA, C., DEAN, R. T., MAY, J. & STOCKER, R. 1995. Human atherosclerotic plaque contains both oxidized lipids and relatively large amounts of alpha-tocopherol and ascorbate. *Arterioscler Thromb Vasc Biol*, 15, 1616-24.
- SUKHOVA, G. K., ZHANG, Y., PAN, J. H., WADA, Y., YAMAMOTO, T., NAITO, M., KODAMA, T., TSIMIKAS, S., WITZTUM, J. L., LU, M. L., SAKARA, Y., CHIN, M. T., LIBBY, P. & SHI, G. P. 2003. Deficiency of cathepsin S reduces atherosclerosis in LDL receptor-deficient mice. *J Clin Invest*, 111, 897-906.
- SULLIVAN, J. L. 1981. Iron and the sex difference in heart disease risk. Lancet, 1, 1293-4.
- SULLIVAN, J. L. 1989. The iron paradigm of ischemic heart disease. *Am Heart J*, 117, 1177-88.
- SULLIVAN, J. L. 1992. Stored iron and ischemic heart disease. Empirical support for a new paradigm. *Circulation*, 86, 1036-7.
- SULLIVAN, J. L. 2003. Are menstruating women protected from heart disease because of, or in spite of, estrogen? Relevance to the iron hypothesis. *Am Heart J*, 145, 190-4.

- SULLIVAN, J. L. 2005. Stored iron and vascular reactivity. *Arteriosclerosis, Thrombosis, and Vascular Biology*, 25, 1532–1535.
- SULLIVAN, J. L. 2007. Macrophage iron, hepcidin, and atherosclerotic plaque stability. *Exp Biol Med.*, 232.
- SUNMAN, W., HUGHES, A. D. & SEVER, P. S. 1993. Free-radical scavengers, thiolcontaining reagents and endothelium-dependent relaxation in isolated rat and human resistance arteries. *Clin Sci (Lond)*, 84, 287-95.
- SURYAWANSHI, N. P., BHUTEY, A. K., NAGDEOTE, A. N., JADHAV, A. A. & MANOORKAR, G. S. 2006. Study of lipid peroxide and lipid profile in diabetes mellitus. *Indian Journal of Clinical Biochemistry*, 21, 126-130.
- TABAS, I. 1999. Nonoxidative modifications of lipoproteins in atherogenesis. *Annu Rev Nutr*, 19, 123-39.
- TABAS, I. 2005. Consequences and therapeutic implications of macrophage apoptosis in atherosclerosis: the importance of lesion stage and phagocytic efficiency. *Arterioscler Thromb Vasc Biol*, 25, 2255-64.
- TABAS, I. & BORNFELDT, K. E. 2016. Macrophage Phenotype and Function in Different Stages of Atherosclerosis. *Circ Res*, 118, 653-67.
- TABAS, I., WILLIAMS, K. J. & BORÉN, J. 2007. Subendothelial lipoprotein retention as the initiating process in atherosclerosis: Update and therapeutic implications. *Circulation*, 116, 1832-1844.
- TAKAGI, H., SHI, D., HA, Y., ALLEWELL, N. M. & THEIL, E. C. 1998. Localized unfolding at the junction of three ferritin subunits. A mechanism for iron release? *J Biol Chem*, 273, 18685-8.
- TAKAGI, M., ONODERA, H., MIYAMOTO, I. & MORITA, N. 1988. Effects of active oxygen scavengers on the peroxidation of linoleic acid catalyzed by dehydro-L-ascorbic acid or its degradation products. *J. Nutr. Sci. Vitarninol* 34, 141-149.
- TAMES, F. J., MACKNESS, M. I., ARROL, S., LAING, I. & DURRINGTON, P. N. 1992. Non-enzymatic glycation of apolipoprotein B in the sera of diabetic and non-diabetic subjects. *Atherosclerosis*, 93, 237-44.
- TANG, L., ZHANG Y FAU QIAN, Z., QIAN Z FAU SHEN, X. & SHEN, X. 2000. The mechanism of Fe(2+)-initiated lipid peroxidation in liposomes: the dual function of ferrous ions, the roles of the pre-existing lipid peroxides and the lipid peroxyl radical. *Biochem J* 352, 27-36.
- TANG, L. X., YANG, J. L. & SHEN, X. 1997. Effects of additional iron-chelators on Fe(2+)initiated lipid peroxidation: evidence to support the Fe2+ ... Fe3+ complex as the initiator. *J Inorg Biochem*, 68, 265-72.

- TANGIRALA, R. K., CASANADA, F., MILLER, E., WITZTUM, J. L., STEINBERG, D. & PALINSKI, W. 1995. Effect of the antioxidant N,N'-diphenyl 1,4-phenylenediamine (DPPD) on atherosclerosis in apoE-deficient mice. *Arterioscler Thromb Vasc Biol*, 15, 1625-30.
- TANO, J. K., LEE, R. H. & VAZQUEZ, G. 2012. Macrophage function in atherosclerosis. *Channels (Austin)*, 6, 141-148.
- TAVORI, H., RASHID, S. & FAZIO, S. 2015. On the function and homeostasis of PCSK9: Reciprocal interaction with LDLR and additional lipid effects. *Atherosclerosis*, 238, 264-270.
- TAWAKOL, A., SINGH, P., MOJENA, M., PIMENTEL-SANTILLANA, M., EMAMI, H., MACNABB, M., RUDD, J. H., NARULA, J., ENRIQUEZ, J. A., TRAVES, P. G., FERNANDEZ-VELASCO, M., BARTRONS, R., MARTIN-SANZ, P., FAYAD, Z.
 A., TEJEDOR, A. & BOSCA, L. 2015. HIF-1alpha and PFKFB3 Mediate a Tight Relationship Between Proinflammatory Activation and Anerobic Metabolism in Atherosclerotic Macrophages. *Arterioscler Thromb Vasc Biol*, 35, 1463-71.
- TAWAKOL, A., T., O., M., G., IT., W. & MA., C. 1997. Hyperhomocyst(e)inemia is associated with impaired endotheliumdependent vasodilation in humans. *Circulation*, 95, 1119-1121.
- TEISSEDRE, P. L., FRANKEL, E. N., WATERHOUSE, A. L., PELEG, H. & GERMAN, J. B. 1996. Inhibition of In Vitro Human LDL Oxidation by Phenolic Antioxidants from Grapes and Wines J Sci Food Agric 70, 55-61.
- TERMAN, A. & KURZ, T. 2013. Lysosomal iron, iron chelation, and cell death. *Antioxid Redox Signal*, 18, 888-98.
- THEIL, E. C. 1987. Ferritin: structure, gene regulation, and cellular function in animals, plants, and microorganisms. *Annu Rev Biochem*, 56, 289–315.
- THEIL, E. C., BEHERA, R. K. & TOSHA, T. 2013. Ferritins for Chemistry and for Life. *Coord Chem Rev*, 257, 579-586.
- THEURL, I., THEURL, M., SEIFERT, M., MAIR, S., NAIRZ, M., RUMPOLD, H., ZOLLER, H., BELLMANN-WEILER, R., NIEDEREGGER, H., TALASZ, H. & WEISS, G. 2008. Autocrine formation of hepcidin induces iron retention in human monocytes. *Blood*, 111, 2392-9.
- THOMAS, M. J., THORNBURG, T., MANNING, J., HOOPER, K. & RUDEL, L. L. 1994. Fatty acid composition of low-density lipoprotein influences its susceptibility to autoxidation. *Biochemistry*, 33, 1828-34.
- THOMPSON, P. D., BUCHNER, D., PINA, I. L., BALADY, G. J., WILLIAMS, M. A., MARCUS, B. H., BERRA, K., BLAIR, S. N., COSTA, F., FRANKLIN, B., FLETCHER, G. F., GORDON, N. F., PATE, R. R., RODRIGUEZ, B. L., YANCEY, A. K. & WENGER, N. K. 2003. Exercise and physical activity in the prevention and treatment of atherosclerotic cardiovascular disease: a statement from the Council on

Clinical Cardiology (Subcommittee on Exercise, Rehabilitation, and Prevention) and the Council on Nutrition, Physical Activity, and Metabolism (Subcommittee on Physical Activity). *Circulation*, 107, 3109-16.

- THOMPSON, S. G., KIENAST, J., PYKE, S. D. M., HAVERKATE, F. & VAN DE LOO, J. C. W. 1995. Hemostatic factors and the risk of myocardial infarction or sudden death in patients with angina pectoris. . N Engl J Med., 332, 635-651.
- TORZEWSKI, M., KLOUCHE, M., HOCK, J., MESSNER, M., DORWEILER, B., TORZEWSKI, J., GABBERT, H. E. & BHAKDI, S. 1998. Immunohistochemical demonstration of enzymatically modified human LDL and its colocalization with the terminal complement complex in the early atherosclerotic lesion. *Arterioscler Thromb Vasc Biol*, 18, 369-78.
- TOWNSEND, N., WICKRAMASINGHE, K., BHATNAGAR, P., SMOLINA, K., NICHOLS, M., LEA, J., R., L.-F. & RAYNER, M. 2012. Coronary heart disease statistics 2012 edition. *In:* WEISSBERG, P. (ed.). London: British Heart Foundation.
- TRAN, T. N., EUBANKS, S. K., SCHAFFER, K. J., ZHOU, C. Y. & LINDER, M. C. 1997. Secretion of ferritin by rat hepatoma cells and its regulation by inflammatory cytokines and iron. *Blood*, 90, 4979-86.
- TSAI, J. C., WANG, H., PERRELLA, M. A., YOSHIZUMI, M., SIBINGA, N. E., TAN, L. C., HABER, E., CHANG, T. H., SCHLEGEL, R. & LEE, M. E. 1996. Induction of cyclin A gene expression by homocysteine in vascular smooth muscle cells. *The Journal of Clinical Investigation*, 97, 146-153.
- TSIMIKAS, S. 2006. Oxidized low-density lipoprotein biomarkers in atherosclerosis. *Curr Atheroscler Rep*, 8, 55-61.
- TSUJI, Y., MILLER, L. L., MILLER, S. C., SV, T. & TORTI, F. M. 1991. Tumor necrosis factor-alpha and interleukin 1-alpha regulate transferrin receptor in human diploid fibroblasts. Relationship to the induction of ferritin heavy chain. *J Biol Chem 1980*, 266, 7257-61.
- TURK, V., TURK, B., GUNCAR, G., TURK, D. & KOS, J. 2002. Lysosomal cathepsins: structure, role in antigen processing and presentation, and cancer. *Adv Enzyme Regul*, 42, 285-303.
- TURNBUL, F. 2003. Effects of different blood-pressure-lowering regimens on major cardiovascular events: results of prospectively-designed overviews of randomised trials. *The Lancet*, 362, 1527-1535.
- UCHIDA, K. & STADTMAN, E. R. 1992. Modification of histidine residues in proteins by reaction with 4-hydroxynonenal. *Proc Natl Acad Sci U S A*, 89, 4544-8.
- UEDA, M., O'BRIEN, K., ROSING, D. R., LING, A., KLETA, R., MCAREAVEY, D., BERNARDINI, I. & GAHL, W. A. 2006. Coronary artery and other vascular calcifications in patients with cystinosis after kidney transplantation. *Clin J Am Soc Nephrol*, 1, 555-62.

- VAN AALST, J. A., ZHANG, D. M., MIYAZAKI, K., COLLES, S. M., FOX, P. L. & GRAHAM, L. M. 2004. Role of reactive oxygen species in inhibition of endothelial cell migration by oxidized low-density lipoprotein. *J Vasc Surg*, 40, 1208-15.
- VAN DER WESTHUYZEN, D. R., GEVERS, W. & COETZEE, G. A. 1980. Cathepsin-Ddependent initiation of the hydrolysis by lysosomal enzymes of apoprotein B from lowdensity lipoproteins. *Eur J Biochem*, 112, 153-60.
- VAN REYK, D. M., BROWN, A. J., JESSUP, W. & DEAN, R. T. 1995. Batch-to-batch variation of Chelex-100 confounds metal-catalysed oxidation. Leaching of inhibitory compounds from a batch of Chelex-100 and their removal by a pre-washing procedure. *Free Radic Res*, 23, 533-5.
- VAN VRE, E. A., AIT-OUFELLA, H., TEDGUI, A. & MALLAT, Z. 2012. Apoptotic cell death and efferocytosis in atherosclerosis. *Arterioscler Thromb Vasc Biol*, 32, 887-93.
- VENDROV, A. E., HAKIM, Z. S., MADAMANCHI, N. R., ROJAS, M., MADAMANCHI, C. & RUNGE, M. S. 2007. Atherosclerosis is attenuated by limiting superoxide generation in both macrophages and vessel wall cells. *Arterioscler Thromb Vasc Biol*, 27, 2714-21.
- VERLANGIERI, A. J., HOLLIS, T. M. & MUMMA, R. O. 1977. Effects of ascorbic acid and its 2-sulfate on rabbit aortic intimal thickening. *Blood Vessels*, 14, 157-74.
- VIIGIMAA, M., ABINA, J., ZEMTSOVSKAYA, G., TIKHAZE, A., KONOVALOVA, G., KUMSKOVA, E. & LANKIN, V. 2010. Malondialdehyde-modified low-density lipoproteins as biomarker for atherosclerosis. *Blood Press*, 19, 164-8.
- VINSON, J. A. & JANG, J. 2001. In Vitro and In Vivo Lipoprotein Antioxidant Effect of a Citrus Extract and Ascorbic Acid on Normal and Hypercholesterolemic Human Subjects. J Med Food, 4, 187-192.
- VIRCHOW, R. 1989. Cellular pathology. As based upon physiological and pathological history LXVI atheromatous affection of atheries 1852. *Nutr Rev.*, 47, 3.
- VOGIATZI, G., TOUSOULIS, D. & STEFANADIS, C. 2009. The role of oxidative stress in atherosclerosis. *Hellenic J Cardiol*, 50, 402-9.
- WALDO, S. W., LI, Y., BUONO, C., ZHAO, B., BILLINGS, E. M., CHANG, J. & KRUTH, H. S. 2008. Heterogeneity of human macrophages in culture and in atherosclerotic plaques. *Am J Pathol*, 172, 1112-26.
- WALLDIUS, G., ERIKSON, U., OLSSON, A. G., BERGSTRAND, L., HADELL, K., JOHANSSON, J., KAIJSER, L., LASSVIK, C., MOLGAARD, J., NILSSON, S. & ET AL. 1994. The effect of probucol on femoral atherosclerosis: the Probucol Quantitative Regression Swedish Trial (PQRST). Am J Cardiol, 74, 875-83.
- WALTER, M. F., JACOB, R. F., BJORK, R. E., JEFFERS, B., BUCH, J., MIZUNO, Y. & MASON, R. P. 2008. Circulating Lipid Hydroperoxides Predict Cardiovascular Events

in Patients With Stable Coronary Artery Disease: The PREVENT Study. *Journal of the American College of Cardiology*, 51, 1196-1202.

- WALTERS, M. J. & WRENN, S. P. 2010. Size-selective uptake of colloidal low density lipoprotein aggregates by cultured white blood cells. *J Colloid Interface Sci*, 350, 494-501.
- WANG, J. C. & BENNETT, M. 2012. Aging and atherosclerosis: mechanisms, functional consequences, and potential therapeutics for cellular senescence. *Circ Res*, 111, 245-59.
- WANG, T., PALUCCI, D., LAW, K., YANAGAWA, B., YAM, J. & BUTANY, J. 2012. Atherosclerosis: pathogenesis and pathology. *Diagnostic Histopathology*, 18, 461-467.
- WANG, X., SUN, Y., YANG, H., LU, Y. & LI, L. 2016. Oxidized Low-Density Lipoprotein Induces Apoptosis in Cultured Neonatal Rat Cardiomyocytes by Modulating the TLR4/NF-κB Pathway. *Scientific reports*, 6, 27866-27866.
- WANNAMETHEE, S. G., SHAPER, A. G. & WALKER, M. 1998. Changes in physical activity, mortality, and incidence of coronary heart disease in older men. *Lancet*, 351, 1603-8.
- WARDMAN, P. 2007. Fluorescent and luminescent probes for measurement of oxidative and nitrosative species in cells and tissues: progress, pitfalls, and prospects. *Free Radic Biol Med*, 43, 995-1022.
- WATSON, A. D. 1997. Structural identification by mass spectrometry of oxidized phospholipids in minimally oxidized low density lipoprotein that induce monocyte/endothelial interactions and evidence for their presence in vivo. *J. Biol. Chem.*, 272, 13597-13607.
- WEBER, C., ERL, W., WEBER, K. & WEBER, P. C. 1996. Increased adhesiveness of isolated monocytes to endothelium is prevented by vitamin C intake in smokers. *Circulation*, 93, 1488-92.
- WEI, Y., MILLER, S. C., TSUJI, Y., TORTI, S. V. & TORTI, F. M. 1990. Interleukin 1 induces ferritin heavy chain in human muscle cells. *Biochem Biophys Res Commun*, 169, 289-96.
- WEN, Y. & LEAKE, D. S. 2007. Low density lipoprotein undergoes oxidation within lysosomes in cells. *Circ Res*, 100, 1337-43.
- WEN, Y., MOHRI, Z., AHMAD, F., WEINBERG, P. D. & D.S., L. submitted. Cysteamine inhibits lysosomal oxidation of low density lipoprotein in human macrophages and reduces atherosclerosis in mice.
- WEN, Y., SATCHELL, L., GIBSON, T. M., WEINBERG, P. D. & LEAKE, D. S. 2015. Low density lipoprotein aggregated by sphingomyelinase is internalised by macrophages and oxidised in lysosomes. *Atherosclerosis*, 232, E5-6.

- WHITE, C. R., BROCK, T. A., CHANG, L. Y., CRAPO, J., BRISCOE, P., KU, D., BRADLEY, W. A., GIANTURCO, S. H., GORE, J. & FREEMAN, B. A. 1994. Superoxide and peroxynitrite in atherosclerosis. *Proceedings of the National Academy* of Sciences of the United States of America, 91, 1044-1048.
- WHO 1996. Hypertension Control: Report of a WHO Expert Committee. WHO technical report series 86. Geneva, Switzerland: World Health Organization.
- WHO 2016. Cardiovascular diseases. https://www.who.int/cardiovascular_diseases/worldheart-day/en/ [Accessed 31st October 2018].
- WILCOX, C. S. & PEARLMAN, A. 2008. Chemistry and antihypertensive effects of tempol and other nitroxides. *Pharmacological reviews*, 60, 418-469.
- WILKINS, G. M. & LEAKE, D. S. 1994. The effect of inhibitors of free radical generatingenzymes on low-density lipoprotein oxidation by macrophages. *Biochim Biophys Acta*, 1211, 69-78.
- WILLIAMS, J. K., ANTHONY, M. S., HONORE, E. K., HERRINGTON, D. M., MORGAN, T. M., REGISTER, T. C. & CLARKSON, T. B. 1995. Regression of atherosclerosis in female monkeys. *Arterioscler Thromb Vasc Biol*, 15, 827-36.
- WILLIAMS, K. J. & TABAS, I. 1998. The response-to-retention hypothesis of atherogenesis reinforced. *Current Opinion in Lipidology*, 9, 471-474.
- WILMER, M. J., KLUIJTMANS, L. A., VAN DER VELDEN, T. J., WILLEMS, P. H., SCHEFFER, P. G., MASEREEUW, R., MONNENS, L. A., VAN DEN HEUVEL, L. P. & LEVTCHENKO, E. N. 2011. Cysteamine restores glutathione redox status in cultured cystinotic proximal tubular epithelial cells. *Biochim Biophys Acta*, 1812, 643-51.
- WILSON, P., KANNEL, W., SILBERSHATZ, H. & D'AGOSTINO, R. 1999. Clustering of metabolic factors and coronary heart disease. ARCHIVES OF INTERNAL MEDICINE, 159, 1104-1109.
- WILSON, P. W., D'AGOSTINO RB FAU LEVY, D., LEVY D FAU BELANGER, A. M., BELANGER AM FAU - SILBERSHATZ, H., SILBERSHATZ H FAU - KANNEL, W. B. & KANNEL, W. B. 1998. Prediction of coronary heart disease using risk factor categories. *Circulation.*, 97, 1837-1847.
- WINTERGERST, E. S., JELK, J., RAHNER, C. & ASMIS, R. 2000. Apoptosis induced by oxidized low density lipoprotein in human monocyte-derived macrophages involves CD36 and activation of caspase-3. *Eur J Biochem*, 267, 6050-9.
- WITZTUM, J. L. 1993. Role of oxidized low density lipoproten in atherogenesis *Br Heart J.*, 69, 12-S 18.
- WITZTUM, J. L. & BERLINER, J. A. 1998. Oxidized phospholipids and isoprostanes in atherosclerosis. *Curr Opin Lipidol*, 9, 441-8.

- WITZTUM, J. L. & STEINBERG, D. 1991. Role of oxidized low density lipoprotein in atherogenesis. J Clin Invest, 88, 1785-92.
- WOO, K. S., CHOOK, P., LOLIN, Y. I., CHEUNG, A. S., CHAN, L. T., SUN, Y. Y., SANDERSON, J. E., METREWELI, C. & CELERMAJER, D. S. 1997. Hyperhomocyst(e)inemia is a risk factor for arterial endothelial dysfunction in humans. *Circulation*, 96, 2542-4.
- WRIGHTING, D. M. & ANDREWS, N. C. 2006. Interleukin-6 induces hepcidin expression through STAT3. *Blood*, 108, 3204-9.
- XU, H. & REN, D. 2015. Lysosomal physiology. Annual review of physiology, 77, 57-80.
- XU, S. & LIN, B. 2001. The mechanism of oxidation-induced low-density lipoprotein aggregation: an analogy to colloidal aggregation and beyond? *Biophysical journal*, 81, 2403-2413.
- XU, X., YUAN, X., LI, N., DEWEY, W. L., LI, P. L. & ZHANG, F. 2016. Lysosomal cholesterol accumulation in macrophages leading to coronary atherosclerosis in CD38(-/-) mice. *J Cell Mol Med*, 20, 1001-13.
- XU, X. X. & TABAS, I. 1991. Sphingomyelinase enhances low density lipoprotein uptake and ability to induce cholesteryl ester accumulation in macrophages. *J Biol Chem*, 266, 24849-58.
- YABU, T., IMAMURA, S., YAMASHITA, M. & OKAZAKI, T. 2008. Identification of Mg2+ -dependent neutral sphingomyelinase 1 as a mediator of heat stress-induced ceramide generation and apoptosis. *The Journal of biological chemistry*, 283, 29971-29982.
- YAMAGUCHI, Y., MATSUNO, S., KAGOTA, S., HAGINAKA, J. & KUNITOMO, M. 2001. Oxidants in cigarette smoke extract modify low-density lipoprotein in the plasma and facilitate atherogenesis in the aorta of Watanabe heritable hyperlipidemic rabbits. *Atherosclerosis*, 156, 109-17.
- YAMASHITA, S., MASUDA, D., OHAMA, T., ARAI, H., BUJO, H., KAGIMURA, T., KITA, T., MATSUZAKI, M., SAITO, Y., FUKUSHIMA, M., MATSUZAWA, Y. & ON BEHALF OF THE, P. S. G. 2016. Rationale and Design of the PROSPECTIVE Trial: Probucol Trial for Secondary Prevention of Atherosclerotic Events in Patients with Prior Coronary Heart Disease. *Journal of Atherosclerosis and Thrombosis*, 23, 746-756.
- YLÄ-HERTTUALA, S., PALINSKI, W., ROSENFELD, M. E., PARTHASARATHY, S., CAREW, T. E., BUTLER, S., WITZTUM, J. L. & STEINBERG, D. 1989. Evidence for the presence of oxidatively modified low density lipoprotein in atherosclerotic lesions of rabbit and man. *Journal of Clinical Investigation*, 84, 1086-1095.

- YLA-HERTTUALA, S., ROSENFELD, M. E., PARTHASARATHY, S., GLASS, C. K., SIGAL, E., WITZTUM, J. L. & STEINBERG, D. 1990. Colocalization of 15lipoxygenase mRNA and protein with epitopes of oxidized low density lipoprotein in macrophage-rich areas of atherosclerotic lesions. *Proc Natl Acad Sci U S A*, 87, 6959-63.
- YLA-HERTTUALA, S., LUOMA, J., VIITA, H., HILTUNEN, T., SISTO, T. & NIKKARI, T. 1995. Transfer of 15-lipoxygenase gene into rabbit iliac arteries results in the appearance of oxidation-specific lipid-protein adducts characteristic of oxidized low density lipoprotein. *Journal of Clinical Investigation*, 95, 2692-2698.
- YOKODE, M., UEYAMA, K., ARAI, N. H., UEDA, Y. & KITA, T. 1996. Modification of high- and low-density lipoproteins by cigarette smoke oxidants. *Ann N Y Acad Sci*, 786, 245-51.
- YOSHIDA, H., QUEHENBERGER, O., KONDRATENKO, N., GREEN, S. & STEINBERG, D. 1998. Minimally oxidized low-density lipoprotein increases expression of scavenger receptor a, CD36, and macrosialin in resident mouse peritoneal macrophages. *Arteriosclerosis, Thrombosis, and Vascular Biology*, 18, 794-802.
- YOSHIDA, H. & KISUGI, R. 2010. Mechanisms of LDL oxidation. *Clin Chim Acta*, 411, 1875-82.
- YOU, S. A., ARCHACKI, S. R., ANGHELOIU, G., MORAVEC, C. S., RAO, S. Q., KINTER, M., TOPOL, E. J. & WANG, Q. 2003. Proteomic approach to coronary atherosclerosis shows ferritin light chain as a significant marker: evidence consistent with iron hypothesis in atherosclerosis. *Physiological Genomics*, 13, 25-30.
- YOU, S. A. & WANG, Q. 2005. Ferritin in atherosclerosis. Clinica Chimica Acta, 357, 1-16.
- YOUNG, I. S. & MCENENY, J. 2001. Lipoprotein oxidation and atherosclerosis. *Biochem Soc Trans*, 29, 358-62.
- YOUNIS, N., SHARMA, R., SORAN, H., CHARLTON-MENYS, V., ELSEWEIDY, M. & DURRINGTON, P. N. 2008. Glycation as an atherogenic modification of LDL. *Curr Opin Lipidol*, 19, 378-84.
- YOUNIS, N., SORAN, H., SHARMA, R., PEMBERTON, P., CHARLTON-MENYS, V., FRANCE, M., ELSWEIDY, M. & DURRINGTON, P. 2009. Glycation of LDL is an important atherogenic modification and opposed by paraoxonase-rich HDL. *Atherosclerosis*, 207, 306.
- YU, Z., PERSSON, H. L., EATON, J. W. & BRUNK, U. T. 2003. Intralysosomal iron: a major determinant of oxidant-induced cell death. *Free Radical Biology and Medicine*, 34, 1243-1252.
- YUAN, X. M. 1999. Apoptotic macrophage-derived foam cells of human atheromas are rich in iron and ferritin, suggesting iron-catalysed reactions to be involved in apoptosis. *Free Radical Research*, 30, 221-231.

- YUAN, X. M., ANDERS, W. L., OLSSON, A. G. & BRUNK, U. T. 1996. Iron in human atheroma and LDL oxidation by macrophages following erythrophagocytosis. *Atherosclerosis*, 124, 61-73.
- YUAN, X. M. & LI, W. 2008. Iron involvement in multiple signaling pathways of atherosclerosis: a revisited hypothesis. *Curr Med Chem*, 15, 2157-72.
- YUSUF, S., PHIL, D., DAGENAIS, G., POGUE, J., BOSCH, J. & SLEIGHT, P. 2000. Vitamin E supplementation and cardiovascular events in high-risk patients. *N. Engl. J. Med.*, 342, 154–160.
- ZAMBON, A., BERTOCCO, S., VITTURI, N., POLENTARUTTI, V., VIANELLO, D. & CREPALDI, G. 2003. Relevance of hepatic lipase to the metabolism of triacylglycerolrich lipoproteins. *Biochem Soc Trans*, 31, 1070-4.
- ZARJOU, A., JENEY, V., AROSIO, P., POLI, M., ZAVACZKI, E., BALLA, G. & BALLA, J. 2010. Ferritin ferroxidase activity: a potent inhibitor of osteogenesis. *J Bone Miner Res*, 25, 164-72.
- ZDRAVKOVIC, S., WIENKE, A., PEDERSEN, N. L., MARENBERG, M. E., YASHIN, A. I. & DE FAIRE, U. 2002. Heritability of death from coronary heart disease: a 36-year follow-up of 20 966 Swedish twins. *J Intern Med*, 252, 247-54.
- ZHANG, X. & LEMASTERS, J. J. 2013. Translocation of iron from lysosomes to mitochondria during ischemia predisposes to injury after reperfusion in rat hepatocytes. *Free Radical Biology and Medicine*, 63, 243-253.
- ZHANG, X., WANG, B., WANG, C., CHEN, L. & XIAO, Y. 2015. Monitoring lipid peroxidation within foam cells by lysosome-targetable and ratiometric probe. *Anal Chem*, 87, 8292-300.
- ZHANG, Y., MIKHAEL, M., XU, D., LI, Y., SOE-LIN, S., NING, B., LI, W., NIE, G., ZHAO, Y. & PONKA, P. 2010. Lysosomal proteolysis is the primary degradation pathway for cytosolic ferritin and cytosolic ferritin degradation is necessary for iron exit. *Antioxid Redox Signal*, 13, 999-1009.
- ZHAO, C. F. & HERRINGTON, D. M. 2016. The function of cathepsins B, D, and X in atherosclerosis. *American journal of cardiovascular disease*, 6, 163-170.
- ZHAO, N., ZHANG, A.-S. & ENNS, C. A. 2013. Iron regulation by hepcidin. *Journal of Clinical Investigation*, 123, 2337-2343.
- ZMIJEWSKI, J. W., MOELLERING, D. R., LE GOFFE, C., LANDAR, A., RAMACHANDRAN, A. & DARLEY-USMAR, V. M. 2005. Oxidized LDL induces mitochondrially associated reactive oxygen/nitrogen species formation in endothelial cells. *Am J Physiol Heart Circ Physiol*, 289, H852-61.

- ZOUAOUI BOUDJELTIA, K., MOGUILEVSKY, N., LEGSSYER, I., BABAR, S., GUILLAUME, M., DELREE, P., VANHAEVERBEEK, M., BROHEE, D., DUCOBU, J. & REMACLE, C. 2004. Oxidation of low density lipoproteins by myeloperoxidase at the surface of endothelial cells: an additional mechanism to subendothelium oxidation. *Biochem Biophys Res Commun*, 325, 434-8.
- ZSCHENKER, O., ILLIES, T. & AMEIS, D. 2006. Overexpression of lysosomal acid lipase and other proteins in atherosclerosis. *J Biochem*, 140, 23-38.

Appendices

Appendix 1: List of chemicals and suppliers

<u>Chemicals</u>	<u>Supplier</u>
Acetic Acid	Sigma-Aldrich Ltd
Acetonitrile (HPLC grade)	Sigma-Aldrich Ltd
Apoferritin from equine spleen (443kDa)	Sigma-Aldrich Ltd
Ammonium molybdate	Sigma-Aldrich Ltd
Ammonium persulphate	Fisher Scientific Ltd
Amphotericin B	Sigma-Aldrich Ltd
Apoptosis Kit	Biolegend
Bathophenanthrolinedisulfonic acid	Sigma-Aldrich Ltd
Bio-Rad DC TM protein assay	Bio-Rad
Bovine serum albumin (BSA)	Invitrogen Ltd
Benzalkonium chloride	Sigma-Aldrich Ltd
Beta-hydroxytoluene	Sigma-Aldrich Ltd
Beta-Mercaptoethanol	Fisher Scientific Ltd
Butanol	Sigma-Aldrich Ltd
Cathepsin B (43 kDa) (Native human cathepsin B)	Abcam
Cathepsin D (45 kDa)(from human liver)	Sigma-Aldrich Ltd
Cell staining buffer	Biolegend
Chelex-100	Sigma-Aldrich Ltd
Cholesterol esterase from Pseudomonas sp.	Sigma-Aldrich Ltd
Cholesterol linoleate hydroperoxide	Cayman chemicals

Coomassie brilliant blue R	BDH Chemicals Ltd
Copper sulphate	Fisher Scientific Ltd
Cysteamine hydrochloride	Sigma-Aldrich Ltd
DAPI (4',6-Diamidino-2-Phenylindole, Dihydrochloride)	Fisher Scientific Ltd
Diethylenetriaminepentaacetate (DTPA)	Sigma-Aldrich Ltd
Dihydroethidium	Sigma-Aldrich Ltd
Dipotassium hydrogen orthophosphate	Fisher Scientific Ltd
Disodium hydrogen orthophosphate	Fisher Scientific Ltd
Dulbecco's modified Eagle's medium	Invitrogen Ltd
Ethanol	Sigma-Aldrich Ltd
Ethylenediaminetetraacetic acid disodium salt (EDTA)	Sigma-Aldrich Ltd
F10 medium	Invitrogen Ltd
Ferric chloride	Sigma-Aldrich Ltd
Ferrous sulphate	Sigma-Aldrich Ltd
Ferritin from equine spleen (440kDa)	Sigma-Aldrich Ltd
Fetal calf serum (FCS)	Sigma-Aldrich Ltd
Fluorescence mounting medium Folin and Ciocalteu phenol reagents	Global Science Fisher Scientific Ltd
Glycerol	Fisher Scientific Ltd
Glycine	Fisher Scientific Ltd
GM-CSF	Sigma-Aldrich Ltd
Hexane (HPLC grade)	Sigma-Aldrich Ltd
Human Hepcidin-25 peptide Human serum	Abcam Sigma-Aldrich Ltd
Hydrochloric acid	Fisher Scientific Ltd
Hydrogen peroxide	Fisher Scientific Ltd
Isopropanol (HPLC grade)	Fisher Scientific Ltd

7-Ketocholesterol	Sigma-Aldrich Ltd
Magnesium chloride Methanol (HPLC grade)	Sigma-Aldrich Ltd Sigma-Aldrich Ltd
3-N-[Morpholinopropane] sulphonic acid	Fisher Scientific Ltd
N,N'-Diphenyl,1,4-phenylenediamine (DPPD)	Sigma-Aldrich Ltd
N, N,N,N'-tetraacetylenediamine	Sigma-Aldrich Ltd
Paraformaldehyde	Sigma-Aldrich Ltd
Penicillin/streptomycin	Sigma-Aldrich Ltd
Potassium acetate	Sigma-Aldrich Ltd
Potassium bromide	Fisher Scientific Ltd
Potassium chloride	Fisher Scientific Ltd
Potassium dihydrogen orthophosphate	Fisher Scientific Ltd
Potassium hydroxide	Fisher Scientific Ltd
Potassium iodide	Sigma-Aldrich Ltd
Potassium sodium tartrate	Fisher Scientific Ltd
RPMI	Invitrogen Ltd
Sodium acetate	Sigma-Aldrich Ltd
Sodium azide	Sigma-Aldrich Ltd
Sodium carbonate	Fisher Scientific Ltd
Sodium chloride	Fisher Scientific Ltd
Sodium deoxycholate	Sigma-Aldrich Ltd
Sodium dihydrogen orthophosphate	Fisher Scientific Ltd
Sodium hydrogen carbonate	Fisher Scientific Ltd
Sodium hydroxide	Fisher Scientific Ltd
Sodium dodecyl sulphate	Sigma-Aldrich Ltd
Sphingomyelinase from Bacillus cereus	Sigma-Aldrich Ltd
Tris-HCl	Fisher Scientific Ltd

Appendix 2: Quantity of iron contained in ferritin.

Ferritin 15 μ l of 53 mg protein/ml obtained from Sigma digested in 1ml nitric acid (5M) and made up to 50 ml was analysed using AAS.

Atomic weight of Fe is 55.84

1M Fe is 55.84 g/l

Top standard (5 mg/l) = $89.54 \mu M$

Linear trend line equation is Y = 0.0358x - 0.0004

0.0358x - 0.0004 = Y

0.0358x = Y + 0.0004

x = (Y + 0.0004) / 0.0358

Where Y=0.07186 (mean absorbance of sample)

x = (0.07186 + 0.0004) / 0.0358 = 2.018 mg/l

 $5 \text{ mg/l} = 89.54 \ \mu M$

 $2.018 \text{ mg/l} = 36.14 \ \mu M$

 $36.14 \ \mu M = 36.14 \ \mu mol/l$

1 l contains 36.14 µmol

50 ml of ferritin contains 1.807 µmol Fe

15 µl of ferritin solution has 1.807 µmol Fe

1 ml of ferritin solution = $(1000 \text{ x } 1.807) / 15 = 120.5 \,\mu\text{mol Fe}$

1 ml of ferritin has 120.5 µmol

The ferritin concentration obtained from Sigma is 53mg protein/ml

53mg has 120.5 µmol Fe

1 g protein = $120.5 \times 1000 / 53 = 2,274 \mu mol Fe$

= 2.274 mmol

Molecular weight of ferritin obtained from Sigma is 440kDA

440,000g protein has 440,000 x 2.274 mmol Fe =1,000,560 mmol Fe = 1,001 mol Fe

Therefore 1,001 atoms of Fe per ferritin particle.

Concentration of iron in ferritin in the spectrophotometer

1 M ferritin = 1001 M Fe

1 μ M ferritin = 1001 μ M Fe 0.05 μ M ferritin contains 50.05 μ M Fe 0.1 μ M ferritin contains 100.1 μ M Fe 0.2 μ M ferritin contains 200.2 μ M Fe
Appendix 3: Quantity of iron contained in Apoferritin.

Apoferritin 35.5 μ l of 25 mg protein/ml obtained from Sigma digested in 1ml nitric acid (5M) and made up to 10 ml mark was analysed using AAS.

Atomic weight of Fe is 55.84

Top standard (5 mg/l) = $89.54 \mu M$

Linear trend line equation is Y = 0.0509x - 0.0031

0.0509x - 0.0031 = Y

0.0509x = Y + 0.0031

x = (Y + 0.0031) / 0.0509

Where Y=0.00646 (mean absorbance of sample)

x = 0.00646 + 0.0031 / 0.0509 = 0.188 mg/l

 $5 \text{ mg/l} = 89.54 \ \mu M$

 $0.188 \text{ mg/l} = 3.36 \mu M$

 $3.36 \,\mu M = 3.36 \mu mol/l$

```
1 l contains 3.36 µmol
```

10 ml of apoferritin nitric digest contains 0.0336 µmol

35.5 µl of apoferritin has 0.0336 µmol

1 ml of apoferritin = $1000 \times 0.0336 / 35.5 = 0.9465 \mu mol$

1 ml of apoferritin has 0.9465 µmol

The apoferritin concentration obtained from Sigma is 25mg protein/ml

25mg apoferritin has 0.9465 µmol Fe

1 g protein = contains $0.9465 \times 1000 / 25 = 37.86 \mu mol Fe$

= 0.03786mmol Fe

1g of protein has 0.03786 mmol of Fe

M Wt of apoferritin= 443kDA

443,000g =443,000 x 0.03786 mmol Fe =16,772 mmol Fe = 17 mol Fe

Therefore 17 atoms of Fe per apoferritin particle

1 M apoferritin = 17 M Fe

1 μ M apoferritin = 17 μ M Fe 0.2 μ M apoferritin contains 3.4 μ M Fe 0.1 μ M apoferritin contains 1.7 μ M Fe 0.05 μ M apoferritin contains 0.85 μ M Fe